Relationships between agrobiodiversity, dietary diversity and nutritional status in Tanzania

Christine Liana Cleghorn

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

The University of Leeds

School of Medicine

School of Food Science and Nutrition

School of Earth and Environment

May 2014

The candidate confirms that the work submitted is her own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

The right of Christine Liana Cleghorn to be identified as Author of this work has been asserted by her in accordance with the Copyright, Designs and Patents Act 1988.

© 2014 The University of Leeds and Christine Liana Cleghorn

Acknowledgements

I would like to thank the respondents and their partners and children in Minyenye and Mbwei for taking part in this research. They very generously gave their time to speak with me and share information about their lives with no direct benefit to themselves.

Thank-you to Mwanane Shabani, the main research assistant of the project and, essentially, my translator. You were patient and hardworking throughout and I really appreciate your dedication to the project in difficult circumstances. Elisha Njoghomi, the biodiversity research assistant in Minyenye and Oscar Bakumbezi, the biodiversity research in Mbwei, your inputs to the project were essential. Kirsty Harper, who volunteered on the study in Mbwei, thankyou for the time and energy you put into the project, your advice and support was reall y appreciated. Thank-you to the local assistants in each sub-village who introduced us to the potential participants.

Many thanks to Professor Joyce Kinabo, Dr Peter Mamiro, the Eco-health team and Sokoine University of Agriculture in Morogoro, Tanzania for facilitating this project and providing equipment, contacts, financial and logistic support, especially for support through the permissions process within Tanzania.

The support I received through the local government was essential for this project to go ahead. Thank-you especially to the Ward Executive Officer (WEO) in Mtinko and the Village Executive Officer (VEO) and sub-village leaders in Minyenye and to the VEO, village Chairman and Assistant Chairman in Mbwei for facilitating the project to take place. Thanks also to my two nutrition contacts for your advice and support, Ester Kawishe in Singida and Neema Shosho in Tanga and to Helen Semu in Dar es Salaam for all your help. Thanks to the National Bureau of Statistics in Dar es Salaam for linking the census enumeration area (EA) number from the Demographic and Health Survey (DHS) to maps and village names, allowing me to locate the two villages I worked in.

Thank-you to my supervisors, Professor Janet Cade, Dr Darren Dahly and Dr Susannah Sallu for all your support, direction, advice and encouragement over the course of this project. Having input for your different backgrounds and perspectives was essential for creating this project and I am very grateful for the time you dedicated to it. Also thanks to George Ellison for stepping in in the last stretch.

I would like to acknowledge the use of data from the DHS project (2010 Tanzanian Demographic and Health Survey) and from the European Space Agency (Globcover 2009).

- iii -

iii

Images from Google Earth were used in the planning of this project. Thanks to UNICEF for providing the Mid Upper Arm Circumference (MUAC) measuring tapes for the study.

- iv -

The data collection was funded by three grants awarded to the PhD student from the Sustainable Agriculture Bursary fund at the University of Leeds (£1500), Africa College at the University of Leeds (£500) and the Dudley Stamp Memorial Award (£500). The flight to Tanzania was paid for by the Ecohealth project. Thanks also to the University of Leeds for my University Research Scholarship.

Thanks to everyone within the University of Leeds that provided information and support over the course of my PhD. This includes Lindsay Stringer, my transfer viva assessor, those within the Epidemiology Division who provided statistical advice and especially to Jerome O'Connell who provided valuable Geographic Information systems (GIS) support. Thanks to Beatrice Paganini and the students who helped with data entry and also to the PhD students in Earth and Environment for your help understanding a new field. Thanks to everyone at the Nutritional Epidemiology Group for making this an enjoyable journey.

Thanks so much to my family for your endless support and encouragement. And finally, thankyou to Kirsty and friends who kept me sane over the course of my PhD.

Abstract

Background: Agrobiodiversity is important for biodiversity conservation and sustainable agriculture. Increasing agrobiodiversity may also improve dietary diversity and nutritional status in low income countries but research is lacking in this area. To fill this knowledge gap, this study explores relationships between agrobiodiversity, dietary diversity and nutritional status in Tanzania. The research investigates 1) the relationships between agrobiodiversity, dietary diversity and nutritional status in children in two villages in rural Tanzania and 2) the relationships between land cover, dietary diversity and nutrition in under five year olds in a nationally representative sample in Tanzania.

Methods: A cross-sectional survey was conducted in 122 randomly selected households in Minyenye village, Singida district and Mbwei village, Lushoto district. Female heads of households were interviewed to collect quantitative and qualitative data on demographics, livelihoods, complementary feeding and household food sources. Dietary diversity was calculated from 24 hour dietary recalls which were collected for the respondent and their oldest child under five. Agrobiodiversity data were systematically collected using the point intercept method. All plants, both intentional crops and other plants, growing on the household's farms at the time of data collection were measured and Shannon Biodiversity Indices were calculated. In addition to these indices, the food sources section of the questionnaire was used to calculate household crop and animal diversity scores. Heights and weights were measured in all family members and MUAC was collected for all children under 15 years old. Relationships between these factors were explored using regression analyses. At the national level relationships between land cover, from GlobCover 2009, and dietary diversity and nutrition, from the 2010 Tanzanian Demographic and Health Survey (DHS), were investigated using spatial and regression analyses.

Results: No significant associations were seen between dietary diversity and nutritional status in either village and dietary diversity was negatively associated with height for age z-scores in the DHS analyses. No significant associations were seen between agrobiodiversity and dietary diversity. Agricultural, but not forest, land cover was associated with dietary diversity. Associations between both agrobiodiversity and land cover and child nutritional status are complex. In Minyenye, agrobiodiversity was positively associated with children's height for age while in Mbwei these were negatively associated with children's body mass index (BMI) zscores. More agricultural and forest land cover was associated with higher weight for height, however more agricultural land cover was associated with lower height for age. Positive

v

associations were seen between eating and rearing animals and dietary diversity but negative or no associations were seen with nutritional status. Selling crops was positively associated with dietary diversity but showed mixed associations with nutritional status at the local village scale.

- vi -

Conclusion: Study results provide a word of caution for those attempting to increase agrobiodiversity to improve diet and nutritional status. The effectiveness of agricultural interventions aimed at improving nutrition through improvements in agrobiodiversity can only be evaluated in light of the multiple determinants of nutritional status. The current study's results illustrate the complexity of the pathway from food production through consumption to nutrient utilization in low income countries.

Table of Contents

- vii -

Acknowledgementsiii
Abstract v
Table of Contents vii
List of Tablesxiv
List of Figuresxvii
List of Abbreviationsxx
Chapter 1: Introduction 1
1.1 Chapter summary1
1.2 Aim and objectives
1.3 Food security 4
1.4 Nutritional status 5
1.5 Dietary diversity10
1.6 Agrobiodiversity, dietary diversity and nutritional status
1.6.1 Agriculture and nutrition11
1.6.2 Agriculture and biodiversity12
1.6.3 Biodiversity and nutrition12
1.6.4 Agriculture, biodiversity and nutrition12
1.7 Food security and nutritional status in Tanzania14
1.8 Foods eaten in Tanzania14
1.9 Outline of thesis and summary of research contributions16
Chapter 2: Literature reviews of the associations between agrobiodiversity, dietary diversity and nutritional status
2.1 Chapter summary18
2.2 Introduction
2.3 Part 1: Dietary diversity and nutritional status in children
2.3.1 Methods19
2.3.1.1 Search strategy19
2.3.1.2 Criteria for inclusion20
2.3.1.3 Included references20
2.3.2 Results and discussion23
2.4 Part 2: Agrobiodiversity and dietary diversity/nutritional status
2.4.1 Methods
2.4.1.1 Search strategy39

2.4.1.2 Criteria for inclusion	40
2.4.1.3 Included references	40
2.4.2 Results and discussion	42
2.4 Conclusion	50
Chapter 3: Pilot, village selection and methods for the project's primary data	
collection	51
3.1 Chapter summary	
3.2 Research design	51
3.3 Village selection for primary data collection	56
3.4 Study site description	63
3.4.1 Minyenye village, Singida district, Singida region	63
3.4.2 Mbwei village, Lushoto district, Tanga region	66
3.5 Mixed method approach	67
3.6 Household questionnaire	69
3.6.1 Dietary recall methods	69
3.6.2 Anthropometric measurements of nutritional status	70
3.6.3 Agrobiodiversity	71
3.6.3.1 Systematic measurement of plant agrobiodiversity	71
3.6.3.2 Plant and animal diversity scores	73
3.7 Sampling strategy	74
3.8 Inclusion criteria	75
3.9 Research process	76
3.9.1 Piloting of methods	76
3.9.2 Ethical approval and permissions	77
3.9.3 Research assistants and translation	77
3.9.4 Recruitment of participants and consent process	
3.10 Data checking and entry	
3.11 Data analysis	
3.11.1 Basic descriptive statistics	79
3.11.2 Linear regression	79
3.11.3 Multiple imputation by chained equations (MICE)	82
Chapter 4: Investigating the associations between dietary diversity and nutrition status in Minyenye and Mbwei, rural Tanzania.	ıal 84
4.1 Chapter summary	
4.2 Introduction	
4.3 Methods	
4.3.1 Data sources	

4.3.1.1 Dietary diversity and food variety
4.3.1.2 Food sources
4.3.1.4 Nutritional status88
4.3.2 Data analysis88
4.3.2.1 Descriptive statistics
4.3.2.2 Linear regression analysis
4.3.2.3 Missing data89
4.4 Results90
4.4.1 Wider social and farming context91
4.4.2 Demographic and household characteristics of sample
4.4.3 Dietary diversity and food variety96
4.4.4 Nutritional status of children100
4.4.5 Breastfeeding and complementary feeding of children
4.4.6 Dietary diversity and food variety in relation to nutritional status104
4.4.7 Complementary feeding, demographics, sanitation and nutritional status
4.5 Discussion
4.5.1 Nutritional status of children110
4.5.2 Dietary diversity and nutritional status110
4.5.3 Complementary feeding, demographics, sanitation and nutritional status
4.5.4 Limitations and strengths115
4.6 Conclusions
Chapter 5: Investigating the associations between plant agrobiodiversity and both dietary diversity and nutritional status in Minyenye and Mbwei, rural Tanzania. 118
5.1 Chapter summary118
5.2 Introduction119
5.3 Methods
5.3.1 Data sources
5.3.1.1 Agrobiodiversity
5.3.1.2 Questionnaire
5.3.2 Data calculations123
5.3.2.1 Agrobiodiversity index123
5.3.3 Data analysis124
5.3.3.1 Descriptive statistics
5.3.3.2 Linear regression analysis125
5.3.3.3 Missing data125

5.4 Results	5
5.4.1 Habitat and farm characteristics126	5
5.4.2 Minyenye village, Singida district134	1
5.4.2.1 Habitat, species present, farm characteristics and plant agrobiodiversity134	1
5.4.2.2 Plant agrobiodiversity and dietary diversity /food variety 142	2
5.4.2.3 Plant agrobiodiversity and nutritional status	1
5.4.2.4 Selling produce and dietary diversity	5
5.4.2.5 Selling produce and nutritional status	3
5.4.3 Mbwei village, Lushoto district150)
5.4.3.1 Habitat, species present, farm characteristics and plant agrobiodiversity150)
5.4.3.2 Plant agrobiodiversity and dietary diversity	5
5.4.3.3 Plant agrobiodiversity and nutritional status	7
5.4.3.4 Selling produce and dietary diversity)
5.4.3.5 Selling produce and nutritional status	L
5.5 Discussion	3
5.5.1 Plant agrobiodiversity and dietary diversity/food variety	3
5.5.2 Plant agrobiodiversity and nutritional status	3
5.5.3 Crop and vegetable diversity scores, dietary diversity and nutritional status	5
5.5.4 Vegetable gardens, dietary diversity and nutritional status	5
5.5.5 Comparison of Minyenye and Mbwei	7
5.5.6 Selling produce, dietary diversity and nutritional status)
5.5.7 Limitations and strengths170)
5.6 Conclusions	L
Chapter 6: Investigating the associations between animal diversity and both dietary diversity and nutritional status in Minyenye and Mbwei, rural Tanzania	3
6.1 Chapter summary	3
6.2 Introduction174	1
6.3 Methods	7
6.3.1 Data collection	7
6.3.2 Data analysis177	7
6.3.2.1 Descriptive statistics	7
6.3.2.2 Linear regression analysis177	7
6.4 Results	3
6.4.1 Animal product consumption and animal diversity in the two villages . 178	3

	6.4.2 Minyenye	. 182
	6.4.2.1 Animal diversity and dietary diversity	. 182
	6.4.2.2 Animal diversity and nutritional status	. 184
	6.4.2.3 Selling animal products and dietary diversity	. 186
	6.4.2.4 Selling animal products and nutritional status	. 188
	6.4.3 Mbwei	. 190
	6.4.3.1 Animal diversity and dietary diversity	. 190
	6.4.3.2 Animal diversity and nutritional status	. 192
	6.4.2.3 Selling animal products and dietary diversity	. 194
	6.4.3.4 Selling animal products and nutritional status	. 196
	6.5 Discussion	. 198
	6.5.1 Animal diversity, dietary diversity and nutritional status	. 198
	6.5.2 Animal rearing for reasons other than consumption	. 199
	6.5.3 Barriers to animal diversity positively impacting nutritional status	. 199
	6.5.4 Selling animals and animal products	. 200
	6.5.5 Limitations and strengths	. 200
	6.6 Conclusions	. 201
Cha	apter 7: Investigating the relationships between land cover, dietary diversity and nutritional status in a national sample in Tanzania	202
		205
	7.1 Chapter summary	203
	7.1 Chapter summary	203
	 7.1 Chapter summary 7.2 Introduction 7.3 Methods 	203 203 204 205
	 7.1 Chapter summary 7.2 Introduction 7.3 Methods 7.3.1 Data sources 	203 203 204 205 205
	 7.1 Chapter summary 7.2 Introduction 7.3 Methods 7.3.1 Data sources 7.3.1.1 Demographic and Health Survey data 	203 203 204 205 205 205
	 7.1 Chapter summary 7.2 Introduction 7.3 Methods 7.3.1 Data sources 7.3.1.1 Demographic and Health Survey data	203 203 204 205 205 205 207
	 7.1 Chapter summary 7.2 Introduction 7.3 Methods	203 203 204 205 205 205 207 207
	 7.1 Chapter summary 7.2 Introduction 7.3 Methods	203 203 204 205 205 205 207 207 207 211
	 7.1 Chapter summary	203 203 204 205 205 205 207 207 207 211
	 7.1 Chapter summary	203 203 204 205 205 205 207 207 207 211 211
	 7.1 Chapter summary	203 203 204 205 205 205 207 207 207 211 211 212
	 7.1 Chapter summary	203 203 204 205 205 205 207 207 211 211 212 213 215
	 7.1 Chapter summary	203 203 204 205 205 205 205 207 207 211 211 211 212 213 215
	 7.1 Chapter summary	203 203 204 205 205 205 205 207 207 211 211 211 212 213 215 215
	 7.1 Chapter summary	203 203 203 204 205 205 207 207 207 211 211 211 212 213 215 215 218

	7.4.4 Dietary diversity and nutritional status	225
	7.4.5 Spatial variation of dietary diversity and nutritional status	227
	7.4.6 Land cover, dietary diversity and nutritional status	230
	7.5 Discussion	235
	7.5.1 Demographic and social factors, dietary diversity and nutritional statu	s235
	7.5.2 Agricultural and dietary factors, dietary diversity and nutritional status	236
	7.5.3 Dietary diversity and nutritional status	237
	7.5.4 Spatial variation of dietary diversity and nutritional status	238
	7.5.5 Land cover, dietary diversity and nutritional status	239
	7.5.6 Limitations and strengths	240
	7.6 Conclusions	241
Chap	ter 8: Overall discussion	242
	8.1 Chapter summary	242
	8.2 Key messages	242
	8.3 Dietary diversity and nutritional status	243
	8.4 Plant agrobiodiversity, dietary diversity and nutritional status	245
	8.5 Animal agrobiodiversity and animal product consumption	250
	8.6 Selling household produce	251
	8.7 Limitations	251
	8.8 Strengths	253
	8.9 Implications and contributions of this study	253
	8.10 Future research and methodological and study design insights	254
Chap	ter 9: Overall conclusion	257
	9.1 How the study fulfils the aim and objectives	257
	9.2 Dietary diversity is not associated with nutritional status in cross-sectional data	257
	9.3 Agrobiodiversity is not associated with dietary diversity in cross-sectional data	258
	9.4 Agrobiodiversity has mixed associations with nutritional status	258
	9.5 Simply increasing agrobiodiversity will not necessarily improve dietary diversity and nutritional status	259
	9.6 Streamline monitoring agrobiodiversity through diversity scores	259
	9.7 Animal production and consumption will not necessarily improve dietary diversity and nutritional status	259
	9.8 Selling produce will not necessarily improve dietary diversity and nutritional status	260
	9.9 Agricultural interventions designed to improve nutritional status need to be piloted, context specific and regularly monitored	260
	xii	

9.10 Interventions need to be broad to successfully address malnutrition	261
References	262
Appendix A: Ethical approval	282
Appendix B: Information sheet and consent form	284
Appendix C: Data collection protocols	286
24 hour recall protocol	286
Anthropometry protocol	287
Agrobiodiversity protocol	289
Appendix D: Household Questionnaire	291
Appendix E: Agrobiodiversity data collection sheet	305

List of Tables

Table 1.1. Methods used to measure nutritional status, their definition and their pros and cons
Table 2.1. Studies investigating the associations between dietary diversity based onnumber of food groups and nutritional status based on anthropometricmeasures
Table 2.2. Studies investigating the association between food variety based onnumber of food items and nutritional status based on anthropometric measures.36
Table 2.3. Studies investigating the associations between agrobiodiversity and dietary diversity 46
Table 2.4. Studies investigating the associations between agrobiodiversity andnutritional status as measured by anthropometry
Table 3.1. Data collection methods and participants involved in meeting each of the study's objectives
Table 3.2. Total number of households and total number of households interviewed by sub-village
Table 4.1. Number of households, children and under five year olds measured in each village
Table 4.2. Demographic information for the family unit for included households inMinyenye and Mbwei and both villages combined
Table 4.3. Characteristics of households in Minyenye and Mbwei and in both villages combined 94
Table 4.4. Mean dietary diversity and food variety scores for respondents and theiroldest child under five in Minyenye, Mbwei and both villages combined
Table 4.5. Nutritional status of all children and specifically children under the age offive in Minyenye, Mbwei and both villages combined
Table 4.6. Breastfeeding and complementary feeding of children under five inMinyenye, Mbwei and both villages combined
Table 4.7. Unadjusted and adjusted linear regression results for dietary diversity andnutritional status in children under five for both villages combined
Table 4.8. Adjusted* linear regression results for complementary feeding andnutritional status in children under five for both villages combined
Table 5.1. Farm characteristics of farms where agrobiodiversity data were and werenot collected in Minyenye and Mbwei127
Table 5.2. Plant agrobiodiversity indices, percentage selling grown food, farm typesand farming issues for Minyenye and Mbwei131
Table 5.3. Abundance of species present on household farms in Minyenye split by foodspecies and non food species

Table 5.4. Associations between plant agrobiodiversity and crop and vegetable diversity scores and dietary diversity/food variety in children under five and respondents in Minyenye 143
Table 5.5. Associations between plant agrobiodiversity and crop and vegetablediversity scores and nutritional status in children under five and all children inMinyenye
Table 5.6. Associations between households selling foods they grew and the respondent's and children's dietary diversity and food variety scores in Minyenye
Table 5.7. Associations between household selling foods they grew and under fiveyear old's and all children's nutritional status in Minyenye
Table 5.8. Abundance of species present on household farms in Mbwei split bywhether they are food species or non food species
Table 5.9. Associations between plant agrobiodiversity and crop and vegetablediversity scores and dietary diversity/food variety and children under five andrespondents in Mbwei
Table 5.10. Association between plant agrobiodiversity and crop and vegetablediversity scores and nutritional status in children under five and all children inMbwei
Table 5.11. Associations between households selling foods they grew and therespondent's and children's dietary diversity and food variety scores in Mbwei160
Table 5.12. Associations between households selling foods they grew and under fiveyear old's and all children's nutritional status in Mbwei162
Table 6.1. Number of households in Minyenye and Mbwei eating, raising, buying,being given and selling animals and animal products in the last calendar year 182
Table 6.2. Associations between the number of different animals eaten by thehousehold and animal diversity scores and dietary diversity and food variety inrespondents and under five year olds in Minyenye
Table 6.3. Association between the number of animals eaten by the household and animal diversity scores and nutritional status in all children and children under five in Minyenye185
Table 6.4. Association between whether animals and animal products are sold and dietary diversity and food variety in respondents and under five year olds in Minyenye
Table 6.5. Association between whether animals and animal products are sold andnutritional status in all children and children under five in Minyenye189
Table 6.6. Association between the number of different animals eaten by thehousehold and animal diversity scores and dietary diversity and food variety inrespondents and under five year olds in Mbwei191
Table 6.7. Association between the number of animals eaten by the household and animal diversity scores and nutritional status in all children and children under five in Mbwei193

Table 6.8. Association between whether animals and animal products are sold anddietary diversity and food variety in respondents and under five year olds inMbwei
Table 6.9. Association between whether animals and animal products are sold andnutritional status in all children and children under five in Mbwei
Table 7.1. The 17 covariates in 10 categories identified as being potential determinants of nutritional status
Table 7.2. Respondent's characteristics information in the Tanzanian 2010 DHS
Table 7.3. Household characteristics in the Tanzanian 2010 DHS
Table 7.4. Respondent's and their youngest child's nutritional status
Table 7.5. Association between household and demographic variables and dietarydiversity in the respondent and their youngest child
Table 7.6. Association between household and demographic variables and nutritional status in the youngest child
Table 7.7. Association between agricultural and dietary variables and dietary diversityin the respondent and their youngest child
Table 7.8. Association between agricultural and dietary variables and nutritionalstatus variables in the youngest child
Table 7.9. Association between dietary diversity scores and its components andnutritional status variables in the youngest children
Table 7.10. Percentage of land cover falling into agriculture, forest and other landcover categories in the 5km surrounding the significantly high and low DHS EAcentre points232
Table 7.11. Association between dietary diversity and nutritional status andpercentage land cover of agriculture, forest or other land cover types in the 5kmsurrounding the DHS centre points

List of Figures

Figure 1.1. UNICEF conceptual framework of the determinants of malnutrition (UNICEF, 1990)
Figure 1.2. Conceptual framework of the links between agriculture and health (Hawkes and Ruel, 2006)11
Figure 1.3. Conceptual framework linking biodiversity conservation and human nutrition in low income countries (Johns and Sthapit, 2004)13
Figure 2.1. Flowchart of inclusion for part 1: Dietary diversity and nutritional status22
Figure 2.2. Flowchart of inclusion for part 2: Agrobiodiversity and dietary diversity/nutritional status41
Figure 3.1. Overall research design, the objectives each phase fulfils and the chapters these are presented
Figure 3.2 a) Conceptual framework for the relationships between agrobiodiversity, dietary diversity and nutritional status54
Figure 3.2 b) Section of the conceptual framework for the relationships between agrobiodiversity, dietary diversity and nutritional status which forms the focus of this thesis
Figure 3.3. A map of the 18 potential enumeration areas determined as malnourished in relation to land cover classification (GlobCover2009) in Tanzania. The key outlines 22 different land cover categories determined in the GlobCover2009 data
Box 3.1 Land use types used to classify high biodiversity sites58
Box 3.2 Land use types used to classify low biodiversity sites59
Figure 3.4. Location of top 6 potential sites, first, second and third choices for the high and low diversity sites60
Figure 3.5. Land cover around EA 62, in Tanga, Northern Tanzania, one of the three potential high diversity EA60
Figure 3.6 Land cover around FA 232 in Singida Central Tanzania one of the three
potential low diversity EA
Figure 3.7. Land cover around EA 376, in Manyara, Northern Tanzania, one of the three potential high diversity EA
 Figure 3.7. Land cover around EA 376, in Manyara, Northern Tanzania, one of the three potential high diversity EA
 Figure 3.7. Land cover around EA 376, in Manyara, Northern Tanzania, one of the three potential high diversity EA
 Figure 3.6. Land cover around EA 376, in Manyara, Northern Tanzania, one of the three potential high diversity EA

Figure 3.12. A typical house and surrounding area in Minyenye village, Singida district (photo taken during data collection, June 2012, dry season)63
Figure 3.13. Google Earth image of the area surrounding Minyenye village (August 2004, dry season) illustrating flat, largely un-vegetated orange land and seasonal rivers/streams. The DHS centre point is marked by the A flag, the red outline is the border of the enumeration area from the DHS (2010)65
Figure 3.14. The hilly landscape of Mbwei's sub-villages (on the opposite hills) with the fertile valley in view (Photo taken during data collection, July 2012, dry season)66
Figure 3.15. Google Earth image of area surrounding Mbwei village (April 2007, Rainy Season). Illustrating mountainous terrain and largely green vegetated lands. The DHS centre point is marked by the A flag, the red outline is the border of the enumeration area from the DHS (2010)67
Figure 3.16. Directed Acyclic Graph (DAG) used in determining potential confounders in the regression models
Figure 4.1. Water sources in Minyenye; a water pump, a borehole, holes dug in an almost dry area of a riverbed and a river. The rivers are used as water sources for both animals and people95
Figure 4.2. Water source in Mbwei; the river and a water spring. The spring water is diverted into channels for water collection and irrigation
Figure 4.3. Percentage of participants (respondents and children combined) in Minyenye and Mbwei consuming the nine food groups that make up the dietary diversity score on the day before the interview
Figure 4.4. Mean household dietary diversity score for the previous calendar year in Minyenye and Mbwei. Score is out of six and based on grown, plant foods only^.99
Figure 4.5. Mean household dietary diversity score for the previous calendar year in Minyenye and Mbwei. Score is out of six and based on grown, wild or gifted plant foods only^100
Figure 5.1. The relationship between planned biodiversity and associated biodiversity and how they promote ecosystem function (presented by (Altieri, 1999), modified from Vandeermeer and Perfecto, 1995)121
Figure 5.2. Households and all measured household farms in the five sub-villages of Minyenye
Figure 5.3. Households and all measured household farms in the seven sub-villages of Mbwei
Figure 5.4. Percentage of households growing different crop types in Minyenye and Mbwei over the previous calendar year
Figure 5.5. A typical area in Minyenye of dry red dirt with a few larger trees
Figure 5.6. The dry riverbed running through the valley in Minyenye, some grass is available for grazing in this area at this time of year135
Figure 5.7. A typical farm in Minyenye135
Figure 5.8. A typical vegetable garden in Minyenye136
Figure 5.9. Number of species; all plants and food plants grown per acre of household farm in Minyenye

Figure 5.10. A typical farm in Mbwei150
Figure 5.11. A typical farm in Mbwei151
Figure 5.12 A typical vegetable garden in Mbwei151
Figure 5.13. Number of species; all plants and food plants grown per acre of household farm in Mbwei
Figure 6.1. Pathway between livestock production and nutritional status (Leroy and Frongillo, 2007)176
Figure 6.2. Animals kept in a pen in a households yard and being taken to get water in Minyenye179
Figure 6.3. A single cow being kept by a household in Mbwei179
Figure 6.4. Percentage of households raising specific animals in the Minyenye and Mbwei
Box 7.1. Land cover types classified as agricultural land cover208
Box 7.2. Land cover types classified as forest land cover209
Box 7.3. Land cover types classified as other land cover209
Figure 7.1. GlobCover2009 land cover map of Tanzania210
Figure 7.2. Globcover 2009 map of Tanzania with DHS EA centre points211
Figure 7.3. Globcover2009 map with the 5km buffer zones around the DHS EA centre point marked in green; the image on the right is a close up look at the central cluster from the image on the left (different colours represent different land cover types)
Figure 7.4. Clusters of high and low dietary diversity in the youngest child and high dietary diversity in the mother
Figure 7.5. Clusters of high and low height for age in the youngest child228
Figure 7.6. Clusters of high and low weight for height in the youngest child229
Figure 7.7. Clusters of high and low BMI in the mother230
Figure 8.1. Conceptual framework of factors impacting on the relationship between agrobiodiversity, dietary diversity and nutritional status248

List of Abbreviations

- XX -

BMI	Body Mass Index
CI	Confidence Interval
COSTECH	Commission for Science and Technology
DDS	Dietary Diversity Score
DHS	Demographic and Health Survey
EA	Enumeration Area
ESA	European Space Agency
ENVISAT	ESA Environmental Satellite
FANTA	Food And Nutrition Technical Assistance project
FAO	Food and Agriculture Organization
FFQ	Food Frequency Questionnaire
FVS	Food Variety Score
GIS	Geographic Information System
GNI	Gross National Income
GPS	Global Positioning System
HAZ	Height for age z-score
IMF	International Monetary Fund
LAZ	Length for age z-score
LCCS	Land Cover Classification System
MAR	Missing At Random
MDG	Millennium Development Goals
MERIS	Medium Resolution Imaging Spectrometer Instrument
MICE	Multiple Imputations using Chained Equations
MUAC	Mid Upper Arm Circumference

хх

MODIS	Moderate Resolution Imaging Spectroradiometer
-------	---

- NBS National Bureau of Statistics
- NDVI Normalised Difference Vegetation Index
- OR Odds Ratio
- RCT Randomised Controlled Trial
- SD Standard Deviation
- SE Standard Error
- SI Shannon Index
- SPOT Probationary System of Earth Observation
- VEO Village Executive Officer
- VCF Vegetation Continuous Fields
- WAZ Weight for age z-score
- WEO Ward Executive Officer
- WHO World Health Organisation
- WHZ Weight for height z-score
- WLZ Weight for length z-score

Chapter 1: Introduction

1.1 Chapter summary

This first chapter introduces the background to this research and the national and regional context in which it took place. This chapter briefly summarises food security issues and nutritional status in sub-Saharan African and the relevance and measurement of dietary diversity as a marker for nutritional status. It defines agrobiodiversity and introduces the potential relevance to dietary diversity and nutritional status. Specifically, the reasons for investigating the relationships between agrobiodiversity, dietary diversity and nutritional status and research into these intersecting areas are discussed. Information is included on food security, nutrition and foods eaten in Tanzania. This introductory chapter ends with an outline of the thesis structure and a summary of the key academic contributions achieved in each chapter.

1.2 Aim and objectives

This thesis reports a study that aimed to **investigate the relationship between agrobiodiversity, dietary diversity and nutritional status in Tanzania**. Broadly, the research investigates 1) the relationships between agrobiodiversity, dietary diversity and nutritional status in children in two villages in rural Tanzania and 2) the relationships between land cover, dietary diversity and nutrition in under five year olds in a nationally representative sample in Tanzania. As a means to fill the knowledge gap on how agrobiodiversity relates to diet and health in subsistence agriculture and to improve understanding of food security in sub-Saharan Africa. - 2 -

The specific objectives of this study are to:

1. Review, design and pilot a conceptual framework and suite of appropriate measurement tools to accurately measure agrobiodiversity, dietary diversity and nutrition in an Eastern Sub-Saharan Africa context.

2. Systematically assess the diversity and abundance of both cultivated and wild plants growing on household agricultural land in two villages in rural Tanzania and investigate how this is related to dietary diversity in under five year old and nutritional status in children in these villages.

3. Measure household crop and animal diversity, dietary diversity of respondents and children under five and nutritional status of children living in these villages, determine whether household produce is sold and investigate how these factors are related.

4. Investigate basic socio-demographic factors affecting dietary intake of children under five and nutrition outcomes in children in these villages.

5. Investigate the socio-demographic determinants of dietary diversity and nutritional status and explore the relationships between land cover, dietary diversity and nutritional status in children under five years at a national level, in Tanzania.

6. Integrate outcomes from this multi-scale investigation of the relationships between agrobiodiversity, dietary diversity and nutritional status, to draw conclusions that inform understandings of food security in sub-Saharan Africa.

Objectives 2, 3 and 4 were met through the primary data collection presented in chapters 4, 5 and 6. Each of these chapters, as well as chapter 7, had specific objectives to ensure they met the overall thesis objectives:

Chapter 4:

- Objective 4A: To present descriptive data on the demographic, social, dietary diversity and nutritional status variables in this population to set the context for the analyses.
- Objective 4B: To investigate whether dietary diversity and food variety are associated with nutritional status in under five year olds.
- Objective 4C: To investigate whether complementary feeding and sanitation are associated with nutritional status in these villages.

Chapter 5:

- Objective 5A: Present descriptive data on habitat, species present, farm characteristics, cross-sectional plant agrobiodiversity and crop and vegetable diversity scores in the two villages.
- Objective 5B: Investigate whether plant agrobiodiversity and crop/vegetable diversity scores are associated with dietary diversity and nutritional status in children.
- Objective 5C: Investigate whether selling staple crops, vegetables, fruit and other produce is associated with dietary diversity and nutritional status in children.

Chapter 6:

- Objective 6A: To assess animal diversity and present descriptive data on animal product consumption and animal rearing practices in the two villages.
- Objective 6B: To examine whether eating animals and animal products and animal diversity are associated with dietary diversity in respondents and under five year olds and nutritional status in children.
- Objective 6C: To examine whether selling animals and animal products is associated with dietary diversity in respondents and under five year olds and nutritional status in children.

Chapter 7:

- Objective 7A: To investigate whether demographic, social, agricultural and dietary factors are associated with dietary diversity and nutritional status in under five year olds in Tanzania.
- Objective 7B: To investigate whether dietary diversity is associated with nutritional status in under five year olds in Tanzania.
- Objective 7C: To investigate whether dietary diversity and nutritional status in under five year olds vary spatially in Tanzania.
- Objective 7D:To investigate whether land cover is related to dietary diversity and
nutritional status in under five year olds in Tanzania.

1.3 Food security

Much of the research in the area of food security cite the Food and Agriculture Organisation (FAO) definition for food security published in 1996: "Food security exists when all people, at all times, have physical and economic access to sufficient, safe and nutritious food to meet their dietary needs and food preferences for an active and healthy life" (FAO, 1996). This definition is of global significance where many low income countries are perhaps first aiming to meet their dietary needs. Food security is often expressed in three levels: Availability, access and utilization (FAO, 1996). Since Sen's essay on entitlement to food was published in 1981 (Sen), food security research in the social sciences has shifted from food production to food access (Webb et al., 2006, Coates et al., 2006). Workshops held by the Food and Nutrition Technical Assistance (FANTA) project in 2004 and 2005 agreed on the following domains of food insecurity: Anxiety/uncertainty about and actual depletion of the household food supply; Insufficient quality, which includes variety, preferences and social acceptability; Insufficient food intake and its physical consequences; Coping strategies to increase household resources (FANTA, 2004).

- 4 -

The International Scientific Symposium "Biodiversity and Sustainable Diets: United Against Hunger" organised by FAO and Bioversity International in 2010 proposed a definition of sustainable diets which encompass concepts of environmental sustainability and food security: "Sustainable diets are those diets with low environmental impacts which contribute to food and nutrition security and to healthy life for present and future generations. Sustainable diets are protective and respectful of biodiversity and ecosystems, culturally acceptable, accessible, economically fair and affordable; nutritionally adequate, safe and healthy; while optimizing natural and human resources" (Burlingame and Dernini, 2012). This shift in the view of food security is essential to ensure food security into the future. Achieving both food security and food systems sustainability simultaneously requires a change in perspective when considering availability and access from agriculture and markets through to consumption and utilization by the individual (Prosperi et al., 2014).

The global food system is under great pressure due to the increasing global population combined with stronger competition for land, water and energy under the largely unknown effects of climate change (Foresight, 2011). Globally there are two billion people who are not getting enough food (Foresight, 2011). These pressures are felt most strongly in low income countries, which are defined as countries with gross national income (GNI) per capita of \$1,035 or less (lower middle income, \$1,036 - \$4,085; upper middle income, \$4,086 - \$12,615; and high income, \$12,616 or more) (The World Bank, 2014). Food security needs to be addressed at

the same time as reducing the impact of food production on the environment (McMichael, 2005). The Millennium Development Goals (MDG) highlight both these areas with "end poverty and hunger" and "ensure environmental sustainability" being two of their eight goals (United Nations, 2014). Researchers are talking about the importance of reducing malnutrition and environmental degradation simultaneously (McMichael, 2005) and sustainable agriculture has a large part to play in this. Increased understanding of the complex relationships between nutrition, agriculture and the environment is essential.

The relationships between agriculture and nutrition are no more relevant globally than in sub-Saharan Africa where 26% of the world's undernourished people live; approximately 239 million people (UNFAO, 2010b). Self employment in agriculture is the most important activity for the rural labour force in sub-Saharan Africa (The World Bank, 2008). In sub-Saharan Africa agriculture and its associated industries are essential to growth and to reducing mass poverty and food insecurity (The World Bank, 2008). The landscape in East Africa is mainly arid and semi-arid, with some sub-humid, desert and highland areas (Barry et al., 2006). These land types characteristically have low uneven rainfall and infertile soils which have led to problems in the food supply of these areas (Barry et al., 2006). The challenge of improving both the quality and amount of food produced is key for the improvement of quality of life for those living in East sub-Saharan Africa on the assumption that increasing food production will increase food security (Larson and Frisvold, 1996). The relationship between food production and consumption is complex and there is concern that an increase in food production will not necessarily lead to an increase in food security (Sen, 1981).

1.4 Nutritional status

Individuals are malnourished if their diet does not provide adequate calories, protein and other essential nutrients for growth or they are unable to use the food they eat due to illness. They are also malnourished if their energy intake exceeds their energy requirements resulting in overweight (UNICEF, 2006). Nutritional status in this thesis refers to undernutrition, which is defined by UNICEF as "the outcome of insufficient food intake and repeated infectious diseases". It includes being underweight or too short for one's age, being too thin for one's height as well as deficiencies in vitamins and minerals (UNICEF, 2006). Nutritional status can be measured in four main ways; through dietary intake, biochemical indices, anthropometry and physical assessment.

- 5 -

Measure	Definition			
Dietary intake				
Nutrient adequacy	Meeting the requirements for energy and all essential nutrients (Ruel, 2003). Can be measured using a variety of dietary assessment tools e.g. Food diaries, 24 hour recall or food frequency questionnaires.			
Pros:	Describes current nutritional status;			
Cons:	Short term; Subject to bias; Does not take absorption or utilisation of nutrients in the body into consideration.			
Biochemical indices				
Biochemical indices	Laboratory assessment of nutrients or other markers of nutritional status in body fluids, most commonly blood and urine (Simko et al., 1995).			
Pros:	Objective.			
Cons:	Invasive; Requires specialised equipment and expertise including refrigeration and access to laboratories; Expensive; Usually specific to individual nutrients.			
Anthropometry				
Height for age z- scores (stunting)*	Moderate to severe stunting is below minus two standard deviations (SD) from the median height for age of the reference population (UNICEF, 2014). Reflects failure to reach linear growth potential due to suboptimal health/nutritional conditions (de Onis et al., 1997).			
Weight for age z- scores (underweight)*	Moderate and severe underweight is below minus two SD from the median weight for age of the reference population (UNICEF, 2014). Reflects body mass relative to age and is influenced by both height and weight (de Onis et al., 1997).			
Weight for height z-scores (wasting)*	Moderate and severe wasting is below minus two SD from the median weight for height of reference population (UNICEF, 2014). Typically indicates a recent and severe process of weight loss (de Onis et al., 1997).			
BMI for age z- scores*	Used as a marker of thinness, especially for children and adolescents between 5 and 19 years of age (Cole et al., 2007).			
Mid Upper Arm Circumference (MUAC)*	Provides estimates of arm soft tissue and wasting (Simko et al., 1995).			
Pros:	Easy to obtain; Objective; Systematic; Reliable; Repeatable.			
Cons:	Potential for measurement error.			
Physical assessment				
Physical assessment	Observation, inspection and measurement of an individual (Simko et al., 1995).			
Pros:	Provides detailed and broad nutritional status information.			
Cons:	Requires medical expertise; Potentially invasive; Time consuming.			

Table 1.1. Methods used to measure nutritional stat	tus, their definition, pros and cons
---	--------------------------------------

*Methods employed in this study to reflect nutritional status.

Table 1.1 outlines these main categories used to define nutritional status and the pros and cons associated with each method, with five more specific sub-categories under anthropometry. More detail is included for this technique as nutritional status in this thesis refers to the anthropometric measurements of height for age, weight for age, BMI for age, weight for height and MUAC.

The nutritional status of people in sub-Saharan Africa remains poor. These countries have considerably higher rates of stunting, wasting and underweight (WHO, 2010b, Onis et al., 2000) and shorter life expectancies than people living in high income countries (Mathers et al., 2001). Rates of stunting are highest in Eastern Africa (Percentage stunted: 50.0 (95% CI: 42.3-57.9) compared to the rest of Africa, Asia and Latin America (Black et al., 2008). Combined statistics from Ghana, Senegal, Guinea Bissau, the Philippines, India, Nepal, Bangladesh and Pakistan show that children who are severely stunted (<-3 Standard Deviations (SD) below World Health Organisation (WHO) Child Growth Standards) have 4.1 (2.6-6.4) greater odds of dying and 4.6 (2.7-8.1) greater odds of dying from diarrhoea compared to children with height for age z-scores above -1 (Black et al., 2008). Malnourished children are also more likely to perform poorly academically (Alderman et al., 2006, Victora et al., 2008, Adair et al., 2013) and have lower capacity for work (Haas et al., 1996) compared to children who are not malnourished.

There is evidence that many different socio-demographic, economic and health factors contribute to the nutritional status of children under five years of age . These include household income (Yang et al., 2012), parental employment (El-Ghannam, 2003), parents' literacy (Fernandez et al., 2002) and maternal education (Kabubo-Mariara et al., 2009, Abuya et al., 2012). Vaccinations (Dancer et al., 2008) sanitation (Fink et al., 2011, Kikafunda et al., 1998), distance to a source of drinking water (Pickering and Davis, 2012) and having crowded living conditions (Yang et al., 2012) are also associated with child nutritionals status in low income countries. The type of complementary foods introduced into an infant's diet (Onyango et al., 1998, Obatolu, 2003) contributes to child growth in young children. The gender dynamics within the household and community also have an important effect on child nutritional status (Khatun et al., 2004). These impact through women's decision making power (Smith et al., 2005), the low value that is placed on women's knowledge (Awumbila, 2003) and maternal education, income and access to resources (Chilton et al., 2007).

In addition, diarrhoea (Victora et al., 1990) and other infection (Fernandez et al., 2002) impacts on nutritional status but being malnourished puts children at increased risk of getting an infection; a dangerous cycle in low income countries (Bhaskaram, 2002). Examples of these cycles are seen with HIV, tuberculosis and malaria in sub-Saharan Africa which, along with

- 7 -

diarrhoeal diseases and respiratory infections are responsible for the highest number of deaths from infectious diseases in Africa (WHO, 2004). Children with HIV are likely to be more severely malnourished (Prazuck et al., 1993) and malnourished children with HIV are more likely to die than those without HIV (Fergusson and Tomkins, 2009). Malnutrition is more common in children with tuberculosis (Karyadi et al., 2000) and malnourished children were more likely to die from tuberculosis (Zachariah et al., 2002) and malaria (Müller et al., 2003) than those that are not malnourished.

- 8 -

A conceptual framework on the determinants of malnutrition developed by UNICEF (1990) (Figure 1.1) outlines a structure for understanding how these and other political, economic and social factors come together to have an impact on malnutrition. The framework highlights that inadequate dietary intake is an important factor caused by many other factors and illustrates that this interacts with disease to cause malnutrition. Although not included in this frame work, agrobiodiversity, acting through dietary diversity, is potentially another factor impacting on access to food and dietary intake. A limitation of this framework is that it makes no reference to environmental or agricultural factors and their impact on access to adequate food, income and exposure to health risks.



Figure 1.1. UNICEF conceptual framework of the determinants of malnutrition (UNICEF, 1990)

Much research has been conducted in low income countries into potential ways of improving food and nutrient intake in order to improve heath in these communities. Accurately measuring food security and nutrient adequacy in a low income country context is an essential step towards improving health outcomes (Keenan et al., 2001) and the associated benefits on educational attainment (Alderman et al., 2006, Victora et al., 2008, Adair et al., 2013) and productivity (Haas et al., 1996). Without a relatively accurate idea of people's access to food and nutrient intake it is difficult to plan and implement interventions that could lead to improvements in nutritional status and health. One of the ways that the quality of the diet is often represented in low income countries is by estimating the diversity in an individuals diet. This is based on the idea that a more diverse diet will provide a wider variety of nutrients to a growing child.

- 9 -

1.5 Dietary diversity

Dietary diversity has been defined as "the number of different foods or food groups consumed over a given reference period" (Ruel, 2003), page 3912S). Dietary diversity is important as the more different kinds of foods an individual consumes the more likely they are to get all the nutrients they need for healthy growth, development and function (Gibson and Anderson, 2009). This is especially true in low income countries where dietary intake is based on starchy staples and the consumption of different food groups such as vitamin A rich fruit and vegetables, meat and eggs are typically low (Arimond, 2004).

Capturing measurements of dietary diversity - the number of different food groups consumed, or food variety - the total number of different foods consumed, over a defined period of time (often 24 hours), as proxies for food security, nutrient adequacy and nutritional status have become popular in low income countries (Ruel, 2003). Dietary diversity scores has been shown to be significantly positively associated with food security (Bukusuba et al., 2007), nutrient adequacy (Arimond, 2004, Ogle et al., 2001, Moursi et al., 2008, Daniels et al., 2007, Hatloy et al., 1998, Kennedy et al., 2007, Torheim et al., 2004, Ponce et al., 2006) and nutritional status in a number of studies (Arimond, 2004, Corbett et al., 1992, Steyn et al., 2006, Garg and Chadha, 2009, Nti and Lartey, 2007, Savy et al., 2005, Savy et al., 2006).

1.6 Agrobiodiversity, dietary diversity and nutritional status

Agrobiodiversity has been defined as the biological diversity on lands used for agricultural purposes (Brookfield and Stocking, 1999). This includes the diversity of plants, animals and micro-organisms at species and ecosystem levels (Cromwell et al., 1999) and covers both cultivated crops and wild plants. The term agrobiodiversity is used to mean plant agrobiodiversity in some studies (Remans et al., 2011a) and both plant and animal agrobiodiversity in others (Ekesa et al., 2008, Walingo and Ekesa, 2013). The focus of this thesis is primarily on plant agrobiodiversity with secondary analysis on a simple measure of animal agrobiodiversity.

Biodiversity has been highlighted as an important factor for sustainable agriculture (Srivastava et al., 1996). It can have a positive effect on soil health which in turn has a positive effect on agricultural output (Chivian, 2002). There have been a number of papers encouraging the agriculture and nutrition (Welch and Graham, 1999, Hawkes and Ruel, 2006) and nutrition and biodiversity fields (Johns and Sthapit, 2004, Frison et al., 2006) to work closer together.

- 10 -

Agriculture, biodiversity and nutrition are all interrelated and a number of publications have investigated how these three areas intersect.

1.6.1 Agriculture and nutrition

In Hawkes and Ruel's (Hawkes and Ruel, 2006) conceptual framework of the links between agriculture and health (Figure 1.2) agriculture is linked with health through labour, environmental change, incomegeneration and access to food, water, land and health re lated services leading to undernutrition as well as a number of other health outcomes.



^a These health conditions are not mutually exclusive – livestock-related illnesses, for example, are also occupational health risks. The list of health outcomes is not necessarily inclusive. Other health conditions are also likely to interact with agriculture, but these have not yet been identified as such in the published scientific literature.

Figure 1.2. Conceptual framework of the links between agriculture and health (Hawkes and Ruel, 2006)

Despite acknowledgement that agriculture and nutrition are linked (Lipton and Kadt, 1988), research in these two fields have run in parallel for many years (Hawkes and Ruel, 2006). In order to improve food security ("Food security exists when all people, at all times, have physical and economic access to sufficient, safe and nutritious food to meet their dietary needs and food preferences for an active and healthy life" (FAO, 1996)), much agricultural research has focused on increasing yields, attempting to fulfil the *availability* aspect of food security (Sen, 1981). While nutrition research has attempted to address the *utilization* aspect of food security. Since Sen's essay on food entitlement in 1981 focus in food security research has turned to include people's *access* to food, addressing some of the complex issues that link food production to consumption (Coates et al., 2006, Webb et al., 2006). Research into the effects of agricultural interventions on nutrition and health outcomes has increased from the 1990's onwards (Berti, 2004). Berti et al (2004) reviewed 30 agricultural interventions that measured impact on nutritional status. They found that most agricultural interventions

increased food production and two thirds of these improved nutrition outcomes in household members.

1.6.2 Agriculture and biodiversity

Except for research looking at conservation of genetic diversity for food production, research into biodiversity has also remained relatively separate from agricultural research until recently (Frison et al., 2011). Agricultural practices can have varying affects on the biodiversity of flora & fauna. These range from concerns about the negative impact of large cash crop farming on biodiversity (Srivastava et al., 1996, Green et al., 2005) to the positive effects agricultural practices can have on biodiversity when sustainable practices are employed (Thrupp, 2000). In the past three decades the concept of agrobiodiversity has been introduced which provides an opportunity for the agriculture and biodiversity research areas to work more closely together.

1.6.3 Biodiversity and nutrition

Very little data is available on the relationship between biodiversity, nutrition and health in low income countries (Frison et al., 2004, Frison et al., 2011). Discussions about the links between biodiversity and nutrition are on the increase (Frison et al., 2004, Johns and Eyzaguirre, 2007, Nakhauka, 2009) and the Food and Agriculture Organisation (FAO) recently published an expert consultation on nutrition indicators for biodiversity (UNFAO, 2010a). Wahlqvist & Specht (1998) outline the many reasons why biodiversity is so important for human health including a diverse food supply and increased resilience.

Frison et al. (2011) highlights the importance of agricultural biodiversity in the sustainable delivery of a more secure food supply. The discussion paper states that more diverse farming systems and crops are more resilient to shocks and changes in the climate (Frison et al., 2011). Additionally, biodiversity influences nutrition through ensuring the sustainable productivity of soils (Chivian, 2002).

1.6.4 Agriculture, biodiversity and nutrition

Few good quality papers have linked agricultural biodiversity with diet and nutrition outcomes. Akrofi (2010) found that the Shannon index (a measure of the number of different plant species and the abundance of these species (Pla, 2004)) of cultivated plants in home gardens was not significantly related to the household dietary diversity score in Ghana. Crop diversity was positively associated with dietary diversity in one of the two populations assessed in

- 12 -

Mexico (Dewey, 1981). Higher agrobiodiversity, as assessed by the number of crops grown by the household, was found to be linked with higher dietary diversity in a large nationally representative sample in India (Bhagowalia et al., 2012) and in two smaller cross-sectional studies in Kenya (Walingo and Ekesa, 2013, Ekesa et al., 2008), although the significance of the associations were not reported in either of the Kenyan papers.

Evidence of an association between agrobiodiversity and nutritional status is even more limited. Shack et al. (Shack et al., 1990) found no association between the number of food crops grown in household gardens and anthropometric measures in children in a cross sectional survey in Papua New Guinea. Crop diversity was significantly correlated with height for age in children in one out of the two villages studied by Dewey (Dewey, 1981) but weight for height was not associated with agrobiodiversity in either village.

A number of discussion papers have outlined how agriculture, biodiversity and nutrition are related. Johns and Sthapit (Johns and Sthapit, 2004) propose the following conceptual framework (Figure 1.3) to illustrate the complex links between biodiversity and health. This framework takes into consideration income generation and socio-cultural traditions and attempts to capture some of the complexity of the pathways between biodiversity and nutrition.





A number of key authors (Bélanger and Johns, 2008, Wahlqvist, 2003, Johns and Eyzaguirre, 2006, Frison et al., 2011) have discussed the ways in which agriculture, biodiversity and nutrition are potentially related. Wahlqvist (2003) states that "while biodiversity is essential for sustainable food diversity, it does not guarantee it" and Johns and Eyzaguirre (2006) call for empirical evidence to prove the association between biodiversity and dietary diversity and health. In much of the other literature in this area there is an underlying assumption, that higher agrobiodiversity will lead to higher dietary diversity (Frison et al., 2005, Frison et al., 2006, Johns, 2003, Deckelbaum et al., 2006, Hillocks, 2011). This assumption appears logical but perhaps does not acknowledge the complexity of the pathway from food production to consumption in Eastern sub-Saharan Africa and globally.

1.7 Food security and nutritional status in Tanzania

A high proportion of the Tanzanian population lack food security, especially those living rurally. Approximately 23% of people were categorised as moderately food secure and 48% as severely food insecure in a study in rural Iringa in Tanzania (Knueppel et al., 2010). The study identified insufficient food quality and insufficient food intake as the two main factors contributing to this food insecurity. Approximately a quarter of surveyed households in Tanzania reported often or always having problems satisfying food needs in the past year in the 2010 DHS survey (Tanzanian NBS and ICF Macro, 2011). High levels of malnutrition among children persist in Tanzania. Recent data show that 44.4% of children in Tanzania below five years of age are stunted and 16.7% are underweight (Gollogly, 2009). There is clearly a need for research into factors that could improve food security and nutritional status in Tanzania.

1.8 Foods eaten in Tanzania

Only approximately 50% of rural households in Tanzania consume three or more meals per day, compared to almost 80% in urban households (Tanzanian NBS and ICF Macro, 2011). The frequency of meals varies depending on the season with less meals being consumed in the rainy season (Kinabo et al., 2006). There has been some initial evidence that Tanzania is moving into the nutrition transition in urban (Njelekela et al., 2002, Bovet et al., 2002) and even rural areas (Keding et al., 2011). However undernutrition still remains the most important component of malnutrition in Tanzania (Abrahams et al., 2011).

The typical meal eaten in Tanzania is *ugali*, a starchy staple made into a stiff porridge. It is mostly made from maize flour but it can also be made from other flours such as cassava
(Mazengo et al., 1997, Kinabo et al., 2006). People consuming *ugali* as their main energy source rely on accompaniments for nutrients such as protein, vitamins and minerals. It is common for people to consume *ugali* with a vegetable dish consisting of green leafy vegetables, boiled or fried with onions and tomatoes (Vainio-Mattila, 2000) and sometimes beans, meat or fish (Kinabo et al., 2006). Meat and fish are not commonly consumed; 60% of households in the 2010 DHS survey for Tanzania consumed no meat in the past week and 50% consumed no fish (Tanzanian NBS, 2011). Traditional diets consumed in rural Tanzania are often not nutritionally adequate; insufficient energy, protein and micronutrient intakes are common (Ecker et al., 2010, Mazengo et al., 1997).

Maize (*Zea Maize Gramineae*) is a very important crop for those living in Tanzania. The area cultivated with maize occupies an area 4.25 times larger than cassava which has the second largest planted area (Tanzanian NBS et al., 2006). A local level study conducted in Katumba ward in Tanzania demonstrated that the local population preferred maize meals and could obtain almost 70% of their energy and over 80% of their required protein through maize meals (Mboya et al., 2011).

In addition to maize, cassava (*Manihot Aspera Crantz Euphorbiaceae*) has become an important crop for food security in Tanzania for a number of reasons. Cassava is a root crop and can be left in the ground until needed for up to three years (Romanoff and Lynam, 1992). This means that, with planning, it can be available when other crops fail (Prudencio and Al-Hassan, 1994). It can grow in poor conditions with limited input and is unusually tolerant to drought compared to other crops grown in sub-Saharan Africa (El-Sharkawy, 2004). However, it requires appropriate processing before consumption to remove cyanide and other antinutritional factors (Padmaja and Steinkraus, 1995).

The use of wild foods is also important in many rural environments across sub-Saharan Africa, including Tanzania. Harris and Mohammed (2003) found the majority of wild foods used in Northern Nigeria to be leaves used for side dishes to the main starchy staple but they also included foods not normally eaten except while waiting for harvest or in times of famine. In Mali, Nordeide et al (1996) report the consumption of wild green leafy vegetables and wild fruit in both rural and urban environments. In Tanzania, Johns et al (1996) and Vainio-Mattila (2000) report that consumption of wild species are common among the Batimi people in Ngorongoro district and the Sambaa people in the Usambara Mountains. *Mlenda (Corchonus Tritocularies Tillaceae*), is an example of a wild food which is an important source of nutrients, such as iron and calcium (Kinabo et al., 2006), in some parts of sub-Saharan Africa. These are some specific examples which illustrate the importance of wild food sources throughout Tanzania.

1.9 Outline of thesis and summary of research contributions

Chapter 2 summarises the available literature relating dietary diversity and food variety to nutritional status as well as agrobiodiversity to dietary diversity/nutritional status in low income countries. The literature review presents strong evidence that measures of dietary diversity and food variety are positively associated with nutritional status in children in low income countries, however the evidence from Eastern Africa is less comprehensive. This chapter also reveals a large gap in the literature relating agrobiodiversity to dietary diversity and nutritional status, it shows that there is very little good quality research that addresses the aim of this thesis.

- 16 -

Chapter 3 of the thesis outlines the methods for the primary data collection; the design of the methodology, the selection of the villages the study took place in and the data collection methods. The results from the primary data collection are reported in chapters 4, 5 and 6. Chapter 4 explores the associations between dietary diversity and food variety and nutritional status in the two villages as well as how additional factors such as complementary feeding and sanitation impact on nutritional status in these communities. The chapter found that neither dietary diversity nor food variety was associated with any of the measures of nutritional status in children in these villages. These results contradict much of the literature outlined in the literature review in chapter 2 highlighting that a positive association between dietary diversity and nutritional status cannot be assumed in all contexts. The chapter also contributes additional knowledge about the importance of appropriate complementary feeding and sanitation in improving nutritional status in low income countries.

Chapter 5 presents the main results of the thesis; how plant agrobiodiversity is related to dietary diversity and nutritional status in children. The results show that in the majority of cases agrobiodiversity is not associated with dietary diversity or nutritional status. It was, however, positively associated with height in Minyenye and negatively associated with BMI in Mbwei. The mixed results of this chapter suggest that simply increasing agrobiodiversity may not be an effective strategy to improve nutritional status. This is only the second piece of research that measures biodiversity in the household farms and relates these measures to both dietary diversity and nutritional status in the household children. It is therefore an important addition to the literature.

Chapter 6 presents data on animal diversity, as measured by the number of animals raised by the household, and how this relates to dietary diversity and nutritional status in household children. The results show that the more different types of animals eaten and raised in the households the poorer the nutritional status of the household children were. Additionally,

households selling animals and animal products had children with the same or worse nutritional status than households not selling these products. This contributes to the literature on how raising animals is related to nutritional status and highlights the complexity of the determinants of diet and health in low income countries.

Chapter 7 links nationally representative dietary diversity data to nutrition outcomes and dietary and nutrition data to national land cover data in Tanzania. This chapter found that dietary diversity was not significantly associated with nutritional status in children. More agricultural land cover in the surrounding area was associated with higher dietary diversity and weight for height but lower height for age. More forest land cover was associated with higher weight for height. There is little research linking land cover and nutritional outcomes and these results provide important information on how these are related. Chapter 8 discusses the overall findings of the thesis, bringing together the primary data collection and national data, and summarises the thesis's main findings. The main conclusions of the thesis are presented in chapter 9.

Due to the constraints on land use and the continuing growth in population worldwide (Godfray et al., 2010) it is important, now more than ever, to establish and attempt to quantify how a greater diversity in agrobiodiversity affects nutrition and health outcomes in a low income country context. This study provides information that is beneficial to researchers interested in health and food security and the overlaps of agricultural and health research. Research that makes links between environment and agricultural practice, food intake and health is rare. Looking at these factors simultaneously has shown the complexities in how these factors are related and suggests these relationships should not be assumed in different contexts. It provides evidence in support of broad, locally driven interventions. This thesis therefore contributes new empirical insights from Tanzania to improve understanding of the relationships between agrobiodiversity, dietary diversity and nutritional status which are important for agricultural, food security, dietary and nutritional research .

- 17 -

Chapter 2: Literature reviews of the associations between agrobiodiversity, dietary diversity and nutritional status

- 18 -

2.1 Chapter summary

Chapter 1 introduced the rationale for investigating associations among agrobiodiversity, dietary diversity and nutritional status, while this chapter summarises published literature on these associations. The chapter presents two narrative synthesis reviews which have used a systematic style approach: the first summarises literature linking dietary diversity with nutritional status in children (Part 1), while the second presents research on the relationships agrobiodiversity has with dietary diversity and nutritional status (Part 2). The main finding presented in part 1 of this chapter is that dietary diversity and food variety are associated with nutritional status in children in low income countries. Part 2 showed that there is limited good quality research on the associations between agrobiodiversity and either dietary diversity or nutritional status. The review reveals a gap in the literature on the relationship between agrobiodiversity and both dietary diversity and nutritional status that this thesis aims to fill.

2.2 Introduction

Measures of dietary diversity have been used in low income countries to reflect diet quality and nutritional status for a number of years. It is generally accepted that more diverse diets in low income country contexts are more nutritionally adequate (Ruel, 2003, Daniels et al., 2007, Hatloy et al., 1998). There is much literature showing that more diverse diets are also associated with better growth outcomes but this is not always the case. Linking agrobiodiversity to dietary diversity and nutritional status is a relatively new area of enquiry and there are limited publications showing associations between these factors. These narrative synthesis reviews summarise all the available literature linking agrobiodiversity, dietary diversity and nutritional status, as measured through anthropometry, in a low income country context.

- 19 -

2.3 Part 1: Dietary diversity and nutritional status in children

2.3.1 Methods

This review focuses on research presenting associations between dietary diversity and nutritional status, as measured by anthropometry, in children in low income countries. Interventions investigating these associations, longitudinal, case control and cross-sectional studies are included in the review.

2.3.1.1 Search strategy

In order to find the maximum number of papers using measures of dietary diversity in low income countries different search techniques were used. Firstly the following electronic databases were searched in February 2014:

- Embase Classic and Embase 1947 to 2014 March 05
- Global Health 1910 to 2014 Week 08
- Ovid Medline 1946 to February Week 4 2014
- Psychinfo 1806 to March Week 1 2014

The following search terms were used, the numbers on the right represent how many papers were identified with each search term from all databases combined.

1	diet\$divers\$.mp.	1655
2	diet\$variet\$.mp.	562
3	food divers\$.mp.	352
4	food variet\$.mp.	504
5	food group divers\$.mp.	23
6	food group variet\$.mp.	33
7	1 or 2 or 3 or 4 or 5 or 6	2797
0		440250
ō	nutritional status.mp.	110250
8 9	growth.mp.	2956808
8 9 10	growth.mp. stunt\$.mp.	110250 2956808 15608
8 9 10 11	growth.mp. stunt\$.mp. wast\$.mp.	110250 2956808 15608 321553
8 9 10 11 12	nutritional status.mp. growth.mp. stunt\$.mp. wast\$.mp. underweight.mp.	110250 2956808 15608 321553 20248

14	malnourish\$.mp.	25669
15	malnutrition.mp.	126573
16	8 or 9 or 10 or 11 or 12 or 13 or 14 or 15	3452136
17	africa\$.mp.	711961
18	low income countr\$.mp.	7773
19	developing countr\$.mp.	823802
20	17 or 18 or 19	1340313
21	7 and 16 and 20	440
22	limit 21 to English language	401
23	limit 22 to humans [valid in Embase and Ovid Medline only]	394
24	limit 23 to yr="1980 -Current"	394

- 20 -

2.3.1.2 Criteria for inclusion

Titles and abstracts were first screened to identify potentially relevant references. The full journal articles of these potentially relevant references were obtained, where possible, and these were then checked for final inclusion in the review. Papers that met the following inclusion criteria were included in the review:

- English language
- Human
- Children under the age of 20 years
- Study relates dietary diversity or food variety to nutritional status as measured by anthropometry
- Based in a low income country as defined by the International Monetary Fund (IMF)

2.3.1.3 Included references

Three hundred and ninety four references were identified through the database searches containing 341 unique references. Based on the titles and abstracts 83 papers were identified as potentially relevant. Thirteen papers were unable to be located. These have been excluded from the review. An additional 25 were excluded because they did not meet the inclusion criteria when full papers were examined. Thirty-five papers were included in the review from this process. Relevant review papers, books and reports were used to obtain two additional relevant references, these are referred to in the inclusion flowchart (figure 2.1) as hand searched references. A total of 37 references are included in the review.

Papers were summarised and details of how dietary diversity and food variety were measured are included. The associations between dietary diversity and nutritional status are summarised in Table 2.1 and food variety and nutritional status in Table 2.2. In keeping with the language used in the papers reviewed this review uses the terms 'dietary diversity (DD)' and dietary diversity score (DDS)' to mean diversity estimated from the number of food groups individuals, or households, have consumed over a particular reference period. The review uses the terms 'food variety (FV) and 'food variety score (FVS)' to mean the number of individual food items individuals, or households, have consumed over the reference period. The FVS is larger than the DDS when people consume more than one different food type within a food group. For example, if both tomatoes and onions are consumed one DD point will be added to the DDS and two FV points will be added to the FVS.





2.3.2 Results and discussion

Thirty-two studies reported associations between dietary diversity and anthropometric measures in children (Table 2.1). Twenty-five of these studies were cross-sectional, five were longitudinal studies and two were interventions. The first, non-randomised, intervention study found an improvement in dietary diversity and weight gain in the intervention arm of the study, in female children only (Kilaru et al., 2005). The second intervention study, which had a randomised cluster design, found that the improved dietary diversity of the younger children in the intervention group was not accompanied by an increase in length for age z-scores (LAZ) (Aboud et al., 2013).

Out of the five longitudinal studies, one found DDS to be higher in children who had better growth (Nti and Lartey, 2007), one found DDS to be positively associated with weight but not length gain (Alvarado et al., 2005), two found DDS to be positively associated with height or length for age z-scores (HAZ/LAZ) (Bork et al., 2012, Ma et al., 2012) and one study found that DDS was not related to the development of kwashiorkor (Lin et al., 2007). Twenty of the cross sectional studies show positive associations between dietary diversity and either LAZ, HAZ, weight for age z-scores (WAZ) or weight for length/height z-scores (WLZ/WHZ) and five found no associations (Hillbruner and Egan, 2008, Nungo et al., 2012, Sullivan et al., 2006, Tessema et al., 2013, Aboussaleh and Ahami, 2009).

The results showing positive associations between dietary diversity and nutritional status are often not entirely conclusive. Some of these 20 cross sectional studies also found no association in particular geographic areas (Dewey, 1981, Ekesa et al., 2011), in rural areas (Hatloy et al., 2000), in girls (Eckhardt et al., 2005) and in some age groups (Benefice et al., 2007, Garg and Chadha, 2009, Sawadogo et al., 2006). Additionally, within these studies dietary diversity was not always associated with both height and weight.

Seven cross- sectional studies, one longitudinal study and one non RCT intervention study reported the association between food variety and nutritional status in children in low income countries (Table 2.2). Gibson et al (2003), whose intervention focused on dietary diversification, found that food variety was not associated with HAZ, WHZ, triceps skinfold zscores or arm fat area z-score after 12 months but MUAC z-scores and arm muscle area zscores were significantly higher in the intervention group. In the longitudinal study, Bork et al (2012) found food variety to be associated with HAZ cross-sectionally but not with height gain over the six months.

Three of the cross-sectional studies found significant positive associations between food variety and nutritional status (Novotny, 1987, Onyango et al., 1998, Steyn et al., 2006).

- 23 -

Additionally, one found positive associations in urban but not rural areas (Hatloy et al., 2000), another found that food variety was positively associated with HAZ in under 23 month olds and WAZ in just 9-11 month olds (Sawadogo et al., 2006) and another found food variety to be associated with the duel burden of malnutrition (overweight mother/underweight child) (Saibul et al., 2009). There was only one cross-sectional study that found no association between food variety and nutritional status (Lamontagne et al., 1998). As with dietary diversity, the evidence suggests that food variety is associated with nutritional status in children in low income countries.

All four studies with samples sizes of over 10 000 children found significant associations between DDS and/or FVS and HAZ (Arimond, 2004), rates of stunting (Li et al., 2011, Marriott et al., 2012, Rah et al., 2010) and rates of underweight (Marriott et al., 2012). These results are supported by seven out of the eight studies with between 1000 and 10 000 participants that were carried out in different countries in Africa and Asia. All of the studies with over 10 000 participants and four other papers (Disha et al., 2012, Eckhardt et al., 2005, Steyn et al., 2006, Zongrone et al., 2012) reported on nationally representative samples. All showed significant associations between dietary diversity and anthropometric outcomes, although Eckhardt et al (2005) found associations in boys only. Studies using large, nationally representative samples are less prone to selection bias than smaller studies and these large studies provide evidence of a true association between dietary diversity and nutritional status. However, these studies were all cross-sectional so cause and effect cannot be implied.

The statistical methods used in the majority of studies compared dietary diversity scores between malnourished children and those that were not malnourished and/or presented the correlation between the diversity indices or assessed association using regression analyses. Almost all of the papers reported the significance of the statistical tests that were carried out. Regression analysis gives the amount of change expected in the dependent variable with each unit change in the independent variable (Montgomery et al., 2012). It is therefore the best statistical method to show how dietary diversity and nutritional status are related crosssectionally, ideally controlled for potential confounders. Twenty-five of the 37 studies used multivariable regression to control for confounders. That such a high number of studies employed these methods indicates the reported results are more reliable.

The type of dietary diversity score and the way it was collected did not alter the relationships seen between dietary diversity and nutritional status. Five out of the 32 studies used household dietary diversity, one of these showed no association but the other four, and the one paper reporting household food variety, found significant associations between the diversity scores and anthropometric measures. The majority of the papers reported collecting

- 24 -

dietary diversity through a 24 hour recall (19/32 dietary diversity studies and 7/9 food variety studies), the other studies used food frequency questionnaires (FFQ), seven day diet recalls, questionnaires, interviews and observation. The size and direction of the results from studies using these other methods were not different from those using 24 hour recalls. The results from this literature review show that the 24 hour recall is the most appropriate method to measure dietary diversity in the current study.

The number of food groups in the dietary diversity scores varied from 6 to 23, the majority of papers used between 7 and 12 food groups. For comparison between studies and across countries it would be useful for studies to use the same number of food groups. Nine food groups (Cereals; roots & tubers; vitamin A rich fruit & vegetables; other fruit; other vegetables; legumes & nuts; meat, poultry & fish; fats & oils; dairy; eggs) were recommended by the FAO workshop in Rome, Italy in October 2004 (FAO/WHO/IFPRI, 2004), however only one study included in this literature review used this grouping (Steyn et al., 2006).

There was, however, some variation by geographic region. There were only 15 studies conducted in Africa, 11 of these conducted in Eastern Africa (in Malawi, Kenya, Tanzania, Ethiopia, Zambia, Burundi). The one dietary diversification intervention study (Gibson et al., 2003), in Malawi, found no association effect on height or weight z-scores. The one longitudinal study (Lin et al., 2007), again in Malawi, found that dietary diversity was not associated with Kwashiorkor. Of the nine cross-sectional studies conducted in Eastern Africa, six found positive associations (Onyango et al., 1998, Corbett et al., 1992, Cordeiro et al., 2012, Disha et al., 2012, Walingo and Ekesa, 2013, Ekesa et al., 2011) and three found no associations (Nungo et al., 2012, Tessema et al., 2013, Sullivan et al., 2006) in Kenya, Ethiopia and Malawi. The evidence in this part of the world is not as convincing as the literature as a whole. This provides evidence towards the relationship between dietary diversity and nutritional status being contextual and suggests that more research into this association in Eastern Africa would be beneficial.

Assessing the combined evidence of the cross-sectional, longitudinal and intervention studies shows that dietary diversity scores and food variety scores are positively associated with growth in children in low income countries. This was the case for both dietary diversity and food variety, using the 24 hour recall to collect the data or other similar methods, using individual or household diversity scores, with a range of food group categories. That results were consistent over these different methods indicates that this is a strong and repeatable association. The association was seen with growth measures in children in a number of different countries and contexts, although the evidence was not as strong in Eastern Africa.

Looking at all the evidence, it is reasonable to hypothesise that dietary diversity and nutritional status, as measured by anthropometry, will be associated in the current study.

Table 2.1. Studies investigating the associations between dietary diversity based on number of food groups and nutritional status based on an	thropometric
measures	

Country	DDS method	Number of food groups: Food groups	Results		
Population subgroup	Individual or household DDS	used in DDS			
	Intervention/study design details				
Bangladesh	24 hr diet recall	7: Grains; legumes; fish & meat; egg;	No significant difference in DD or LAZ between the 2		
Children, 4- Individual DDS vegetables; fruit; milk. groups. DD improved intervention on vegetables; fruit; milk. groups. groups. DD improved intervention on vegetables; fruit; milk. groups. groups. DD improved intervention on vegetables; fruit; milk. groups. groups. groups. DD improved intervention on vegetables; fruit; milk. groups.	groups. DD improved in younger age group but LAZ did				
	Parenting intervention on				
Interventio n N=226, control N=237	health, nutrition, communication and play. Randomised stratified cluster design with control group.				
	15 mo follow up				
India	24 hr diet recall	7: Dairy; cereal; protein; fruit;	Percentage feeding at least 5 different food groups was		
Infants 5-11 mo	Individual DDS	vegetables; oil & fat; sugar & savoury snacks.	significantly higher in the intervention group (42%) than the control group (19%), p=0.01. Weight velocity was		
	Country Population subgroup Bangladesh Children, 4- 14 mo at recruitment Interventio n N=226, control N=237 India Infants 5-11 mo	CountryDDS methodPopulation subgroupIndividual or household DDSIntervention/study design detailsBangladesh24 hr diet recallChildren, 4- 14 mo at recruitmentIndividual DDSIntervention n N=226, control N=237Parenting intervention on play. Randomised stratified cluster design with control group.India24 hr diet recallIndiaIndividual DDSIndiaNetrolet recall play. Randomised group.IndiaStratified cluster design with control group.IndiaNutrition education	Country Population subgroupDDS methodNumber of food groups: Food groups used in DDSPopulation subgroupIndividual or household DDSused in DDSIntervention/study design detailsIntervention/study design detailsBangladesh24 hr diet recall7: Grains; legumes; fish & meat; egg; vegetables; fruit; milk.Children, 4- 14 mo at recruitmentIndividual DDSParenting intervention on Intervention n N=226, communication and control N=237Parenting intervention and edign with control group.N=226, communication and control ylay. Randomised N=237play. Randomised stratified cluster design with control group.India24 hr diet recall7: Dairy; cereal; protein; fruit; vegetables; oil & fat; sugar & savoury snacks.India24 hr diet recall7: Dairy; cereal; protein; fruit; vegetables; oil & fat; sugar & savoury snacks.		

	Interventio n N=69, control N=69	with increasing DD as 1/5 focuses, with control group (not an RCT).		mo of age in the intervention group compared to the control group (Multivariable regression).
		1 year follow up		
Longitudinal st	tudies			
(Bork et al., 2012)*	Senegal Children, 6- 36 mo N=1060	List based FFQ, 24 hours. Individual DDS 2 visits, 6 mo apart	7: Animal milk products; animal-based foods; cereals & tubers; pulses & nuts; fruit & vegetables; vitamin A–rich foods; food with fat added.	DDS significantly positively associated with HAZ at 6-12 mo, 12-18 mo and 12-18 mo. DDS not associated with length/height increments (mixed models with adjustments).
(Ma et al., 2012)	China Children 5-7 mo N=180	7 day diet recall Individual DDS 3 visits every 6 mo over 12 mo	8: Starchy staples; legumes & nuts; dairy; meat, poultry & liver/organ meats; fish, shrimp & crabs; eggs; fruits; vegetables.	DDS at visit 1 was not significantly associated with LAZ, WAZ or WLZ at visit 3. DDS at visit 3 was significantly associated with LAZ at visit 3 (regression coefficient: 0.156, p=0.036) (Multivariable regression).
(Lin et al., 2007)	Malawi Children, 1– 3yrs N=1651	2 mo FFQ Individual DDS Followed for 10 wks	7: Starchy staples; legumes; dairy; meat, fish & eggs; vitamin A rich foods; other fruits & vegetables; foods rich in fats. (minimum portion size: 1 serving)	DDS was not associated with the development of Kwashiorkor over the 10 weeks (Multivariable regression).
(Nti and Lartey, 2007)	Ghana Children, 6- 12 mo	Monthly behavioural observation visit	10: Cereal & cereal products; roots, tubers & plantain; meat products; fish & seafood; eggs; milk & milk products; margarine; legumes, nuts & pulses; fruits;	Positive deviant children (growth above the norm) had significantly higher DDS (mean (SD): 6.3 (0.6)) than negative deviant children (growth below the norm) (3.7

	N=100	Individual DDS	soups & stews.	(1.1)) (p=0.001, independent t-test).
		Over6 mo		
(Alvarado et	Columbia	ia Week FFQ	21: intake frequency of food/beverages in previous week. Score represents both food frequency and food diversity.	DDS positively related to weight gain in non-breastfed
al., 2005)	Afro-	Individual DDS		children (regression coefficient: 0.14kg/mo, p=0.03). No significant association with length gain. (multivariable regression)
	Columbian Children 5-7 mo at baseline	Followed for 18 mo		
	N=133			
Cross sectio	nal surveys and ca	ase control studies		
(Tessema	Ethiopia	24 hr diet recall	Notstated	No significant difference in odds of being stunted
et al., 2013)	Children 0-23 mo	Individual DDS		between those with DDS<4 and those with DDS>4. Crude Odds Ratio (OR) (95% Confidence Intervals (CI)): 0.90 (0.53, 1.3) (Multivariable regression)
	N=575			
(Walingo	Western Kenya	24 hr diet recall	23 food items within 8 food groups:	DDS was positively associated with stunting (r2 = 0.036,
and Ekesa, 2013)	Youngest pre- school child, 12-60 mo	Individual DDS	Cereals; roots & tubers; pulses & nuts; vegetables; fruits & vegetables; meat, fish & meat products; milk & milk products; fats & oils.	significance not reported) and significantly associated with wasting (r2 = 0.081).
	N=164		Based on only the first 23 healthy foods consumed over a period of 2 mo, established using food checklists.	

(Cordeiro et al., 2012)	Tanzania Adolescents 10-19 yrs. N=670	24 hr diet recall Household DDS	12: Cereals; fish and seafood; roots and tubers; pulses, legumes, nuts; vegetables; milk and milk products; fruits; oil, fats; meat, poultry, offal; sugar, honey; eggs; miscellaneous.	Each additional food group consumed at the household level decreased the odds of an adolescent being undernourished by 14% (OR = 0.86 (95%CI: 0.74, 0.99); P < 0.05) (multivariable regression). Undernutrition defined as BMI < 5 th percentile.
(Disha et al., 2012)	Ethiopia and Zambia Children, 0-23 mo N=4322	24 hr diet recall Individual DDS DHS	7: grains, roots, tubers; legumes & nuts; dairy products; flesh foods; eggs; vitamin A-rich fruits & vegetables; other fruits & vegetables.	DDS was associated with HAZ (OLS regression coefficient: 0.23, p<0.001) and WAZ (0.17, p<0.001) in Ethiopia and with HAZ (0.12, p<0.01) in Zambia. DDS was not associated with WHZ in either country. (Multivariable regression)
(Marriott et al., 2012)	14 low income countries Children 6-24 mo Weighted N=79 423	24 hr diet recall Individual DDS DHS	6: Grains, roots, tubers; legumes & nuts; dairy products; meat & eggs; vitamin A- rich fruits & vegetables; other fruits & vegetables.	DDS was significantly associated with stunting (OR: 0.79 (0.72, 0.86), p<0.001) and underweight (0.78 (0.71, 0.86), p<0.001) (Multivariable regression).
(Nungo et al., 2012)	Kenya Children under 5 yrs N=232	24 hr diet recall Household DDS	12: Not stated	There was no significance correlation between DDS and nutrition status.
(Paudel et al., 2012)	Nepal Children 6-60	Not stated Not stated	Not stated	DD below the WHO standard was associated with stunting. Unadjusted OR: 7.28 (4.09-12.94), adjusted OR: 4.06 (1.70, 9.67) (Multivariable regression).

	mo	Case control study		
	Cases N=118, Control N=236			
(Zongrone	Bangladesh	24 hr diet recall	6: Grains, roots & tubers; legumes &	DDS was significantly associated with HAZ (0.08, p=
et al. <i>,</i> 2012)	Youngest	Individual DDS	nuts; dairy products; flesh foods & eggs; vitamin-A rich fruits & vegetables: other	0.006) and WAZ (0.04, p= 0.045) (Multivariable regression).
,	children	DHS	fruits & vegetables.	
	N=2096			
(Ekesa et al., 2011)	Burundi and Democratic Republic of Congo (DRC)	24 hr diet recall (validated by FFQ) Household DDS	12: Cereals; root, tubers, bananas; pulses & legumes; milk & milk products; eggs; meat & offal; fish & sea foods; oil & fats; sugar &honey fruits; vegetables; spices & condiments.	There were no significant relationships between DDS and HAZ, WHZ or WAZ in DRC. DD was significantly related to HAZ and WAZ in Burundi but the coefficients of determination were very small (r2 = 0.051 and 0.030 respectively).
(Li et al.,	China	3 consecutive days of	13: Rice and products (0.5); wheat &	Children with normal weight and height had
2011)	Children, 2-17 yrs	24 hr diet recalls	products (0.5); corn, coarse grains &	significantly higher DDS (Mean (Standard Error (SE)): 4.18 (0.01)) than those who were stunted (3.77 (0.03))
		Individual DDS	(0.5); red meat & products (0.5); poultry	and those who were stunted and overweight (3.75
	N=13 770	China National Nutrition and Health Survey	& game (0.5); egg (0.5); fish & shellfish (0.5); legumes & products (1.0); milk & dairy products (1.0); dark-coloured vegetables (1.5); light-coloured vegetables (1.0); fruit (1.5).	(0.14)) and significantly lower DDS than those who were overweight (4.53 (0.05)) (p<0.05 for all associations) (Multivariable regression).
			Weighting in brackets for a total of 10. (25g minimum portion size)	
(Niranjala	Sri Lanka	24 hr diet recall	18: Not stated	50.5% of those with DD <5 had BMI below the 5 th

and	Females 13 to	Individual DDS		percentile for age compared to 36.4% of those with
ena, 2011)	N=525			(μ-0.01).
(Rah et al., 2010)	Bangladesh	Interview	9: Rice; lentils; green leafy vegetables; yellow/orange fruits; eggs; fish; chicken;	After adjusting for all potential confounders high DDS was associated with reduced odds of being stunted
	mo	National Surveillance	meat other than chicken; milk.	among children aged 6–11 mo (OR (95% CI): 0.85 (0.76– 0.94), 12–23 mo (0.74 (0.69–0.79) and 24–59 mo (0.69
	N=165 111	Project	groups were consumed in the previous week.	(0.66–0.73) (Multivariable regression).
(Aboussale	Morocco	7 day FFQ	12: Meat; poultry; fish; legumes; green &	Mean (SD) DDS was significantly higher in stunted
h and Ahami, 2009)	School aged children	Individual DDS	other vegetables; fruits that are a source of vitamin C; other fruits; cereals & derivatives; dairy products; fats; sweets & sweetened tea.	(8.06±0.96) than in non-stunted (7.75±1.08) children when fats, sweets and sweetened tea was excluded (p=0.03). This association disappeared when area of residence or parent education were controlled for. No significant relationship was seen with wasting (t-test).
2005)	N=263			
(Garg and	India	24 hr diet recall	6: Cereals, grains & tubers; pulses; milk;	DD of infants 9-12 months had significant associations
Chadha, 2009)	Infants, 6-12 mo	Individual DDS	green leaty vegetables & vitamin A rich fruits; eggs; other.	(p<0.01) with WAZ, LAZ and WLZ. No associations in the 6-8 month olds were seen (Multivariate regression).
	N=151			
(Zhang et	China	24 hr diet recall	6: Cereals & tubers; beans; animal milk;	No association was seen between DDS and LAZ. DDS
al., 2009)	Children 6-11	Individual DDS	egg; meat & fish; other foods.	was significantly positively associated with WAZ (p=0.026) and WLZ (p=0.017) (Multivariable regression).
	mo.		1-3 FG: 1 point	
	N=501		4-6 FG: 2 points	

(Hillbru	iner Bangladesh	24 hr diet recall	9: Cereals; pulses; vegetables; fruit;	No significant association between DD and wasting (OR:
and Ega	n, Children, 6-72	Individual DDS	meat; fish; dairy; eggs; miscellaneous.	1.2: 0.8-1.8) or not achieving expected growth (OR:
2008)	mo			0.95: 0.65-1.39) (logistic fixed-effects modelling).
	N=555			
(Benef	ce Bolivia	24 hr diet recall and	7: Fish(4); cereals(3); tubers (1); plantain	No difference in DDS in normal weight and overweight
et al.,	Children, 0-15	questionnaire on the	(1); fruits (2); meat & milk (4).	groups of children.
2007)	yrs	frequency of foods	Weighted scores in brackets.	DDS associated with HAZ in 0-5 yrs only (regression
	N=452	and weekly basis		coefficients): 0-5 yr olds: (0.07, p=0.05), 5-10 yr olds:
				(0.04, ns), >10 yr olds: (0.03, ns) (multivariable
		Household DDS		regression).
(Sawad	ogo Burkina Faso	24hr diet recalls	8: Cereals; roots & tubers; nuts & pulses;	DDS was significantly positively associated with HAZ in
et al.,	Children, 6-35	Individual DDS	fruits & vegetables; meat & fish; eggs;	6-11 (p=0.002) and 12-23 (p=0.0003) mo olds, this
2006)*	mo		mink & dairy products; fats.	association was not significantly associated with WA7 at any age
	N=2466			(Multivariable regression).
/ <u>(</u> ()+0)/()	at Couth Africa	24 br diatra call	Or Coroaler roots & tuborer vitamin A rich	DDC was significant correlated with UA7 r=0.10
al 200	s) *	24 nr diet recail	9: Cereals; roots & tubers; vitamin A rich	(p<0.0001) WAZ r=0.21 ($p<0.0001$) and WHZ r=0.1
ai., 200	Children, 1-8	Individual DDS	vegetables: legumes & nuts: meat.	(p<0.0001), where -0.21 , $(p<0.0001)$ and where -0.1 , $(p<0.0001)$.
	yrs	National Food	poultry & fish; fats & oils; dairy; eggs	
	N=2200	Consumption Study	(tea, sugar & sweets not included)	
(Sulliva	n et Malawi	FFQ	7: Starchy staples; legumes; dairy; meat,	DDS was not significantly different in children with
al. <i>,</i> 2006)	⁶⁾ Children.	Individual DDS	fish & eggs; vitamin A-rich foods; other	Kwashiorkor (Mean (SD)): (5.02 (1.10)) compared to
	under 5 yrs	Case control study	fruits & vegetables; foods rich in fats.	children with Marasmus (5.06 (0.99)), which acted as a
	Case N=145.	Case control study	(minimum portion size: 1 serving/day)	control group.

	control N=46			
(Eckhardt et al., 2005)	Philippines Children, repeated measures at 8.5, 11.5, 15.5, & 18.5 yrs N=2029	Usual intake questionnaire Individual DDS Cebu Longitudinal Health and Nutrition Study	8: Fish; animal source foods; staple cereals; other starches; vegetables; fruits; beans & nuts; dairy.	DDS significantly associated with height in boys (regression coefficients (SE) 0.33 (0.06), p<0.05) but not girls (-0.01 (0.05), NS) (Multivariable regression).
(Arimond, 2004)	11 across Africa, S/SE Asia and Latin America/Carib bean Children, 6-23 mo N=22 065	7 day diet recall Individual DDS DHS	7: Starchy staples; legumes; dairy; meat, poultry, fish & eggs, vitamin A rich fruit & vegetables; other fruit & vegetables; foods made with oil, fat or butter. Included in DDS if consumed on 3 or more of the past 7 days	Significant association between DDS and HAZ was found in 9/11 countries. This association was seen in 7/11 countries when SES was controlled for (multivariable regression).
(Hatloy et al., 2000)*	Mali Children, 6-59 mo N=2315	24 hr FFQ Household DDS	10: Staples; vegetables; oil & sugar; fruit; nuts & pulses; Meat, milk; fish; leaves & gathered foods; eggs.	Compared to those with the highest DDS, children from households with the lowest DDS had increased risk for being underweight (OR (95% CI): 2.4 (1.3-4.6) or stunted (2.2 (1.1, 4.2) in urban but not rural areas. DD was not related to the prevalence of wasting in either urban or rural areas (multivariable regression).
(Corbett et al., 1992)	Kenya Children, 5-24	Questionnaire on consumption over	7 : Standard diet = any combination of	HAZ was significantly greater in those who reported the consumption of 'non-standard' food during the

	yrs the past 7 days maize,	maize, pulse, vegetables & milk.	previous week. No significant association was seen with	
	N=362	Individual DDS	Non standard diet = additional elements such as animal protein (excluding milk) rice or bread.	WHZ and skinfold thickness (ANOVA).
(Dewey,	Mexico	2 24 hr diet recalls	Number of food groups not stated: DDS	DD significantly correlated with WHZ (r=0.36, p<0.05)
1981)	Children, 2-4 yrs N=149	Individual DDS	was calculated using an adaptation of a commonly used index of species diversity (Shannon index) using calories contributed by food category/total calories consumed.	but not HAZ in one village. No significant associations were seen in the other village.

*used both food groups and food items and is reported in table 2.1 and 2.2.

Mo: Months; Hrs: Hours; Yrs: years; Wk: Weeks; DD: Dietary Diversity; DDS: Dietary Diversity Score; LAZ: Length for age z-score; HAZ: Height for age z-score; WAZ: Weight for age z-score; WHZ: Weight for height z-score; BMI: Body mass index; FFQ: Food frequency questionnaire; SD: Standard deviation; SE; Standard error; CI: Confidence in terval; OR: Odds ratio; S/SE; South/South East; DRC: Democratic Republic of Congo; WHO; World Health Organisation.

Table 2.2. Studies investigating the association between food variety based on number of food items and nutritional status based on anthropometric measures

Author and year	Country	Food variety method	riety method Number of individual food	Results	
	Population subgroup	Individual or household food variety score	items (Mean (SD))		
		Intervention/study design details			
Interventi	ion studies				
(Gibson et al., 2003)	Malawi Stunted children, 30-90 mo Interventio n N=200, control N=81	2 validated interactive 24 hr diet recall Individual FVS Community based dietary diversification /modification intervention. Quasi- experimental design with a non equivalent control group (not RCT).	Mean (1 st , 3 rd quartile) FVS Intervention: 6 (5,7) Control: 5 (4,7)	No significant difference between 2 groups after 12 months in HAZ; WHZ; triceps skinfold z-scores; arm fat area z-score. MUAC z-score and arm muscle area z-score were significantly higher in intervention group (p<0.001). (Multivariable regression).	
Longitudi	nal studies				
(Bork et	Senegal	List based FFQ, 24 hrs.	The FVS included 20 individual	FVS was significantly positively associated with HAZ at 6-12 mo, 12-	
al. <i>,</i> 2012)*	Children 6- 36 mo	Children 6- Individual FVS 36 mo 2 vicits 6 mo apart	foods or food groups (fresh milk, powered milk, sour milk, fresh fish, dried or smoked fish	18 mo and 12-18 mo. FVS not associated with length/height increments (Multivariable regression).	
	N=1060		eggs, meat, organ meats, chicken, groundnuts, other		

			legumes, vegetables, leaves, fruit, vitamin A–containing food, tubers, roots, millet gruel, milk-based millet gruel, millet couscous, millet porridge, rice, fat-containing foods, bread, biscuits) & "other foods".	
Cross secti	onal surveys			
(Saibul et	Malaysia	3 24hr diet recalls	Children: 6.9(1.9)	Duel burden of malnutrition (overweight mother/underweight
al., 2009)	Children, 2-	Individual FV		child) was significantly associated (Adjusted odds ratio and 95%
	9 yrs			regression).
	N=284			
(Sawadog	Burkina	24 hr diet recalls	Total: 16 items	FVS was significantly positively associated with HAZ in 6-11
o et al., 2006)*	Faso	Individual FVS		(p=0.0001) and 12-23 (p=<0.0001) mo olds, this association was not significant in 24-35 mo olds. EV was significantly negatively
2000)	Children, 6-			associated with WAZ in 6-11 (p=0.01) mo olds but was not
				significantly associated with WAZ at age 12-23 and 24-35 month
	N=2466			olds (multivariable regression).
(Steyn et	South	24hr diet recall	5.5(2.5)	Significant correlation between FVS and HAZ ($r=0.21$, $p<0.0001$),
ar., 2006)*	AIrica	Individual FVS		waz (r=0.23, p<0.0001) and wHz (r=0.1, p<0.0001).
20007	Children, 1-	National Food		
	0 yis NI-2200	Consumption Study		
	IN=2200			

(Hatloy et al., 2000)*	Mali Children, 6- 59 mo N=2315	24hr FFQ Household FVS	Urban: 19.6 (6.1) Rural: 14.3 (5.2)	Compared to those with the highest FVS, children from households with the lowest FVS had increased risk for being underweight (OR (95% CI): 2.3 (1.3-4.0) or stunted (1.7 (1.0, 3.1) in urban but not rural areas. FVS was not related to the prevalence of wasting in either urban or rural areas (Multivariable regression).
(Lamonta gne et al., 1998)	Nicaragua Children, 12-18 mo N=80	Observation and recall combined to give 24hrs of intake Individual FVS	Not reported	No significant correlations between FVS and HAZ, WAZ, WHZ were found.
(Onyango et al., 1998)	Kenya Children, 12-36 mo N=154	3 24 hr diet recall Individual FVS	Mean (SE): 6.0 (0.2)	FVS was significantly and positively associated with WAZ (regression coefficient: 0.19, p=0.001), HAZ (0.17, p=0.008), WHZ (0.12, p=0.01), triceps skinfolds (0.24, p=0.05) & MUAC (0.17, p=0.006) (Multivariable regression).
(Novotny, 1987)	Ecuador Children 12 mo - 5 yrs N=146	24 hr diet recall Individual FVS	7 (3) Range 2-16	FVS significantly associated with HAZ (regression coefficient): 0.159, p<0.05) and WAZ (0.232, p<0.05) (Multivariable regression).

*used both food groups and food items and is reported in table 2.1 and 2.2.

SD: Standard deviation; SE; Standard error; CI: Confidence interval; OR: Odds ratio; Mo: Months; Hrs: Hours; Yrs: years; FFQ: Food frequency questionnaire; FV: Food variety; FVS: Food variety score; HAZ: Height for age z-score; WAZ: Weight for age z-score; WHZ: Weight for height z-score; MUAC: Mid upper arm circumference.

2.4 Part 2: Agrobiodiversity and dietary diversity/nutritional status

2.4.1 Methods

This second review focuses on research presenting associations between plant or plant and animal agrobiodiversity and either dietary diversity or nutritional status, as measured by anthropometry, in low income countries. Due to the limited amount of research available in this area research in both children and adults is included. Interventions investigating these associations, longitudinal, case control and cross-sectional studies are included.

2.4.1.1 Search strategy

The same methodology was followed for part 2 of this chapter. The following electronic databases were searched in February 2014:

- Embase Classic and Embase 1947 to 2014 February 14
- Global Health 1973 to 2014 Week 06
- Ovid medline 1946 to February week 1 2014
- Psychinfo 1806 to February Week 2 2014

The following search terms were used. The numbers on the right represent how many papers were identified with each search term from all databases combined.

1	diet\$divers\$.mp.	1635
2	diet\$variet\$.mp.	557
3	food divers\$.mp.	350
4	food variet\$.mp.	500
5	food group divers\$.mp.	23
6	food group variet\$.mp.	33
7	1 or 2 or 3 or 4 or 5 or 6	2769
8	nutritional status.mp.	107066
9	growth.mp.	2884272
10	stunt\$.mp.	14925
11	wast\$.mp.	313836
12	underweight\$.mp.	19545
13	undernutrition.mp.	16079

14	malnourish\$.mp.	24195
15	malnutrition.mp.	116572
16	8 or 9 or 10 or 11 or 12 or 13 or 14 or 15	3360729
17	7 or 16	3362592
18	agrobiodiversity.mp.	93
19	biodiversity.mp.	48164
20	18 or 19	48194
21	17 and 20	7031

- 40 -

22	africa\$.mp.	651952
23	low income countr\$.mp.	7651
24	developing countr\$.mp.	705500
25	22 or 23 or 24	1209473
26	21 and 25	568
27	limit 26 to English language	546
28	limit 27 to humans [valid in Embase and Ovid Medline only]	317

2.4.1.2 Criteria for inclusion

Titles and abstracts were first screened to identify potentially relevant references. Where possible, the full journal articles of these potentially relevant references were obtained and these were checked for final inclusion in the review. Papers that met the following inclusion criteria were included in the review:

- **English language** •
- Human

- Measures plant agrobiodiversity or number of crops grown
- Study relates agrobiodiversity to dietary diversity or measure of nutritional status •
- Based in a low income country as defined by the IMF •

2.4.1.3 Included references

Database searches identified 317 references of which 295 were unique (figure 2.2). Based on the titles and abstracts 32 papers were identified as potentially relevant. Four papers were unable to be found and have been excluded from the review (authors were emailed where contact details could be found). On review of the full journal articles 25 were excluded as they did not meet the inclusion criteria. Three papers were included in the review from those found through database searches. An additional four hand-searched references were identified through Google scholar and the reference lists of included papers, relevant review papers, books and reports. A total of seven references are included in the review.

Papers were summarised and research relating agrobiodiversity to dietary diversity or food variety are presented in Table 2.3, those relating agrobiodiversity to nutritional status are presented in Table 2.4. Agrobiodiversity is used to reflect the diversity of plants, the diversity or crops grown or the diversity of both crops grown and animals raised depending on the different definitions used by the researchers. These different definitions are outlined in the tables.



Figure 2.2. Flowchart of inclusion for part 2: Agrobiodiversity and dietary diversity/nutritional status

2.4.2 Results and discussion

All six references presented in table 2.3 were cross-sectional surveys assessing associations between agrobiodiversity and dietary diversity at one time point. Four studies defined agrobiodiversity as the diversity of plants or crops and two based their definitions of agrobiodiversity on both plant foods and animals. Four of the six studies showed some positive association between agrobiodiversity and dietary diversity (Walingo and Ekesa, 2013, Dewey, 1981, Bhagowalia et al., 2012, Ekesa et al., 2008). Of these studies, Walingo and Ekesa (2013) were the only ones to use a biodiversity index of plants and animals and showed that agrobiodiversity was positively associated with child dietary diversity (coefficient of determination: 0.496). Despite using a biodiversity index, the methods on how the agrobiodiversity data was collected and what this index reflects was not reported. Additionally, the quality of this study was lowered by the statistical methods used and the omission of information on statistical significance. Ideally regression analysis adjusting for potential confounders would have been used, with associated p-values or confidence intervals reported.

- 42 -

Ekesa et al (2008) also used a count of both food crops grown and animals reared in their definition of agrobiodiversity. A positive correlation between agrobiodiversity and dietary diversity was reported but no indication of whether this was significant was given (Ekesa et al., 2008). Dewey (1981) found the farm crop diversity (details not presented) to be significantly correlated with child dietary diversity in one out of two areas studied. Again details of how crop diversity was measured were not reported making it difficult to understand the associations reported.

The final paper showing a positive association between agrobiodiversity and dietary diversity use a count of crops grown as reported by a household member to represent agrobiodiversity. A count of crops grown over a reference period may be a useful tool in determining food availability but it is not agrobiodiversity which is defined as the biological diversity on lands used for agricultural purposes; it includes the diversity of plants, animals and micro-organisms at species and ecosystem levels and covers both cultivated crops and wild plants (Brookfield and Stocking, 1999, Cromwell et al., 1999). Never the less, a large, nationally representative survey in India found the reported number of crops grown to be significantly associated with the number of food groups the household had consumed in the previous 30 days (Bhagowalia et al., 2012). Regression analysis was used but no potential confounders were adjusted for so the associations seen could be affected by the socio-economic status of the household, the education of the household members or a number of other factors. As this was a large, nationally representative sample using mainly appropriate statistical methods this paper provides important evidence on the association between the number of crops the household

grows and dietary diversity. As a result of weak methodology and statistical methods, the overall evidence of an association between agrobiodiversity and dietary diversity provided by these four references is poor.

Two papers found no association between agrobiodiversity and dietary diversity (Remans et al., 2011a, Akrofi et al., 2010). Akrofi et al (2010) found no significant correlation between home garden Shannon biodiversity indices and household dietary diversity. The details reported on the methods used to assess agrobiodiversity were limited and agrobiodiversity was not measured in the household farms. Remans et al (2011a) found no significant correlation between nutritional functional diversity of the farm (the diversity of nutrients provided by the farm) and household dietary diversity. Although the methodology of this paper was high quality and well reported, this study measured something slightly different from agrobiodiversity. It provides evidence that suggests there is no association between agrobiodiversity and dietary diversity but does not address the question directly. The six studies linking agrobiodiversity to dietary diversity are collectively inconclusive.

Two studies employed a cross sectional design to link plant agrobiodiversity with nutritional status (Table 2.4) (Dewey, 1981, Shack et al., 1990). The first found no association between the number of food crops grown in household gardens and children's nutritional status using multivariable regression (Shack et al., 1990). The study did, however, find a significant association between agrobiodiversity and the mothers BMI. Agrobiodiversity was recorded directly by a researcher but it appears that it is a count of food crops grown in the household gardens, rather than a method capturing the diversity of plants grown in the garden. The second found a significant correlation between crop diversity and height for age in children in one of the two villages studied. Interestingly it was not the village that had a significant correlation between agrobiodiversity and dietary diversity (Dewey, 1981).

In addition to the cross sectional studies presented in table 2.3 and 2.4 Talukder et al (2010) reported on a Helen Keller International intervention to improve vegetable gardens in Bangladesh, Cambodia, Nepal and the Philippines. It compared the frequency of consumption of vegetables in households with no garden, traditional (seasonal with a few traditional fruits and vegetables), improved (produce more varieties of fruits and vegetables) and developed (produce a wide variety of fruits and vegetables that are available year round) vegetable gardens after the intervention. Children from households with developed gardens consumed vegetables 5/week compared to 4/week in households with improved gardens and 3/week in those with traditional or no vegetable gardens (significance not stated). There was no significant difference in anaemia prevalence between the program and the control

- 43 -

communities. Similar results were seen in the pilot study of this project published in Talukder et al (2000).

- 44 -

Land cover has previously been used as a proxy for biodiversity (Skidmore et al., 2003, Walker et al., 1992, Tucker and Sellers, 1986), details of this are presented in section 7.2, chapter 7. Two papers were identified which looked at land cover and dietary diversity or nutritional status. Johnson et al (2013) presented the Vegetation Continuous Fields (VCF) and the Normalized Difference Vegetation Index (NDVI) measured using satellites in Malawi. They found that communities experiencing loss in forest cover over the previous ten years, based on these two indices, were 19% less likely to have a diverse diet (OR: 0.813, p=0.05) but were no more likely to be stunted (OR: 1.113, p=0.64) than children living in areas with no change in forest cover. The odds of having poor dietary diversity and of being stunted did not vary depending on the current percentage forest cover. Tree cover and NDVI were measured in the East-Usambara mountains in Tanzania using Landsat eTM+ satellite images by Powell et al. (2011). This data was not directly related to dietary diversity in mothers and their youngest child between the ages of two and five years. They did, however, report that households with greater nearby tree cover were more likely to consume forest foods and individuals using forest foods had higher dietary diversity.

Overall there is very little published data on the association between agrobiodiversity and dietary diversity or nutritional status. The data that is available is inconclusive and the quality of the literature is relatively poor. The methodological limitations in how agrobiodiversity was assessed and the statistical methods used to assess these associations make it difficult to compare the results of the studies and to assess the reliability of the results and conclusions of the studies. There is some evidence from a large intervention study that improvements in agrobiodiversity lead to improvements in vegetable consumption but unfortunately this was not linked to measures of dietary diversity. There is also limited data showing associations between more biodiverse land cover and dietary diversity but not child growth. This review identified four potentially relevant papers that could not be located. In addition to these it is likely that there is other relevant work on improving nutrition through agrobiodiversity that has not been published in peer reviewed journals. This review, therefore, may not cover all the research in this area. The conclusions are based on the small number of papers obtained and it is possible that this additional research would alter the conclusions of the review. Additional limitations of both this review and the dietary diversity and nutritional status review are those commonly associated with narrative synthesis reviews. These reviews included no formal quality assessment. There is a greater likelihood of bias than in a systematic review following

strict protocols. These reviews do, however, have transparent methods which would allow replication of these results.

To the students knowledge, there has been no published literature using both systematic observations of the biodiversity of household farms and reliable statistical methods to assess the association between agrobiodiversity and dietary diversity or nutritional status to date. It appears that none of the published studies have assessed the biodiversity of both the crops grown and wild plants present in the gardens or on the farms. Additionally, those studies directly observing crops grown focused on vegetable gardens rather than household farms. Due to the limited number of relevant studies and the quality of these studies, the literature to date is inconclusive. This literature review has revealed a substantial gap in the literature on how agrobiodiversity is related to dietary diversity and nutritional status that the current project aims to address. The gaps in the literature found by this literature review consolidated the decision to measure agrobiodiversity of household farms systematically rather than relying on counts of foods grown.

Author	Country	Agrobiodiversity measurement	Dietary diversity score details	Results	
and year	Population subgroup				
(Walingo	Western Kenya	Agrobiodiversity was measured by	Qualitative 24 hr diet recall.	Agrobiodiversity was positively	
and Ekesa, 2013)	Individual DDS (Youngest child, 12-60 mo)	assessing the variety of food plants grown and animals reared by households.	The score was based on only the first 23 healthy foods consumed over a period of 2 months, established using food checklists.	associated with DDS (r2 = 0.496). (significance was not reported).	
	N=164	A Shannon Index (SI) was calculated to reflect species diversity.	8 : Cereals; roots & tubers; pulses & nuts; vegetables; fruits & vegetables; meat & meat products (included fish); milk & milk products; fats &oils.		
(Bhagow	India	Number of crops grown by the	Number of food groups consumed over the	Number of crops grown by the	
alia et al., 2012)	India Human Development Survey	household, information collected through the survey.	past 30 days.	household is significantly associated with dietary diversity (regression coefficient: 0.32 (p<0.01). Associations remain significant when farms are broken down into their different sizes ranging from <0.5 hectares to >5 hectares.	
	Household DDS				
	N=19 000				
(Remans	Malawi, Kenya and	All crop, plant and tree species on	24 hr diet recall.	Correlations between nutritional	
et al.,	Uganda	the farms and vegetable gardens	15: Cereals; vitamin A rich vegetables &	functional diversity and household dietary diversity were not significant.	
20110)	Household DDS	documented. Only plants that were	tubers; white tubers, roots & plantains; green leafy vegetables; other vegetables;		

- 46 -

Table 2.3. Studies investigating the associations between agrobiodiversity and dietary diversity

	N=170	edible and consumed in the village were considered for this study.	vitamin A rich fruits; other fruits; legumes & nuts; oils & fat; meat; fish; eggs; milk; sweets; spices & tea.	
		Species richness was defined by the number of identified edible species per farm. Nutritional functional diversity summarises the diversity of nutrients provided by the farm. It was calculated based on farm species composition and species nutritional composition using 17 nutrients from 77 crops.		
(Akrofi et	Ghana	An inventory of the cultivated plant species was compiled for home gardens. The number of individual plants of each species recorded. Shannon index (SI) was calculated based on these cultivated species.	Household DDS as food groups based on 24	There was no significant correlation between the home garden SI and the household DDS ($r = 0.17$, $p = 0.14$).
al., 2010)	HIV positive and HIV		14: Cereals; vitamin A-rich vegetables & tubers; white tubers & roots; dark green leafy vegetables; other vegetables; vitamin A-rich fruits; other fruits; organ meat (iron rich); flesh meats; eggs; fish; legumes, nuts & seeds; milk & milk products; oils, fats & red palm products.	
	N-80			
	Plant species were categ human food, medicine a feed. Livestock reared in was also recorded.	Plant species were categorised as human food, medicine and animal feed. Livestock reared in the gardens was also recorded.		
(Ekesa et	Eastern Kenya	Variety of food crops grown, animals	7 day food variety score.	The number of animals kept and plants grown was correlated with children's DDS (r=0.697) (significance
al., 2008)	Individual DDS	reared for food and wild food items collected through gathering, hunting	7: Breads & cereals; roots & tubers; pulses & nuts; vegetables & fruits; meat & meat	
	(Pre school children)	and trapping. Total number of food		. ,

	N=144	types, plant and animal combined, were used.	products (included fish); milk & milk products; fats & oils.	was not reported).
(Dewey,	Mexico	Crop diversity in family plots	2 24 hr diet recalls	Socios: Households with >=5 crops in
1981)*	Socios: an area of organised farming	(number of species). Further details not presented.	DDS was calculated using an adaptation of a commonly used index of species diversity (SI) using calories contributed by food category/total calories consumed.	the family plot had higher DDS. R=0.25 (p<0.05)
	Nearby village: Tecominoacan			Tecominoacan: no relationship between crop diversity and DDS.
	Individual DDS (Children, 2-4 yrs)			
	N=149			

*presented associations with both dietary diversity and nutritional status and is reported in table 2.3 and 2.4.

DDS: Dietary diversity score; SI: Shannon index; Hr: Hour; Yrs: Years.

Author	Country	Agrobiodiversity	Nutritional status	Results	
and year	Population subgroup	measurement	measurement		
(Shack et	Papua new Guinea	Number of food crops grown in the household gardens as observed by a researcher.	Weight, height, arm circumference, and triceps and subscapular skinfolds in mothers and children.	The number of different types of crops planted in the garden	
al., 1990)	Mothers and their			was not related to nutritional status in the children.	
	youngest non-breastfed child 2-6 yrs			The number of different types of crops planted was positively associated with the mothers BMI (Regression	
	N=56			coefficient 0.002, p<0.05) (Multivariable regression).	
(Dewey,	Mexico	Crop diversity in family plots (number of species). Further details not presented	Height for age and weight for height in children were reported.	Socios: no significant correlation between crop diversity and	
1981)*	Socios: an area of			HAZ or WHZ.	
	organised farming.			Tecominoacan: Crop diversity was significantly correlated to	
	Nearby village:	presented.		HAZ in children: r=0.29 (p<0.05) but not WHZ.	
	Tecominoacan.				
	Children, 2-4 yrs				
	N=149				

Table 2.4. Studies investigating the associations between agrobiodiversity and nutritional status as measured by anthropometry

*presented associations with both dietary diversity and nutritional status and is reported in table 2.3 and 2.4.

HAZ: Height for age z-scores; WHZ; Weight for height z-scores; Yrs: Years; BMI: Body mass index.

2.4 Conclusion

Current research indicates that dietary diversity and food variety are associated with nutritional status as measured using anthropometry in children in low income countries. There were a number of papers using methods that varied in the way the data was collected and presented which found dietary diversity and food variety to be related to these anthropometric measures. The results seen from studies carried out in Eastern Africa do not show such a clear cut association and further research in this area would be beneficial. The associations between agrobiodiversity and both dietary diversity and nutritional status are less clear. There is limited research in this area and, due to methodological limitations, the results seen to date are inconclusive. This literature review has revealed a substantial gap in current knowledge about the relationships between agrobiodiversity and dietary diversity and nutritional status in low income countries.
Chapter 3:

Pilot, village selection and methods for the project's primary data collection.

3.1 Chapter summary

This chapter describes the overarching methodological approach and outlines the research design employed to investigate the relationships between agrobiodiversity, dietary diversity and nutritional status in Tanzania. This chapter describes the design and application of the methodological tools used during the primary data collection phase of the project. The project described is a cross-sectional survey in two villages which collected data through household interviews, anthropometry and a systematic agrobiodiversity survey. Important aspects of the research process, including sampling design, piloting methods, ethical procedures, training of research assistants, randomisation and participant recruitment are described. Information on the two villages where this project collected data is presented. Methodological specifics, particularly the analytical approach, associated with each empirical chapter (chapters 4, 5 and 6) are covered subsequently in these chapters. The methods used to investigate the relationships between land cover, dietary diversity and nutrition, the DHS and land cover analysis phase of the project, are described in chapter 7.

3.2 Research design

This research was conducted in four phases of work (Figure 3.1). Phase 1 involved literature review, design of methodological tools and piloting of these tools to meet objective 1. Phase 2 involved selection of study site villages, primary data collection and analysis of data to investigate the relationships between agrobiodiversity, dietary diversity and nutritional stat us in rural Tanzania to fulfil objectives 2, 3 and 4. Phase 3 involved analysis of Demographic and Health Survey (DHS) data and land cover data to investigate the relationships between land cover, dietary diversity and nutritional status and fulfil objective 5. Phase 4 involved the integration of outcomes from the empirical and analytical work conducted in phases 2 and 3 to generate discussion and conclusions (chapters 8 and 9) that inform understanding of food security in sub-Saharan Africa and thus fulfil objective 6.

Phase:

Objectives:

Phase 1: Literature review & pilot

1. Review, design and pilot a suite of appropriate measurement tools to accurately measure agrobiodiversity, dietary diversity and nutrition in an Eastern Sub-Sahara Africa context.

2,3

Phase 2: Primary data collection

2. Systematically assess the diversity and abundance of both cultivated and wild plants growing on household agricultural land in two villages in rural Tanzania and investigate how this is related to dietary diversity in under five year old and nutritional status in children in these villages. 3. Measure household crop and animal diversity, dietary diversity of respondents and children under five and nutritional status of children living in these villages, determine whether household produce is sold and investigate how these factors are related.

4. Investigate basic socio-demographic factors affecting dietary intake of children under five and nutrition outcomes in children in these villages.

3, 4, 5, 6

Chapters:



Phase 3: DHS

& land cover

analysis

5. Investigate the

determinants of

dietary diversity

and nutritional

status and explore

the relationships

nutritional status

in children under

national level, in

7

between land

cover, dietary

diversity and

five years at a

Tanzania.

socio-demographic

multi-scale investigation of the relationships between agrobiodiversity, dietary diversity and nutritional status, to draw conclusions that inform understandings of food security in sub-Saharan Africa.

8,9

Phase 4:

Integration



The conceptual framework was designed using other relevant conceptual frameworks presented in section 1.4, 5.1 and 6.1 ((UNICEF, 1990, Altieri, 1999, Leroy and Frongillo, 2007) and attempts to graphically display a simple representation of how agrobiodiversity would be expected to be associated with individual dietary diversity and childhood nutritional status. The main three pathways illustrated are its potential relationship to individual dietary diversity, to household income and to production, through improvements to ecosystem functioning (Figure 3.2a)).

The impact agrobiodiversity potentially has on dietary diversity and household income through improvements in ecosystem functioning leading to increased production is beyond the scope of this study. The relationships between agrobiodiversity and dietary diversity/nutritional status and between dietary diversity and nutritional status are addressed through both the literature reviews presented in chapter 2 and the primary data collection. These relationships are also explored at a national level relating average group dietary diversity and nutritional status and land cover surrounding these geographically clustered groups.

Figure 3.2 b) illustrates the elements of the framework that the empirical study addresses. The main focus of the primary data collection are the relationships between agrobiodiversity, dietary diversity and nutritional status. Data was also collected on specific socio-demographic characteristics and whether households produce was sold in order to see if household income from these practices influenced child dietary diversity and nutritional status. As these factors are not the primary focus of the thesis, these are displayed less prominently in the framework (figure 3.2 b)).



Figure 3.2 a). Conceptual framework for the relationships between agrobiodiversity, dietary diversity and nutritional status



Figure 3.2 b). Section of the conceptual framework for the relationships between agrobiodiversity, dietary diversity and nutritional status which forms the focus of this thesis

3.3 Village selection for primary data collection

Two villages were selected in rural Tanzania for collection of primary data. Areas where the population were, on average, likely to be malnourished were selected, as explained below, to allow for factors associated with malnutrition to be more easily identified. Areas with different biophysical (eg. soil, vegetation type and topography and thus land cover) and climate characteristics are likely to have different levels of agrobiodiversity (Hadgu et al., 2009). Differing contexts allow for contrasting relationships between agrobiodiversity, diet ary diversity and nutritional status to be investigated. One site in an area of high biodiversity with two rainfall seasons and one site in an area of low biodiversity with one rainfall season were selected. The following sections outline the criteria that were used for selecting Enumeration Areas (EA), areas defined by the census, with high proportions of malnourished children in high and low biodiversity areas.

- 56 -

Areas of poor nutritional status were determined using data from the 2010 Demographic and Health Survey (DHS) obtained free from the DHS program (The DHS Program, 2014b). Age, sex, height and weight of children under the age of five years were used to calculate their individual height and weight z-scores. These scores were added to ArcGIS, a piece of software based on geographic information systems (GIS) for working with mapped data and analysing geographic information. Scores averaged by EA were added using the Global Positioning System (GPS) co-ordinates that were collected at the centre point of each EA cluster. ArcGIS was then used to select those EA clusters that had both average height and weight z-scores below negative 2. These cut offs were used to identify EA that had a high prevalence of malnutrition.

Of 474 EA included in the 2010 DHS survey, 61 had average weight z-scores below negative 2 and 36 had average height z-scores below negative 2. Twenty had both height and weight zscores below negative 2. Due to the agricultural focus of this project, rural environments were the most appropriate areas to collect data. EA that were classed as urban in the DHS survey were therefore excluded leaving 18 classed as rural and malnourished. The land cover data used, GlobCover2009, was obtained from the European Space Agency (ESA) GlobCover Portal (European Space Agency, 2010) for free. This data was added to ArcGIS, the 18 EA classified as malnourished were then added and a map generated (Figure 3.3). Five km buffers around each of these 18 EA were created and land cover classifications within these boundaries assessed to determine areas of high and low biodiversity.



GlobCover2009 land cover categories



Figure 3.3. A map of the 18 potential enumeration areas determined as malnourished in relation to land cover classification (GlobCover2009) in Tanzania. The key outlines 22 different land cover categories determined in the GlobCover2009 data

The GlobCover2009 project was able to differentiate between different types of land cover with 67.5% accuracy (Bontemps et al., 2011). Land cover classification characteristics were used as a proxy for biodiversity. Previous research has shown that satellite imagery can identify biodiverse forested areas (Skidmore et al., 2003, Walker et al., 1992, Tucker and Sellers, 1986) (Austin et al., 1996, Homer et al., 1993, Miller and Conroy, 1990). Due to the focus on agrobiodiversity in this project, sites of high biodiversity were defined as areas including both 'Croplands' and 'Forests' in the buffer surrounding the EA (Box 3.1).

Box 3.1 Land use types used to classify high biodiversity sites Croplands

The following land cover types were included in the definition of 'Croplands' in land cover assessment. These correspond to the Land Cover Classification System (LCCS) of 'cultivated terrestrial areas and managed lands':

- 1) Post-flooding or irrigated croplands
- 2) Rainfed croplands

3) Mosaic Cropland (50-70%) / Vegetation (grassland, shrubland, forest) (20-50%)

4) Mosaic Vegetation (grassland, shrubland, forest) (50-70%) / Cropland (20-50%)

Forests

The following land cover types were included in the definition of 'Forests' in land cover assessment. These match the LCCS of 'Natural and semi-natural terrestrial vegetation – Woody-Trees' and 1 of the classes of 'natural and semi-natural aquatic vegetation'.

- 1) Closed to open (>15%) broadleaved evergreen and/or semi-deciduous forest (>5m)
- 2) Closed (>40%) broadleaved deciduous forest (>5m)
- 3) Open (15-40%) broadleaved deciduous forest (>5m)
- 4) Closed (>40%) needleleaved evergreen forest (>5m)
- 5) Open (15-40%) needleleaved deciduous or evergreen forest (>5m)
- 6) Closed to open (>15%) mixed broadleaved and needleleaved forest (>5m) (100)

7) Mosaic Forest/Shrubland (50-70%) / Grassland (20-50%)

8) Mosaic Grassland (50-70%) / Forest/Shrubland (20-50%)

9) Closed (>40%) broadleaved forest regularly flooded - Fresh water

Sites of low biodiversity were defined as areas with land cover types that were not associated with high biodiversity and did not include 'Croplands' or 'Forests' (Box 3.1). Box 3.2 summarises the land use types used to classify low biodiversity sites.

Box 3.2 Land use types used to classify low biodiversity sites

Low diversity

The following land cover types were included in the definition of 'Low diversity' in land cover assessment. These correspond to the LCCS of 'Natural and semi-natural terrestrial vegetation – shrub and herbaceous' and 'artificial surfaces' and 'bare areas':

- 1) Closed to open (>15%) shrubland (<5m)
- 2)Closed to open (>15%) grassland
- 3) Sparse (>1. 5%) vegetation (woody vegetation shrubs, grassland)
- 4) Artificial surfaces and associated areas (urban areas >50%)
- 5) Bare areas
- 6) Permanent snow and ice

Two villages, representing areas of low nutrition with 1) high and 2) low biodiversity were chosen from maps that were produced for each of the 18 potential EA. These decisions were made by examining the maps to find the most diverse land cover and the least diverse land covers by comparing the proportion of land that fell into 'Cropland', 'Forest' and 'Low diversity' categories. No strict cut-offs were used in the distance from the centre point of the EA as EA vary in size depending on population size. The following criteria, obtained through the 2010 Tanzanian DHS and Google maps, were also considered to determine suitable sites: agroecological zone, how close the nearest villages and main roads were, how close protected areas were, which districts they fell in, elevation, percentage of population with agricultural land, average hectares of agricultural land, accessibility by public transport, whether they were in a protected area and wealth index.

For each site, three areas, ranked as first, second and third choices, were chosen based on their land cover and this additional information. The location of the six potential sites are shown in Figure 3.4. The land cover surrounding the six potential EA which was used to help

decide on the final two sites are shown in figures 3.5 to 3.10. The figures on the left (Figure 3.5, 3.7 and 3.8) are the high diversity sites and the figures on the right (Figure 3.6, 3.8 and 3.10) are the low diversity sites.

- 60 -



Figure 3.4. Location of top 6 potential sites, first, second and third choices for the high and low diversity sites.





Figure 3.5. Land cover around EA 62, in Tanga, Northern Tanzania, one of the

Figure 3.6. Land cover around EA 232, in Singida, Central Tanzania, one of the

three potential low diversity EA



three potential high diversity EA



Figure 3.7. Land cover around EA 376, in Manyara, Northern Tanzania, one of the three potential high diversity EA



Figure 3.9. Land cover around EA 160, in Mtwara, Southern Tanzania, one of the three potential high diversity EA

Figure 3.8. Land cover around EA 34, in Arusha, Northern Tanzania, one of the three potential low diversity EA



Figure 3.10. Land cover around EA 149, in Lindi, Southern Tanzania, one of the three potential low diversity EA

The borders of the six potential EA were obtained from the Director of AfricaScope (Africascope, 2014). These EA borders and their identifying numbers were taken to the Bureau of Statistics in Dar es Salaam, Tanzania. The GIS expert who worked on the DHS 2010 linked this information with the original paper copies of the census which defined these borders and the villages within the borders were identified. Based on the information obtained from the Bureau of Statistics and information obtained from pre data collection visits to the two villages, the first choice areas were confirmed as the data collection sites.

- 62 -

The low diversity EA in Singida covered one sub-village of a village called Minyenye which was made up of five different sub-villages. Five different EAs covered these five sub-villages. Each EA did not correspond to a sub-village and the edge of the EA would be difficult to locate on the ground. As the rest of the village was unlikely to significantly differ in nutritional status it was decided to collect data in the whole of Minyenye. Similarly for the high diversity EA in Lushoto, five EA made up Mbwei village but it was decided that data collection would take place in all of the seven sub-villages.



Figure 3.11. Map of Tanzania with Minyenye, Singida and Mbwei, Lushoto

3.4 Study site description

The low biodiversity village was Minyenye in Singida district, Singida region, located in the central plains of Tanzania (Figure 3.11). The village identified as a high biodiversity area was Mbwei in Lushoto district, Tanga region, situated in the West Usambara mountains close to the coast of Tanzania.

3.4.1 Minyenye village, Singida district, Singida region



Figure 3.12. A typical house and surrounding area in Minyenye village, Singida district (photo taken during data collection, June 2012, dry season)

Minyenye is situated in the relatively flat region of Singida, in central Tanzania (Figure 3.13). The region of Singida lies 1200m to 1500m above sea level, is semi-arid and experiences low rainfall, short rainy seasons and commonly drought (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). Singida's average rainfall is 700mm a year (Tanzanian NBS and Singida Regional Commissioner's Office, 2005) and the average maximum temperature in June is 26.5 degrees Celsius (National Bureau of Statistics, 2011). The long dry season typically lasts from April to November followed by the short rainy season from December to March (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). Data was collected in Minyenye in June of 2012; the dry season (Figure 3.12). Singida has a high average land area per capita (4.5 hectares) (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). Agriculture is the main economic activity in the region producing 60% of the areas goods and services (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). Ninety percent of the people of Singida region rely on agriculture as their main livelihood (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). The main food crops are maize, sorghum, bullrush millet, sweet potatoes and beans and the main cash crops are sunflower, groundnut, simsim, pigeon peas, onion and cotton (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). People here rely on sorghum and maize as their main staples with bullrush millet and sweet potatoes also used as starchy foods (Tanzanian NBS and Singida Regional Commissioner's Office, 2005).

The average household size in Singida region 2002 was 5.0 people (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). Thirty three percent of the households in this region were in the lowest quintile for wealth index of the country and only 6% were in the highest quintile (Tanzanian NBS and ICF Macro, 2011). In Singida region, approximately 14% of the female heads of households had no education and the median years of education was 6.3 (Tanzanian NBS and ICF Macro, 2011). Regionally, 75% of women were literate, 93% were currently employed and 83% worked in agriculture (Tanzanian NBS and ICF Macro, 2011). Nine percent of male heads of households had no education and the median years of education for men was 6.4 (Tanzanian NBS and ICF Macro, 2011). Eighty-two percent of the male heads of households were literate, 99% were currently employed and 92% were employed in agriculture (Tanzanian NBS and ICF Macro, 2011).



Figure 3.13. Google Earth image of the area surrounding Minyenye village (August 2004, dry season) illustrating flat, largely un-vegetated orange land and seasonal rivers/streams. The DHS centre point is marked by the A flag, the red outline is the border of the enumeration area from the DHS (2010)

3.4.2 Mbwei village, Lushoto district, Tanga region



Figure 3.14. The hilly landscape of Mbwei's sub-villages (on the opposite hills) with the fertile valley in view (Photo taken during data collection, July 2012, dry season)

Mbwei is in the West Usambara mountains and is situated in Lushoto district in Tanga region (Figure 3.15). The Usambara mountains rise to a maximum altitude of 2300 meters above sea level (National Bureau of Statistics, 2011). In the mountainous zone, where Mbwei lies, temperature ranges between 21 and 28 degrees and the rainfall ranges between 800 and 2000mm. In contrast to Minyenye village's single rainy season, there are two rainy seasons in Mbwei. In the North and Costal regions of Tanzania, the short rains typically fall in November to April and the long rains in March to May (Tanzanian NBS and ICF Macro, 2011). Data was collected in Mbwei in July of 2012; the dry season (Figure 3.14). In 2010, 40.9% of the agricultural land in Tanga region was planted in the short rainy season (National Bureau of Statistics, 2011). Mbwei village therefore had one additional planting season each year compared to Minyenye. The main crops grown in the mountainous zone of Tanga region are coffee, tea, cardamom, maize, round potatoes, banana, beans, spices, fruits and vegetables (Tanzanian NBS and Tanga Regional Commissioner's Office, 2008).

The average household size for Tanga region in 2002 was 4.6 people (Tanzanian NBS and Tanga Regional Commissioner's Office, 2008). Only 16% of households in this region were in the lowest quintile of wealth index nationally and 25% were in the highest wealth index (Tanzanian NBS and ICF Macro, 2011) suggesting Mbwei residents would be less poor than Minyenye residents. In Tanga region, 19.6% of female heads of households and 5.5% of male heads of households had no education (Tanzanian NBS and ICF Macro, 2011) . The median years of completed education for this region was 6.4 for both woman and men (Tanzanian NBS and ICF Macro, 2011), similar to Singida region. Seventy percent of the women were literate, 71% were currently employed and 48% worked in agriculture compared to 83% literacy in men, 82% currently employed and 59% employed in agriculture (Tanzanian NBS and ICF Macro, 2011). Mbwei's region had a lower rate of employment overall and in agriculture compared to Minyenye's.



Figure 3.15. Google Earth image of area surrounding Mbwei village (April 2007, Rainy Season). Illustrating mountainous terrain and largely green vegetated lands. The DHS centre point is marked by the A flag, the red outline is the border of the enumeration area from the DHS (2010)

3.5 Mixed method approach

During phase 2, an interview based, mixed methodology approach was used to collect primary data. Four key methods were applied; household questionnaire, 24 hour dietary recall, anthropometric measures and agrobiodiversity survey. Table 3.1 outlines what data collection

methods were used to meet each of the study's objectives and which participants were involved in these methods.

able 3.1. Data collection methods and participants involved in meeting each of the stud	y's
objectives	

Objectives		Data collection methods employed	Participants involved
1	Review, design and pilot a conceptual	Literature reviews	None
	framework and suite of appropriate measurement tools to accurately measure agrobiodiversity, dietary diversity and nutrition in an Eastern Sub-Saharan Africa context.	Quadrat sampling and point intercept method	None
		Pilot household questionnaire	Female respondent (pilot study)
		24 hour recall	Female respondent (pilot study)
			Oldest child under five (pilot study)
2	Systematically assess the diversity and abundance of both cultivated and wild plants growing on household agricultural land in two villages in rural Tanzania and investigate how this is related to dietary diversity in under five year old and nutritional status in children in these villages.	Point intercept method	None
(1)	Measure household crop and animal diversity, dietary diversity of respondents and children under five and nutritional status of children living in these villages, determine whether household produce is sold and investigate how these factors are related.	Household questionnaire	Female respondent
		24 hour recall	Female respondent
			Oldest child under five
		Anthropometric measurements of nutritional status	All children
			Female respondent
			Respondents husband
4	 Investigate basic socio-demographic factors affecting dietary intake of children under five and nutrition outcomes in children in these villages. 	Household questionnaire	Female respondent
		Observation	None
	Investigate the socio-demographic determinants of dietary diversity and nutritional status and explore the relationships between land cover, dietary diversity and nutritional status in children under five years at a national level, in Tanzania.	Statistical analyses on secondary data	None
6	Integrate outcomes from this multi- scale investigation of the relationships between agrobiodiversity, dietary diversity and nutritional status, to draw conclusions that inform understandings of food security in sub-Saharan Africa.	Synthesis and discussion of thesis's findings	Based on data collected for objective 1-4 and secondary data for objective 5

3.6 Household questionnaire

Much of the data for this research was collected using a questionnaire which consisted of a mixture of closed and open ended questions generating both quantitative and qualitative data. The majority of questions collected quantitative data. Qualitative data was coded into categories to be used in the data analysis. The questionnaire was made up of three main section: Demographic and livelihood information; Dietary questions; Food sources. A detailed description of the dietary recall, nutritional status and agrobiodiversity methods follow. Protocols for these methods are included in Appendix C.

3.6.1 Dietary recall methods

The dietary diversity data was captured through a three pass 24 hour dietary recall (see Appendix C for the protocol followed). The dietary recall was designed to capture the diversity of what the respondent, and her oldest child under the age of five years, ate the previous day. It asked about all foods, drinks and snacks the respondent has eaten for the 24 hour period from when they got up the day before the interview to when they got up the day of the interview. This was then repeated for items consumed by her oldest child under the age of five. As the diet recall was designed to capture diversity of the diets it did not collect the amount of food consumed. The 24 hour recall was conducted in three passes (Johnson et al., 1996):

- 1. A list of all foods, drinks and snacks consumed.
- 2. Details of all the foods, drinks and snacks consumed including cooking methods, ingredients in recipes and additions.
- 3. A review of all the foods, drinks and snacks consumed and final check the recall is complete.

Foods were subsequently broken down into nine food groups (1. Cereals, roots and tubers, 2. Vitamin A rich vegetables, tubers and fruit, 3. Other vegetables, 4. Other fruits, 5. Flesh meats, organ meats, fish and insects, 6. Eggs, 7. Legumes, nuts and seeds, 8. Milk and milk products, 9. Oils, fats and sweets), as recommended by the FAO dietary diversity workshop that was held in Rome, Italy in October 2004 (FAO/WHO/IFPRI, 2004). The dietary diversity score for each individual was therefore calculated out of nine. The number of different individual food items were also tallied to give a food variety score.

Dietary diversity is a quick and simple measure that is commonly used in a low income country setting. Dietary diversity has been shown to be significantly associated with both measures of

food security (Bukusuba et al., 2010, Leroy et al., 2008, Thorne-Lyman et al., 2010) and nutrient adequacy (Arimond, 2004, Ogle et al., 2001, Moursi et al., 2008). The Food and Nutrition Technical Assistance project (FANTA) has indicated household dietary diversity to be an indicator of food security (Swindale and Bilinsky, 2006) which links food security and nutrition together and should help provide a comprehensive view of food access and intake. A 24 hour recall was chosen to collect this data as it is the most common method used to collect dietary diversity data in low income countries, as shown in chapter 2. Additionally, a food frequency style questionnaire for previous day consumption of the nine food groups was tested during the pilot phase of the study and was not as well understood by the participants as the 24 hour recall.

3.6.2 Anthropometric measurements of nutritional status

Anthropometry was used to measure nutritional status of participants and their families. Heights and weights of all children in the family unit (defined as the respondent, her husband or partner and their children) were measured to calculate their height for age, weight for age and BMI for age z-scores using the 2006 WHO Child Growth Standards. MUAC (Mid Upper Arm Circumference) was measured in all children under the age of 15 as an additional measure of nutritional status. Height, weight and BMI z-scores and MUAC are used to reflect nutritional status of the children. Stunting and underweight were defined as height for age and weight for age z-scores below minus two standard deviations from the WHO international reference median values (de Onis et al., 1997).

Specific protocols (Appendix C) were followed to measure the height and weight of the respondent and her husband and all children in the family unit who lived in the household. For infants under the age of two years length was measured instead of height (de Onis et al., 1997). A considerable effort was made to get anthropometric measures for those individuals who were in the village, at school or work, but were not present at the interview. Researchers returned to the household at a more convenient time, family members came to other participating households to be measured and the researchers visited two local schools in Minyenye to get as complete anthropometric records as possible.

SECA Digital weighing scales (SECA GMBH & Co Germany, Model 881, Max Weight 150 kg, Precision 100g) were used to measure weight. Weight was measured without shoes or heavy jackets but otherwise participants remained clothed. A portable wooden infant/child/adult length board (Shorr Productions, Perspective Enterprises, Portage, Missouri) was used to measure height and length. MUAC was measured using non-stretch MUAC measuring tapes

- 70 -

provided by UNICEF. The lead researcher (PhD student) and the translator worked together to take these measures.

The pros and cons of the different methods that can be used to measure nutritional status (Table 1.1) were considered and anthropometry was identified as the most appropriate approach to measuring nutritional status in this setting. MUAC is a simple, cheap, acceptable method for detecting malnutrition (Myatt et al., 2006). Height for age, weight for age and BMI for age z-scores are standard anthropometric measures which take into consideration the child's age and gender and are widely recognized as the best indicators of nutritional status in children (de Onis et al., 1997). All these methods are simple, non-invasive measures that are commonly used in nutrition research in low income countries. They will be easily understood and provide a simple and objective measure of undernutrition.

3.6.3 Agrobiodiversity

Agrobiodiversity is defined as the biological diversity of plants, animals and micro-organisms on lands used for agricultural purposes (Brookfield and Stocking, 1999, Cromwell et al., 1999). The primary focus of this thesis is on plant agrobiodiversity. Animal agrobiodiversity is addressed as a secondary analysis and micro-organism agrobiodiversity is not assessed by this project.

3.6.3.1 Systematic measurement of plant agrobiodiversity

The research team aimed to collect agrobiodiversity data on as many of the household farms as possible. In Mbwei, where the households had many parcels of land, some of them very far away from the house, it was not possible to collect data on every farm. The lead researcher collected details on all household farms from the respondent and then selected representative farms for the biodiversity research assistant to visit. This information included estimates of farm sizes, whether they were far from the household (defined as more than 30 minutes walk), garden locations and crop types grown. If more than one farm had the same characteristics the closer farm was selected. The selected farms were then agreed with the household member who volunteered to show the research assistant the farms.

The diversity of all plants, both intentional crops and other plants, growing on the household's farms at the time of data collection were measured using the point intercept method (U.S. Department of Agriculture, 1999). The method used in this project was based on the method outlined in the 'Sampling vegetation attributes' interagency technical reference put out by the US Bureau of Land Management (U.S. Department of Agriculture, 1999).

- 71 -

The methods was designed to measure all plants growing on the farms including crops, wild plants, grasses and trees. Firstly, a string baseline was laid along the top of the farm. String transect lines were laid down the farm every 20 metres along the baseline. A meter long metal pin was inserted into the ground every 10 meters down the transect, starting at a random starting point at either 1, 2, 3, 4 or 5 meters in from the baseline. Family, genus and species of all plants present at these intercepts were recorded and number of individuals tallied. The agrobiodiversity protocol is included in Appendix C (See Appendix E for the data collection sheet).

After the first day of data collection the relative abundance of each species and the total number of species observed from data collected every 20 metres down the transect was compared to data collected every 5 metres. It was found that the abundance of individual species was very similar between these two different intervals. Data collected every 5 metres identified approximately twice the number of species compared to data collected every 20 meters. The missed species, however, were only 2% or less of the overall species encountered. Based on this information it was decided that collecting species data every 10 meters would provide sufficient detail.

In Mbwei, additional data were collected on transects laid every 10 metres along the baseline, due to the smaller size of the farms. Collecting data on transects laid every 20 metres in Minyenye captured enough data to represent the biodiversity of the farm, due to their large size. But in Mbwei the farms were much smaller and laying transect lines every 20 metres may not have captured enough data to reflect the biodiversity of their farms. Biodiversity in vegetable gardens was measured in Mbwei only. Because the vegetable gardens were so much smaller than the farms transects were laid every 5 metres with species data collected every 1 metre.

These data were used to calculate Shannon diversity indices for the household farms. The Shannon index is a species diversity index (Akrofi, 2010) and provides an indication of agrobiodiversity at the household level. The Shannon diversity index takes into account the number of different plant species and the abundance of these species (Pla, 2004). Shannon indices were calculated using the number of different species found and the frequency they were encountered in the household farms. It is a commonly used measure in biodiversity research and has previously been used to compare agrobiodiversity to dietary diversity (Akrofi, 2010, Dewey, 1981). These calculations are described in detail in chapter 5.

There is no one recognised way to measure agrobiodiversity but many different techniques were reviewed and considered for this project. The point intercept method was chosen because it provides systematic, precise and repeatable data on plant agrobiodiversity suitable

72

- 72 -

for household farms in rural Tanzania. The point intercept method records individual plant species, total cover and species composition by cover ((U.S. Department of Agriculture, 1999). The method is particularly well suited for vegetation less than 1.5 meters in height (U.S. Department of Agriculture, 1999). This method is more repeatable and produces more precise measurements than cover estimates acquired through quadrat sampling (U.S. Department of Agriculture, 1999). The method is also more efficient than line intercept methods. The point intercept method is a good method for determining cover of dominant species but is not as well suited to picking up the minor species present (U.S. Department of Agriculture, 1999).

Farm size was calculated in ArcGIS from the GPS coordinates collected at farm borders. Farm size is only available for farms that were visited by the biodiversity research assistants when they were collecting the agrobiodiversity measures. Notes were made by the biodiversity research assistants on the condition of the farm, how well the farms were maintained, whether some crops had already been harvested that year and markers of land degradation. Markers of land degradation were: 'presence of species associated with poor soil'; 'erosion'; 'low soil fertility'; 'fertile soil eroded by rainfall'; 'steep or very steep slope'; 'farm not well maintained/poor condition'; 'rocks and/or stones on the farm'. These qualitative data were coded into categories to be used in the analysis.

3.6.3.2 Plant and animal diversity scores

In addition to the systematic measurement of plant agrobiodiversity outlined above, information on the plant and animal foods consumed throughout the previous year (January to December) were collected through the food sources section of the household questionnaire.

This section collected information on all foods consumed by the household and where these foods had come from; grown or reared by the household, bought by the household, found or hunted in the wild or gifted to the household. For plant foods that were grown, information on when the crops were available for consumption by the household and whether the household sold any of the crops was recorded. For animal foods, whether the household sold animals or animal products was also captured. The data collected through the food sources section allows analysis of foods consumed year round and provides the animal agrobiodiversity data for the project. The results from this section are presented as crop diversity, vegetable diversity and animal diversity scores in order to differentiate this information from the plant agrobiodiversity data. These scores represent the number of different types of plants/vegetables/animals intentionally grown or reared for household consumption in the previous calendar year.

- 73 -

3.7 Sampling strategy

A target of 60 households in each village was set in order to measure 100 children under the age of five years old. This would give enough power to detect a difference of 0.5 in mean weight for age z-scores or height for age z-scores among the communities in a two village design (power=0.85, alpha=0.05, between village variance = 0.04, within village variance = 1.0). These calculations were based on the 2010 Tanzania DHS data.

- 74 -

Households were selected randomly using the following procedure in each village. In Minyenye, Singida household lists were collated for this project by the Village Executive Officer (VEO) and the five sub-village leaders who have a good knowledge of the people living in each sub village. The lists included the name of the sub-village, the name of the head of household and the number of children under five years living in the household. Only households with at least one child under five were included in the list. The average number of children under five per household was 1.57. To meet the target of 100 children under five, 65 households would need to be interviewed. A 10% refusal/drop out rate was allowed and 72 households were randomised using the random number generator function in Microsoft excel (=RANDBETWEEN(bottom,top).

In Mbwei, Lushoto, household lists were collated by the Ward Executive Officer (WEO) who was acting as the VEO for Mbwei, the Chairman and the Assistant Chairman for the village. The list consisted of the seven sub-villages and the names of all the head of household in these sub-villages. Whether the household had under five year olds was not included in the Mbwei list. More household were randomised in order to reach the target 100 children under five years of age. Initially 70 households were randomised but when this number was not sufficient a subsequent additional 25 households were randomised. The percentage of the total number of households in each village that were interviewed is presented in Table 3.2.

	Total number of households	Total number of households with >=1 under five year old	Number of households interviewed	Percentage of total households interviewed*
Minyenye	-	206	64	31.1
Bwali	-	41	16	39.0
Mwangozo	-	29	5	17.2
Amani	-	19	5	26.3
Jamida	-	38	16	42.1
Kujitegemea	-	79	22	27.8
Mbwei	880	-	58	6.6
Nekrasi	117	-	7	6.0
Kwemeaganga	96	-	6	6.3
Mntindii	113	-	6	5.3
Zagati	109	-	10	9.2
Vugiri	156	-	10	6.4
Pongwe	138	-	8	5.8
Mbunguni	151	-	11	7.3

Table 3.2.	. Total number of households and total number of households interviewed by sub-
villa	age

*For Minyenye this is the percentage of households with at least 1 under five year old, for Mbwei it is a percentage of all households

3.8 Inclusion criteria

Households were included in the survey if they had a woman with a child under the age of five years living in the house. As the preferred option the female head of household would be interviewed. The female head of household was defined as the female in the household who was responsible for the preparation of food for that household. If this female did not have a child under five, was unavailable to be interviewed or did not want to be interviewed then any other female who lives in the household that had a child under five years was asked. If no eligible females were willing to be interviewed then the research team moved onto the next household. If an eligible woman lived in the household but was not home or unavailable when we called we visited three additional times before excluding them from the study. This happened in one households in Mbwei and none in Minyenye.

Women were chosen as participants as the project focused on both diet and agriculture. Women are primarily responsible for food preparation and farming (Hyder et al., 2005) in rural sub-Saharan Africa and would therefore be the most appropriate household member to answer the projects questions. Respondents were asked questions about the family unit. The family unit was defined as the respondent, her husband or partner, referred to as her husband for simplicity, and their children. This definition included step-children or grandchildren that the respondent was raising.

- 76 -

Children under five years of age were the main focus of this study as these first five years are when children are most at risk of malnutrition and not reaching their growth potential. Victora et al (2010) showed rapid growth faltering in height for age in children from 54 different low and middle income countries until 24 months of age. Declines in weight for age z-scores were seen throughout the first five years of life. Inadequate growth in the first 1000 days after conception impacts on the physical, cognitive and socio-emotional well-being of people throughout all lifecycles into adulthood and old age (Hoddinott et al., 2013).

Nutritional status was measured in all children in the household rather than just in under five year olds for two main reasons. Firstly it is possible that the environment these children have grown up in are stable over time and malnutrition seen in the older children could be a result of factors captured in this study. Secondly, weight and BMI for age is affected by recent health and diet and it is expected that measuring these outcomes in older children will provide additional information on the nutritional status of those living in these communities. Demographic data were collected on the respondents husband to give social and financial context to the children's lives and heights and weights were collected in order to control for parental weight in the data analyses.

3.9 Research process

3.9.1 Piloting of methods

The questionnaire and agrobiodiversity methods were piloted during Phase 1 of the research process in Shebomeza village, Muheza district, Tanga region in northern Tanzania. The pilot study took place between the 9th, 10th and 11th of April 2011. Household questionnaires were conducted with six participants (two women and four men) who represented different wealth groups as defined by a field assistant from the village. Different agrobiodiversity survey approaches (including point intercept methods and quadrat sampling) were trialled in the farms of the households. An additional two farms were visited and only agrobiodiversity measures were taken. For these households one male and one female household member were questioned on the utility of crops as part of the agrobiodiversity measurements.

The piloting of method highlighted a number of issues with the original design of data collection methods, including repetition, concepts not translating well, missing interesting information due to the structure of questions and specifically, that the agrobiodiversity data collected was unsystematic and therefore unrepresentative. The pilot activities also provided ideas for how the methods could be improved. Questionnaire questions were modified to improve participant understanding and the dietary diversity score sheet changed to a basic 24 hour dietary recall as detailed above in section 3.5.2. A questionnaire designed to measure food security was removed and more direct questions about the household's ability to feed itself were included. The food sources section was expanded to capture more detail about where specific foods that individuals within households were eating came from. Agrobiodiversity methods were changed from quadrat sampling (Zarin et al., 1999) to the point intercept method. These were better suited to the farms in rural Tanzania.

3.9.2 Ethical approval and permissions

Ethical approval was obtained from the University of Leeds through the Faculty of Medicine and Health (Reference number: HSLTLM/11/031). The application for this approval included the following provisions. Informed verbal consent was obtained from the respondent and the children's free and voluntary assent to participate was sought. The research assistant was trained in how to measure height, weight and MUAC to ensure measures taken on children were done correctly and in the presence of one of the children's parents or older siblings. All data was anonymised and stored securely.

National level permission to conduct the research was obtained from the Commission for Science and Technology (COSTECH) in Tanzania. Letters of permission were also obtained at the regional, district, ward and village level following meetings with the appropriate individuals. The next level of permissions needed was advised at each level and varied between the two sites. A summary of the permissions granted and a copy of the COSTECH permit is included in Appendix A.

3.9.3 Research assistants and translation

A total of three research assistants were employed during this research; A translator and two different biodiversity research assistants, one in Minyenye, Singida and one in Mbwei, Lushoto. In addition, a local contact who knew the potential participants and where they lived was hired in each sub-village.

77

- 77 -

Both the Minyenye and the Mbwei biodiversity research assistants were experts in the measurement of biodiversity and had experience using similar methods to the point intercept method. They were hired through the project's contacts with Sokoine University of Agriculture. They were assisted by the local contact and sometimes the farmer. A student volunteer from the UK helped collect this data in Mbwei.

The questionnaire used for data collection was written in English and translated to Swahili (the national language which was widely spoken in the research areas) by the project's translator before data collection began. This was then back translated into English by the Minyenye biodiversity research assistant and the lead researcher discussed any loss of meaning that occurred in translation with the translator. Modifications to the Swahili version of the questionnaire were made as necessary before it was administered. The same process was used for the information sheet and consent form.

The translator was trained on how to conduct a 24 hour recall, how to carry out the anthropometric measurements and how to use the research questionnaire by the lead researcher in the days leading up to data collection.

3.9.4 Recruitment of participants and consent process

The local contact in each sub-village escorted the research team up to the households, briefly introduced the team and project to the head of the household and asked if the team could speak with the mother of the under five year olds. The translator would then briefly introduce the research team and the project before reading out the detailed information sheet. The information sheet was then offered to the participants for them to keep. Once the partici pants had had the opportunity to ask questions the translator asked for consent for the various aspects of the research:

- Opportunity to ask questions?
- Do you agree to take part in this research?
- Can we record some of the interview using this tape recorder?
- Can we measure your height and weight?
- Can we measure the height, weight and mid upper arm circumference of your children?
- Can we measure the location of your house and farm using this global positioning system?

Can we take pictures of your health records?

Verbal consent was obtained from the mother (the primary research participant). The aspects of the research the participant gave consent for was recorded on a tick list that was retained by the researcher (An English version of the Swahili information sheet and consent form is included in Appendix B).

3.10 Data checking and entry

Questionnaires were checked by the PhD student at the time of data collection in order for clarification and additions to be made while the participants were still available. Agrobiodiversity data was checked and data was clarified with the biodiversity research assistant at the end of the day when possible or soon after. Both biodiversity research assistants were available after data collection for clarification and corrections of species data when necessary. The majority of the data was entered by the PhD student with some being entered by Masters students and a nutrition research volunteer under close supervision of the PhD student.

3.11 Data analysis

3.11.1 Basic descriptive statistics

For chapters 4, 5 and 6, basic descriptive statistics were performed in Stata version 12. Means with 95% Confidence Intervals and percentages are presented at the beginning of the results chapters. In order to detect differences between the villages and between sub-groups of interest (eg. gender) Mann-Whitney tests, for differences in continuous variables, and Chi-squared tests, for differences in proportions in categorical variables were used. Qualitative data were coded into categories in order to be used in the analyses.

3.11.2 Linear regression

Linear regression was used to estimate relationships among variables of interest. The terminology 'independent variable' was used to indicate the effecting or exposure variable and the term 'dependent variable' was used to indicate the affected or outcome variable. For example dietary diversity would be the independent variable and height for age z-scores would be the dependent variable.

Regression results are presented for both villages combined, controlling for village, in the study of the relationship between dietary diversity, complementary feeding and sanitation and nutritional status (chapter 4). This is because the larger sample size is beneficial in detecting relationships between dietary diversity and nutritional status. As the landscape and farming practices varied substantially between the two villages results of plant and animal agrobiodiversity (chapters 5 and 6) are presented for the two villages separately. This was so factors specific to the two villages could be investigated without their effects being lost in combined models. Including the village in the multiple regression as a covariate in chapter 4 addressed the effect of clustering at the village level. Analysing the data by village in chapter 5 and 6 eliminating this potential issue for these analyses.

Both unadjusted models, with just the independent and the dependent variable in the model, and adjusted models are presented. The adjusted models include other variables that could be confounding the relationship between the independent variable and the dependent variable. For each model a list of potential confounders were determined using a Directed Acyclic Graph (DAG). These potential confounders were identified through reviewing the literature and local knowledge. DAGs map causal associations between variables which allows easy identification of variables that may be related to both the independent and the dependant variable. This helps to ensure that appropriate confounders are selected for multivariable regression (Glymour, 2006). The DAG used for the regression between agrobiodiversity and dietary diversity and agrobiodiversity and nutritional status is presented in Figure 3.16.

Once potential confounders had been identified each confounder was added to the unadjusted model one by one to assess if they affected the regression coefficients for the independent variable of interest. If the potential confounder modified the regression coefficient substantially (criteria varied from approximately >0.02 to >0.05 depending on the size of the regression coefficients) then they were included in the adjusted model.

- 80 -



Figure 3.16. Directed Acyclic Graph (DAG) used in determining potential confounders in the regression models

In order to determine if the regression models met the assumptions of linear regression a number of tests were undertaken (UCLA: Statistical Consulting Group, 2014). To test if the residuals were normally distributed Kernel density estimate plots were compared with normal distribution curves. For the assumption of linearity in the unadjusted models scatterplots of the continuous independent and dependant variables were checked to make sure there were a random scatter of points. For the multivariable regression standardised residuals were plotted against each of the continuous predictor variables in the regression model. Again, these were checked that they were a random scatter of points. To test the assumptions of homoscedasticity the residuals were plotted against the fitted values and the scatter plot was checked for randomness. Additionally, the null hypothesis that the residuals are homogenous was tested using the Whites test and the Breusch-Pagan test. Multi-collinearity was tested by checking the variance inflation factors were in the acceptable range. The regression models presented in chapters 4 to 6 met the regression assumptions for linearity, normality of residuals, homoscedasticity and multi-collinearity.

3.11.3 Multiple imputation by chained equations (MICE)

For the adjusted regression multiple imputations by chained equations were used within Stata to estimate missing data in covariates. Leaving out individuals with missing data from analyses can lead to bias and the resulting decrease in sample size leads to loss of power (Sterne et al., 2009). Multiple imputation has been devised to try to deal with missing data. It aims to account for the uncertainty in the missing data by creating different possible data sets based on the available data and combines the results from each of these data sets (Sterne et al., 2009). The command creates multiple copies of the data set and replaces missing values with imputed values. Variability in the imputed variables is created to allow for uncertainty in predicting the missing values. The model is then fitted to each of the imputed datasets. The estimated associations will differ because of the variation in the imputed variables, these are averaged to give an overall estimated association. Rubin's rule (Rubin, 1987) is used to calculate standard errors, this takes into account this variability (Sterne et al., 2009). For all the MICE regression analyses missing variables were assumed to be missing at random (MAR). Values were missing mainly because participants and their husbands were unable to remember their date of births and because the husbands were unavailable to be measured. Ten imputed datasets were used and 100 iterations were carried out. The variables used in the imputation models are reported in the data analysis sections in chapters 4 and 6.

For all results chapters any regression results discussed in the results section but not presented in a table will include the regression coefficients and their 95% Confidence intervals

82

- 82 -

within the text. When results are discussed and no regression results are included they will be presented in the subsequent table. Also, significant regression coefficients, in both the text and the tables, are presented in bold.

The next chapter is the first of the four results chapters of this thesis and focuses on the main results of the primary data collection described in this chapter; the relationship between dietary diversity and nutritional status in Minyenye, Singida and Mbwei, Lushoto.

Chapter 4:

Investigating the associations between dietary diversity and nutritional status in Minyenye and Mbwei, rural Tanzania.

4.1 Chapter summary

This chapter is the first of four results chapters; the first three are based on the primary data collected in rural Tanzania. This chapter explores the relationship between dietary diversity and nutritional status in two Tanzanian villages: Minyenye in Singida district and Mbwei in Lushoto district. This chapter fulfils part of objective 3 (Measure household crop and animal diversity, dietary diversity of respondents and children under five and nutritional status of children living in these villages, determine whether household produce is sold and investigate how these factors are related) and 4 (Investigate basic socio-demographic factors affecting dietary intake of children under five and nutrition outcomes in children in these villages) of this thesis. The chapter objectives are:

- Objective 4A: To present descriptive data on the demographic, social, dietary diversity and nutritional status variables in this population to set the context for the analyses.
- Objective 4B: To investigate whether dietary diversity and food variety are associated with nutritional status in under five year olds.
- Objective 4C: To investigate whether complementary feeding and sanitation are associated with nutritional status in these villages.

The major findings of this chapter include that dietary diversity and food variety scores are not significantly associated with any of the nutritional status variables. No significant differences were seen between the two villages in dietary diversity but those in Mbwei had lower food variety and height and weight for age z-scores than those in Minyenye. Children who had liquids, specifically multiple flour porridge with additions and millet juice or solids introduced to their diets had poorer nutritional status as did children from households not boiling drinking water and households with open pit latrines. This chapter adds to the literature on the relationship between dietary diversity and nutritional status, providing additional evidence on representative samples from two different villages in Tanzania which conflicts with the majority of evidence in this area so far. It also provides additional data on specific complementary feeding and sanitation factors that are associated with nutritional status in these communities which adds to the body of literature on the determinants of nutritional status in low income countries.

4.2 Introduction

Those living in sub-Saharan Africa have high rates of malnutrition; 26% of the world's undernourished people live in the region (UNFAO, 2010b). In Tanzania specifically, 44.4% of children under the age of five are stunted and 16.7% are underweight (Gollogly, 2009). Childhood malnutrition has been linked with a range of negative consequences. Malnourished children tend to achieve lower academic attainment (Alderman et al., 2006, Victora et al., 2008, Adair et al., 2013), grow into smaller adults (Rivera et al., 1995, Alderman et al., 2006, Victora et al., 2008, Adair et al., 2013), may be at increased risk for chronic disease later in life (Adair et al., 2013) and have a lower capacity for work (Haas et al., 1996) than their well-nourished counterparts. In addition, severe malnutrition in childhood has been linked with overall increased rates of mortality due to infection (Chen et al., 1980).

Many researchers and development organisations have attempted to improve nutritional status in low income countries through various dietary interventions (Bhutta et al., 2013) such as food supplementation (Rivera et al., 1995, Super et al., 1990), agricultural interventions (Berti, 2004) and context specific, tailored nutrition programmes (Berti et al., 2010). Despite success in many of these research projects these lessons have not been translated into population wide decreases in stunting and underweight rates. This is partly because malnutrition is caused by a multitude of interacting factors (Bhutta et al., 2008) making it a difficult issue to tackle.

Dietary diversification has been proposed as a holistic and sustainable nutrition intervention that may have multiple health benefits in low income countries (Gibson et al., 2003). The term dietary diversity refers to the number of different food groups or individual foods consumed over a defined reference period (Ruel, 2003). A summary score of dietary diversity has been proposed as a marker for diet quality as it is relatively easy to collect (Arimond, 2004). Versions of this summary score have been shown to be significantly associated with both measures of food security (Bukusuba et al., 2010, Leroy et al., 2008, Thorne-Lyman et al., 2010) and

- 85 -

nutrient adequacy (Arimond, 2004, Ogle et al., 2001, Moursi et al., 2008, Daniels et al., 2007, Hatloy et al., 1998, Kennedy et al., 2007, Torheim et al., 2004, Ponce et al., 2006).

Research has also demonstrated links between increased dietary diversity and improved nutritional status in children (Arimond, 2004, Corbett et al., 1992, Steyn et al., 2006, Garg and Chadha, 2009, Nti and Lartey, 2008) and adults (Savy et al., 2005, Savy et al., 2006). Other studies, however, have shown mixed results (Eckhardt et al., 2005, Sawadogo et al., 2006) or no association (Hillbruner and Egan, 2008, Lin et al., 2007). The evidence for this association is not as strong in Eastern Africa with five of the eleven studies identified in chapter 2 showing no association between dietary diversity and nutritional status. This study will add to the existing literature, providing additional evidence on these associations in Eastern Africa, by presenting the relationship between dietary diversity and nutritional status in these two villages in rural Tanzania.

4.3 Methods

This section covers how dietary diversity and height for age, weight for age and BMI for age zscores were calculated and the statistical analysis methods used in this chapter. Additional methodological details are presented in chapter 3. A description of the questionnaire and how the anthropometric data was collected is also included in that chapter. Results are presented for the family unit which is defined as the respondent, her husband or partner and her children.

4.3.1 Data sources

4.3.1.1 Dietary diversity and food variety

Dietary diversity scores (the number of different food groups consumed over the previous 24 hours) and food variety scores (the total number of different foods consumed within that timeframe) were calculated for both the respondent and her oldest child under five. The oldest child under five was chosen as the project was specifically interested in under five year olds but younger children were more likely to be breastfed, limiting the other foods they consumed. The dietary diversity scores were derived through the questionnaire from two 24 hour recalls provided by the respondent; one for herself and one for her child. The diet recall asked the respondent about all foods and drinks consumed for a 24 hour period from waking the day before to waking on the day of the interview (see section 3.6.1, page 61 and appendix
C). The design of the recall prompted for details of cooking methods, foods added during cooking, snacks and drinks consumed between meals and food consumed outside the home. These prompts attempted to ensure that all the different foods eaten were captured in the recalls.

Foods were subsequently coded by the researcher and categorised into nine food groups ('Cereals, roots and tubers'; 'vitamin A rich vegetables, tubers and fruit'; 'other vegetables'; 'other fruits'; 'flesh meats, organ meats, fish and insects'; 'eggs'; 'legumes, nuts and seeds'; 'milk and milk products'; 'oils, fats and sweets'), as recommended by the FAO (2004) dietary diversity workshop (FAO/WHO/IFPRI, 2004). This gave each individual a dietary diversity score out of nine for the 24 hour period. The number of different individual food items were also calculated to give a food variety score (Hatloy et al., 1998). This is a tally of all foods consumed in the 24 hour period. For example, if the individual consumed *ugali* (a very common, usually maize based, staple served as a stiff porridge) with spinach, tomatoes and onions as their main meal this would contribute four points towards their food variety score.

4.3.1.2 Food sources

In order to further investigate how households obtain their food and the effect this has on their diet and nutritional status the questionnaire also included questions about the sources of the household's food. The sources were: grown by the household; bought; obtained from the wild and gifted, as defined by the respondent. This data collection technique was designed by the author and aimed to capture a level of detail about household food sources not usually seen in nutrition research. It was an in-depth and time consuming method which was reduced after the first ten interviews in order to capture the necessary information in the shortest possible time.

Which months during the last calendar year each individual food type had been available from each of these sources was ascertained. The respondent was asked for a list of all the food consumed by the household and then questioned about the source of each individual food type and the months it was available. This information was used to calculate a dietary diversity score out of six for each month of the year in order to illustrate the annual variation in dietary diversity. The score was based on the following six categories: 'Cereals, roots and tubers'; 'vitamin A rich vegetables; tubers and fruit'; 'other vegetables'; 'other fruits'; 'legumes, nuts and seeds' and 'oils, fats and sweets'. This differs from the nine item score in that it does not include the animal based categories: 'flesh meats, organ meats, fish and insects'; 'eggs' and 'milk and milk products'. In order to reduce the time taken for this section monthly availability information was not collected on animal source food as participants were unable to say which

87

- 87 -

months meat and animal products were available for consumption. The score was calculated separately for grown food only and again for grown, wild and gifted food.

Data on how long children were breastfed, when complementary liquids and foods were introduced and what kind of liquids and foods were first introduced into the infant's diets was captured through the questionnaire for all children in the family unit under the age of five.

4.3.1.4 Nutritional status

Heights and weights were collected for all individuals in the family unit that lived in the household and were present at the time of interview. If eligible individuals were not present at the time of the interview the researchers attempted to meet these individuals at a later time or day in order to take these measurements. This information, combined with the age and sex of the children collected through the questionnaire, was used to calculate age adjusted *z*-scores for height, weight and BMI using an excel add-on (WHO, 2010a). These *z*-scores have been developed by WHO and are based on a pooled sample of breastfed infants from Brazil, Ghana, India, Norway, Oman and the USA. It is considered an international standard suitable for use in all countries (WHO, 2010a). Photographs of the health records of all family unit children were taken if the family agreed and were able to provide them. These health records provided the date of birth of the children and this was used instead of reported date of births when they were available.

Means of height, weight and BMI z-scores were calculated and presented for all children. Individuals with z-scores below negative five or above five were excluded from the analysis. The proportion of children who were stunted and underweight were calculated. Stunting and underweight were defined as below minus two standard deviations from median height for age and weight for age of the reference population (UNICEF, 2014).

4.3.2 Data analysis

4.3.2.1 Descriptive statistics

Basic descriptive statistics of crude associations were performed in Stata version 12 (chapter objective 4A). Means (95% Confidence Intervals) of continuous variables and percentages within groups for categorical variables are presented for the descriptive results. Mann-Whitney tests were used to test differences in means and Chi-squared tests were used to detect differences in proportions between the subgroups of interest, e.g. village and gender.

4.3.2.2 Linear regression analysis

Linear regression (95% Confidence Intervals) was used to estimate the relationship between dietary diversity scores and food variety scores and the nutritional status variables (chapter objective 4B). Both unadjusted and adjusted models are presented. Adjusted models include potential confounders which were identified using a directed acyclic graph (DAG) as described in chapter 3, page 73 (Figure 3.16).

The following variables were controlled for in the adjusted model investigating the relationship between dietary diversity/food variety and nutritional status: village; parent's age; parents highest level of education; parent's height, weight or BMI; parental ethnicity; whether the household has a mobile phone; household takes action to make water safe and whether the parents had an alternative source of income.

Linear regression was used to determine whether complementary feeding and sanitation were associated with nutritional status in under five year olds (chapter objective 4C). Both unadjusted and adjusted regression models are presented for the relationship between complementary feeding and nutritional status. These multivariable regression models were adjusted for: village; parent's age; parents ethnicity; highest level of education; parent's height; weight or BMI; whether the household has a mobile phone; household takes action to make water safe; whether the parents have an alternative source of income.

Unadjusted and adjusted models are presented for the relationships between sanitation and nutritional status. These models were adjusted for village; parent's age; husband's frequency of employment. No potential confounders were identified for the relationships between parental ethnicity, height and weight and the nutritional status variables in the children. Unadjusted models only are therefore presented for these associations.

4.3.2.3 Missing data

Multiple imputations using chained equations (MICE) were used to estimate parameters under the assumption that any missing data were MAR (see section 3.11.3, page 74). MICE were used in the adjusted model investigating associations between dietary diversity/food variety and nutritional status. The missing values were imputed based on the complete variables in the model: village; respondent's weight; whether the household has a mobile phone; respondent's highest level of education; whether the respondent earns extra income; household takes action to make water safe. Imputations were based on under five year olds only.

MICE were used in the adjusted model investigating associations between complementary feeding and nutritional status. Imputations were based on under five year olds only. The missing values were imputed based on the complete variables in the model: village; respondent's weight; whether the household has a mobile phone; respondent's highest level of education; whether the respondent earns extra income; household takes action to make water safe. This analysis was repeated excluding children who were still being breastfed in case this influenced the results.

MICE were again used for the multivariable regression analyses of sanitation and nutritional status. Imputations were done separately for those under five and for all children. The missing values were imputed based on the complete variables in the model and other complete employment variables: village; whether the household has a mobile phone; respondent's highest level of education; whether the respondent works in agriculture; whether the respondent has an additional small business; whether the respondent earns extra i ncome.

4.4 Results

No households in Minyenye and two households in Mbwei declined to take part in the project. One household in Mbwei was excluded as the potential respondent was absent on all three occasions that the researchers visited. Sixty-four households were interviewed in Minyenye. Heights and weights were measured in 295 children (of whom 106 were under the age of five), 64 women and 43 men. Additionally 252 children under the age of 15 had their mid upper arm circumference (MUAC) measured. Fifty-eight households were interviewed in Mbwei; 170 children of whom 104 were under the age of five, 58 women and 35 men were measured. 180 MUAC was measured in 180 children under 15 years old. The proportion of under five year olds that were male and female, under three years and under one year old is presented in table 4.1.

		Minyenye	Mbwei
Number of households		64	58
Number of children		295	170
Number of under fives		106	104
Percentage of under five years	Male	47.2	41.4
	Female	52.8	58.7
	Under three years	58.5	56.7
	Under one year	17.0	22.1

Table 4.1. Number of households, children and under five year olds measured in each village

- 91 -

4.4.1 Wider social and farming context

Detailed information about the climate, agriculture and socio demographics about the two regions are included in section 3.4.1 and 3.4.2. Based on observations made in Minyenye during data collection, the village appeared similar to the surrounding district in the lands characteristics, the farming context and socially. The area was flat and dry, appeared relatively poor and the main economic activity in the village was agriculture. Mbwei appeared to differ from the surrounding district in a number of key aspects. The majority of the land was relatively dry with small shrubs and sparse trees with farms that struggled to grow the planted crops, with the exception of cassava. There was a band of land within the village which followed the river that reflected the crop growth typical of Lushoto district. The way land was farmed could potentially impact on the health and wellbeing of those living in Mbwei. Small pockets of land large distances apart meant more work and energy expenditure for the amount of food harvested. Further information on the land characteristics, habitat and farm types of the two areas and how this relates to agrobiodiversity and farming in the two areas is presented in section 5.4.1, 5.4.2.1 and 5.4.3.1

4.4.2 Demographic and household characteristics of sample

The two villages were similar in many demographic and household factors (Table 4.2). However husbands were significantly older, by approximately eight years, in Mbwei compared to Minyenye and children were significantly younger. Respondents in Mbwei were more likely to be married. The majority (97%) of participants' highest level of education was primary school. 70% of participants over ten were self-employed with approximately 25% defining themselves as unemployed and 3% running an additional small business.

- 92 -

	All	Minyenye	Mbwei	P-value for difference*
Total number of individuals (N)	773	420	353	
Age (mean(95% CI))				
Respondents	31.5 (29.4, 33.5)	30.2 (28.0, 32.4)	34.4 (29.9 <i>,</i> 39.0)	0.058
Husband	40.6 (37.3, 43.9)	37.3 (33.5, 41.1)	45.6 (40.1, 51.1)	0.011
Children	7.6 (7.1, 8.1)	8.0 (7.4, 8.7)	7.0 (6.2, 7.8)	0.052
Mean number of children/household	4.1	4.5	3.6	
Respondent's ethnicity (%)				<0.001
Nyantulu	50	95.3	0	
Pare	33.6	0	70.7	
Sambaa	9.8	0	20.7	
Other	4.9	3.1	6.9	
Missing	1.6	1.6	1.7	
Married/engaged (%)^^	62.1	57.7	67.4	0.011
Attended school(%)^	85	86.1	83.7	0.43
Highest schooling (%)^^^				0.449
Primary	96.9	95.8	98.3	
Middle/secondary	3.1	4.2	1.7	
Employment(%)^^				0.633
Unemployed	24.4	26.6	21.9	
Employed	3.7	3.2	4.4	
Self employed	69.1	68	70.5	
Self employed farming and small business	2.5	2.3	2.7	
Doesn't know	0.3	0	0.6	

Table 4.2. Demographic information for the family unit for included households in Minyenye and Mbwei and both villages combined

* Mann-Whitney tests used for difference in means, chi squared used for difference in proportions between the two villages

Total number of respondents is 122 (64 in Minyenye, 58 in Mbwei). Age of respondents is based on 54 individuals (38 in Minyenye, 16 in Mbwei). Age of husbands is based on 33 individuals (20 in Minyenye, 13 in Mbwei). Age of children is based on 499 individuals (288 in Minyenye, 211 in Mbwei).

^only those over 5 years (N=554, 309 and 245 for all, Minyenye and Mbwei)

^^only those over 10 years (N=405/406, 222 and 184/183 for all, Minyenye and Mbwei)

^^^only those who are no longer attending school (N=261, 144 and 117 for all, Minyenye and Mbwei)

Approximately half of the households had a radio and half had a mobile phone, while no households had electricity or a landline (Table 4.3). On average, it took 18 minutes longer to get water in Mbwei than Minyenye (58 vs. 40 minutes; P = 0.003). The majority of Mbwei households got their drinking water from a river (90%) (Figure 4.2), while 56% and 33% of those in Minyenye got their water from a borehole/dug well and through a piped water pump respectively (Figure 4.1). However, people in Mbwei were significantly more likely to take action to make water safe, with 66% of households boiling their drinking water compared to 31% in Minyenye, possibly due to these differences in water sources.

Table 4.3. Characteristics of households in Minyenye and Mbwei and in both villages combined

	All	Minyenye	Mbwei	P-value for difference*
Ν	122	64	58	
Radio in household (%)	50.0	45.3	55.2	0.277
Mobile phone in household (%)	55.7	57.8	53.5	0.628
Raised animals	77.9	85.9	69.0	0.024
Owned animals	27.9	4.7	53.5	<0.001
Time to get water minutes (mean(95% CI))	48.7 (42.5, 54.9)	39.9 (31.9, 47.8)	58.4 (49.2, 67.6)	0.003
Water source (%)				<0.001
piped water	17.2	32.8	0.0	
spring	4.9	0.0	10.3	
river	48.4	10.9	89.7	
borehole or dug well	29.5	56.3	0.0	
Take action to make water safe (%)				0.001
Boil	47.5	31.3	65.5	
Other eg. Strain, let settle	4.9	7.8	1.7	
Nothing	47.5	60.9	32.8	
Type of toilet (%)				0.056
closed pit latrine	18.0	9.4	27.6	
open pit latrine	80.3	89.1	70.7	
other	1.6	1.6	1.7	

* Mann-Whitney tests used for difference in means, chi squared used for difference in proportions between the two villages



Figure 4.1. Water sources in Minyenye; a water pump, a borehole, holes dug in an almost dry area of a riverbed and a river. The rivers are used as water sources for both animals and people



Figure 4.2. Water source in Mbwei; the river and a water spring. The spring water is diverted into channels for water collection and irrigation

4.4.3 Dietary diversity and food variety

On average, people consumed approximately five out of the nine food groups (Table 4.4). No significant differences in dietary diversity scores were seen between the villages, by gender or between mother and child. Respondents and their oldest child under five consumed on average 7.5 food items in the previous 24 hours. Those in Mbwei consumed significantly more food items (8.5) compared to those in Minyenye (6.5). This was seen in respondents and their oldest child under five. When the children were analysed separately by gender the difference between the two villages was only significant for the male children. In both villages combined, no significant differences in food item variety were seen between mother and children or between female and male children. These results did not differ when respondents and children who reported the past 24 hours as 'not typical' were excluded.

		All	Minyenye		Mbwei	P-value for	
	Ν	Mean (95% CI)	Ν	Mean (95% CI)	Ν	Mean (95% CI)	difference*
Dietary diversity scores							
All	244	4.8 (4.6, 5.0)	128	4.8 (4.6, 5.0)	116	4.8 (4.6, 5.1)	0.652
Respondent	122	4.9 (4.7, 5.1)	64	4.9 (4.6, 5.2)	58	4.9 (4.6 <i>,</i> 5.3)	0.859
Oldest child<5yrs^	122	4.7 (4.5 <i>,</i> 5.0)	64	4.7 (4.3, 5.0)	58	4.8 (4.4, 5.1)	0.665
Female children	67	4.8 (4.5, 5.1)	34	4.9 (2.0, 7.0)	33	4.7 (1.0, 7.0)	0.461
Male children	55	4.6 (4.1, 5.0)	30	4.4 (0.0, 7.0)	25	4.8 (1.0, 7.0)	0.255
Food variety scores							
All	244	7.5 (7.1, 7.8)	128	6.5 (6.1, 7.0)	116	8.5 (8.0, 9.1)	< 0.001
Respondent	122	7.6 (7.1, 8.1)	64	6.6 (59, 7.2)	58	8.7 (8.0 <i>,</i> 9.4)	< 0.001
Oldest child<5yrs	122	7.4 (6.8, 7.9)	64	6.5 (5.8 <i>,</i> 7.2)	58	8.3 (7.6, 9.1)	< 0.001
Female children	67	7.4 (6.7, 8.1)	34	6.8 (2.0, 16.0)	33	8.0 (2.0, 13.0)	0.061
Male children	55	7.4 (6.5, 8.2)	30	6.1 (1.0, 12.0)	25	8.8 (2.0, 14.0)	0.001

Table 4.4. Mean dietary diversity and food variety scores for respondents and their oldest child under five in Minyenye, Mbwei and both villages combined

* Mann-Whitney tests used for difference in means between the two villages

^19 children had breast milk (included in food variety score but not in dietary diversity score)

Table 4.4 shows that overall, dietary diversity scores were very similar between the two villages. Figure 4.3 shows how the villages compare in their intakes of the food groups that make up the dietary diversity score. Mbwei respondents and their oldest children under five years were significantly more likely to consume 'oils and sweets', 'beans, nuts and seeds' and 'milk and dairy products' the day before the interview. While participants in Minyenye were significantly more likely to consume 'vitamin A rich vegetables' and 'other vegetables' than those in Mbwei.

There were a number of differences in the specific foods eaten between the two villages. For example, data from the respondents on the foods eaten over the last calendar year shows that cassava root was consumed by households more frequently in Mbwei (98%) than in Minyenye (55%). *Mlenda*, a wild green leafy vegetable, was consumed in all Minyenye households but only in 16% of households in Mbwei. Milk was more commonly consumed in Mbwei (83%) compared with in Minyenye (55%).



Figure 4.3. Percentage of participants (respondents and children combined) in Minyenye and Mbwei consuming the nine food groups that make up the dietary diversity score on the day before the interview

Figure 4.4 shows the number of different kinds of plant grown by the household, for each month, over the previous year, as reported by the respondent. Figure 4.5 shows plant food grown by the household, gifted to the household or found wild. For both of the graphs the diversity of food available peaks around June and July with the lowest variety of food available between November and February. When considering only grown food Mbwei has a significantly higher diversity of food available in May and June as compared to Minyenye. When wild and gifted food is added in this significant difference disappears and Minyenye then has significantly more variety of food available to them in January, February, April and November compared to Mbwei.



Figure 4.4. Mean household dietary diversity score for the previous calendar year in Minyenye and Mbwei. Score is out of six and based on grown, plant foods only^

^Dietary diversity score includes the following six categories: 'Cereals', 'vitamin A rich vegetables or fruit', 'other vegetables', 'other fruit', 'beans, nuts and seeds' and 'oils and sweets'.

4.00 Househod dietary diversity score out of 6 3.50 3.00 2.50 2.00 Minyenye 🖬 1.50 📕 Mbwei 1.00 0.50 difference between 0.00 Minvenve Jan Feb Mar Apr May June July Aug Sept Oct Nov Dec and Mbwei: *p<0.05 Months of previous year **p<0.01

- 100 -

Figure 4.5. Mean household dietary diversity score for the previous calendar year in Minyenye and Mbwei. Score is out of six and based on grown, wild or gifted plant foods only^

^Dietary diversity score includes the following six categories: 'Cereals', 'vitamin A rich vegetables or fruit', 'other vegetables', 'other fruit', 'beans, nuts and seeds' and 'oils and sweets'.

4.4.4 Nutritional status of children

On average, height was 2.05 z-scores and weight was 1.57 z-scores below the reference population median (WHO Child Growth Standards) in children under the age of five (Table 4.5). Mean BMI for age z-scores for these children was 0.14 below the population median. MUAC was approximately 15cm in children under five years of age. Mbwei had significantly lower height and weight for age z-scores with a corresponding higher rate of stunting and underweight in children under five compared to Minyenye. There were no significant differences between the two villages in BMI z-scores or MUAC for children under five.

A similar trend was seen in all children, with Mbwei children having a significantly lower average height z-score and a lower (non-significant) average weight z-score. Mbwei had a significantly higher average BMI z-score than Minyenye in all children. MUAC was on average 16cm in all children under the age of 15 years with Mbwei children having a significantly lower MUAC than children in Minyenye. There were no significant differences between females and males in average z-scores or MUAC.

	All	Minyenye	Mbwei	P-value for difference*
Children under 5 years	204	103	101	
Height z-scores (mean (95% CI))	-2.05 (-2.23, -1.88)	-1.76 (-1.99, -1.54)	-2.36 (-2.61, -2.10)	0.001
Weight z-scores (mean (95% CI))	-1.57 (-1.72, -1.42)	-1.38 (-1.58, -1.18)	-1.76 (-1.98 <i>,</i> -1.54)	0.012
BMI z-scores (mean (95% CI))	-0.14 (-0.28, -0.01)	-0.23 (-0.42, -0.04)	-0.06 (-0.24, 0.13)	0.200
Stunted (%)	53.0	40.6	66.0	0.002
Underweight (%)	31.5	23.0	40.0	0.034
MUAC(cm)^ (mean (95% Cl))	14.9 (14.7 <i>,</i> 15.0)	15.0 (14.8, 15.2)	14.7 (14.5, 15.0)	0.166
All children	432	255	180	
Height z-scores (mean (95% CI))	-1.85 (-1.96, -1.74)	-1.61 (-1.73, -1.49)	-2.22 (-2.40, -2.03)	< 0.001
Weight z-scores (mean (95% CI))	-1.74 (-1.84, -1.64)	-1.67 (-1.79, -1.55)	-1.84 (-2.02, -1.67)	0.086
BMI z-scores (mean (95% CI))	-0.66 (-0.75, -0.56)	-0.83 (-0.96, -0.71)	-0.39 (-0.54, -0.24)	< 0.001
Stunted (%)	43.0	30.7	62.0	< 0.001
Underweight (%)	38.3	34.7	43.7	0.105
MUAC(cm)^ (mean (95% CI))	16.0 (15.8, 16.2)	16.2 (15.9, 16.4)	15.8 (15.5, 16.0)	0.039

Table 4.5. Nutritional status of all children and specifically children under the age of five in Minyenye, Mbwei and both villages combined

 MUAC(cm)^ (mean (95% Cl))
 16.0 (15.8, 16.2)
 16.2 (15.9, 16.4)
 15.8 (15.5, 16.0)
 0.039

 *T tests used for difference for normally distributed means, Mann-Whitney tests used for non-normally distributed means, chi squared used for difference in proportions between the two villages

^MUAC only collected in children up to 15 years of age

4.4.5 Breastfeeding and complementary feeding of children

Almost 100% of children under the age of five years had been breastfed (Table 4.6). These children were breastfed until approximately 23 months in Min yenye and, significantly longer (27 months) in Mbwei. Despite children being breastfed longer, liquids were introduced significantly earlier in Mbwei; at an average of 4.5 months compared to 5.3 months in Minyenye. The opposite trend was seen for solids; they were introduced at 10.5 months in Mbwei and 8.3 months in Minyenye.

Children in Minyenye were more likely to have been weaned onto a porridge made from multiple grains with or without additions such as oil, beans or fish while children in Mbwei were more likely to have been weaned onto a single flour porridge without additions. Millet juice was used as a weaning liquid in Minyenye only. *Ugali* was more likely to be introduced in Minyenye and other staples such as cassava and potato were more likely to be introduced in Mbwei. Vitamin A rich vegetables and other vegetables and fruit were used more frequently as weaning foods in Minyenye compared to in Mbwei.

	All	Minyenye	Mbwei	P-value for difference*
Breastfeeding (N)	208	106	102	
Breastfed(%)	99.5	100.0	99.0	0.307
Age breastfed until (months, mean(95% CI))	25.1 (24.1, 26.1)	23.3 (22.3, 24.2)	27.2 (25.5, 29.0)	< 0.001
Complementary feeding: Liquids (N)	201	104	97	
Age introduced liquids (months, mean(95% CI))	4.9 (4.6, 5.2)	5.3 (5.0, 5.6)	4.5 (4.0, 5.1)	0.002
Introduced as first liquids(%):				
Single flour porridge(incl. with sugar/salt)	49.8	27.9	73.2	< 0.001
Multiple flour porridge eg. Maize, millet (incl. with				
sugar/salt)	6.5	12.5	0.0	< 0.001
Single flour porridge with additions eg. Beans, oil.	13.9	8.7	19.6	0.025
Multiple flour porridge with additions eg. Beans, oil.	21.9	40.4	2.1	< 0.001
Cow's milk	14.9	11.5	18.6	0.163
Milletjuice	11.9	23.1	0.0	< 0.001
Otherliquids	14.4	19.2	11.3	0.122
Complementary feeding: Solids (N)	181	96	85	
Age introduced Solids (months, mean(95% CI))	9.3 (8.7, 10.0)	8.3 (7.7, 8.9)	10.5 (9.2, 11.7)	0.001
Introduced as first solids(%):				
Ugali	91.2	97.9	83.5	0.001
Other staples eg. rice, potatoes, cassava, yams	25.4	11.5	41.2	< 0.001
Vitamin A rich vegetables	13.3	25	0.0	< 0.001
Other fruit and vegetables	17.7	24.0	10.6	0.019
Beans, meat, fish, eggs	15.5	19.8	10.6	0.087
Other solids eg. biscuits	7.7	8.3	7.1	0.749

Table 4.6. Breastfeeding and complementary feeding of children under five in Minyenye, Mbwei and both villages combined

*Mann-Whitney tests used for difference in means, chi squared used for difference in proportions

- 104 -

4.4.6 Dietary diversity and food variety in relation to nutritional status

The dietary diversity score was not significantly associated with any of the nutritional status variables in either the unadjusted or the adjusted regression models (Table 4.7). Additionally the food variety score was not significantly associated with height, weight or BMI z-scores or MUAC in the adjusted model. The borderline statistically significant negative association seen between the food variety score and height for age z-scores disappears in the adjusted model. Similar results were seen when children who were still being breastfed were excluded from the analysis. The dietary diversity and food variety scores of the respondents were not significantly associated with BMI (results not shown).

Some of the food group components that make up the dietary diversity score are significantly associated with the nutritional status variables. In unadjusted models children who consumed vitamin A rich fruit or vegetables the previous day had higher height z-scores by **0.48** (Regression coefficient (95% confidence intervals):**0.48 (0.02, 0.94)**). Children eating eggs the previous day had higher weight **(1.31 (0.01, 2.62))** and BMI **(1.38 (0.09, 2.66))** z-scores. Children who consumed meat the day before had **0.46** lower height z-scores **(-0.46 (-0.91, -0.01))**. Similarly children consuming milk the previous day had MUACs approximately 0.8cm lower than those not consuming milk **(-0.76 (-1.45, -0.08))**.

When all the individual foods in the dietary diversity score are added to the same model the only two associations that remain significant are egg consumption and BMI z-scores (1.45 (0.12, 2.79)) and cow milk consumption and MUAC (-0.76 (-1.48, -0.04)). No significant associations are found after adjustment for potential confounders.

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient (95% CI)	Regression coefficient (95% Cl)	Regression coefficient (95% CI)	Regression coefficient (95% CI)
Dietary diversity score				
Unadjusted model^	-0.09 (-0.24, 0.06)	-0.12(-0.24, 0.01)	-0.07(-0.19 <i>,</i> 0.06)	0.01(-0.14, 0.15)
Adjusted model^^	-0.08(-0.23, 0.07)	-0.011(-0.24, 0.03)	-0.05 (-0.18, 0.08)	0.05 (-0.10, 0.20)
Food variety score				
Unadjusted model^	-0.08(-0.15, -0.01)	-0.06(-0.12, 0.00)	0.01(-0.05, 0.07)	-0.01(-0.08, 0.06)
Adjusted model^^	-0.05(-0.13, 0.03)	-0.04 (-0.11, 0.04)	0.00 (-0.07, 0.07)	0.04(-0.05, 0.11)

Table 4.7. Unadjusted and adjusted linear regression results for dietary diversity and nutritional status in children under five for both villages combined

^N for unadjusted model was 112 for height, weight and BMI z-scores and 115 for MUAC.

^^adjusted for village, parents age, highest level of education, height, weight or BMI, whether the household has a mobile phone, household takes action to make water safe, whether the parents have an alternative source of income. N for adjusted model was 112 for height, weight and BMI z-scores and 115 for MUAC. Of these variables the following had missing values (number of missing values in brackets after each variable): respondent's age (23); husband's age (57); husband's highest level of education (9); respondent's height (2); husband's height (77); husband's weight (77); respondent's BMI (2); husband's BMI (77); parental ethnicity (9) and whether the husband earns extra income (8) out of a total of 210 individuals.

4.4.7 Complementary feeding, demographics, sanitation and nutritional status The factor that was most strongly associated with the nutritional status variables was complementary feeding variables. In unadjusted regression models, for each month extra the child was breastfed their weight z-score was 0.01 lower (Regression coefficient (95% confidence intervals): (-0.01 (-0.03, -0.00)) and MUAC was 0.04 higher (0.04 (0.03, 0.05)).
Children who had already had liquids introduced to their diet at the time of interview had lower height (-1.57 (-2.49, -0.64)), weight (-1.75 (-2.54, -0.96)) and BMI z-scores (-0.75 (-1.47, -0.24)) and higher MUAC (1.36 (0.47, 2.24)) compared to those who had not had liquids introduced. The age liquids were introduced is significantly associated with MUAC (0.16 (0.09, 0.23)). For every additional month of age the child was when they had liquids introduced MUAC increased by 0.16cm.

Children receiving multiple flour porridge as their first foods with or without additions had higher heights (with additions: 0.67 (0.25, 1.08), without additions: 0.73 (0.02, 1.43)). However, children receiving single flour porridge with additions had lower heights (-0.65 (-1.14, -0.16)). Children receiving multiple flour with additions had lower BMI z-scores (-0.52 (-0.84, -0.20)). Those consuming cow's milk as one of the first liquids introduced to the diet had significantly lower height z-scores (-0.52 (-1.02, -0.02)). While those consuming other liquids had significantly higher BMI z-scores (0.65 (0.28, 1.02)). Children who had received solids at the time of interview had lower height (-0.72 (-1.24, -0.20)), weight (-1.05 (-1.48, -0.62)) and BMI (-0.68 (-1.07, -0.28)) z-scores but higher MUAC (1.24 (0.78, 1.69)). The older the children were when solids were introduced the lower their height (-0.04 (-0.08, -0.01)) and weight (-0.05 (-0.08, -0.02)) z-scores were and the higher their MUAC was (0.07 (0.04, 0.10)).

Associations between these complementary feeding variables and nutritional status are adjusted for potential confounders and presented in Table 4.8. Children who were breastfed for longer had lower weight z-scores but higher MUAC. The older the child was when liquids were introduced the lower their height and the higher their MUAC were. For each additional month of age the child was when liquids were introduced height was 0.07 z-scores lower and MUAC was 0.17cm larger. Children who received multiple flour porridge with additions as a first complementary food had lower BMI z-scores and MUAC and children receiving single flour porridge had higher MUAC. Children receiving millet juice had lower height and weight zscores. Children receiving other liquids such as water, fruit juice , tea or soda as the initial complementary foods had lower BMI z-scores.

In the adjusted model those children who had already had solids introduced into their diets had lower height, weight and BMI z-scores and higher MUAC. For every additional month of age at which solids were introduced height decreased by 0.03 and weight decreased by 0.04

while MUAC increased by 0.07cm. As with the unadjusted regression no specific solids were significantly associated with any of the nutritional status variables.

Table 4.8. Adjusted* linear regression results for complementary feeding and nutritional status in children under five for both villages combined

	Height z-scores	Weight z-scores	BMI 7-scores	Μυρο
	Regression coefficient	Regression coefficient	Regression coefficient	Regression
	(95% CI)	(95% CI)	(95% CI)	coefficient (95% CI)
Age breastfed until (months)	-0.01 (-0.02, 0.01)	-0.02 (-0.03, -0.00)	-0.01 (-0.02, 0.00)	0.04 (0.02, 0.05)
Liquids introduced (Y/N)	-1.35 (-2.21, -0.48)	-1.80 (-2.55, -1.05)	-0.89 (-1.60, -0.18)	1.17 (0.30, 2.04)
Age introduced liquids (months)	-0.07 (-0.15, -0.00)	-0.05 (-0.11, 0.02)	0.02 (-0.04, 0.08)	0.17 (0.10, 0.24)
Single flour porridge (incl. with sugar/salt) (Y/N)	0.35 (-0.07, 0.77)	0.09 (-0.27, 0.45)	-0.19 (-0.53, 0.15)	0.66 (0.25, 1.06)
Multiple flour porridge e.g. Maize, millet (incl. with	/			
sugar/salt) (Y/N)	0.01 (-0.75, 0.76)	0.11 (-0.53, 0.75)	0.04 (-0.55, 0.64)	-0.03 (-0.77, 0.71)
Single flour porridge + additions e.g. Beans, fish (Y/N)	-0.28 (-0.80, 0.24)	-0.06 (-0.51, 0.38)	0.27 (-0.14, 0.69)	0.08 (-0.44, 0.60)
Multiple flour porridge + additions e.g. Beans (Y/N)	0.43 (-0.04, 0.90)	0.10 (-0.32, 0.52)	-0.43 (-0.81, -0.05)	-0.49 (-0.96, -0.02)
Cow's milk (Y/N)	-0.36 (-0.89, 0.13)	-0.14 (-0.57, 0.29)	0.03 (-0.37, 0.44)	-0.05 (-0.54, 0.44)
Millet juice (Y/N)	-0.86 (-1.39, -0.33)	-0.54 (-1.01, -0.07)	0.16 (-0.28, 0.61)	-0.17 (-0.72, 0.39)
Other liquids (water, fruit juice, tea, soda) (Y/N)	-0.20 (-0.66, 0.27)	0.08 (-0.32, 0.48)	0.55 (0.16, 0.93)	-0.04 (-0.52, 0.44)
Solids introduced (Y/N)	-0.72 (-1.21, -0.22)	-1.10 (-1.51, -0.69)	-0.73 (-1.12, -0.33)	1.11 (0.65, 1.57)
Age introduced solids (months)	-0.03 (-0.06, -0.00)	-0.04 (-0.07, -0.01)	-0.03 (-0.05, 0.00)	0.07 (0.04, 0.10)
Ugali (Y/N)	-0.06 (-0.69, 0.57)	0.06 (-0.47, 0.59)	0.09 (-0.42, 0.59)	0.59 (-0.04, 1.23)
Other staples (rice, potatoes, cassava, yams) (Y/N)	0.38 (-0.04, 0.79)	0.18 (-0.17, 0.52)	-0.21 (-0.53, 0.12)	-0.10 (-0.51, 0.32)
Vitamin A rich vegetables (Y/N)	-0.38 (-0.93, 0.17)	-0.17 (-0.63, 0.29)	0.18 (-0.25, 0.62)	-0.22 (-0.77, 0.33)
Other fruit and vegetables (Y/N)	-0.06 (-0.52, 0.41)	-0.00 (-0.39, 0.38)	-0.08 (-0.45, 0.29)	0.05 (-0.41, 0.52)
Beans, meat, fish, eggs (Y/N)	-0.05 (-0.54, 0.43)	0.03 (-0.38, 0.43)	-0.05 (-0.43, 0.33)	-0.18 (-0.66, 0.30)
Other solids (biscuits, mandazi) (Y/N)	-0.18 (-0.81, 0.45)	-0.14 (-0.67, 0.38)	0.04 (-0.46, 0.54)	-0.61 (-1.23, 0.02)

*Adjusted for village, parents age, parents ethnicity, highest level of education, height, weight or BMI, whether the household has a mobile phone, household takes action to make water safe, whether the parents have an alternative source of income. N was 193 for height, 195 for weight, 196 for BMI z-scores and 199 for MUAC. Of these variables the following had missing values (number of missing values in brackets after each variable): respondent's age (23); husband's age (57); husband's highest level of education (9); respondent's height (2); husband's height (77); husband's weight (77); respondent's BMI (2); husband's BMI (77); parental ethnicity (9) and whether the husband earns extra income (8) out of a total of 210 individuals.

In unadjusted regression analysis for each additional cm in the respondent's height their under five child's height increased by **0.05cm (95% CI: 0.03, 0.08**). This figure was **0.08cm (0.04, 0.11)** for the husband's height. Similarly, for each additional kg of respondent's weight their under five year old's weight was **0.04kg (0.02, 0.06)** higher. For each additional point of respondent BMI the under five year old's BMI was **0.06kg/m2 (0.02, 0.10)** and their MUAC was **0.07cm (0.01, 0.13)** higher. For each additional point of the husband's BMI their under five year old's BMI was **0.06kg/m2 (0.02, 0.10)** and their under five year old's MUAC was **0.07cm (0.01, 0.13)** higger.

Parents ethnicity was significantly associated with under five year old's height (-0.23 (-0.40, - 0.07) and BMI z-scores (0.15 (0.03, 0.27) but not weight (-0.07 (-0.21, 0.07)) or MUAC (0.02 (-0.13, 0.18)). Height for age z-scores were highest when both parents were Nyantulu (-1.72) and similar if both parents were Pare (-2.32), Sambaa (-2.29), or mixed/other (-2.39). BMI for age z-scores were highest for under five year olds with mixed or other parental ethnicity (0.21), followed by those with Pare parents (-0.14), with Nyantulu (-0.30) and Sambaa (-0.33) parents having children with similar BMI z-scores.

Households taking action to make water safe for drinking, for example boiling, had under five year olds with **0.18 (0.03, 0.33)** higher weight z-scores in the unadjusted model. This association was no longer significant when adjusted for confounders (village, respondent's and her husband's age and husband's frequency of employment) (0.16 (-0.01, 0.34). The source of drinking water was significantly associated with under five year old's height z-scores in the unadjusted model only (-0.15, (-0.27, -0.17). Height z-scores were -2.27 for households getting water from rivers or lakes, -1.88 in households using piped water, -1.84 where a borehole was the source of water and -1.59 for households using dug wells.

In the adjusted model whether the household did something to make water safe was significantly associated with weight z-scores; those households boiling water had under five year olds with weight z-scores 0.28 higher than those using other methods and 0.56 higher than those doing nothing (regression coefficients (95%CI): -0.28 (-0.54, -0.02) . Compared to in houses with open pit latrines, under five year olds in houses with closed pit latrines had 0.37(0.04, 0.70) higher MUAC and 0.19 (0.01, 0.36) higher weight for age z-scores. Source of drinking water, taking action to make water safe, time to get water and type of toilet were not significantly associated with other measures of nutritional status in either the unadjusted or the adjusted models (of the variables in the model the following had missing values in under five year olds (number of missing values in brackets after each variable): respondent's age (23); husband's age (57); husband's frequency of employment (9) out of a total of 210 individuals. These number were: respondent's age (59); husband's age (128); husband's frequency of employment (24) out of a total of 532 individuals for all children.)

Seventy-one percent of all respondents (77% in Minyenye and 66% in Mbwei) felt their family did not get enough food and 83% (88% in Minyenye and 78% in Mbwei) felt they did not get enough variety of food.

4.5 Discussion

4.5.1 Nutritional status of children

Growth rates of the children in this study were poor. On average under five year olds were stunted, as defined as two standard deviations (SD) below the median of the international WHO Child Growth Standards in height for age z-scores. The rates of stunting seen in this study were higher than rates seen in the Tanzanian 2010 demographic and health survey (STAT compiler, 2014) (chapter 7, Table 7.4) but broadly similar to those seen in other Tanzanian articles (Beasley et al., 2000, de Onis et al., 2012).

Both villages, but especially Mbwei, had high rates of low weight for age z-scores. Low height for age, or stunting, reflects a failure to reach the expected linear growth for the child's age (de Onis et al., 1997). This occurs over a longer time period than low weight for age or underweight which reflects a lower than average weight for a given age and is influenced by short term factors such as diarrhoea (Rowland et al., 1988). The high rates of both stunting and underweight seen in this study indicates long term adverse conditions where the children have not reached their height potential combined with current adverse conditions where they are not as heavy as would be expected for their already reduced heights. MUAC reflects short term nutritional status. The WHO standards for the definition of severe acute malnutrition is a MUAC of below 11.5cm (Gollogly, 2009). The mean and 95% Confidence Intervals of MUAC showed it to be at acceptable levels in both villages.

4.5.2 Dietary diversity and nutritional status

Participants in this study had average dietary diversity, five out of nine food groups, but low food variety, a total of 7.5 different food items. In much of the literature in this area dietary diversity (Arimond, 2004, Corbett et al., 1992, Steyn et al., 2006, Garg and Chadha, 2009, Nti and Lartey, 2008, Savy et al., 2005) and food variety (Hatloy et al., 2000, Sawadogo et al., 2006, Saibul et al., 2009, Onyango et al., 1998) are significantly, positively associated with nutritional status, something that was not found in this study. In the adjusted regression models no significant associations were seen between dietary diversity or food variety and height, weight or BMI z-scores or MUAC. This result does not support the pathway from dietary diversity to nutritional status illustrated in figures 3.2 a) and b).

There are a number of possible reasons for this. Firstly, there are many factors which have an impact on nutritional status. Infectious diseases in the first five years of life has a significant impact on children's nutritional status (Victora et al., 1990). Many individuals enter into a cycle of malnutrition and infection each exacerbating the other leaving the individual more malnourished and more at risk of infection (Chen et al., 1980). Foods introduced into a child's diet when they are being breastfed have an important impact on their nutritional status (Onyango et al., 1998, Obatolu, 2003). Additionally, vaccinations (Dancer et al., 2008), access to clean drinking water and living with poor sanitation (Fink et al., 2011, Kikafunda et al., 1998) are associated with stunting in children under the age of five. Nutritional status in children has also been liked to their parents' literacy (Fernandez et al., 2002) and maternal education (Kabubo-Mariara et al., 2009, Abuya et al., 2012). These other factors may be enough to offset or mask the effect of dietary diversity on nutritional status in this study. Duration of breastfeeding, introduction of complementary foods and sanitation were assessed in the current study and will be discussed further in the next section. Additionally, dietary diversity varied little across households in the villages, 95% of the study population had dietary diversity between 4.6 and 5.0. There may not be enough variation in dietary diversity to detect an effect on nutritional status.

Secondly, some methodological factors may have affected results. Dietary diversity was assessed cross-sectionally at the same time the nutritional status measures were taken. This assumes that dietary diversity is static and a cross sectional measure can represent dietary diversity of the past. This may be too great an assumption; the study design limits the likelihood an association between dietary diversity and nutritional status would be detected, even if it did exist in this population. However, a number of other studies have found significant associations between dietary diversity (Arimond, 2004, Garg and Chadha, 2009, Savy et al., 2005, Sawadogo et al., 2006, Hatloy et al., 2000) and food variety (Onyango et al., 1998, Hatloy et al., 2000, Saibul et al., 2009, Sawadogo et al., 2006) and nutritional status outcomes measured cross-sectionally.

Lastly, the data was collected in June and July, times of relative food plenty. In the previous year food diversity dropped from approximately 3.5/6 in June to 1.5/6 in November, December and January when looking at foods available to the household through foods grown, found wild and gifted to the household. Perhaps if dietary diversity was measured at a time of

111

- 111 -

food shortage an association between dietary diversity and nutritional status would have been seen.

- 112 -

4.5.3 Complementary feeding, demographics, sanitation and nutritional status

Some complementary feeding, demographic and sanitation factors are associated with nutritional status in these villages. Children under five in Mbwei were more likely to be stunted than children in Minyenye. There were a number of differences between the two villages that may explain why those in Mbwei had poorer nutritional status. Ethnicity was significantly associated with height and weight so the biophysical characteristics of different tribes living in the two villages may explain some of the difference seen between the villages. Habicht et al (1974) found that the effect of ethnicity on the growth of young children in a range of low and high income countries was small compared to environmental factors. While Ebomoyi et al (1991) found ethnicity to be significantly associated with birth weight in Nigeria and a Tanzanian study showed fathers ethnicity to be significantly associated with perinatal mortality (Habib et al., 2008). Proos called for local reference growth data for different ethnic groups and regions to be developed (Proos, 1993). Ethnicity, in this study could have impacted on the nutritional status results but it is difficult to remove its effects from the effect of the village as the majority of people from Minyenye were Nyantulu and the majority of those from Mbwei were Pare.

World Vision has been working in Singida district since 2004 and Minyenye village has been a target for their education and nutrition programs (World Vision Tanzania, 2009). One of their main achievements outlined in their 'Essential nutrition package Mtinko and Kinampanda annual report 2011' is the improvements made to the porridge used in complementary feeding (World Vision Tanzania, 2011). This is reflected in the results of this project with Minyenye being more likely to use a multiple grain porridge and to make additions of oil, beans and fish to their porridge. Although respondents in Minyenye were more likely to report better weaning practices such as introducing multiple grain porridge, porridge with high nutrient additions, vitamin A rich vegetables and fruits and vegetables, this did not have a significant impact on children's nutritional status, in this study.

On the contrary, the practice of introducing multiple flour porridge with additions and millet juice, which was carried out almost exclusively in Minyenye, was negatively associated with the nutritional status outcomes. Respondents in Mbwei introduced liquids to breastfed infants significantly earlier than those in Minyenye. These liquids were typically porridge made with water, as was the porridge and millet juice in Minyenye. This may introduce a risk of infection

to the children earlier which could impact on their nutritional status negatively (Motarjemi et al., 1993). Gross et al (2000) found Vietnamese children who had early introduction of complementary foods (typically rice water and rice porridge) to have a higher rate of infection and poorer anthropometric outcomes than children that continued to be exclusively or predominantly breastfed at ages 1-12 months. Children in Mbwei did not receive solids until later than their Minyenye counterparts leaving them at risk of infection from unclean drinking water without sustenance from solid food for longer. This may contribute to the lower height and weight z-scores in Mbwei.

There is an established association between infection and nutritional status in under five year olds (Stephensen, 1999). Although this study did not collect data on markers of infection there are a number of reasons why Mbwei might be expected to have a higher rate of infection which may have contributed to the higher rates of stunting in this village. Firstly the majority of participants in Mbwei collected their drinking water from rivers which were also used as watering holes for livestock. Many infectious organisms are transmitted via the faecal-oral route through contamination of drinking water (Fayer et al., 2000). Different sources of water were not, however, associated with nutritional status. Some of the increased risk of infection may be offset by the higher proportion of people boiling water in Mbwei as wheth er the household took action to make water safe to drink was associated with weight in under five year olds. Open pit latrines, which were the most common type of toilet in both villages, was associated with poorer nutritional status. Poorer sanitation has previously been linked to poorer health outcomes in low income countries (Fink et al., 2011, Esrey, 1996). Making water safe to drink and the type of toilet the household had, along with the difference in complementary feeding, may still have impacted on rates of infection and nutritional status in these two villages.

There were no significant differences seen in dietary diversity between the two villages. Perhaps using a summary score of dietary diversity means we lose important dietary information. It may be the information that makes up these scores that is most valuable. When looking at the components of the dietary diversity score a higher proportion of those in Minyenye consumed 'vitamin rich fruit and vegetables' and 'other vegetables' and a lower proportion consumed 'oils and sweets', 'beans, nuts, seeds', and 'milk and dairy products'. A significant positive relationship between 'vitamin A rich fruit and vegetables' and height and a negative relationship between milk consumption and MUAC were found in unadjusted models. When other dietary diversity components were taken into account only egg consumption and milk consumption showed significant associations and when potential confounders were adjusted for these associations disappeared. Only two children consumed eggs the previous

day so it is not appropriate to generalise this result to the wider population. However, the difference in vitamin A rich food and milk intake between the two villages may have contributed to their different rates of stunting.

- 114 -

Vitamin A deficiency has been shown to be causally associated with poor growth (Tarwotjo et al., 1992). A possible mechanism for this association is the protective effect vitamin A has against infection (Sommer et al., 1984). Milk consumption has not been shown to be associated with malnutrition, in fact animal products are often recommended to improve nutritional status in low income countries (Gibson et al., 2003). Kikafunda et al. (1998) found children in Uganda who had never consumed cow's milk to have a significantly higher rate of underweight that those who had consumed milk. Grillenberger et al (2006) showed in Kenya that growth was improved by providing a milk supplement to school children. However a number of respondents mentioned during the course of data collection that milk is often watered down with drinking water in these areas. This practice could introduce a risk of infection.

Other dietary components and practices may also help account for this difference in nutritional status. The majority of participants in Minyenye consumed Mlenda (Corchonus tritocularies Tillaceae), a wild green leafy vegetable that grew freely in the area. It was a very important part of their diet with many participants consuming only ugaliand mlenda as their main meal. *Mlenda* is eaten fresh but is also dried and ground to be used throughout the year. *Mlenda* is high in iron and calcium (Kinabo et al., 2006) and having this food available year round is expected to be an important contributor to the food security of people living in Minyenye. This is illustrated in the difference between Figure 4.4 and 4.5 which shows that Mbwei has more household dietary diversity than Minyenye in May and June, this difference disappears when wild food is included in Figure 4.5. Minyenye has a higher diversity of food available in January, February, April and November when wild food is taken into consideration, highlighting how important wild food can be to dietary intake. The importance of wild food to those living in rural sub-Saharan Africa has been extensively discussed in the literature (Harris and Mohammed, 2003, Nordeide et al., 1996, Johns et al., 1996, Vainio-Mattila, 2000, Bharucha and Pretty, 2010) and has been shown to be an important source of energy and micro-nutrients (Nordeide et al., 1996), especially at times of food scarcity (Harris and Mohammed, 2003).

Households in Mbwei were more likely to grow and eat cassava compared to Minyenye. Cassava typically provides enough calories but inadequate protein, iron, zinc and vitamin A (Stephenson et al., 2010). Gregios et al (2010) found the proportion of the diet made up of cassava was inversely correlated with vitamin A, zinc and iron intake. Cassava, however, is an

important crop for food security as it can be stored in the ground, it grows in poor conditions with limited input and is unusually tolerant (El-Sharkawy, 2007). It can help maintain energy intake during the hungry season and provide an in ground food store in case of crop failure (Prudencio and Al-Hassan, 1994). Both the higher growth and higher consumption of cassava in Mbwei may indicate poorer food security in this area.

Minyenye had significantly lower rates of stunting than Mbwei but their under five year olds were still 41% stunted, rates higher than outlined in the millennium targets for 2015 (United Nations, 2014). Reasons for this high rate of stunting found in both villages, in addition to infection and the types of foods introduced to breastfed children discussed above, includ e the high level of poverty and low food security found in rural Tanzania (Hadley et al., 2007). The households participating in this study were poor with none of the houses having electricity and only half reporting owning a radio. Approximately three quarters of the participants reported not getting enough food as well as not getting enough variety of food. All these factors are likely to contribute to the poor nutritional status seen in these villages. These factors relating to poverty and acting through diet and infection, were illustrated by the social/economic/care/health factors section of the conceptual framework (Figure 3.2 a) and b)). In comparison to agrobiodiversity, these factors have a large impact on child nutritional status.

4.5.4 Limitations and strengths

This study has a number of limitations that need to be acknowledged, in addition to the cross sectional nature of the study and the seasonal effects discussed above. As with all studies that rely on a translator there will be inaccuracies in the information translated and interesting detail in the participants responses may have been lost. The cross-cultural dynamic of the study may have affected the honesty of the responses from the participants (Twyman et al., 1999). The researcher was white and from a high income country and it is possible that some of the respondents exaggerated the difficulties they faced in the hope that they would receive aid from the research project. It is also possible that the participants underplayed the difficulty of their circumstances if they felt embarrassed in front of the research team. The cross-cultural dynamic could have affected the answers the participants gave, the way this was translated and how this has been interpreted by the researcher in a number of different ways (Twyman et al., 1999). This needs to be taken into consideration when interpreting the results of the study. The household questionnaire collected mainly quantitative data, collecting more qualitative data may have provided additional insights into the determinants of nutritional status in these communities.

- 115 -

Additionally the interview often took place within hearing distance of other family and community members and this may have affected how the participants answered the questions. The interview was detailed and took between 1 and 1.5 hours to complete. The researcher expects the quality of the responses to have decreased over the course of the interview.

Information collected on complementary feeding would have been affected by recall bias as it was collected on all children under the age of five years. Anthropometric measures were taken outside the family home, almost exclusively on mud ground. There were no truly flat surfaces to take the height and weight measures which may have decreased the accuracy of the results. Similarly some of the young children were upset, making taking accurate anthropometric measures more difficult. The researchers did their best to minimise the effect of the above limitations on the data collected but expect them still to have had an effect. The effect of clustering at the village level was taken into consideration for the sample size calculation; the study is powered to detect differences between the two villages. This calculation did not however allow for clustering at the household level. As approximately 55% of the households in Minyenye and 65% of Mbwei households had more than one under five year old, this may have decreased the studies power to detect differences in nutritional status between the two villages.

Using a set of indicators to represent household wealth, such as the DHS wealth index (Rutstein et al., 2004) is a useful approach in a low income country setting. However, wealth was not addressed directly by this study as it was outside the scope of the study's objectives. Using individual variables as proxies for wealth was therefore considered sufficient to act as potential confounders in regression analyses. Factors potentially associated with wealth were added to data analyses to try and control for wealth in these villages. The only variable that impacted on the regression coefficients, indicating it was a confounding variable, was whether the household owned a mobile phone. This variable was therefore used as a proxy for wealth in a number of the data analyses.

The limitation of this approach is that mobile phone ownership may be confounding the relationships for reasons other than its relationship to household wealth. For example, improved communication may be having a positive impact on nutritional status. The other limitation is that, if it is representing wealth, it is just one aspect of wealth when a wealth index summarises many aspects of wealth making it a more accurate estimate.

This study comprehensively collected paired data on dietary diversity and nutritional status on a randomly selected sample within the two villages. This data and information on other potential determinants of nutritional status in these villages was collected specifically for this

project and data collection methods were tailored to collect the data needed to meet project objectives.

4.6 Conclusions

Malnutrition was high in these communities but typical for Tanzania. This is likely to be due to high levels of poverty, low food security, poor access to clean water and poor complementary feeding leading to high rates of infection among other inter-related factors. This study identified factors that may have been responsible for the higher rates of stunting in Mbwei. These included differences in ethnicity between the two villages, the early introduction of liquids and late introduction of solids in Mbwei, Mbwei's poorer access to clean drinking water and greater time taken to collect water. With the exception of ethnicity, all of these factors raise the risk of infection in children which, along with food intake, has been identified as the major determinant of nutritional status in under five year olds in low income countries.

Participants in Minyenye and Mbwei had average dietary diversity but low food variety. There was no relationship found between dietary diversity or food variety and nutritional status in these communities and no difference in dietary diversity or food variety between the two villages. It should be acknowledged that these results may have been influenced by design, methodological and seasonal limitations of the study. However, other dietary factors may have contributed to the difference in height and weight between Minyenye and Mbwei. Specifically, the higher intake of vitamin A rich fruit and vegetables and lower intake of potentially contaminated milk in Minyenye may have played a part. Additionally, the high proportion of households consuming wild green leafy vegetables in Minyenye may have buffered food intake at times of food shortage while the higher proportion of households in Mbwei eating cassava may indicate worse food security in this village.

The high rates of stunting in children under five years old shows that there is still a great need for interventions to improve nutritional status these villages, and most likely others like them. The multiple determinants of nutritional status discussed in this chapter highlight the difficulty of intervening in order to improve nutrition and health outcomes. This chapter provides evidence to support broad nutrition interventions which address the wide range of factors , such as sanitation and complementary feeding, shown to be related to nutritional status.

Chapter 5:

Investigating the associations between plant agrobiodiversity and both dietary diversity and nutritional status in Minyenye and Mbwei, rural Tanzania.

5.1 Chapter summary

The previous chapter outlined the relationship between dietary diversity and nutritional status as well as the associations between other demographic, social and dietary factors and nutritional status. This chapter aims to investigate the relationship between plant agrobiodiversity and both dietary diversity and nutritional status in households reliant on subsistence farming in rural Tanzania. It reports on plant agrobiodiversity measured crosssectionally in the household farms as well as the number of crops and vegetables grown over the previous calendar year. These second sets of measures of plant agrobiodiversity will be referred to as the crop and vegetable diversity scores. Whether selling different types of crops is associated with dietary diversity and nutritional status is also reported in this chapter.

This chapter meets objective 2 (Systematically assess the diversity and abundance of both cultivated and wild plants growing on household agricultural land in two villages in rural Tanzania and investigate how this is related to dietary diversity in under five year old and nutritional status in children in these villages) and contributes to meeting objective 3 (Measure household crop and animal diversity, dietary diversity of respondents and children under five and nutritional status of children living in these villages, determine whether household produce is sold and investigate how these factors are related) and objective 4 (Investigate basic socio-demographic factors affecting dietary intake of children under five and nutrition outcomes in children in these villages) of this thesis. The objectives of this chapter are to:

- Objective 5A: Present descriptive data on habitat, species present, farm characteristics, cross-sectional plant agrobiodiversity and crop and vegetable diversity scores in the two villages.
- Objective 5B:Investigate whether plant agrobiodiversity and crop/vegetable diversityscores are associated with dietary diversity and nutritional status in

children.

Objective 5C: Investigate whether selling staple crops, vegetables, fruit and other produce is associated with dietary diversity and nutritional status in children.

Agrobiodiversity is not associated with dietary diversity or food variety in Minyenye or Mbwei. Households with higher annual crop or vegetable diversity scores, however, had individuals with more diversity in their diets. Associations between agrobiodiversity and crop diversity scores and nutritional status are mixed in Minyenye but negative in Mbwei. Individuals from households selling produce they grew had higher dietary diversity in both villages. Households selling produce had the same or better nutritional status in Mbwei but results in Minyenye were mixed.

Only one study (Dewey, 1981) has linked these factors together within a study, tracing associations between food production, consumption and growth. This study therefore provides important data on the association between agrobiodiversity, dietary diversity and nutritional status and shows that these associations are not to be assumed. This study informs future interventions intending to use improvements in plant agrobiodiversity as a tool to improve health and encourages researchers and development agencies to explore some of the barriers to the pathway from diversity in crops grown to diversity in diets to health outcomes.

5.2 Introduction

Agrobiodiversity has been defined as the biological diversity on lands used for agricultural purposes (Brookfield and Stocking, 1999). It includes all aspects of biological diversity which affect agriculture and food; the diversity of plants, animals and micro-organisms at species and ecosystem levels (Cromwell et al., 1999). This chapter focuses on plant agrobiodiversity. There has been discussion in the literature about the importance of improving agrobiodiversity to improve food security, diet and nutrition (Thrupp, 2000, Frison et al., 2011). An increase in the types of crops grown also opens up the potential for these crops to be sold, supplementing the households income and potentially improving nutritional status through this pathway (Shack et al., 1990). Crops may also be specifically grown for sale with none of the grown crops being eaten within the household.

Agrobiodiversity has been highlighted as essential in the sustainable delivery of a more secure food supply (Frison et al., 2011, Thrupp, 2000). According to Frison et al (2011) the more

diverse farming systems and crops are, the more resilient farming systems are to shocks and changes in the climate. Thrupp (2000) outlines the problems associated with agrobiodiversity loss. These are most relevant for plant agrobiodiversity: disruption of ecosystem services including water retention, nutrient cycling and decomposition which leads to decreases in productivity; erosion of genetic resources of crops and livestock leading to increased risk and decreased food security; erosion of insect diversity leading to decreased pollination and increased susceptibility; erosion of soil diversity leading to fertility loss and decreases in productivity; loss of habitat diversity including wild foods and loss of indigenous methods and biodiversity knowledge.

Ecosystem services provided by biodiversity are considered in detail by Altieri (1999) who discusses how the disruption in these ecosystem services are linked to reductions in food production. Tilman et al (1996) provided experimental evidence supporting the diversity-sustainability theory that the sustainability of soil fertility is reliant on plant biodiversity. They found that ecosystem productivity increased and nitrogen loss decreased with higher plant diversity.

Two distinct types of plant-based biodiversity are present in the majority of agricultural systems. The first is the biodiversity of the crops planted by the farmer. The second is the wild plants as well as the soil flora and fauna, pollinators, decomposers, herbivores and carnivores associated with this planned biodiversity (Altieri, 1999). How these two different kinds of biodiversity interact and impact on ecosystem function is outlined in Figure 5.1.



Figure 5.1. The relationship between planned biodiversity and associated biodiversity and how they promote ecosystem function (presented by (Altieri, 1999), modified from Vandeermeer and Perfecto, 1995)

Alongside conservation rationales, one of the most important reasons cited for improving agrobiodiversity is to improve food security (Thrupp, 2000) and dietary diversity (Frison et al., 2006) with the hope that this will lead to improvements in nutritional status and health in malnourished populations. For these reasons, it has become an important focus of discussions around delivering a sustainable food supply (Wahlqvist and Specht, 1998, Chivian, 2002, Frison et al., 2006, Gotor, 2010). To date there has been very little empirical research into how plant agrobiodiversity is related to consumption and health.

Some evidence points to a relationship between agrobiodiversity, dietary diversity and nutritional status. Increased agricultural diversity has been linked with a greater production of essential nutrients (Marten and Abdoellah, 1988). Dewey found farmers cultivating more diverse farms to have higher dietary diversity and nutritional status than those farming cash crops (Dewey, 1981). Ekesa et al (2008) found agrobiodiversity to be positively correlated with dietary diversity in a cross-sectional survey in Kenya. However, agrobiodiversity was estimated using number of crops grown rather than accepted measures of biodiversity and the project did not go as far as to link these factors to nutritional status. Other studies have failed to find associations between agrobiodiversity and nutritional status. Shack et al. found no association between number of food crops grown in home gardens and nutritional status in Papua New Guinea (Shack et al., 1990). Kidala et al. (2000) found communities that received a nutrition education intervention leading to the establishment of household and school gardens had higher green leafy vegetable intakes but lower serum retinol concentrations compared to control areas receiving no intervention. The authors concluded that these results were confounded by *helminths* infection. A long term study in Senegal found no improvements in nutritional intake after the establishment of home vegetable gardens, despite a positive impact on women's income (Brun et al., 1989).

The assumption that increased agrobiodiversity leads to increased dietary diversity and improved nutritional status appears logical but perhaps does not acknowledge the complexity of the pathway from food production through consumption to nutrient utilization in people living in low income countries. However, the potential that increased agrobiodiversity can improve the diets and health of subsistence farmers while making an important contribution to biodiversity conservation (Chivian, 2002) is an important research area to be investigated. To contribute further understanding of these relationships this chapter will explore the associations between agrobiodiversity, dietary diversity and nutritional status variables in Minyenye and Mbwei in rural Tanzania.

5.3 Methods

This study aimed to systematically assess the diversity and abundance of cultivated and wild plants on the households agricultural land. In order to meet this objective a systematic measurement of plant biodiversity was chosen as a proxy indicator of agrobiodiversity. The diversity of crops grown by the household over the last calendar year, as reported by the respondent, supplements this data. This is referred to as the crop diversity score. Vegetable diversity scores, referring to the number of different types of vegetables grown over the previous calendar year, are also reported. Data collection methods have been described in detail in chapter 3. Specific data sources, plant biodiversity index calculations and statistical analysis methods are described below.
5.3.1 Data sources

5.3.1.1 Agrobiodiversity

The number of farms the household had, where they were and which were going to be measured were established. The biodiversity research assistant left the interview with a family member, often the husband, to measure the agrobiodiversity on the chosen farms. The diversity of plant species were collected as an indicator of agrobiodiversity using the point intercept method (Coulloudon et al., 1999). Details of how data was collected is presented in chapter 3, section 3.5.4.1. The agrobiodiversity protocol and the data collection sheets are included in appendix C and E.

5.3.1.2 Questionnaire

Additional quantitative data were collected using the questionnaire, including the number of crops the household grew in the last calendar year. This information was used to calculate the crop and vegetable diversity scores. These scores represent the number of different types of crops and vegetables intentionally grown for household consumption in the previous calendar year. Whether they sold any staple crops, vegetables, fruit or other produce (oil, beans, honey or sugarcane) was also collected. Data on potential confounding variables, such as husband's type of employment, was also collected through the questionnaire (See Appendix D for a copy of the questionnaire).

5.3.2 Data calculations

5.3.2.1 Agrobiodiversity index

The Shannon diversity index was used to provide a measure of the number of different plant species and the abundance of these species (Pla, 2004). The higher the index number the more diverse the area. Shannon indices were calculated using plant species data from farms collected using the point intersect method described in chapter 3 and provide an indication of plant agrobiodiversity at the household level. The following Shannon diversity index formula was used:

$$\hat{H} = -\sum_{i=1}^{\hat{S}} \hat{p}_i \ln \hat{p}_i,$$

H= the Shannon diversity index

pi = fraction of the entire population made up of species i

S=number of species encountered

The Shannon index calculations were carried out in excel. The excel spreadsheet contained the list of all the plant species encountered on all the farms measured for a household and the number of times they were encountered through the data collection method. *Pi*was calculated by dividing the total number of times the individual species was encountered by the number any species was encountered. For each species this was multiplied by the natural log of *pi* and these numbers were added together to give the Shannon indices per household. This plant agrobiodiversity index was calculated separately for 1) all plant species and 2) food plant species (species either intentionally grown for food or that could be used for food). Due to the different methods used, diversity indices for farms and vegetable gardens were calculated separately.

5.3.3 Data analysis

5.3.3.1 Descriptive statistics

Qualitative data on indicators of land degradation, recorded by the biodiversity research assistant, and changes in crop production, collected through the questionnaire, were coded into categories to be used in the analysis. Basic descriptive statistics including the number of farms, abundance data and average plant agrobiodiversity indices (the Shannon indices) were calculated for all plants and food plants for each village (chapter objective 5A). Mann Whitney and Chi-squared tests were used to detect significant differences between the villages in plant agrobiodiversity indices and farm characteristics.

5.3.3.2 Linear regression analysis

Linear regression was used to answer the main aim of this study; to estimate the relationship between plant agrobiodiversity, dietary diversity and nutritional status (chapter objective 5B). Relationships between selling staples, vegetables, fruit and other produce and dietary diversity and nutritional status was investigated using linear regression (chapter objective 5C). The relationship between farm characteristics and dietary diversity and nutritional status was also investigated using linear regression in order to provide more information on the context of the agrobiodiversity measures.

Data collected on transects laid every 20 metres was used to calculate Shannon indices for both villages to make like for like comparisons. However, for the regression analysis, when villages were analysed separately, data collected on transects every 10 metres was used for Mbwei as this more detailed level of data collection was more appropriate for Mbwei's smaller farms.

Both unadjusted and adjusted regression models are presented. For adjusted regression models potential confounders were identified using the Directed Acyclic Graph (Glymour, 2006) (Figure 3.15) and confounders were selected as described in chapter 3 (page 73). Variables included in the dietary diversity/food variety regression models included: husband's type of employment; husband's frequency of employment; number of farms; average farm size. The village the participant lived in was the only variable identified through the DAG which substantially altered the regression coefficients in the agrobiodiversity nutritional status regression models. As the regression coefficients varied substantially between the two villages presenting regression results separately for Minyenye and Mbw ei was felt to be more informative than using multivariable regression.

5.3.3.3 Missing data

Multiple imputations using chained equations were used for the multivariable regression analyses for the associations between selling crops and dietary diversity. Imputations were done separately by village for respondents and those under five. The missing values were imputed based on the complete variables in the model: whether the household has a mobile phone; respondent's highest level of education; whether the respondent works in agriculture; whether the respondent earns extra income; number of farms and number of vegetable gardens the household has.

Multiple imputations using chained equations were also used for the multivariable regression analyses for the associations between selling crops and nutritional status. Imputations were

125

done separately by village for children under five and all children. The missing values were imputed based on the complete variables in the model: whether the household has a mobile phone; respondent's highest level of education; whether the respondent works in agriculture; whether the respondent earns extra income and whether the husband has an additional small business.

5.4 Results

5.4.1 Habitat and farm characteristics

A total of 163 farms were sampled, representing 96% of farms in Minyenye and 44% of farms in Mbwei (Table 5.1). All farms near the house and three quarters of farms far from the house had plant agrobiodiversity data collected in Minyenye. Three quarters of farms close to the house and approximately 20% of those far from the house had data collected on them in Mbwei. The farms where agrobiodiversity data were not collected were estimated to be larger than the other farms, this difference was most pronounced in Minyenye. In farms where agrobiodiversity data were collected, Minyenye respondents reported a higher number of crop types grown, especially staple crops as compared to the farms where agrobiodiversity data were not collected. In Mbwei, the reported number of crop types did not vary between farms where agrobiodiversity data were collected and where this data were not collected. This indicates the farms where data were collected were representative of all household farms in terms of crop types grown. The distribution of the households and their farms is displayed in Figures 5.2 and 5.3.

The farm types were very different in Minyenye compared to Mbwei. In Minyenye the majority of participants had one medium to large farm very close to their household and sometimes an additional farm further away from their house. In Mbwei it was typical for the household to have a number of smaller farms, often quite far away from the household, in the valley, in the mountains and sometimes in both areas.

126

	Miny	enye	Mb	owei
	ABD data collected	No ABD data	ABD data collected	No ABD data
Number of farms	95	4	68	88
Percentage of total farms	96.0	4.0	43.6	56.4
Farms near the house (%)	100.0	0.0	75.0	25.0
Farms away from the house (%)	75.0	25.0	21.1	78.9
Estimated size of farm (acres, mean (95% CI))	1.8 (1.6, 2.1)	3.0 (0.0, 7.4)	1.1 (0.8, 1.3)	1.5 (1.1, 1.9)
Reported number of crop types grown (mean (95% Cl))	2.4 (1.4, 3.3)	1.0 (1.0, 1.0)	2.8 (2.5, 3.0)	2.9 (2.7, 3.1)
Reported number of staple crops types grown* (mean (95% CI))	1.4 (0.6, 2.1)	0.3 (0.0, 1.8)	1.6 (1.4, 1.7)	1.5 (1.4, 1.7)
Reported number of fruit/vegetables types grown** (mean (95% Cl))	0.2 (0.0, 0.5)	0.0 (0.0, 0.0)	1.1 (0.9, 1.3)	1.3 (1.1, 1.4)

Table 5.1. Farm characteristics of farms where agrobiodiversity data were and were not collected in Minyenye and Mbwei

ABD: agrobiodiversity data

*included maize, cassava, millet, finger millet and sorghum

**included tomatoes, beans, onions, pumpkin leaves, cabbage, potatoes, yams, pumpkin, sweet potato, green pepper, banana, papaya



Figure 5.2. Households and all measured household farms in the five sub-villages of Minyenye



Figure 5.3. Households and all measured household farms in the seven sub-villages of Mbwei.

In Mbwei, in 51% of households, indicators of land degradation or characteristics that were likely to be associated with land degradation were observed on at least one of the household farms. These included farm 'not well maintained/poor condition' in 9%, 'erosion' in 7% and 'low soil fertility' in 7% of the households. This figure was only 16% in Minyenye. The slope of at least one of the household farms was described as steep or very steep in 40% of the households in Mbwei. Household farms in Minyenye were significantly more diverse than in Mb wei with an average plant agrobiodiversity Index of 2.5 compared to 1.1 in Mbwei (Table 5.2). This difference was even more marked when looking at the agrobiodiversity of food plants (2.3 compared to 0.6). This is the opposite of what would be expected based on the site selection; Minyenye was intended to be the low diversity village and Mbwei was the high diversity village. Mbwei, however, had significantly higher crop and vegetable diversity scores (7.9 and 4.3) compared to Minyenye (6.0 and 2.1). Mbwei had significantly more farms far from their houses (more than 30 minutes walk) contributing to a significantly higher number of total farms per household. However, Minyenye, on average, had significantly bigger farms (mean (95% CI):1.5 (01.3, 1.6) acres) than those in Mbwei (0.4 (0.3, 0.5) acres). The mean (95% CI) total amount of land used for farming in Minyenye was 2.2 (1.8, 2.6) acres. Unfortunately, due to the number of farms not visited in Mbwei, the equivalent statistic was not able to be calculated for Mbwei. Figures 5.2 and 5.3 illustrate the difference in farm size of the measured farms.

A significantly higher proportion of households had vegetable gardens in Mbwei compared to Minyenye. Those households in Mbwei with vegetable gardens, approximately 35% of participating households, had an mean (95% CI) plant agrobiodiversity index of 1.3 (1.2, 1.5) and 1.3 (1.1, 1.5) for all plants and food plants respectively within the vegetable gardens. - 131 -

	Minyenye	Mbwei	P-value for difference*
Plant agrobiodiversity by household (mean (95% CI))			
All plants agrobiodiversity index	2.5 (2.5, 2.6)	1.1 (1.0, 1.3)	< 0.001
Food plants agrobiodiversity index	2.3 (2.2, 2.3)	0.6 (0.5 <i>,</i> 0.8)	< 0.001
Crop diversity score	6.0 (5.2, 6.8)	7.9 (6.7, 9.2)	0.031
Vegetable diversity score	2.1 (1.7, 2.5)	4.3 (3.4, 5.1)	< 0.001
Household sold foods they grew (%)			
Sold staples	26.6	43.1	0.055
Sold vegetables	12.5	44.8	< 0.001
Sold fruit	10.9	20.7	0.138
Sold other	42.2	12.1	< 0.001
Farm types			
Grow food on their own land (%)	89.1	91.4	0.826
Vegetable gardens (%)	18.8	34.5	0.049
Farms per household (mean(95% CI))	1.4 (1.2, 1.5)	2.9 (2.5, 3.2)	< 0.001
Nearby farms per household (mean(95% CI))	1.1 (1.0, 1.2)	0.9 (0.7, 1.2)	0.045
Far away farms per household (mean(95% CI))	0.2 (0.1, 0.3)	1.9 (1.6, 2.3)	< 0.001
Total area of farms (acres, mean(95% CI))	2.2 (1.8, 2.6)	-	
Average area of farms (acres, mean(95% CI))	1.5 (1.3, 1.6)	0.4 (0.3, 0.5)	< 0.001
Farming issues (%)			
Grew food not eaten or sold	3.2	75.9	< 0.001
Food not eaten/sold: not enough rain	50.0	72.7	0.485
Food not eaten/sold: wrong kind of rain	50.0	0.0	0.050
Food not eaten/sold: too much sun	0.0	15.9	0.054
Food not eaten/sold: land not fertile	0.0	11.4	0.621

Table 5.2. Plant agrobiodiversity indices, percentage selling grown food, farm types and farming issues for Minyenye and Mbwei

*Mann Whitney tests used for difference in means, chi squared used for difference in proportions between the two villages

The majority of households in both settlements grow Maize (over 85%) and a high proportion of households in Minyenye grow sorghum (86%) and in Mbwei grow cassava root (90%) (Figure 5.4). Many Mbwei households grow pumpkin, beans and pumpkin or potato leaves to eat and approximately 40% of households in this village grow sweet potatoes, tomatoes, *mchicha* (spinach) and potatoes. Common vegetables grown in Minyenye included pumpkins and cassava leaves. Between approximately 15% and 30% of those in Mbwei grow guava, avo cado, banana and papaya while the main fruit grown in Minyenye, in 25% of households, was guava. Forty-four percent and 52% of households grow groundnuts and sunflowers in Minyenye while households in Mbwei were more likely to grow sugarcane (45%).

In the last year, 2011, 71% (72% in Minyenye and 69% in Mbwei) said that there were no changes in what they grew and harvested as compared to the last five years. Twenty percent of respondents in Minyenye said they harvested less crops and 3% said they harvested more crops. While in Mbwei 26% said they harvested less and 4% said they harvested more crops. Seventy-six percent of those in Mbwei compared to only 3% of those in Minyenye said they grew food that was not eaten or sold. The majority of participants said this was because there was not enough rain (72%), too much sun (15%) or that the land was not fertile (11%).



Figure 5.4. Percentage of households growing different crop types in Minyenye and Mbwei over the previous calendar year

5.4.2 Minyenye village, Singida district

5.4.2.1 Habitat, species present, farm characteristics and plant agrobiodiversity

- 134 -

Minyenye, the 'low biodiversity' village, is situated in the central plains of Tanzania. Although Singida region is approximately 1200 to 1500m above sea level it is not a steep area. The landscape has gentle hills and the majority of the farms were relatively flat. The earth in this area is a rich red colour and the area is dry and susceptible to drought (Tanzanian NBS and Singida Regional Commissioner's Office, 2005). The majority of the land is dirt with brush and trees (Figure 5.5) but there are areas, where the river runs in the rainy season that has grass (Figure 5.6). It was common for farms in Minyenye to be covered in different types of grasses in amongst the staple crops. It was not uncommon for Minyenye farms to be overgrown (Figure 5.7). In 16% of households the biodiversity research assistant noted indicators of land degradation on one or more of the household farms. These characteristics most commonly included perceived low quality soil in this village.



Figure 5.5. A typical area in Minyenye of dry red dirt with a few larger trees



Figure 5.6. The dry riverbed running through the valley in Minyenye, some grass is available for grazing in this area at this time of year



Figure 5.7. A typical farm in Minyenye



Figure 5.8. A typical vegetable garden in Minyenye

The majority of households in Minyenye had between 5 and 15 species per acre growing on their farms (Figure 5.9). Approximately 10% of households have more than 20 species per acre. Around 28% of households grow between 0 and 5 species per acre and over 55% of households grow between 5 and 10 food species per acre. 84% of households grow less than 10 species per acre of farmland. On average households in Minyenye had 9 food species and 12 species in total per acre of farmland.



Figure 5.9. Number of species; all plants and food plants grown per acre of household farm in Minyenye

Table 5.3 shows all the species present on the Minyenye household farms where agrobiodiversity data was captured. It shows the total count for each species made for the agrobiodiversity index calculations and percentage abundance. Fifty-nine different food species and 29 different non-food species were found on the household farms in Minyenye. The most common food species found were Sorghum (*Sorghum Bicolor Gramineae*), followed by *Gafinda* (*Commelina African Commelinaceae*) and Maize (*Zea Maize Gramineae*) at 6%, 5% and 4% of the total species found. Finger millet (*Eleusine Africana Gramineae*) only made up 1% of the total species found and Cassava was not encountered. *Fagio* (*Elerngia Cordifolia Compositae*) and *Ighimbi* (*Eleusine Jaegeri Gramineae*) were the two most common non-food species encountered at 12% and 5% of all species.

Species genus name	Family name	Common (local) name	Number of individuals*	Abundance (%)**
Food species***				
Sorghum bicolor	Gramineae	Sorghum (Mtama)	162	5.8
Commelina african	Commelinaceae	(Gafinda)	131	4.7
Zea maize	Gramineae	Maize (Mahindi)	108	3.9
Corchorus trilocularis	Tiliaceae	(Mlenda)	105	3.8
Helianthus annuus	Compositae	Sunflower (Alizeti)	101	3.6
Eleusine indica	Gramineae	(Busai)	91	3.3
Cleome hirta	Capparidaceae	(Mnyisira)	86	3.1
Clotalaria cylindro stachys	Papilionaceae	(Mukuku)	61	2.2
Bidens pilosa	Compositae	(Mpangwe)	57	2.1
Setaria verticilata	Gramineae	(Kinasa nguo)	56	2.0
Hibiscus diversifolius	Malvaceae	(Inkongwa)	55	2.0
Ceratotheca sesamoides	Pedaliaceae	(Mbata)	50	1.8
Combretum collinum	Combretaceae	(Mlahaa)	49	1.8
Cynodon nlemfuensis	Gramineae	(Tahai)	49	1.8
Sonchus luxurians	Compositae	(Mchunga)	49	1.8
Cajanus Cajan	Papilionaceae	(Kunde)	45	1.6
Combretum zeyheri	Combretaceae	(Mhanyati)	45	1.6
Amaranthus hybridus	Amaranthaceae	(Mchicha)	40	1.4
Cucumis pepo	Cucurbitaceae	Pumpkin (Boga)	38	1.4
Panicum miliaceum	Gramineae	Millet (Uwele)	35	1.3
Oxygonum sinuatum	Polygonaceae	(Mbigili)	34	1.2

 Table 5.3. Abundance of species present on household farms in Minyenye split by food species and non food species

- 138 -

Dichrostachys cinerea	Mimosaceae	(Mtunduru)	32	1.2
Cucumis dipsace us	Cucurbitaceae		31	1.1
Citrullus lanatus	Cucurbitaceae	(Tikiti)	30	1.1
Eleusine africana	Gramineae	Finger millet (Ulezi)	25	0.9
Trichodesma zeylenicum	Boragnaceae	(Majani washa)	24	0.9
Triumfetta rhomboidea	Tiliaceae	(Mululi)	22	0.8
Dactyloctenium aegyptium	Gramineae	(Talanje)	20	0.7
Solanum lycopersicum	Solanaceae	(Nyanya)	18	0.6
Dalbergia nitidula	Papilionoideae		17	0.6
Azanza garckeana	Malvaceae	(Mtogo)	15	0.5
Solanum incanum	Solanaceae	(Ntula)	13	0.5
Solanum villosum	Solanaceae		11	0.4
Physalis peruviana	Solanaceae	(Vitunda)	10	0.4
Vigna subterranea	Fabaceae		9	0.3
Abelmoschus esculentus	Malvaceae	Ocra	8	0.3
Ipomea batatas	Convolvulaceae	Sweet potato	8	0.3
Markhamia obtusifolia	Bignoniaceae		8	0.3
Combretum molle	Combretaceae		6	0.2
Solanum incanum	Solanaceae		6	0.2
Hibiscus sabdariffa	Malvaceae	(Choya)	5	0.2
Solanum villosum	Solanaceae		5	0.2
Balanites aegyptiaca	Zygophyllaceae		4	0.1
Caylusea abyssinica	Redeceae	(Ngwiba)	4	0.1
Syzygium cumini	Myrtaceae		4	0.1
Albizia harveyi	Mimocaceae		3	0.1
Amaranthus spinosus	Amaranthaceae	(Mchicha pari)	3	0.1

Ampelocissus africanaVitaceae20.1Cordia monoicaBoraginaceae20.1Emilia coccineaCompositae20.1Manihot esculentaEuphorbiaceae20.1Mucuna pruriensPhytolaccaceae20.1Mucuna pruriensConvolvulaceae20.1Bauhinia fassoglensisConvolvulaceae20.1Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Cordia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Salvadora persicaGramineae10.0Elerugia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae291.0Polemonium viscosumCompositae291.0Polemonium viscosumConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomeea hyoscyamineConvolvulaceae(Irag'anga)120.4	Manihotglaviozii	Euphorbiaceae		3	0.1
Cordia monoicaBoraginaceae20.1Emilia coccineaCompositae20.1Manihot esculentaEuphorbiaceae20.1Mucuna pruriensPhytolaccaceae20.1Vangueria infaustaRubiaceae20.1Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Crythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food species10.00Elengia cordifoliaCompositae10.0Digitaria scalarumGramineae(Fagio)34212.3Eleusine jaegeriGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumComyoitae190.7Corchorus kirkiiMalvaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhonkey)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irag'anga)120.4	Ampelocissus africana	Vitaceae		2	0.1
Emilia coccineaCompositae20.1Manihot esculentaEuphorbiaceae20.1Mucuna pruriensPhytolaccaceae20.1Vangueria infaustaRubiaceae20.1Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Commiphora africanaBurseraceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Salvadora persicaSalvadoraceae10.0Non food species10.00Elerngia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.51Leucas martinicensisLamiaceaeSimbilili)893.2Polemonium viscosumCompositae190.70Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.70Corchorus kirkiiMalvaceae(Ikhombe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Cordia monoica	Boraginaceae		2	0.1
ManihotesculentaEuphorbiaceae20.1Mucuna pruriensPhytolaccaceae20.1Mucuna pruriensRubiaceae20.1Vangueria infaustaRubiaceae20.1Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Erythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food species10.01Elensia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.51Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irag'anga)120.4	Emilia coccinea	Compositae		2	0.1
Mucuna pruriensPhytolaccaceae20.1Vangueria infaustaRubiaceae20.1Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Erythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food species10.00.0Elengia cordifoliaCompositae(Fagio)342Digitaria scalarumGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomea ahyoscyamineConvolvulaceae(Irang'anga)120.4	Manihotesculenta	Euphorbiaceae		2	0.1
Vangueria infaustaRubiaceae20.1Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Erythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food species10.0Elengia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Mucuna pruriens	Phytolaccaceae		2	0.1
Bauhinia fassoglensisConvolvulaceae10.0Commiphora africanaBurseraceae10.0Erythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food species10.0Elengia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineLonvolvulaceae(Irang'anga)120.4	Vangueria infausta	Rubiaceae		2	0.1
Commiphora africanaBurseraceae10.0Erythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food speciesElerugia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomea byoscyamineConvolvulaceae(Irang'anga)120.4	Bauhinia fassoglensis	Convolvulaceae		1	0.0
Erythrina abyssinicaFabaceae10.0Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food speciesElerngia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Commiphora africana	Burseraceae		1	0.0
Grewia bicolorTiliaceae10.0Salvadora persicaSalvadoraceae10.0Non food speciesElerngia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Erythrina abyssinica	Fabaceae		1	0.0
Salvadora persicaSalvadoraceae10.0Non food speciesElerngia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Grewiabicolor	Tiliaceae		1	0.0
Non food speciesElerngia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Salvadora persica	Salvadoraceae		1	0.0
Elerngia cordifoliaCompositae(Fagio)34212.3Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Non food species				
Eleusine jaegeriGramineae(Ighimbi)1324.8Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Elerngia cordifolia	Compositae	(Fagio)	342	12.3
Digitaria scalarumGramineae973.5Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineLabiate(Irang'anga)120.4	Eleusine jaegeri	Gramineae	(Ighimbi)	132	4.8
Leucas martinicensisLamiaceae973.5Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Digitaria scalarum	Gramineae		97	3.5
Dolithia uniflorusPapilionaceae(Simbilili)893.2Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Leucas martinicensis	Lamiaceae		97	3.5
Polemonium viscosumCompositae291.0Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4	Dolithia uniflorus	Papilionaceae	(Simbilili)	89	3.2
Ipomea bilobaConvolvulaceae(Ikhombe)250.9Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4Hinnearratea panyifilaraLabiata(Mdima mpahi)110.4	Polemonium viscosum	Compositae		29	1.0
Borreria strictaLabiate190.7Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4Hinnearratea panuifloraLabiata(Mdima mpahi)110.4	Ipomea biloba	Convolvulaceae	(Ikhombe)	25	0.9
Corchorus kirkiiMalvaceae(Ikhandaghii)190.7Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4Hinnegratea papuifloraLabiata(Mdima mpabi)110.4	Borreria stricta	Labiate		19	0.7
Perotis hildebrandtiiGramineae(Ginkhokwe)180.6Astro Ipomoea hyoscyamineConvolvulaceae(Irang'anga)120.4Hinnegratea partificaraLabiata(Mdima machi)140.4	Corchorus kirkii	Malvaceae	(Ikhandaghii)	19	0.7
Astro Ipomoea hyoscyamine Convolvulaceae (Irang'anga) 12 0.4	Perotis hildebrandtii	Gramineae	(Ginkhokwe)	18	0.6
Vinneerates parviflers (Mdime machi) 11 04	Astro Ipomoea hyoscyamine	Convolvulaceae	(Irang'anga)	12	0.4
הויףסטרמנפי (איטווזיא דער במסומנפי (איטווזיא דער 11 ס.4	Hippocratea parviflora	Labiate	(Mdima mpahi)	11	0.4
Indigofera spicata Papilionaceae 10 0.4	Indigofera spicata	Papilionaceae		10	0.4

panicum trichocladium	Gramineae	(Iraangimba)	7	0.3
Tridax procumbens	Asteraceae		7	0.3
Cynodon dactylon	Gramineae		6	0.2
Leonotis leonurus	Labiate		6	0.2
Acacia drepanolobium	Mimosaceae		5	0.2
Rothia hirsuta	Leguminosae		5	0.2
Senecio vulgaris	Compositae		5	0.2
Striga asiatica	Orabansiaceae		5	0.2
Albizia amara	Mimosaceae		4	0.1
Eucalyptus maideni	Myrtaceae		4	0.1
Hypochoeris glabra	Asteraceae		4	0.1
Turea amoena	Meliaceae		3	0.1
Lonchocarpus bussei	Papilionaceae		2	0.1
Albizia gummifera (Gmel) C.A Smith	Mimosoidea		1	0.0
Markhamia lutea	Bignoniaceae		1	0.0
Sennasenguena	Papilionaceae		1	0.0

* Number of individuals is the number of times the specific species was encountered along the transects for all the farms in Minyenye

**Number of times this specific species was encountered as a percentage of the total number of encountered plants

***Identified as plant species by the biodiversity research assistant, books (Pendaeli, 2010, Dharani, 2002, Peters et al., 1992)) and reputable internet sites ((FAO, 2014a, JSTOR, 2014, World Agroforestry Centre, 2014).

5.4.2.2 Plant agrobiodiversity and dietary diversity /food variety

Plant agrobiodiversity is not associated with dietary diversity or food variety in Minyenye. No significant associations were seen between all plant or food plant agrobiodiversity indices and the nine food group dietary diversity score or the individual food variety scores of respondents or children under five years in regression models (Table 5.4).

In Minyenye, the more crop types grown in the previous year the higher the respondent's dietary diversity scores were in the unadjusted but not the adjusted model. The higher the crop diversity score the higher the respondent's and their oldest child under five's food variety score was in both the unadjusted model and when the model was adjusted for husband's type of employment; husband's frequency of employment; number of farms; average farm size. For example, for each additional crop the household grew in the previous calendar year the number of foods the child ate in the previous 24 hours increased by **0.31** items (95% CI: **0.12**, **0.50**) in the adjusted model.

The more different types of vegetable grown the higher the respondent and their oldest child under five's food variety was in the unadjusted model. This association remained significant with the under five year olds in the adjusted model.

None of the farm characteristics listed in Table 5.2 were significantly associated with dietary diversity or food variety in Minyenye.

	Dietary diversity score		Food var	iety score
	Child <5 years	Respondent	Child<5 years	Respondent
Unadjusted model^				
All plants agrobiodiversity	-0.82 (-2.24, 0.60)	-0.36 (-1.54, 0.82)	0.39 (-2.29, 3.06)	0.42 (-2.13, 2.96)
Food plants agrobiodiversity	-0.36 (-1.65, 0.93)	-0.09 (-1.16, 0.97)	0.37 (-2.03, 2.78)	0.49 (-1.80, 2.78)
Crop diversity score	0.09 (-0.02, 0.21)	0.12 (0.03, 0.21)	0.35 (0.14, 0.55)	0.32 (0.12, 0.51)
Vegetable diversity score	0.09 (-0.15, 0.33)	0.18 (-0.01, 0.36)	0.57 (0.16, 0.99)	0.54 (0.15, 0.93)
Adjusted model^^				
All plants agrobiodiversity	-0.86 (-2.18, 0.46)	-0.65 (-1.87, 0.57)	0.02 (-2.60, 2.65)	-0.34 (-2.97, 2.29)
Food plants agrobiodiversity	-0.38 (-1.60, 0.84)	-0.33 (-1.45, 0.78)	0.21 (-2.19, 2.60)	0.02 (-2.37, 2.41)
Crop diversity score	0.08 (-0.02, 0.18)	0.09 (-0.01, 0.18)	0.31 (0.12, 0.50)	0.24 (0.05, 0.44)
Vegetable diversity score	0.06 (-0.16, 0.27)	0.11 (-0.08, 0.30)	0.46 (0.05, 0.87)	0.35 (-0.05, 0.75)

^Unadjusted model. N was 61 for children under five and 64 for respondents.

^^Adjusted for husband's type of employment; husband's frequency of employment; number of farms; average farm size. N was 55 for children under five and 58 for respondents.

5.4.2.3 Plant agrobiodiversity and nutritional status

Plant agrobiodiversity and crop diversity scores show mixed associations with nutritional status in Minyenye. Higher agrobiodiversity indices for all plants and for food plants were significantly associated with higher height z-scores in all children (Table 5.5). The more different types of crops and vegetables grown by the household the lower the MUAC of children under the age of five years.

For each additional farm far from the house a household has under five year old BMI z-scores decreases by 0.50 (regression coefficient (95% confidence intervals): -0.50 (-0.93, -0.07)). Those households reporting crops that could not be eaten or sold had children with weight z-scores 0.81 lower than those households not reporting this (-0.81 (-1.59, -0.04)). There were no other significant associations seen between farm characteristics and nutritional status in children in Minyenye.

Table 5.5. Associations between plant agrobiodiversity and crop and vegetable diversity scores and nutritional status in children under five and all children in Minyenye

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient (95% CI)	Regression coefficient (95% Cl)	Regression coefficient (95% CI)	Regression coefficient (95% CI)
Children <5 years^				
All plants agrobiodiversity	0.64 (-0.27, 1.55)	0.05 (-076, 0.87)	-0.57 (-1.36, 0.21)	-0.17 (-1.06, 0.71)
Food plants agrobiodiversity	0.70 (-0.11, 1.50)	0.08 (-0.64, 0.80)	-0.61 (-1.31, 0.09)	-0.09 (-0.88, 0.70)
Crop diversity scores	-0.00 (-0.08, 0.07)	-0.04 (-0.10, 0.02)	-0.05 (-0.11, 0.02)	-0.07 (-0.14, -0.00)
Vegetable diversity scores	0.01 (-0.13, 0.16)	-0.04 (-0.16, 0.09)	-0.06 (-0.18, 0.06)	-0.14 (-0.27, -0.00)
All Children^^				
All plants agrobiodiversity	0.75 (0.26, 1.23)	0.22 (-0.26, 0.70)	-0.23 (-0.73, 0.28)	-0.31 (-1.32, 0.70)
Food plants agrobiodiversity	0.66 (0.24, 1.08)	0.23 (-0.19, 0.64)	-0.21 (-0.65, 0.23)	-0.16 (-1.04, 0.71)
Crop diversity scores	-0.01 (-0.05, 0.03)	-0.01 (-0.05, 0.03)	0.01 (-0.03, 0.04)	-0.04 (-0.12, 0.04)
Vegetable diversity scores	-0.04 (-0.12, 0.03)	-0.04 (-0.11, 0.04)	0.01 (-0.07, 0.08)	-0.13 (-0.29, 0.02)

Unadjusted model only is presented as no potential confounders were identified.

^Unadjusted model. N was 101 for height, 100 for weight, 102 for BMI z-scores and 103 for MUAC in children under five. ^^Unadjusted model. N was 251 for height, 251 for weight, 255 for BMI z-scores and 252 for MUAC in all children.

5.4.2.4 Selling produce and dietary diversity

Individuals from households selling staples, vegetables and other produce had higher dietary diversity in Minyenye. Focusing on the results for the adjusted models shows households who sold other produce such as oil, beans, honey or sugarcane had children and respondents with higher dietary diversity and children with higher food variety (Table 5.6). Households selling staples and vegetables had children under five years of age with higher food variety.

	Dietary diversity score		Food vari	ety score
	Child <5 years*	Respondent**	Child<5 years*	Respondent**
Unadjusted models^				
Sold staples	0.89 (0.07, 1.71)	0.71 (0.05, 1.37)	2.23 (0.76, 3.70)	1.72 (0.31, 3.12)
Sold Vegetables	0.86 (-0.37, 2.10)	0.55 (-0.35, 1.45)	3.28 (1.11, 5.46)	1.79 (-0.13, 3.70)
Sold Fruit	-0.01 (-1.18, 1.17)	1.09 (0.16, 2.01)	1.00 (-1.16, 3.17)	2.10 (0.08, 4.11)
Sold Other	1.16 (0.46, 1.85)	0.89 (0.33, 1.46)	1.87 (0.55, 3.19)	1.78 (0.55, 3.02)
Adjusted models^^				
Sold staples	0.76 (-0.22, 1.74)	0.66 (-0.11, 1.44)	1.91 (0.16, 3.66)	1.42 (-0.28, 3.11)
Sold Vegetables	0.69 (-1.03, 2.41)	0.07 (-1.16, 1.30)	3.36 (0.41, 6.30)	1.09 (-1.53, 3.71)
Sold Fruit	-0.19 (-1.80, 1.42)	1.06 (-0.11, 2.23)	0.82 (-2.12, 3.77)	2.08 (-0.53, 4.68)
Sold Other	1.28 (0.39, 2.17)	0.83 (0.09, 1.57)	1.75 (0.05, 3.44)	1.46 (-0.18, 3.11)

Table 5.6. Associations between households selling foods they grew and the respondent's and children's dietary diversity and food variety scores in Minyenye

^Unadjusted model; N was 61 for children under five and 64 for respondents.

^^Adjusted for respondent's and husband's age, whether the household has a mobile phone; respondent and husband's highest level of schooling; respondent's and husband's type of employment (farming/other); husband has extra small business; respondent's and husband's frequency of employment; Respondent or husband earns extra income; number of farms; number of vegetable gardens; average farm size. N for children under five was 61 and N for respondents was 64.

*The following variables had missing values (number of missing values in brackets after each variable): husband's age (17); husband's highest level of education (7); whether the husband works in agriculture (6); whether the husband has an additional small business (6); respondent (1) and husband's frequency of employment (7) and whether the husband earns extra income (6) out of a total of 106 individuals.

**The following variables had missing values: husband's age (12); husband's highest level of education (6); whether the husband works in agriculture (5); whether the husband has an additional small business (5); respondent (1) and husband's frequency of employment (6) and whether the husband earns extra income (5) out of a total of 64 individuals.

5.4.2.5 Selling produce and nutritional status

Selling produce showed mixed associations with nutritional status in Minyenye. The majority of the associations between whether a household sold produce and children's nutritional status were not significant (Table 5.7). Focusing on the adjusted models; households that sold vegetables and fruit had children with **0.42** and **0.49** higher height z-scores. Those selling other produce such as oil, beans, honey or sugarcane had children with lower height for age and weight for age z-scores.

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient (95% CI)			
Under five year olds*				
Unadjusted [^]				
Sold staples	0.14 (-0.38, 0.66)	-0.29 (-0.74, 0.17)	-0.48 (-0.93, -0.04)	-0.16 (-0.66, 0.34)
Sold Vegetables	0.42 (-0.25, 1.09)	0.28 (-0.32, 0.87)	-0.05 (-0.63, 0.54)	-0.64 (-1.27, -0.00)
Sold Fruit	0.08 (-0.65, 0.80)	0.19 (-0.44, 0.83)	0.16 (-0.47, 0.79)	-0.27 (-0.96, 0.43)
Sold Other	-0.19 (-0.64, 0.27)	-0.25 (-0.65, 0.15)	-0.15 (-0.54, 0.24)	-0.05 (-0.48, 0.39)
Adjusted^^				
Sold staples	0.03 (-0.54, 0.61)	-0.40 (-0.91, 0.10)	-0.53 (-1.01, -0.05)	-0.18 (-0.74, 0.38)
Sold Vegetables	0.59 (-0.16, 1.34)	0.33 (-0.34, 1.00)	-0.13 (-0.77, 0.52)	-0.68 (-1.41, 0.04)
Sold Fruit	-0.08 (-0.87, 0.71)	0.14 (-0.56, 0.84)	0.21 (-0.46, 0.88)	-0.33 (-1.10, 0.45)
Sold Other	-0.34 (-0.85, 0.16)	-0.36 (-0.81, 0.09)	-0.17 (-0.60, 0.25)	-0.02 (-0.52, 0.47)
All children**				
Unadjusted^^^				
Sold staples	0.08 (-0.19, 0.36)	-0.09 (-0.36, 0.18)	-0.16 (-0.44, 0.12)	-0.11 (-0.67, 0.45)
Sold Vegetables	0.24 (-0.13, 0.60)	0.25 (-0.11, 0.61)	0.11 (-0.26, 0.48)	-0.24 (-0.99, 0.51)
Sold Fruit	0.50 (0.11, 0.88)	0.44 (0.06, 0.81)	0.10 (-0.30, 0.50)	0.32 (-0.47, 1.10)
Sold Other	-0.47 (-0.71, -0.23)	-0.37 (-0.60, -0.13)	-0.06 (-0.31, 0.20)	-0.30 (-0.80, 0.19)
Adjusted^^^^				
Sold staples	0.08 (-0.21, 0.38)	-0.09 (-0.37, 0.20)	-0.12 (-0.43, 0.18)	-0.21 (-0.80, 0.37)
Sold Vegetables	0.42 (0.01, 0.83)	0.27 (-0.13, 0.67)	-0.00 (-0.43, 0.42)	-0.43 (-1.24, 0.39)
Sold Fruit	0.49 (0.08, 0.90)	0.37 (-0.03, 0.77)	0.09 (-0.34, 0.53)	-0.09 (-0.90, 0.71)
Sold Other	-0.50 (-0.77, -0.23)	-0.40 (-0.66, -0.14)	-0.09 (-0.38, 0.19)	-0.27 (-0.79, 0.26)

Table 5.7. Associations between household sellir	g foods they grew and under five y	/ear old's and all children's nutritional status in Miny	enye
--	------------------------------------	--	------

^Unadjusted model. N was 101 for height, 100 for weight, 102 for BMI z-scores and 103 for MUAC in children under five.

^Adjusted model. ^A and ^^A were adjusted for respondent's and husband's age, mobile phone in the household, respondent's and husband's highest level of school, husband employed in farming or other and plus small business, frequency of respondent and husband employment, respondent and husband earns extra income. N was 101 for height, 100 for weight, 102 for BMI z-scores and 103 for MUAC in children under five. ^^Unadjusted model. N was 251 for height, 251 for weight, 255 for BMI z-scores and 252 for MUAC in all children.

^^^Adjusted model. N was 251 for height, 251 for weight, 255 for BMI z-scores and 252 for MUAC in children under five.

*The following variables had missing values: husband's age (17); husband's highest level of education (7); whether the husband works in agriculture (6); whether the husband has an additional small business (6); respondent (1) and husband's frequency of employment (7) and whether the husband earns extra income (6) out of a total of 106 individuals.

**The following variables had missing values: hus band's age (51); hus band's highest level of education (26); whether the husband works in agriculture (18); whether the husband has an additional small business (18); respondent (3) and husband's frequency of employment (21) and whether the husband earns extra income (18) out of a total of 296 individual

5.4.3 Mbwei village, Lushoto district

5.4.3.1 Habitat, species present, farm characteristics and plant agrobiodiversity

Mbwei, the 'high biodiversity' village, is situated in the West Usambara mountains at approximately 2300m above sea level. Much of the surrounding area is forest but the area of Mbwei and the area immediately surrounding the village is open with scrubby vegetation. The area was quite dry at the time of data collection. In Mbwei, in 51% of households, characteristics that were likely to be associated with land degradation, such as presence of specific species associated with poor soil or farm on a very steep slope were observed on at least one of the household farms. The slope of at least one of household farms was described as steep or very steep in 40% of the households in Mbwei. Figure 5.10 and 5.11 show typical farms in Mbwei and 5.12 shows a typical vegetable garden.



Figure 5.10. A typical farm in Mbwei



Figure 5.11. A typical farm in Mbwei



Figure 5.12 A typical vegetable garden in Mbwei

Approximately 50% of households in Mbwei have between 5 and 15 species per acre growing on their household farms (Figure 5.13). Approximately 15% have between 15 and 20 species and approximately 25% of households have more than 20 species per acre. Over 30% of households grow between 0 and 5 food species per acre and another 30% grow between 5 and 10 species per acre. A similar distribution to what was seen in Minyenye. 64% of households 151 grew less than 10 species per acre of farmland. On average households in Mbwei had 10 food species and 17 species in total per acre of farmland a similar number of food species to Minyenye but more non food species.



Figure 5.13. Number of species; all plants and food plants grown per acre of household farm in Mbwei

A total of 45 species were found on the Mbwei household farms that were measured, 22 food species and 23 non food species (Table 5.8), much less than was found in Minyenye. The most common three food species present on Mbwei farms were Cassava (*Manihot Aspera Crantz Euphorbiaceae*), *Mpangwe (Bidens Pilosa Compositae*), and Maize (*Zea Maize Gramineae*) at 25%, 13% and 8%. Finger millet and Sorghum were not encountered. *Enneapogon Cenchroides Gramineae* (12%) and *Albizia Gummifera (Gmel) C.A Smith Mimosoidea* (4%) were the two most common non food species found on Mbwei household farms.

		Common (local)	Numberof	Ahundanca
Species genus name	Family name	Common (local)	individuale*	
		name	Individuals*	(%)**
Food species***				
Manihot aspera Crantz.	Euphorbiaceae	Cassava	146	24.9
Bidens pilosa	Compositae	(Mpangwe)	77	13.1
Zea maize	Gramineae	Maize (Mahindi)	49	8.3
Phaseolus vulgaris	Fabaceae	Beans	20	3.4
Saccharum officinarum L.	Gramineae	Sugarcane	19	3.2
Musa sapientum L	Musaceae	Banana	16	2.7
Amaranthus hybridus L.	Amaranthaceae	Spinach (Mchicha)	11	1.9
Cucurbita moschata	Cucurbitaceae	Pumpkin	10	1.7
Ipomea batatas	Convolvulaceae	Sweet potato	7	1.2
Xanthosoma violaceum Schott	Araceae	Yam	6	1.0
Lycopersium esculentum Mill.	Solanaceae	Tomato	5	0.9
Solanum tuberosum L.	Solanaceae	Potato	5	0.9
Sonchus luxurians	Compositae	(Mchunga)	4	0.7
Telfairia pedata (sims)Hook.	Cucurbitaceae		3	0.5
Cajanus Cajan	Papilionaceae	Pigeon pea (Kunde)	2	0.3
Combretum molle	Combretaceae		2	0.3
Lablab purpureus (L) Sweet	Papilionaceae		2	0.3
Vangueria madagascariensis Gmail.	Rubiaceae		2	0.3
Aphloia theiformis (Vahl) Benne	Flacourtiaceae		1	0.2
Helianthus annuus	Compositae	Sunflower (Alizeti)	1	0.2
Mangifera indica L.	Anacardiaceae	Mango	1	0.2
Persea Americana Mill.	Lauraceae	Avocado	1	0.2
Non food species				

 Table 5.8. Abundance of species present on household farms in Mbwei split by whether they are food species or non food species

- 153 -

Enneapogon cenchroides	Gramineae	72	12.3
Albizia gummifera (Gmel) C.A Smith	Mimosoidea	22	3.7
Aristida adscensionis	Gramineae	16	2.7
Grevillea robusta A.	Proteaceae	16	2.7
Chloris virgata	Gramineae	15	2.6
Acacia microphylla	Mimosoidea	10	1.7
Croton megalocarpus Hutch.	Euphorbiaceae	8	1.4
Catha edulis	Celastraceae	6	1.0
Ricinus communis	Euphobiaceae	5	0.9
Vernonia galamensis	Compositae	4	0.7
Dissotis sp	Melastomataceae	3	0.5
Panicum sp	Gramineae	3	0.5
Turraea robusta Gürke	Meliaceae	3	0.5
Clerodendrum rotundifolium	Verbenaceae	2	0.3
Eragrostis aspera	Gramineae	2	0.3
Eucalyptus camaldulensis	Myrtaceae	2	0.3
Markhamia lutea	Bignoniaceae	2	0.3
Anise Pappus buchwald H.wild	Compositae	1	0.2
Dombeya shupangae K.schum	Sterculiaceae	1	0.2
Euclea divinorum Hiern	Ebenaceae	1	0.2
Olea chrysophylla Lam	Oleaceae	1	0.2
Psiadia punctulata (Dc) vatke	Asteraceae	1	0.2
Tithonia diversifolia (hemsl)A.Gray	Compositae	1	0.2

* Number of individuals is the number of times the specific species was encountered along the transects for all the farms in Minyenye

**Number of times this specific species was encountered as a percentage of the total number of encountered plants

***Identified as plant species by the biodiversity research assistant, books (Pendaeli, 2010, Dharani, 2002, Peters et al., 1992)) and reputable internet sites ((FAO, 2014a, JSTOR, 2014, World Agroforestry Centre, 2014).

- 155 -

5.4.3.2 Plant agrobiodiversity and dietary diversity

Crop and vegetable diversity scores, but not plant agrobiodiversity indices were associated with dietary diversity in Mbwei. As in Minyenye, no significant associations were seen between either all plant or food plant agrobiodiversity indices and any of the dietary diversity or food variety measures in Mbwei (Table 5.9). No significant associations were seen between vegetable garden agrobiodiversity indices and dietary diversity or food variety in Mbwei.

The higher the crop diversity score the higher the children's and respondent's dietary diversity and food variety scores were in the unadjusted model. When the model was adjusted for husband's type of employment; husband's frequency of employment; number of farms; average farm size, these associations were only significant for the respondent's dietary diversity and food variety scores. Similar results were seen for the vegetable diversity score; the more vegetables grown the higher the dietary diversity and food variety of both children under five and the respondents were in the unadjusted model. In the adjusted model the higher the vegetable diversity score the higher the under five year old's dietary diversity score and the respondent's dietary diversity and food variety scores. These results are similar to those found in Minyenye.

As in Minyenye, none of the farm characteristics listed in Table 5.2 were significantly associated with dietary diversity or food variety.

Table 5.9. Associations between plant agrobiodiversity and crop and vegetable diversity scores and dietary diversity/food variety and children under five an
respondents in Mbwei

	Dietary diversity score		Food variety score	
	Child <5 years	Respondent	Child<5 years	Respondent
Unadjusted model^				
All plants agrobiodiversity	0.23 (-0.59, 1.05)	0.43 (-0.29, 1.14)	-0.49 (-2.20, 1.23)	-0.15 (-1.63, 1.34)
Food plants agrobiodiversity	0.44 (-0.33, 1.21)	0.38 (-0.30, 1.06)	0.60 (-1.01, 2.22)	0.29 (-1.11, 1.70)
Crop diversity score	0.09 (0.01, 0.17)	0.11 (0.04, 0.18)	0.20 (0.04, 0.36)	0.26 (0.13, 0.39)
Vegetable diversity score	0.14 (0.02, 0.27)	0.16 (0.06, 0.27)	0.31 (0.05, 0.57)	0.35 (0.14, 0.57)
Adjusted model^^				
All plants agrobiodiversity	0.14 (-0.73, 1.00)	0.45 (-0.32, 1.21)	-0.66 (-2.40, 1.09)	-0.11 (-1.63, 1.41)
Food plants agrobiodiversity	0.41 (-0.40, 1.22)	0.41 (-0.31, 1.14)	0.34 (-0.31, 2.00)	0.16 (-1.28, 1.61)
Crop diversity score	0.09 (-0.00, 0.17)	0.13 (0.05, 0.20)	0.18 (-0.00, 0.35)	0.28 (0.13, 0.42)
Vegetable diversity score	0.14 (0.00, 0.28)	0.19 (0.07, 0.30)	0.28 (-0.00, 0.57)	0.38 (0.15, 0.61)

^Unadjusted model. N was 52 for children under five and 55 for respondents.

^^Adjusted for husband's type of employment; husband's frequency of employment; number of farms; average farm size. N was 51 for children under five and 54 for respondents.

5.4.3.3 Plant agrobiodiversity and nutritional status

Higher plant agrobiodiversity was linked with poorer nutritional status in Mbwei. Higher food plant agrobiodiversity indices were associated with lower BMI z-score in under five year olds and all children (Table 5.10). For each unit increase in agrobiodiversity indices, BMI z-scores decreased by 0.37 and 0.46 units respectively. This is in contrast to the higher height seen with increased agrobiodiversity in Minyenye. No other significant associations were seen between agrobiodiversity indices or crop/vegetable diversity scores and nutritional status variables in Mbwei.

For vegetable gardens in Mbwei the higher all plant diversity indices were the lower children's height z-scores in all children (Regression coefficient (95% confidence intervals): -0.86 z-scores (-1.57, -0.15)) and children under five (-0.95 (-1.85, -0.04). Higher food plant diversity was associated with lower height in all children only (-0.83 (-1.52, -0.14)). The more diverse the vegetable gardens, the lower the children's height z-scores. No significant associations with the other nutritional status variables were found.

The larger the average size of the farms the lower height z-scores were in all children (-0.78 (-1.54, -0.02)). For each acre increase in average farm size height z-scores decreased by 0.78 units. Children in household reporting that they grew crops that they could not eat or sell had significantly lower BMI z-scores compared to children in other households (-0.51 (-0.87, -0.15)). Reasons given for not being able to harvest all crops mainly included 'not enough rain' 'too much sun' and 'land not being fertile'. BMI z-scores were found to be significantly related to reasons given for crops not being eaten of sold. Children in households reporting the reason for crops not being eaten or sold was that the land was not fertile had significantly lower BMI z-scores than those citing other reasons (-0.65 (-1.20, -0.10)). Under five year olds in households citing not enough rain as the reason for this had significantly higher BMI z-scores (0.69 (0.23, 1.15)).

157

Table 5.10. Association between plant agrobiodiversity and crop and vegetable diversity scores and nutritional status in children under five and all children in Mbwei

	Height	Weight	BMI	MUAC
	z-scores	z-scores	z-scores	
	Regression coefficient	Regression coefficient	Regression coefficient	Regression coefficient
	(95% CI)	(95% CI)	(95% CI)	(95% CI)
Children <5 years^				
All plants agrobiodiversity	0.40 (-0.12, 0.91)	0.08 (-0.36, 0.51)	-0.27 (-0.63, 0.09)	0.07 (-0.42, 0.56)
Food plants agrobiodiversity	0.33 (-0.15, 0.80)	0.06 (-0.36, 0.47)	-0.37 (-0.71, -0.04)	-0.02 (-0.48, 0.44)
Crop diversity score	-0.04 (-0.09, 0.13)	-0.02 (-0.07, 0.02)	0.01 (-0.03, 0.05)	-0.02 (-0.07, 0.03)
Vegetable diversity score	-0.05 (-0.13, 0.03)	-0.03 (-0.10, 0.05)	0.02 (-0.04, 0.08)	-0.03 (-0.11, 0.06)
All Children^^				
All plants agrobiodiversity	0.09 (-0.28, 0.46)	-0.08 (-0.43, 0.26)	-0.28 (-0.58, 0.02)	0.27 (-0.30, 0.85)
Food plants agrobiodiversity	0.20 (-0.16, 0.55)	-0.09 (-0.43, 0.25)	-0.46 (-0.75, -0.18)	0.25 (-0.29, 0.79)
Crop diversity score	-0.02 (-0.05, 0.02)	-0.01 (-0.05, 0.02)	0.00 (-0.03, 0.03)	0.05 (-0.01, 0.10)
Vegetable diversity score	-0.01 (-0.06, 0.05)	-0.00 (-0.05, 0.05)	0.00 (-0.04, 0.05)	-0.09 (-0.00, 0.18)

Unadjusted model only is presented as no potential confounders were identified.

^Unadjusted model. N was 92 for height, 95 for weight, 94 for BMI z-scores and 96 for MUAC in children under five. ^^Unadjusted model. N was 154 for height, 158 for weight, 160 for BMI z-scores and 169 for MUAC in all children.
5.4.3.4 Selling produce and dietary diversity

As in Minyenye, selling produce was associated with better dietary diversity in Mbwei. Households selling vegetables had respondents and under five year olds with higher food variety scores in both the unadjusted and adjusted models (Table 5.11).

	Dietary diversity score		Food varie	ety score
	Child <5 years*	Respondent**	Child<5 years*	Respondent**
Unadjusted models^				
Sold staples	0.25 (-0.54, 1.04)	0.19 (-0.52, 0.90)	1.50 (-0.11, 3.12)	1.75 (0.38, 3.12)
SoldVegetables	0.98 (0.23, 1.73)	0.82 (0.15, 1.49)	2.51 (0.98, 4.04)	2.38 (1.09, 3.67)
Sold Fruit	0.61 (-0.35, 1.58)	0.82 (-0.02, 1.66)	2.09 (0.10, 4.08)	2.76 (1.16, 4.37)
Sold Other	-0.03 (-1.29, 1.23)	0.89 (-0.16, 1.94)	0.30 (-2.36, 2.95)	2.59 (0.51, 4.67)
Adjusted models^^				
Sold staples	0.15 (-0.77, 1.06)	-0.09 (-0.93, 0.74)	1.32 (-0.51, 3.15)	1.11 (-0.43, 2.66)
SoldVegetables	0.94 (-0.05, 1.93)	0.88 (-0.01, 1.74)	2.43 (0.42, 4.44)	2.61 (1.08, 2.66)
Sold Fruit	0.35 (-0.92, 1.62)	0.49 (-0.67, 1.64)	1.14 (-1.48, 3.76)	1.71 (-0.41, 3.82)
Sold Other	-0.01 (-1.64, 1.63)	0.54 (-0.86, 1.94)	-0.49 (-3.84, 2.87)	1.51 (-1.08, 4.12)

Table 5.11. Associations between households selling	g foods they grew and the res	pondent's and children's dietary	diversity and food variety	y scores in Mbwei
---	-------------------------------	----------------------------------	----------------------------	-------------------

^Unadjusted model; N was 55 for children under five and 58 for respondents.

^^ Adjusted for respondent's and husband's age, whether the household has a mobile phone; respondent and husband's highest level of schooling; respondent's and husband's type of employment (farming/other); husband has extra small business; respondent's and husband's frequency of employment; Respondent or husband earns extra income; number of farms; number of vegetable gardens; average farm size. N for children under five was 55 and N for respondents was 58.

*The following variables had missing values: respondent's age (23); husband's age (40); husband's highest level of education (2); whether the husband works in agriculture (2); whether the husband has an additional small business (2); husband's frequency of employment (2); whether the husband earns extra income (2) and average farm size (5) out of a total of 104 individuals.

**The following variables had missing values: respondent's age (11); husband's age (20); husband's highest level of education (1); whether the husband works in agriculture (1); whether the husband has an additional small business (1); respondent (0) and husband's frequency of employment (1); whether the husband earns extra income (1) and average farm size (3) out of a total of 58 individuals.

5.4.3.5 Selling produce and nutritional status

As in Minyenye, the majority of associations between whether households sold produce and nutritional status were non-significant (Table 5.12). In adjusted models, households selling fruit however, had children with 1.13cm higher MUAC.

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient (95%	Regression coefficient (95%	Regression coefficient (95%	Regression coefficient (95%
	CI)	CI)	CI)	CI)
Under five year olds*				
Unadjusted [^]				
Sold staples	-0.07 (-0.59, 0.45)	0.06 (-0.39, 0.51)	0.24 (-0.13, 0.62)	0.01 (-0.50, 0.52)
Sold Vegetables	-0.61 (-1.11, -0.11)	-0.42 (-0.86, 0.02)	-0.02 (-0.39, 0.35)	-0.34 (-0.84, 0.16)
Sold Fruit	-0.12 (-0.74, 0.50)	0.06 (-0.48, 0.60)	0.26 (-0.19, 0.70)	0.01 (-0.60, 0.62)
Sold Other	-0.35 (-1.15, 0.46)	-0.27 (-0.95, 0.42)	0.03 (-0.56, 0.62)	-0.77 (-1.53, -0.00)
Adjusted^^				
Sold staples	-0.08 (-0.68, 0.52)	0.05 (-0.46, 0.55)	0.24 (-0.17, 0.64)	0.11 (-0.45, 0.67)
Sold Vegetables	-0.60 (-1.22, 0.02)	-0.17 (-0.70, 0.36)	0.24 (-0.18, 0.65)	-0.07 (-0.64, 0.51)
Sold Fruit	-0.10 (-0.87, 0.67)	0.24 (-0.40, 0.88)	0.40 (-0.11, 0.92)	0.43 (-0.26, 1.13)
Sold Other	-0.04 (-1.13, 1.04)	0.02 (-0.89, 0.93)	0.10 (-0.59, 0.79)	-0.30 (-1.23, 0.64)
All children**				
Unadjusted^^^				
Sold staples	0.01 (-0.36, 0.38)	0.03 (-0.32, 0.38)	0.10 (-0.21, 0.40)	0.30 (-0.27, 0.88)
Sold Vegetables	-0.25 (-0.62, 0.12)	-0.23 (-0.58, 0.11)	-0.04 (-0.34, 0.26)	0.25 (-0.32, 0.82)
Sold Fruit	-0.07 (-0.49, 0.35)	-0.03 (-0.43, 0.36)	-0.00 (-0.35, 0.34)	0.83 (0.18, 1.49)
Sold Other	-0.03 (-0.54, 0.48)	-0.15 (-0.62, 0.32)	-0.21 (-0.62, 0.21)	0.32 (-0.51, 1.14)
Adjusted^^^^				
Sold staples	0.03 (-0.42, 0.47)	0.04 (-0.36, 0.45)	0.12 (-0.21, 0.44)	0.49 (-0.16, 1.14)
Sold Vegetables	-0.23 (-0.66, 0.19)	-0.09 (-0.49, 0.30)	0.11 (-0.21, 0.43)	0.27 (-0.37, 0.91)
Sold Fruit	-0.13 (-0.68, 0.41)	0.01 (-0.49, 0.51)	0.06 (-0.36, 0.47)	1.13 (0.34, 1.92)
Sold Other	-0.04 (-0.71, 0.62)	-0.20 (-0.80, 0.40)	-0.25 (-0.76, 0.27)	0.28 (-0.75, 1.30)

Table 5.12. Associations between households selling foods they grew and under five year old's and all children's nutritional status in Mbwei

^Unadjusted model. N was 97 for height, 100 for weight, 99 for BMI z-scores and 101 for MUAC in children under five.

^Adjusted model. ^A and ^^^ were adjusted for mobile phone in the household, respondent's and husband's highest level of school, husband employed in farming or other and plus small business, frequency of respondent and husband employment, respondent and husband earns extra income. N was 97 for height, 100 for weight, 99 for BMI z-scores and 101 for MUAC in children under five

^^^Unadjusted model. N was 163 for height, 167 for weight, 169 for BMI z-scores and 1680 for MUAC in all children.

^^^Adjusted model. N was 163 for height, 167 for weight, 169 for BMI z-scores and 180 for MUAC in children under five

*The following variables had missing values (number of missing values in brackets after each variable): respondent's age (23); husband's age (40); husband's highest level of education (2); whether the husband works in agriculture (2); whether the husband has an additional small business (2); husband's frequency of employment (2) and whether the husband earns extra income (2) out of a total of 104 individuals.

**The following variables had missing values: respondent's age (59); husband's age (77); husband's highest level of education (3); whether the husband works in agriculture (3); whether the husband has an additional small business (3); husband's frequency of employment (3) and whether the husband earns extra income (3) out of a total of 236 individuals.

5.5 Discussion

5.5.1 Plant agrobiodiversity and dietary diversity/food variety

No significant associations were seen between the plant agrobiodiversity measures and dietary diversity and food variety scores in either village. There are a number of factors that could contribute to the lack of association seen. Firstly, seasonal variation; the agrobiodiversity was only measured at one time point. The availability of foods that have already been harvested may be related to dietary diversity when agrobiodiversity is not. This will be discussed in the section on annual plant diversity. Secondly the agrobiodiversity measure does not account for crop failure. Intentional crops may be planted but have been unable to be harvested and therefore unable to contribute to the households diets. Crop failure was common in Mbwei; 76% of households said that they had crops that they were unable to either eat or sell, but not in Minyenye. Thirdly, planted crops may also not be eaten by household members as they are sold. In Minyenye, 13% of households sold some of the vegetables they grew, 11% sold fruit and 42% sold other produce like oil or sugar cane. In Mbwei 45% of households sold vegetables, 21% sold fruit and 12% sold other produce. And finally, the agrobiodiversity in the household farms may not have impacted on the diversity of the diets of the respondent and her oldest child under the age of five due to the distribution of food within the household (Engle and Nieves, 1993, Gittelsohn et al., 1997).

More generally, there are many factors which could be influencing diets in these communities. For instance food intake is determined partly by cultural norms (Gittelsohn et al., 1997) and, to a lesser extent in low income countries, individual food preferences. Socioeconomic status and the money available to buy food supplementary to grown food will also affect the foods eaten (Shack et al., 1990). Knowledge about what are healthy and age appropriate foods for under five year olds may impact on food choice and dietary diversity (Caulfield et al., 1999). In the context of these factors, growing a more diverse set of crops at a household level may not improve the diversity of foods eaten in the household.

5.5.2 Plant agrobiodiversity and nutritional status

Associations between plant agrobiodiversity and crop diversity scores and nutritional status are mixed in Minyenye but negative in Mbwei. In Minyenye, biodiversity indices were positively associated with height in children i.e. households with farms with higher plant agrobiodiversity had children who were less chronically malnourished. As the plant

agrobiodiversity indices were not associated with the diet variables measured in this study it may be that higher biodiversity is associated with nutritional status through another pathway or it may be that dietary diversity measured at one time point is not a good enough measure of diet to capture these associations; A possibility that was raised in chapter 4.

- 164 -

In Mbwei, higher plant agrobiodiversity was associated with poorer height and BMI in children. Additionally, diversity in vegetables gardens in Mbwei was negatively related to children's height z-scores. These results do not support the main pathway from agrobiodiversity to individual dietary diversity to child nutritional status illustrated in the conceptual framework (figure 3.2 a) and b). Based on discussions in the literature, finding negative relationships between biodiversity and nutritional status in this study was surprising. On the contrary, there has been much discussion about biodiversity and its potential benefits on diet and health in the literature (Wahlqvist and Specht, 1998, Chivian, 2002, Frison et al., 2004, Frison et al., 2006, Gotor, 2010).

However, despite these discussion papers there have been few studies actually investigating these relationships. There have been a number of interventions designed to improve dietary diversity and nutritional status through agriculture. A review by Berti et al. identified 30 such projects (Berti, 2004) and Masset et al identified 23 (Masset et al., 2012). Few of these focused explicitly on techniques to increase the diversity of crops grown but many focused on additions of or improvements in vegetable gardens. In the first review anthropometry outcomes improved in three out of the four vegetable garden interventions collecting these outcomes (Berti, 2004). In the review by Masset et al (Masset et al., 2012) eight interventions presented prevalence of stunting, wasting and underweight post intervention and only three of these found improvements in these markers of nutritional status.

There have been a few studies, while not actually measuring agrobiodiversity, which suggest that improved agrobiodiversity would be associated with improved dietary intake. Mixed cropping fields and home gardens were identified as major sources of essential minerals and vitamins in communities studied in Java. The production of these nutrients were greater in farms and gardens with greater diversity of crops (Marten and Abdoellah, 1988). The study linked crops grown to supplied nutrients but did not go so far as to assess actual consumption and resulting health. A novel ecological tool, nutritional functional diversity, was developed by Remans et al. to reflect the nutrients available from all edible species available on household farms (Remans et al., 2011a). The study found that this metric was related to dietary diversity and iron and vitamin A levels in the blood in three villages in sub-Saharan Africa at the village but not household level.

In terms of research explicitly investigating agrobiodiversity, Akrofi et al (2010) had similar findings to the current study. They found no significant correlation between the Shannon indices in the home gardens and the household dietary diversity scores. Dewey (1981) had mixed findings were households with more than five crops in the family farm had greater dietary diversity in the Socios but there was no difference seen in the other village studied. Interestingly, there was no association between agrobiodiversity and nutritional status in the Socios but in the nearby village crop diversity was significantly correlated with height for age. Unfortunately the details of how agrobiodiversity of crops growing on the household farms, as in this study, or it could be a count of crops grown by the household reported by a household member. Shack et al (1990) found the number of food crops grown in the households vegetable garden, as observed by a researcher, to be significantly associated with mother BMI but not children's nutritional status. Despite the logical pathway from agrobiodiversity through dietary diversity to nutritional status, strong evidence for this pathway occurring in rural areas of low income countries is stilllacking.

5.5.3 Crop and vegetable diversity scores, dietary diversity and nutritional status

Crop diversity scores were associated with measures of dietary diversity. Plant agrobiodiversity was not positively associated with dietary diversity but the crop and vegetable diversity scores were related to dietary diversity and food variety in both villages. Annual plant diversity was negatively associated with MUAC in under five year olds in Minyenye but no associations with nutritional status were found in Mbwei.

Studies attempting to link agrobiodiversity to dietary diversity usually use a simple count, captured at one time point, of the number of crops the household is growing or has grown over the course of the year (Ekesa et al., 2008, Shack et al., 1990, Walingo and Ekesa, 2013, Bhagowalia et al., 2012). Both Ekesa et al. (2008) and Walingo and Ekesa (2013) found agrobiodiversity, defined as the variety of animals kept and plants grown for food over an unspecified time period, to be positively correlated to dietary diversity in a cross-sectional survey in Kenya (Ekesa et al., 2008). Unfortunately neither of these studies reported statistical significance associated with these correlations. Bhagowalia et al (2012) found the number of crops grown by a household to be significantly associated with household dietary diversity over the previous 30 days.

When agrobiodiversity is defined by the number of crops grown by the household over the previous year it is often found to be significantly associated with dietary diversity, as was

found in this study. As this method captures the number of crops grown over the entire year as opposed to those grown only at data collection, perhaps this is a more appropriate method for measuring agrobiodiversity. There are both benefits and disadvantages to this approach. It is quicker and easier to capture but it relies on memory and honesty rather than observation of the crops grown.

5.5.4 Vegetable gardens, dietary diversity and nutritional status

Having a vegetable garden is not associated with improved dietary diversity or nutritional status. In this study farm diversity indices did not include diversity from vegetable gardens. These were more common in Mbwei so it can be expected that vegetable gardens contribute to the diversity of foods available to households in about a third of the households in Mbwei and a fifth of Minyenye households. Households with vegetable gardens had higher crop diversity scores than those households without vegetable gardens but there were no significant differences in any of the dietary diversity or nutritional status variables between households with vegetable gardens and those without.

No significant associations were seen between the biodiversity indices measured in the vegetable gardens of Mbwei and dietary diversity or food variety but it should be noted that the sample size for these analyses were likely to be too small to detect any associations that may exist. Diversity in vegetables gardens in Mbwei was, however, negatively related to children's height z-scores. The more diverse the vegetable gardens were, the lower the heights. This is comparable to the overall results showing no or negative associations between agrobiodiversity of household farms and nutritional status. The reasons for this will be explored in the section comparing Minyenye and Mbwei.

Research using vegetable gardens to improve agricultural diversity with the intention of improving diet and nutritional status have shown mixed results. Some have had positive results; Cabalda et al. showed that having a fruit or vegetable garden was positively associated with dietary diversity in children under five in the Philippines although not with household food security (Cabalda et al., 2011). In Bangladesh, children in households with more developed vegetable gardens which were producing vegetables year round consumed more vitamin A rich vegetables than those with the traditional, limited vegetable gardens (Talukder et al., 2000). English et al. found a decrease in diarrhoeal and respiratory infections in Vietnam in a controlled multi-intervention nutrition trial which improvement of vegetable gardens was one of the components of the intervention (English et al., 1997).

However others have failed to find an association between number of crops grown in vegetable gardens and nutritional status (Shack et al., 1990) or have shown no improvement in nutritional intake (Brun et al., 1989) or biochemical markers of nutritional status (Kidala et al., 2000) on the establishment of household vegetable gardens in Senegal and Tanzania respectively.

5.5.5 Comparison of Minyenye and Mbwei

It may be possible to get some insights into the results of this study by examining some of the differences between the two villages. The land cover in the area surrounding the DHS Enumeration Area (EA) centre point as defined by the Globcover 2009 project (Bontemps et al., 2011) was used to select Minyenye as a low biodiversity village and Mbwei as a high biodiversity village.

The agrobiodiversity measurements of the household farms revealed a very different picture to what would be expected from the surrounding land cover. Minyenye showed slightly greater diversity than Mbwei for overall plant agrobiodiversity (2.5 vs. 2.3). When the agrobiodiversity index was broken down into just plants that can be used for food this difference was even more pronounced (1.1 vs. 0.6).

To try and understand the discrepancy between the surrounding land cover and the agrobiodiversity of the household farms, the habitat observed in the villages and the characteristics of the farms in the two villages were examined. On average households in Mbwei had more farms than households in Minyenye but Minyenye households tended to have larger farms than those in Mbwei. Minyenye is likely to have more land available for growing food on.

The soil quality of the farms is likely to also play a part in this difference in diversity. Approximately 50% of households had markers of land degradation on one or more of their farms in Mbwei compared to only 16% in Minyenye. This is reflected in the fact that respondents in Mbwei were much more likely to report not being able to eat or sell crops that they grew, mainly due to the rain, sun and fertility of the land.

It is likely that the poorer quality of the land in Mbwei is partly due to the topography of the area. Mbwei is situated in the West Usambara mountains and the seven sub villages that make up Mbwei were situated on hill-tops or in valleys with 40% of their farms on steep slopes. It was noted in Mbwei that the farmers mentioned that rains had carried off fertile soil from their lands on two occasions. This issue has been identified in the literature as an important

agricultural problem in the Usambara mountains (Ezaza, 1988, Stahl, 1993). Steep topography is linked to poor soil fertility which causes decreased food production (Ezaza, 1988).

Farmers in Mbwei raised less animals than those in Minyenye, which means they would have less natural fertiliser available to improve the quality of the soil on their farms, especially considering the greater distance between the farms in Mbwei. The poorer quality of the land may help to explain the lower biodiversity seen on Mbwei's household farms. As low biodiversity has also been linked with poor soil fertility (Tilman et al., 1996) these factors combined may be having a negative impact of food production in Mbwei. The higher quality of the farms in Minyenye could also help explain the positive association seen between agrobiodiversity and height in Minyenye when no or negative associations were seen in Mbwei. Barriers such as crop failure could be preventing the crops grown from being consumed by the household. This may be a significant finding as such a high proportion of households in Mbwei (76%) reported growing food that they could neither eat or sell and these households had children with lower BMI z-scores. Additionally, respondents in Mbwei

Minyenye had a far greater number of species encountered on their household farms (88 species) compared to Mbwei (45 species). The variety of food species was also far greater, 59 compared to 22 in Mbwei. Another important difference between the two villages is the growth of staple crops. Sorghum, maize and finger millet made up 6%, 4% and 1% of all the plants found on household farms in Minyenye. In Mbwei, cassava made up 25%, maize made up 8% and finger millet and sorghum were not encountered. Minyenye's greater number of and variety of staples grown could have a beneficial effect on their food security and could contribute to the better nutritional status seen in this village. Additionally, the nutrient content of the staple crops grown in Minyenye are higher than the main crop grown in Mbwei, cassava (Gegios et al., 2010).

Interestingly, when one considers the crop diversity scores and the proportion of households growing specific crops it appears that Mbwei does, in fact, grow a greater variety of food compared to Minyenye. This was particularly pronounced when looking at the vegetables grown by the two villages in Figure 5.4. The plant and vegetable diversity scores include foods grown on vegetable gardens, these numbers may have been higher in Mbwei partly because vegetable gardens were more common in Mbwei. This result may also reflect a higher number of crops grown throughout the year which is not reflected by the agrobiodiversity indices as these were taken at one time point.

The negative association between agrobiodiversity and nutritional status in Mbwei in both the farms and in the vegetable gardens may be partly explained by energy expenditure through 168

work and time taken from care of the children. Kumar and Hotchkiss (1988) found that the more time children spent in agricultural work such as grazing animals and working on farms the poorer the nutritional status of their preschool age siblings.

Mbwei has children with poorer nutritional status, more farms far from the house, higher crop failure and a negative association seen between agrobiodiversity and nutritional status. Perhaps, in the context of increased workload and increased time away from caring for the children, for presumably less food, a relationship does not exist between agrobiodiversity and nutritional status. Whereas in Minyenye, farms are nearer the house and crop failure is less common suggesting a better workload to food production ratio compared to Mbwei. In this context, the higher agrobiodiversity translates into better nutritional status.

It is also possible that families that are more at risk are more likely to grow more different types of crop to minimise the risk to their families if a crop fails (Altieri, 2009). As this was a cross sectional study there is no way to tell the direction of the observed effect.

It may be that there are too many other factors contributing to undernutrition in these villages for a difference in agrobiodiversity measured at one time point, to be more strongly associated with nutritional status. From agricultural factors such as the quantity of food grown and the annual variation in the diversity and amount of food grown to diet and health factors as outlined in chapter 4 such as infection (Chen et al., 1980, Alvarado et al., 2005), complementary feeding (Onyango et al., 1998, Obatolu, 2003) and maternal education (Kabubo-Mariara et al., 2009, Abuya et al., 2012). The pathway from the diversity of crops grown in the household farms through dietary diversity to nutritional status, shown in the conceptual framework (figure 3.2 a) and b)), may be too long with too many confounding factors to be seen in a cross-sectional sample.

5.5.6 Selling produce, dietary diversity and nutritional status

Whether households sold vegetables, fruit or other produce was positively associated with dietary diversity and food variety. There are two likely reasons for this. Firstly, the sale of produce generated an increased income which had a positive effect on the households diets or, secondly, being able to sell crops indicates that there was enough, of that particular crop, to feed the household members.

Previous research has shown that an increased income is positively associated with nutritional status (Shack et al., 1990, Yang et al., 2012). This study found mixed relationships between selling produce and nutritional status; selling fruit and vegetables appears to have a positive

effect while selling other produce such as oil, beans, honey or sugarcane was negatively associated with nutritional status. This difference can be explained through the division of labour down gender lines. Subsistence foods such as vegetables and fruit are thought of as 'women's crops' while cash crops fall into men's domain (Due and Gladwin, 1991, Berinyuy and Fontem, 2011). These results suggest that the money raised by selling fruit and vegetables is used to buy food while the income from cash crops is not. Fruit that was most commonly sold was avocado, banana and guava. These fruits are easy to grow and pick, not requiring much additional time and effort and providing not only a good source of nutrients for the household members but additional money for the household. Consuming and selling fruit has been identified as a coping strategy in times of food shortage in Kenya (Thorlakson et al., 2012).

5.5.7 Limitations and strengths

The true association between agrobiodiversity and dietary diversity may have been masked by some of the study's methodological limitations. There are a number of potential issues in the way the agrobiodiversity measures were collected in this project. Firstly the biodiversity data were collected by two different research assistants for the two different villages. Although both research assistants followed the same data collection protocol there may still be differences in the way this data were collected leading to differences in the resulting biodiversity measurements.

The timing of the data collection would also have impacted on the agrobiodiversity results. Data collection took place in June in Minyenye and July in Mbwei. Mbwei falls in the bimodal part of Tanzania (Kabanda and Jury, 1999) and it was expected that it would have two harvests and that data collection would take place before the main harvest. This was unfortunately not the case in many households and a high proportion of participants reported only one harvest. The Village Executive Officer indicated that Mbwei was in a micro-climate within Lushoto and did not experience the same weather patterns as the surrounding area. The weather tended to be hotter and drier than what was typical in Lushoto. The result of this was that Mbwei did not represent the high diversity area as planned and the agrobiodiversity measures were taken post harvest for a number of households, reducing the diversity indices for this village.

Due to the greater total number of farms and greater number of farms far from the household a smaller proportion of total farms were measured in Mbwei compared to in Minyenye. So, while in Minyenye the biodiversity of almost all of the farms used by the household were measured, in Mbwei only a selection of farms were measured and those far from the

- 170 -

household were under-represented. This may introduce error into the biodiversity measurements, but the reported size and types of crops grown on the measured farms did not differ markedly from the farms where no data were collected. This should minimise the bias introduced by this limitation.

Despite these limitations this project has brought together detailed data on plant agrobiodiversity, dietary diversity and nutritional status, something which is rare within one project. Plant agrobiodiversity is measured systematically and biodiversity indices have been calculated. This is the first study to compare biodiversity indices on households farms to nutrition outcomes. This study has also collected detailed supplementary data which provides important insights into the context of this work. The results have highlighted some of the complexities of relating factors with multiple determinants to each other and has produced some interesting insights into how agricultural, dietary and health factors are related.

5.6 Conclusions

This project aimed to investigate the relationship between agrobiodiversity, dietary diversity and nutritional status in villages in rural Tanzania. Chapter 4 illustrated a lack of association between dietary diversity and nutritional status in these villages. This chapter has found no significant associations between agrobiodiversity and dietary diversity and a limited number of associations between agrobiodiversity and nutritional status variables. Due to the push for improving agrobiodiversity to improve dietary diversity and health, this is an important finding.

A simple count of the foods that a household grows was significantly associated with dietary diversity and food variety while a more systematic measure of biodiversity in the household farms was not. This has implications for nutrition focused research both in terms of measurement of agrobiodiversity indices and in terms of targets for dietary improvement. If researchers or development organisations are interested in improving diets in a subsistence farming setting asking the household members what they grow over the course of a calendar year may give a better representation of food availability compared to employing time consuming cross-sectional agrobiodiversity measures. Project targets could focus on adding a certain number of crops to the households annual planting cycles, simplifying monitoring and evaluation procedures.

The results of this study also provide a word of caution for those attempting to increase agrobiodiversity on the assumption that this will lead to improvements in diet and nutritional status. In a number of cases in this study measures of biodiversity and numbers of crops grown were negatively associated with nutritional status. Households with greater diversity in their household farms and vegetable gardens as well as households which sold staples, vegetables or other food had children with poorer nutritional status. The results from this study illustrates some of the complexities of the relationships between factors along the pathway from food production through consumption to nutrient utilization in low income countries.

Chapter 6:

Investigating the associations between animal diversity and both dietary diversity and nutritional status in Minyenye and Mbwei, rural Tanzania

6.1 Chapter summary

This is the last results chapter for the primary data collection. The previous chapter focused on the relationships between plant agrobiodiversity and both dietary diversity and nutritional status. This chapter will build on this work by investigating the relationships between these health outcomes and animal diversity. The previous chapter reported plant agrobiodiversity as measured in the household farms and as the number of crops grown over the previous calendar year. This chapter reports animal agrobiodiversity using the number of animals raised over the previous year. This will be referred to as the animal diversity score to maintain consistency with the terminology used chapter 5. This chapter also reports how eating meat and animal products and selling animals and animal products are associated with dietary diversity and nutritional status.

This chapter helps meet objective 3 (Measure household crop and animal diversity, dietary diversity of respondents and children under five and nutritional status of children living in these villages, determine whether household produce is sold and investigate how these factors are related) and objective 4 (Investigate basic socio-demographic factors affecting dietary intake of children under five and nutrition outcomes in children in these villages) of this thesis. The objectives of this chapter are:

- Objective 6A:To assess animal diversity and present descriptive data on animalproduct consumption and animal rearing practices in the two villages.
- Objective 6B: To examine whether eating animals and animal products and animal diversity are associated with dietary diversity in respondents and under five year olds and nutritional status in children.
- Objective 6C:To examine whether selling animals and animal products is associatedwith dietary diversity in respondents and under five year olds and

nutritional status in children.

The results presented in this chapter have shown that households eating more animal products have members with higher dietary diversity in Minyenye and Mbwei. Higher animal diversity scores are associated with higher dietary diversity in Minyenye only. Households eating more animal products and households with higher animal diversity scores have poorer nutritional status in Minyenye only. Selling animals and animal products is not associated with dietary diversity in either village. However, Minyenye households selling animals and animal products have children with poorer nutritional status.

These results contradict much of the literature on the consumption of animal products in low income countries (Krebs et al., 2011). These results have shown that eating and raising animals is not necessarily beneficial to diet and nutritional status of subsistence farmers living in environments such as rural Tanzania. This is an important finding that should encourage caution in those working to improve nutritional status through animal rearing focused interventions.

6.2 Introduction

Meat and animal products have the potential to make an important contribution to dietary diversity and nutritional status and increasing animal protein in the diets of those living in rural sub-Saharan Africa has long been heralded as a potential avenue for improving undernutrition in these communities (Gibson et al., 2003, Bwibo and Neumann, 2003).

There is evidence in the literature that meat supplementation or higher intakes of animal products are linked to better dietary quality and improvements in micronutrient status but there is limited evidence of the impact on growth. Despite finding a significant inverse association between meat consumption and stunting in populations in Guatemala, Democratic Republic of Congo, Zambia, and Pakistan (Krebs et al., 2011), a year long trial using 30 to 45g of meat per day failed to decrease the rates of stunting in this population (Krebs et al., 2012). Gibson et al. (Gibson et al., 2003) developed a dietary diversification intervention which included increasing the consumption of animal source food, mainly fish, and showed significant improvements in MUAC z-scores and arm muscle area. There were no improvements in height or weight after 12 months. Providing a daily snack containing meat to Kenyan school children for 21 months showed significantly improved cognitive function over a milk or energy supplement and the control group in a randomised trial (Whaley et al., 2003). Dietary quality

(Murphy et al., 2003), plasma vitamin B-12 (Siekmann et al., 2003), mid upper-arm muscle area and mid-upper-arm fat area (Neumann et al., 2013) were increased in the meat and milk groups. But there were no changes in height (Grillenberger et al., 2003).

Cross-sectional data provides some evidence of the link between animal product consumption and growth in communities not involved in development programs or interventions. Marquis et al (Marquis et al., 1997) found linear growth to be positively associated with intake of animal products but only in children with low intakes of complementary foods. Consumption of animal foods was significantly correlated with weight and height z-scores in Mexican children 18 to 30 months of age (Allen et al., 1992).

Literature on the effects of animal agrobiodiversity and animal rearing more generally on diet and nutritional status is lacking. Two of the papers reporting a positive associations between agrobiodiversity and dietary diversity used a score which combined plant and animal agrobiodiversity (Correlation of 0.697 (Ekesa et al., 2008) and coefficient of determination=0.496 (Walingo and Ekesa, 2013) (significances not reported)). Leroy and Frongillo (2007) conducted a review on the effects of promoting animal production in order to combat undernutrition in which the majority of studies found improved dietary intake with animal production. The authors suggest this is not necessarily through the consumption of animals raised but, more likely, through increased income (See Figure 6.1 taken from Leroy and Frongillo (2007)). All the projects that measured nutritional status integrated different food production interventions and nutrition education and showed improvements in the prevalence of iron deficiency, serum retinol and ferritin and child growth. As these were multicomponent interventions it is not possible to determine how much of this effect was due to improvements in animal production.



Figure 6.1. Pathway between livestock production and nutritional status (Leroy and Frongillo, 2007)

There is some evidence to suggest that increasing the intake of animal products in low income countries can improve measures of health (Gibson et al., 2003, Whaley et al., 2003, Murphy et al., 2003, Siekmann et al., 2003, Neumann et al., 2013), especially in the context of multi-component interventions (Gibson et al., 2003, Leroy and Frongillo, 2007). However, due to the complex set of determinants of diet and nutritional status in these communities, it is important to investigate whether animal diversity and consumption of meat and animal products is linked in communities receiving no food production intervention. This chapter will outline the associations between meat and animal product intake and animal diversity and both dietary diversity and nutritional status. Along with providing additional evidence of the affect of animal product intake in low income environments this chapter provides complementary data for the last chapter on plant agrobiodiversity, dietary diversity and nutritional status.

6.3 Methods

6.3.1 Data collection

The food sources section of the household questionnaire was used to collect the animal food sources and animal diversity data for this chapter (see chapter 3 for full details of methods). Details on how dietary diversity and nutritional status was measured and transformed for analysis were described in chapter 3 and 4. The specific information collected on animal diversity, animal rearing, animal and animal product consumption and the statistical analysis methods used in this chapter are described below.

The respondent was asked if members of the household had consumed chickens, eggs, cows, milk, sheep, goats, pigs, ducks or fish in the past calendar year. For positive responses the respondents were asked if the meat or animal products came from animals raised in the household or if they were bought, gifted to the household or obtained from the wild. For animals that were raised by the household additional questions about the number of animals the household raised and whether the animals, meat or animal products were sold by the household were also asked. The number of different types of animals that were raised by the household – the animal diversity score, was used to reflect animal agrobiodiversity.

6.3.2 Data analysis

6.3.2.1 Descriptive statistics

Descriptive data of crude associations are presented as means (95% Confidence Intervals) of continuous variables and percentages within groups for categorical variables (chapter objective 6A).

6.3.2.2 Linear regression analysis

Linear regression was used to estimate the relationship between the number and type of animals raised and eaten by the household and dietary diversity/food variety and nutritional status (chapter objective 6B). Again, a DAG was used to identify potential confounders to these relationships (Glymour, 2006) (See description on page 73, chapter 3 and Figure 3.16). These variables were added into the model one by one to assess if they affected the regression coefficients for the independent variable of interest. When the potential confounder modified the regression coefficient substantially they were included in the model. Variables included in

the regression models investigating the relationships between animal consumption and rearing and dietary diversity or food variety were: household mobile phone; respondent and husband's highest level of education; whether husband is employed as a farmer or other employment; respondent and husband's frequency of employment and number of farms the household uses for growing food.

In the multivariable regression analyses investigating the relationships with nutritional status these potentially confounding variables were included: respondent and husband's highest level of education; whether husband is employed as a farmer or other employment and respondent's frequency of employment.

Unadjusted and multivariable regression were used to assess the relationship between selling animals, meat and animal products and dietary diversity and nutritional status (chapter objective 6C). Analyses for dietary diversity were adjusted for: household mobile phone; respondent and husband's highest level of education; whether respondent and husband are employed as a farmer or other employment; respondent has small business on top of farming; respondent earns extra income; number of farms the household uses for growing food; average size of household farms. Analyses for nutritional status were adjusted for: household mobile phone; respondent and husband's highest level of education; respondent and husband's frequency of employment; average size of household farms.

6.4 Results

6.4.1 Animal product consumption and animal diversity in the two villages

In Minyenye, the average animal diversity score was 1.42 (95% CI: 1.21, 1.63), this is the number of types of animals raised by the household. Of 64 households in Minyenye, the majority raised chickens, approximately 40% raised cows and less than 10% raised sheep, goats, pigs or ducks (Figure 6.4). On average those households raising sheep, goats or pigs (only 6 households) raised approximately 11 of these animals, the 43 households raising cows raised, on average, about 7 cows and those 53 households raising chickens raised about 10 chickens per household.



Figure 6.2. Animals kept in a pen in a households yard and being taken to get water in Minyenye

In Mbwei, the mean animal diversity score was 1.12 (95% CI: 0.85, 1.40). Approximately 55% of the 58 households raised chickens, 35% raised cows and 15% raised ducks. No households in either village kept fish. Only one household in Mbwei reported raising sheep and goats and they raised 5 of these animals in total. Thirty-five households raised cows and each household, on average raised 2.5 cows. Of those 45 households raising chickens each household raised on average 5 chickens.



Figure 6.3. A single cow being kept by a household in Mbwei



Figure 6.4. Percentage of households raising specific animals in the Minyenye and Mbwei

100% of households ate animals in the past year and 87.5% (95% CI: 79.2, 95.8) raised animals. 4.03 (95% CI: 3.74, 4.32) different types of animals were eaten by the household in the previous calendar year. Of those households raising animals 62.0% (95% CI: 48.1, 75.9) of households sold animals or meat in the past calendar year.

A high proportion of households in Minyenye ate chicken (83%) and eggs (69%), with most of these foods coming from animals they raised in the household (Table 6.1). Approximately 85% of households ate beef with almost all of those households buying this meat. Just over half the households drank milk, with 80% of this milk coming from cows they raised in the household. All households ate fish in the last calendar year, approximately three quarters of the households ate meat from goats, 40% ate meat from sheep, 13% meat from pigs and 5% duck meat. The majority of this meat was bought by the household rather than raised and a small proportion was gifted to the household. Sixteen percent of households ate no chicken, 8% ate no meat from goats, sheep or pig and 20% ate no milk or eggs in the last calendar year. No households in Minyenye consumed any animals or animal products from wild sources (data not shown).

Forty-eight percent of all the Minyenye households sold meat and 29% sold eggs or milk. Of those households raising chickens 60% also sold them and approximately 40% sold eggs for additional income. Only one household sold a cow for income and five households sold milk from cows they raised. Three households sold goat, one sold pig while none of the households raising sheep or ducks sold these animals for additional income.

In the previous calendar year in Mbwei, 4.40 (95% CI: 3.87, 4.92) types of animals were eaten by the household. In the past calendar year 94.8% (95% CI: 89.0, 100) of households ate animals and 69.0% (95% CI: 56.7, 81.2) raised animals. Of those households raising animals 21.9% (95% CI: 6.7, 37.0) of households sold some of the animals or meat.

Compared to Minyenye, a lower proportion of households in Mbwei raised animals. Sixtythree percent raised chickens and 16% raised goats, sheep. A high proportion of households ate both chicken and eggs in the last calendar year. Approximately three quarters of households ate meat from cows and all of this meat was bought by the households. Eighty percent of households consumed milk, approximately 40% got milk from cows they raised in the household and approximately 70% bought milk. Approximately 70% of households consumed sheep and goat meat in the previous calendar year with the majority of this meat being bought. Just over half the households consumed duck meat with a quarter of households raising the ducks themselves and three quarters buying this meat. About 90% of households ate fish with all households buying this fish. Twenty one percent of households ate no chicken, 24% ate no meat from goats, sheep and 14% ate no milk or eggs in the last calendar year. No households in Mbwei ate meat from pigs. As in Minyenye, no households in Mbwei consumed any animals or animal products from wild sources (data not shown).

A much lower proportion of Mbwei households, as compared to Minyenye, sold animals or animal products. Twelve percent of Mbwei households reported selling some meat and 18% sold eggs or milk. Of those households raising specific animals six households sold chicken, eggs and milk and one household sold sheep and goat. Seven households who raised ducks sold some of the animals or meat.

		Of those who ate:			Of those
					that raised:
	Ate	Raised	Bought	Gift	Sold
	(N)	(N)	(N)	(N)	(N)
Minyenye (N=64)					
Chickens	53	50	3	0	30
Eggs	44	42	3	0	16
Cows	54	2	53	0	1
Milk	35	28	6	3	5
Sheep	25	2	22	1	0
Goat	49	5	42	2	3
Pig	8	1	6	1	1
Duck	3	1	2	0	0
Fish	64	0	63	1	0
Mbwei (N=58)					
Chickens	46	30	18	0	6
Eggs	36	24	13	0	6
Cows	43	0	43	0	0
Milk	48	21	35	0	6
Sheep	42	2	40	0	1
Goat	40	2	38	0	1
Pig	0	0	0	0	0
Duck	33	8	24	1	7
Fish	51	0	51	0	0

Table 6.1. Number of households in Minyenye and Mbwei eating, raising, buying, being given and selling animals and animal products in the last calendar year

*Respondents could respond with more than one source

6.4.2 Minyenye

6.4.2.1 Animal diversity and dietary diversity

Eating animals and animal products and animal diversity is positively associated with dietary diversity in Minyenye. The more types of animals the households had consumed in the past calendar year and the higher the animal diversity score the higher the food variety scores of the respondents and their oldest child under the age of five (Table 6.2). The more different types of animals eaten (in the unadjusted model only) and animal diversity (in both the unadjusted and the adjusted models) the higher the respondent's dietary diversity scores. For example, for each additional type of animal raised by the household the respondent's dietary diversity increased by 0.42.

Table 6.2. Associations between the number of different animals eaten by the household and animal diversity scores and dietary diversity and food variety in respondents and under five year olds in Minyenye

	Dietary di	Dietary diversity score		riety score
	Child <5 years	Respondent	Child<5 years	Respondent
	Regression coefficient	Regression coefficient	Regression coefficient	Regression coefficient
	(95% CI)	(95% CI)	(95% CI)	(95% CI)
No. animals/animal p	roducts eaten			
Unadjusted [^]	0.19 (-0.02, 0.40)	0.23 (0.07, 0.40)	0.67 (0.31, 1.03)	0.62 (0.27, 0.96)
Adjusted^^	0.15 (-0.07, 0.37)	0.20 (-0.01, 0.40)	0.68 (0.27, 1.08)	0.57 (0.13, 1.01)
Animal diversity score	es			
Unadjusted [^]	0.01 (-0.45, 0.43)	0.37 (0.03, 0.72)	0.70 (-0.09, 1.49)	1.14 (0.43, 1.85)
Adjusted^^	0.08 (-0.37, 0.52)	0.42 (0.03, 0.81)	1.00 (0.17, 1.83)	1.31 (0.50, 2.13)

^Unadjusted model. N was 61 for children under five and 64 for respondents.

^^Adjusted for household mobile phone, respondent and husband's highest level of education, whether husband is employed as a farmer or other employment, respondent and husband's frequency of employment, number of farms the household uses for growing food. N was 54 for children under five and 57 for respondents.

6.4.2.2 Animal diversity and nutritional status

Eating animals and animal products and animal diversity is negatively associated with nutritional status in Minyenye. In the adjusted models the more different types of animals eaten by the household the lower the BMI z-scores of all children and those children under the age of five (Table 6.3). The higher the animal diversity score, the lower the BMI z-scores in all children only. Additionally, in children under five only, the higher the animal diversity score the lower the children's MUAC in both unadjusted and adjusted models. No other significant associations were seen between the number of animals eaten and the animal diversity score and nutritional status in children.

Table 6.3. Association between the number of animals eaten by the household and animal diversity scores and nutritional status in all children and children under five in Minyenye

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient	Regression coefficient	Regression coefficient	Regression coefficient
	(95% CI)	(95% CI)	(95% CI)	(95% CI)
Children <5				
No. animals eaten				
Unadjusted [^]	0.05 (-0.08, 0.18)	-0.03 (-0.15, 0.09)	-0.10 (-0.21, 0.02)	-0.10 (-0.22, 0.03)
Adjusted^^	0.05 (-0.11, 0.21)	-0.07 (-0.21, 0.06)	-0.15 (-0.28, -0.02)	-0.14 (-0.29, 0.01)
Animal diversity sco	ores			
Unadjusted [^]	0.07 (-0.20, 0.34)	-0.05 (-0.28, 0.19)	-0.14 (-0.37, 0.08)	-0.27 (-0.52, -0.02)
Adjusted^^	0.04 (-0.29, 0.36)	-0.13 (-0.40, 0.13)	-0.23 (-0.48, 0.03)	-0.35 (-0.65, -0.06)
All children				
No. animals eaten				
Unadjusted [^]	0.02 (-0.06, 0.09)	-0.02 (-0.09, 0.05)	-0.01 (-0.09, 0.06)	-0.03 (-0.18, 0.11)
Adjusted^^	0.04 (-0.04, 0.13)	-0.05 (-0.13, 0.03)	-0.09 (-0.17, -0.01)	-0.07 (-0.24, 0.09)
Animal diversity sco	ores			
Unadjusted [^]	0.01 (-0.14, 0.15)	-0.08 (-0.22, 0.06)	-0.08 (-0.22, 0.07)	-0.06 (-0.35, 0.23)
Adjusted^^	0.03 (-0.14, 0.19)	-0.13 (-0.28, 0.02)	-0.17 (-0.32, -0.02)	-0.06 (-0.38, 0.27)

^Unadjusted model. N was 101 for height, 100 for weight, 102 for BMI z-scores and 103 for MUAC in children under five. N was 251 for height, 251 for weight, 255 for BMI z-scores and 252 for MUAC in all children.

^^Adjusted for respondent and husband's highest level of education, whether husband is employed as a farmer or other employment, respondent's frequency of employment. N was 93 for height, 93 for weight, 95 for BMI z-scores and 95 for MUAC in children under five. N was 230 for height, 231 for weight, 235 for BMI z-scores and 232 for MUAC in all children.

6.4.2.3 Selling animal products and dietary diversity

Selling animal products is not associated with dietary diversity in Minyenye. No significant associations were seen between whether a household sold milk or eggs and the dietary diversity and food variety scores of the respondents or their oldest children under five (Table 6.4).

Selling animals or meat was positively associated with respondent's food variety scores. In the unadjusted model those households who sold animals or meat from animals had respondents and children with higher food variety scores and respondents with higher dietary diversity scores. When the models were adjusted for potential confounders the association between whether animals were sold and the respondent's food variety score was the only coefficient that remained significant. Respondents from households selling animals or meat had food variety scores approximately two points higher than those in households not selling animals or meat.

Table 6.4. Association between whether animals and animal products are sold and dietary diversity and food variety in respondents and under five year olds in Minyenye

	Dietary diversity score		Food variety score	
	Child <5 years	Respondent	Child<5 years	Respondent
Sold milk and/or eggs				
Unadjusted [^]	-0.43 (-1.43, 0.57)	0.31 (-0.45, 1.07)	-0.66 (-2.48, 1.16)	0.04 (-1.62, 1.70)
Adjusted^^	-0.40 (-1.31, 0.52)	0.31 (-0.48, 1.09)	-0.97 (-2.73, 0.78)	-0.02 (-1.89, 1.86)
Sold animals/meat				
Unadjusted [^]	0.86 (-0.03, 1.75)	0.78 (0.09, 1.46)	2.06 (0.47, 3.65)	2.44 (1.09, 3.80)
Adjusted^^	0.18 (-0.76, 1.11)	0.72 (-0.11, 1.55)	0.91 (-0.95, 2.76)	2.19 (0.55, 3.83)

^Unadjusted model. N was 43 for children under five and 46 for respondents for sold milk and/or eggs. N was 47 for children under five and 50 for respondents for sold animals/meat.

^^Adjusted for household mobile phone, respondent and husband's highest level of education, whether respondent and husband are employed as a farmer or other employment, respondent has small business on top of farming, respondent earns extra income, number of farms the household uses for growing food, average size of household farms. N was 40 for children under five and 43 for respondents for sold milk and/or eggs. N was 44 for children under five and 47 for resp ondents for sold animals/meat.

6.4.2.4 Selling animal products and nutritional status

Selling animals and animal products is negatively associated with nutritional status in Minyenye. Households selling milk and/or eggs had under five year olds with significantly lower height z-scores compared to those in households not selling milk and/or eggs (Table 6.5). Looking at all children shows that households selling milk and/or eggs have children with lower height and weight z-scores in both unadjusted and adjusted models.

Households selling animals or meat from animals they raised had under fives and all children with significantly lower BMI z-scores than those not selling animals or meat. In the adjusted model these households had children with significantly lower weight z-scores and MUAC.

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient	Regression coefficient	Regression coefficient (95%	Regression coefficient
	(95% CI)	(95% CI)	CI)	(95% CI)
Under 5				
Sold milk and/or eggs				
Unadjusted [^]	-0.74 (-1.35, -0.14)	-0.45 (-0.98, 0.08)	0.19 (-0.34, 0.72)	-0.07 (-0.62, 0.48)
Adjusted^^	-0.75 (-1.46, -0.03)	-0.45 (-1.05, 0.16)	0.18 (-0.40, 0.76)	-0.19 (-0.83, 0.46)
Sold animals/meat				
Unadjusted [^]	0.12 (-0.45, 0.68)	-0.25 (-0.70, 0.19)	-0.44 (-0.88, -0.01)	-0.42 (-0.95, 0.10)
Adjusted^^	-0.00 (-0.72, 0.72)	-0.43 (-0.98, 0.11)	-0.56 (-1.08, -0.04)	-0.62 (-1.26, 0.01)
All Children				
Sold milk and/or eggs				
Unadjusted [^]	-0.32 (-0.65, -0.00)	-0.35 (-0.65, -0.05)	-0.19 (-0.48, 0.11)	0.24 (-0.35, 0.83)
Adjusted^^	-0.39 (-0.74, -0.04)	-0.33 (-0.65 <i>,</i> -0.01)	-0.14 (-0.46, 0.17)	0.06 (-0.59, 0.70)
Sold animals/meat				
Unadjusted [^]	0.04 (-0.25, 0.33)	-0.21 (-0.47, 0.05)	-0.35 (-0.62, -0.09)	-0.50 (-1.05, 0.05)
Adjusted^^	0.03 (-0.33, 0.38)	-0.32 (-0.64, -0.00)	-0.45 (-0.76, -0.13)	-0.92 (-1.57, -0.26)

Table 6.5. Association between whether animals and animal products are sold and nutritional status in all children and children under five in Minyenye

^Unadjusted model. N was 68 for height, 67 for weight, 69 for BMI z-scores and 70 for MUAC in children under five and was 175 for height, 175 for weight, 179 for BMI z-scores and 176 for MUAC in all children for sold milk and/or eggs. N was 76 for height, 76 for weight, 78 for BMI z-scores and 78 for MUAC in children under five and was 207 for height, 208 for weight, 212 for BMI z-scores and 207 for MUAC in all children for sold animals/meat.

^^Adjusted for household mobile phone, respondent and husband's highest level of education, respondent and husband's frequency of employment, average size of household farms. N was 64 for height, 64 for weight, 66 for BMI z-scores and 66 for MUAC in children under five and was 163 for height, 164 for weight, 168 for BMI z-scores and 165 for MUAC in all children for sold milk and/or eggs. N was 73 for height, 73 for weight, 75 for BMI z-scores and 75 for MUAC in children under five and was 198 for height, 199 for weight, 203 for BMI z-scores and 199 for MUAC in all children for sold animals/meat.

6.4.3 Mbwei

6.4.3.1 Animal diversity and dietary diversity

Eating animals and animal products but not animal diversity scores were positively associated with dietary diversity in Mbwei. The more animals eaten by the household in the past calendar year the higher the respondent's and the under five year old's dietary diversity and food variety in the previous 24 hours in both the unadjusted and the adjusted analyses (Table 6.6), as was the case for food variety in Minyenye. The higher the animal diversity score the higher the respondent's and her oldest child under five's food variety score in the unadjusted model only. This association was seen in both the unadjusted and the adjusted models in Minyenye.

	Dietary diversity score		Food va	riety score
	Child <5 years	Respondent	Child<5 years	Respondent
	Regression coefficient	Regression coefficient	Regression coefficient	Regression coefficient
	(95% CI)	(95% CI)	(95% CI)	(95% CI)
No. animals/animal produ	ucts eaten			
Unadjusted [^]	0.18 (0.04, 0.32)	0.22 (0.09, 0.34)	0.57 (0.29, 0.85)	0.54 (0.30, 0.78)
Adjusted^^	0.22 (0.08, 0.37)	0.23 (0.09, 0.36)	0.58 (0.29, 0.88)	0.51 (0.26, 0.76)
Animal diversity scores				
Unadjusted [^]	0.34 (-0.02, 0.70)	0.31 (-0.02, 0.64)	1.09 (0.36, 1.81)	0.85 (0.20, 1.50)
Adjusted^^	0.37 (-0.08, 0.82)	0.35 (-0.06, 0.76)	0.89 (-0.05, 1.84)	0.62 (-0.18, 1.42)

^Unadjusted model. N was 55 for children under five and 58 for respondents.

^^Adjusted for household mobile phone, respondent and husband's highest level of education, whether husband is employed as a farmer or other employment, respondent and husband's frequency of employment, number of farms the household uses for growing food. N was 54 for children under five and 57 for respondents.

6.4.3.2 Animal diversity and nutritional status

Eating animals and animal products and animal diversity scores were not associated with nutritional status in Mbwei. No significant associations were seen between the number of animals eaten by the household or the animal diversity scores for the last calendar year and the nutritional status variables of the children (Table 6.7). This differs from the results seen in Minyenye where eating animals and animal diversity was negatively associated with nutritional status.

Table 6.7. Association between the number of animals eaten by the household and animal diversity scores and nutritional status in all children and children under five in Mbwei

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient	Regression coefficient	Regression coefficient	Regression coefficient
	(95% CI)	(95% CI)	(95% CI)	(95% CI)
Children < 5				
No. animals eaten				
Unadjusted [^]	-0.07 (-0.17, 0.03)	-0.01 (-0.10, 0.07)	0.02 (-0.05, 0.10)	-0.04 (-0.14, 0.06)
Adjusted^^	-0.07 (-0.18, 0.04)	0.00 (-0.09, 0.09)	0.04 (-0.04, 0.11)	-0.04 (-0.14, 0.07)
Animal diversity scores				
Unadjusted [^]	-0.10 (-0.33, 0.13)	-0.02 (-0.23, 0.18)	0.04 (-0.13, 0.21)	-0.06 (-0.29, 0.17)
Adjusted^^	-0.04 (-0.31, 0.22)	0.08 (-0.14, 0.31)	0.12 (-0.06, 0.30)	0.03 (-0.22, 0.29)
All children				
No. animals eaten				
Unadjusted [^]	-0.01 (-0.09, 0.06)	0.01 (-0.06, 0.07)	0.00 (-0.05, 0.06)	0.03 (-0.08, 0.14)
Adjusted^^	-0.01 (-0.09, 0.06)	0.01 (-0.06, 0.08)	0.01 (-0.05, 0.07)	0.04 (-0.08, 0.15)
Animal diversity scores				
Unadjusted [^]	-0.06 (-0.24, 0.11)	0.01 (-0.16, 0.17)	0.07 (-0.08, 0.21)	0.04 (-0.24, 0.31)
Adjusted^^	-0.01 (-0.21, 0.18)	0.10 (-0.08, 0.27)	0.14 (-0.01, 0.29)	0.09 (-0.21, 0.38)

^Unadjusted model. N was 197 for height, 100 for weight, 99 for BMI z-scores and 101 for MUAC in children under five. N was 163 for height, 167 for weight, 169 for BMI z-scores and 180 for MUAC in all children.

^^Adjusted for respondent and husband's highest level of education, whether husband is employed as a farmer or other employment, respondent's frequency of employment. N was 95 for height, 98 for weight, 97 for BMI z-scores and 99 for MUAC in children under five. N was 161 for height, 165 for weight, 167 for BMI z-scores and 178 for MUAC in all children.

6.4.2.3 Selling animal products and dietary diversity

Selling animals and animal products is not associated with dietary diversity in Mbwei. No significant associations were seen between whether households sold milk, eggs or meat and respondent's and children's dietary diversity and food variety scores (Table 6.8). Similar non-significant associations were seen in Minyenye in the majority of associations between selling animal produce and measures of diversity in the diet.
Table 6.8. Association between whether animals and animal products are sold and dietary diversity and food variety in respondents and under five year olds in Mbwei

	Dietary	Dietary diversity score		riety score
	Child <5 years	Respondent	Child<5 years	Respondent
Sold milk and/or eggs				
Unadjusted [^]	0.44 (-0.67, 1.56)	-0.11 (-1.15, 0.93)	0.35 (-1.93, 2.62)	0.47 (-1.34, 2.27)
Adjusted^^	0.53 (-1.18, 2.24)	0.50 (-0.98, 1.97)	0.20 (-3.11, 3.51)	0.70 (-1.84, 3.24)
Sold animals/meat				
Unadjusted [^]	0.10 (-1.15, 1.35)	0.15 (-0.84, 1.14)	0.88 (-1.54, 3.30)	1.66 (0.01, 3.31)
Adjusted^^	0.22 (-1.24, 1.68)	0.28 (-0.71, 1.26)	0.67 (-2.39, 3.73)	1.36 (-0.29, 3.00)

^Unadjusted model. N was 31 for children under five and 34 for respondents for sold milk and/or eggs. N was 30 for children under five and 32 for respondents for sold animals/meat.

^^Adjusted for household mobile phone, respondent and husband's highest level of education, whether respondent and husband are employed as a farmer or other employment, respondent has small business on top of farming, respondent earns extra income, number of farms the household uses for growing food, average size of household farms. N was 29 for children under five and 32 for respondents for sold milk and/or eggs. N was 29 for children under five and 31 for respondents for sold animals/meat.

6.4.3.4 Selling animal products and nutritional status

Selling animals and animal products shows mixed associations with nutritional status in Mbwei. Those households selling milk and/or eggs had under five year olds with weight zscores approximately 0.8 z-scores lower than those not selling milk and/or eggs in the adjusted model (Table 6.9). This is similar to the negative associations seen between selling animal produce and nutritional status in Minyenye. Households selling animals or meat in Mbwei, however, had children with approximately 1cm larger MUAC in both unadjusted and adjusted models.

	Height z-scores	Weight z-scores	BMI z-scores	MUAC
	Regression coefficient	Regression coefficient	Regression coefficient	Regression coefficient
	(95% CI)	(95% CI)	(95% CI)	(95% CI)
Under five year olds				
Sold milk and/or eggs				
Unadjusted [^]	-0.60 (-1.38, 0.19)	-0.57 (-1.25, 0.12)	-0.20 (-0.76, 0.35)	-0.27 (-1.06, 0.51)
Adjusted^^	-0.73 (-1.65, 0.19)	-0.82 (-1.62, -0.03)	-0.34 (-0.98, 0.29)	-0.30 (-1.17, 0.56)
Sold animals/meat				
Unadjusted [^]	-0.46 (-1.26, 0.34)	-0.23 (-1.00, 0.54)	0.24 (-0.44, 0.92)	0.30 (-0.56, 1.16)
Adjusted^^	-0.56 (-1.55, 0.43)	-0.36 (-1.29, 0.58)	0.24 (-0.57, 1.04)	0.29 (-0.68, 1.25)
All children				
Sold milk and/or eggs				
Unadjusted [^]	-0.27 (-0.81, 0.26)	-0.27 (-0.77, 0.23)	-0.16 (-0.58, 0.26)	-0.18 (-1.05, 0.70)
Adjusted^^	-0.51 (-1.13, 0.11)	-0.53 (-1.11, 0.05)	-0.23 (-0.73, 0.26)	-0.33 (-1.36, 0.71)
Sold animals/meat				
Unadjusted [^]	-0.11 (-0.61, 0.39)	-0.03 (-0.55, 0.49)	0.01 (-0.45, 0.47)	1.04 (0.07, 2.00)
Adjusted^^	-0.23 (-0.82, 0.36)	-0.08 (-0.69, 0.53)	0.10 (-0.44, 0.64)	1.14 (0.02, 2.26)

Table 6.9. Association between whether animals and animal products are sold and nutritional status in all children and children under five in Mbwei

^Unadjusted model. N was 58 for height, 60 for weight, 59 for BMI z-scores and 60 for MUAC in children under five and was 105 for height, 108 for weight, 107 for BMI z-scores and 112 for MUAC in all children for sold milk and/or eggs. N was 55 for height, 57 for weight, 56 for BMI z-scores and 57 for MUAC in children under five and was 96 for height, 99 for weight, 98 for BMI z-scores and 103 for MUAC in all children for sold animals/meat.

^^Adjusted for household mobile phone, respondent and husband's highest level of education, respondent and husband's frequency of employment, average size of household farms. N was 55 for height, 57 for weight, 56 for BMI z-scores and 57 for MUAC in children under five and was 98 for height, 101 for weight, 100 for BMI z-scores and 103 for MUAC in all children for sold milk and/or eggs. N was 53 for height, 55 for weight, 54 for BMI z-scores and 55 for MUAC in children under five and was 92 for height, 95 for weight, 94 for BMI z-scores and 97 for MUAC in all children for sold animals/meat.

6.5 Discussion

6.5.1 Animal diversity, dietary diversity and nutritional status

Animal diversity and consumption of animals and animal products is linked with higher dietary diversity but not nutritional status. The majority of households in Minyenye and Mbwei both raised animals and consumed some meat and animal products in the last calendar year. The more different types of animals eaten by the households the higher the food variety score of the respondents and the under five year olds were. This did not however convert into better nutritional status. If anything, the opposite was true; in Minyenye either no association or a negative association with nutritional status was seen; the greater the number of animals eaten, the poorer the children's BMI z-scores were. In Mbwei, there were no significant associations seen with any of the nutritional status variables.

A similar pattern was seen with the animal diversity score and dietary diversity and nutritional status in Minyenye. Animal diversity was positively associated with dietary diversity scores in the respondents and food variety scores in both the respondents and the children. However, either no associations were seen with nutritional status or negative associations were seen. In Mbwei, there were no associations seen between the animal diversity score and either the dietary diversity/food variety scores or nutritional status variables. These results indicate that there is no benefit to nutritional status of households raising animals in these communities. These results contradict much of the research in this area (Marquis et al., 1997, Allen et al., 1992, Ekesa et al., 2008, Walingo and Ekesa, 2013).

The negative association seen between meat consumption and BMI z-scores seen in Minyenye may be due to an increased risk of infection introduced by the meat, for example from hookworm (Pasricha et al., 2008), Salmonella or Campylobacter (Pouillot et al., 2012). The negative association seen between animal diversity and nutritional status seen in Minyenye may simply be because those buying meat are better off financially which is associated with better nutritional status (Shack et al., 1990) or it may actually have something to do with raising these animals.

There are a number of differences between these two villages that could help explain the different associations seen. Due to the dry nature of the land in Minyenye, at the time of survey, there was not very much grass available for animals to eat and it was often the children's responsibility to take the animals to graze elsewhere. This may have impacted on the children's energy expenditure and requirements. Kumar and Hotchkiss found that how much children were involved in collection, grazing and agricultural activities was negatively

- 198 -

related to their nutritional status in Nepal (Kumar and Hotchkiss, 1988). Despite Mbwei households being more likely to own animals (53% in Mbwei compared to 5% in Minyenye), Minyenye households were more likely to raise animals (86% in Minyenye compare to 69% in Mbwei). Households in Minyenye who were raising animals kept a higher number of those animals compared to Mbwei. Time and energy spent and resource use for animal rearing is

therefore higher in Minyenye.

Interestingly, despite these two villages being quite different (Table 4.2, chapter 4 and Table 6.1, chapter 6) associations between animal diversity and eating animal products and meat and dietary diversity and nutritional status are similar in this study. There is evidence in the literature that supplementing children's diets with animal products in a low income country context leads to better diet quality (Murphy et al., 2003), micronutrient status (Siekmann et al., 2003) and cognition (Whaley et al., 2003, Murphy et al., 2003, Siekmann et al., 2003, Neumann et al., 2013). However, this study illustrates that in communities with no outside intervention, households eating meat and animal products do not necessarily have children with better nutritional status. This, along with the results from chapter 5, provide evidence against the pathway illustrated in figure 3.2 a) and b): increases in agrobiodiversity do not necessarily lead to improved nutritional status through increased dietary diversity.

6.5.2 Animal rearing for reasons other than consumption

In attempting to understand why this might be it is important to consider the reasons why people were raising animals. It has been documented (Fafchamps et al., 1998, Dovie et al., 2006) that it is common for animals to be kept, not to be slaughtered for food, but to sell when the household needs money. Animals act as insurance against times of extreme food insecurity or can be sold to cover the cost of school fees (Mazzeo, 2011). Animals raised also fulfil the important task of providing fertiliser for household farms (Powell and Williams, 1994). Often a household would raise animals that they did not own, and therefore could not slaughter for food, in order to get fertiliser for their farms. In these villages it was common for households to raise chickens, not to provide eggs for consumption, but to eat or sell the chicken s the eggs would develop into.

6.5.3 Barriers to animal diversity positively impacting nutritional status

Poultry diseases are common in these areas, reducing the number of animals available to eat and sell. It is also common for animals in these villages, especially chickens, to roam in and out of houses and to live in the yards outside the house (personal observation). According to

199

Upton (2000) it is difficult to control the spread of infectious diseases among such widely dispersed system of poultry production. These practices can lead to zoonosis (Angulo et al., 2000, Butzler, 2004), which impacts negatively on children's nutritional status through infection (Stephensen, 1999).

The impact on caregivers time, although not assessed by this study, should also be considered. It is possible that the amount of time it takes the female head of household to care for these animals takes time away from food preparation and childcare (Guèye, 2005, Kumar and Hotchkiss, 1988).

6.5.4 Selling animals and animal products

There is published evidence that raising animals improves the household income (Nielsen et al., 2003) which, in turn, has been shown to be linked positively to children's nutritional status (Shack et al., 1990, Yang et al., 2012). The results of this study do not support this hypothesis. In Minyenye, either no association or positive associations were seen between animals, meat, eggs or milk being sold and dietary diversity and food variety and either no association or negative associations were seen with nutritional status variables. Eggs/milk being sold was negatively related to height z-scores and animals or meat being sold was negatively associated with weight, BMI and MUAC. In Mbwei, all associations between animals and their products being sold and dietary diversity and food variety were non-significant. Milk and/or eggs being sold was negatively associated with weight in the under five year olds and meat or animals being sold was positively associated with MUAC in children. That selling animals and their products was not positively associated with nutritional status variables would suggest that if households were making extra money by raising and selling animals, this was not getting fed back into buying household food.

6.5.5 Limitations and strengths

A number of the limitations outlined in chapter 4, such as translation, study dynamics and the research environment are also relevant to the results in this chapter. Translation errors may have affected the accuracy of the studies results recorded. Cross cultural dynamics and having family members present at the interview could have affected the honesty of the responses from the participants. Additionally, results are cross-sectional and based on respondent's memory of household consumption and animal rearing practices over the previous calendar year. This recall bias may have affected the accuracy of the results given by the participant. However, the respondent was well placed to answer questions about both diet and farming

practices due to her central role in food production and preparation. The questionnaire collected very thorough information on the animal foods consumed and the animals raised by the household. Data was translated at the time of data collection and recorded by the researcher. This allowed information to be checked on collection to minimise missing data.

6.6 Conclusions

This chapter presents detailed, household level information on animal consumption and animal diversity of households and whether this animal produce was sold. It tests associations between these practices and both dietary diversity and nutritional status in order to inform understanding on how animal food production practices are related to diet and nutritional status in these rural villages.

The key results of this chapter are that, despite having higher dietary diversity, households eating more animal products and households with higher animal diversity scores had poorer nutritional status in Minyenye. Mbwei households eating more animal products have members with higher dietary diversity but no associations were seen with nutritional status. Additionally, no associations were seen with animal diversity and dietary diversity or nutritional status. Selling animals and animal products is not associated with dietary diversity in either village but households in Minyenye selling animals and animal products have children with poorer nutritional status. These results go against much of the literature published in this area, which generally encourages the consumption of animal products in low income countries in order to improve nutritional status.

The mixed associations between raising animals and nutritional status raise questions about how successful interventions which solely encourage households to raise animals in order to improve nutritional status are likely to be in low income countries. Rural households reporting eating and raising animals were worse off nutritionally than those not eating and raising animals in both villages despite having different animal rearing, dietary and social practices.

Additionally, exploring the reasons why supplying animal foods to communities has a different effect on diet and health than local improvements in animal production could provide some interesting lessons on effective interventions. Reasons for raising animals, the uses these animals are put to and animal rearing practices need to be more fully investigated before interventions designed to increase animal rearing in rural areas are implemented.

Meat and animal products are rich sources of protein and essential micronutrients. As such, they have potential to improve nutritional status in populations that are malnourished. The 201 projects, perhaps surprising, results again highlight the complexity of the determinants of die t and nutritional status in these contexts. Viewing this project's results in the context of the wider literature, which has shown animal rearing interventions to be more successful as part of a multi component projects, should encourage researchers and development organisations into more thorough approaches to combating malnutrition. The results presented in this chapter suggest caution and in-depth research into local contexts and practices before outside intervention by researchers and development organisations attempting to improve nutritional status through encouraging the rearing of livestock.

Chapter 7:

Investigating the relationships between land cover, dietary diversity and nutritional status in a national sample in Tanzania

7.1 Chapter summary

The previous three results chapters have presented the results and analyses of primary data collected at local village scales. This chapter broadens analysis to investigate the factors of interest at the national level. Due to data availability at the national level there are a number of differences in the methodology and resulting data as compared to chapter 5. This earlier chapter focused on agrobiodiversity as measured at the level of household farms, this chapter has taken a broader view and links land cover to individual diet and nutrition variables. This chapter meets objective 5 of the thesis (Investigate the socio-demographic determinants of dietary diversity and nutritional status and explore the relationships between land cover, dietary diversity and nutritional status in children under five years at a national level, in Tanzania). The chapter objectives are:

- Objective 7A: To investigate whether demographic, social, agricultural and dietary factors are associated with dietary diversity and nutritional status in under five year olds in Tanzania.
- Objective 7B: To investigate whether dietary diversity is associated with nutritional status in under five year olds in Tanzania.
- Objective 7C: To investigate whether dietary diversity and nutritional status in under five year olds vary spatially in Tanzania.
- Objective 7D: To investigate whether land cover is related to dietary diversity and nutritional status in under five year olds in Tanzania.

The results in this chapter show that those children with higher dietary diversity had lower height for age. The later children had complementary foods introduced to their diet the lower their height for age. Children from households using piped, tank, rain or bottled water for drinking or taking action to make water safe to drink had higher height for age. Those recently experiencing diarrhoea had poorer nutritional status. The more agricultural land cover surrounding the DHS EA centre point the higher the respondent and child dietary diversity scores were but the lower the children's height. More agricultural and forest land cover was associated with a higher weight for height.

These results contribute knowledge about the determinants of nutritional status in Tanzania. In contradiction to much of the literature carried out in low income countries, dietary diversity was negatively related to nutritional status. To date there has been little research linking land cover to diet and nutritional status, this research provides some important data to fill this gap.

7.2 Introduction

Chapters 4 to 6 have suggested and discussed some potential determinants of nutritional status in low income countries that were identified by work carried out in two villages in Tanzania. To broaden the usefulness of this work this chapter will conduct similar analyses using data that is less detailed but from a much larger, nationally representative study conducted in Tanzania in 2010; the Demographic and Health Survey (DHS). In order to explore the relationship between diet, nutritional status and agrobiodiversity at this level, land cover maps are used as a proxy for agrobiodiversity. Relationships between land cover as well as social, demographic, agricultural and dietary factors and children's nutritional status will be explored.

Satellite data comes from environmental sensors that have been placed in orbit to observe the earths surface (Brown, 2009). Data from this remote sensing technique has been used to determine land cover. Land cover has been defined as the observed (bio)physical cover on the earth's surface by Di Gregorio and Jansen (1998). There are a number of reasons why land cover might be related to diet and health. One of these reasons is that land cover and biodiversity are related. There is a strong precedence for using satellite data to estimate biodiversity of plant populations (Skidmore et al., 2003, Walker et al., 1992, Tucker and Sellers, 1986) and to estimate vegetation or land cover in order to locate animal habitats (Austin et al., 1996, Homer et al., 1993, Miller and Conroy, 1990).

Few studies, however, have used satellite data to link land use or land cover to diet and human health (Xu et al., 2008, Brown, 2009, Johnson et al., 2013, Powell et al., 2011). There are many ways in which changes in land use can impact on human health including risk of flash flooding, risk of malaria and changes to food production (Xu et al., 2008). Brown (2009) outlines how remote sensing can be used to identify conditions which may lead to famine in particular spatial, temporal and social contexts. Relevant conditions identified through remote sensing

204

include climate, specifically rainfall, and plant growth which, Brown states, can be used to estimate agricultural production.

A study published in 2013 (Johnson et al., 2013) used Normalized Difference Vegetation Indices (NDVI) and Vegetation Continuous Fields (VCF) based on Moderate Resolution Imaging Spectroradiometer (MODIS) satellite data collected for Malawi and linked percentage forest cover to dietary diversity, intake of vitamin A rich foods, incidence of diarrhoea and rates of stunting. They found that children living in areas with more forest cover consumed more vitamin A rich foods and were less likely to have diarrhoea. Powell et al (Powell et al., 2011) calculated tree cover from Landsat eTM+ and Probationary System of Earth Observation (SPOT) satellite data to investigate forest cover and dietary intake in the East Usambara mountains. While the relationship between dietary diversity and tree cover were not directly examined in this study they did find that households with greater tree cover nearby were more likely to consume forest foods and individuals using forest foods had higher dietary diversity.

Linking large scale measures of vegetation or land cover to health outcomes is a relatively new area of inquiry. This chapter will add to this area by providing data on the association between land cover, dietary diversity and nutritional status at a national level in Tanzania. This chapter will also provide additional data on how dietary diversity and nutritional status are related in Tanzania and on other social and demographic factors associated with nutritional status to add to the data from the primary data collection. This national level data will strengthen or contend the hypotheses discussed in the previous chapters. How the primary data collection and this national level analysis come together will be discussed further in chapter 8.

7.3 Methods

7.3.1 Data sources

7.3.1.1 Demographic and Health Survey data

The 'Measure DHS' project (Demographic and Health Surveys) has been collecting nationally representative data on maternal and child health, gender, and nutrition across the world since 1984 (Measure DHS, 2014). The project is funded by the US Agency for International Development (USAID). DHS is a source of good quality data on a wide range of demographic and health factors in Tanzania (The DHS Program, 2014a). The DHS survey carried out in Tanzania in 2010 (DHS VI) was used for these analyses. GPS co-ordinates were captured at the centre point of each of the 475 clusters included in this survey. Clusters were based on Enumeration Areas (EA) defined by the 2002 population and housing census. These are the

205

- 205 -

units of randomisation for the DHS and household data collection takes place within the selected EA, an average of 15 households per EA. Seventy-six percent of clusters were rural and 24% were classed as urban. 19% of the clusters were on Zanzibar or Pemba and 5% were in Dar es Salaam.

This research project was registered with 'Measure DHS'. The 'Individual recode dataset', for the respondent and child data, and 'Household recode dataset', for household level data were downloaded (The DHS Program, 2014c) as Stata files. Variables relevant to this project were selected from these datasets and combined into one Stata file using the household ID to merge the variables from the two different datasets together. Average dietary diversity, nutrition, demographic and social variables at DHS EA level were linked to the GPS coordinates through the EA number present in each original dataset.

The outcomes of interest are nutritional status and dietary diversity. Nutritional status measures that were used are BMI of the respondents and height for age and weight for height z-scores in the respondent's youngest child.

In the DHS, dietary diversity was calculated from a 24 hour recall for the respondents and their youngest child. The dietary diversity data available through the DHS is presented as a 'yes', 'no' or 'don't know' for each food category. This was recoded into a dietary diversity score based on the food groups used in chapters 4 to 6 ('Cereals, roots and tubers'; 'vitamin A rich vegetables, tubers and fruit'; 'other vegetables'; 'other fruits'; 'flesh meats, organ meats, fish and insects'; 'eggs'; 'legumes, nuts and seeds'; 'milk and milk products'; 'oils, fats and sweets'), as recommended by the FAO (2004) dietary diversity workshop (FAO/WHO/IFPRI, 2004). Unfortunately, in the DHS questionnaire other vegetables and other fruit were asked in one question so the dietary diversity score used for this chapter is out of eight food categories instead of nine.

The DHS collected information on basic demographics and background, reproduction, pregnancy, breastfeeding, health, marriage, husband's background, women's work and residence throughout Tanzania. Potentially confounding variables were identified from these categories using the DAG described in chapter 3 (page 73, Figure 3.16). Whether these variables confounded the relationships between land cover and dietary diversity and nutritional status was checked by adding them to the model one by one and seeing if they modified the regression coefficient substantially (criteria varied from approximately >0.02 to >0.05 depending on the size of the regression coefficients). They were included in the multivariable model if they confounded the relationships. Questions on the number of meals per day, number of days in the last seven days meat and fish were eaten and some food preparation data were also collected. DHS VI has specific questions on agriculture directed at 206 the women respondents that were used in the analysis. These included whether they had land usable for agriculture and how much land.

7.3.1.2 Land cover

The Land Cover Classification System (LCCS) was developed by the United Nations Food and Agriculture Organisation (UNFAO). It is an attempt to develop an internationally agreed reference base for land cover. It is a hierarchical system based on a set of independent diagnostic criteria eg. life form, cover, height and density (Di Gregorio and Jansen, 1998). It is a flexible system with two main phases. The first phase distinguishes eight major land cover types:

- Cultivated and Managed Terrestrial Areas
- Natural and Semi-Natural Terrestrial Vegetation
- Cultivated Aquatic or Regularly Flooded Areas
- Natural and Semi-Natural Aquatic or Regularly Flooded Vegetation
- Artificial Surfaces and Associated Areas
- Bare Areas
- Artificial Water bodies, Snow and Ice, and
- Natural Water bodies, Snow and Ice.

In the second phase classes are created using pre-defined classifiers which are specific to each of the above land cover types. The result is a system that allows comparisons of land cover over different smaller areas and projects but is flexible enough to accurately categorise different areas. For example, within the 'Cultivated and Managed Terrestrial Areas' land cover type sits the following classes: 'Post-flooding or irrigated croplands (or aquatic)', 'Rainfed croplands', 'Mosaic cropland (50-70%) / vegetation (20-50%)' and 'Mosaic vegetation (50-70%) / cropland (20-50%)'. The LCCS was used to define land cover for land cover maps of Tanzania produced by GlobCover 2009.

7.3.1.3 GlobCover 2009 data

A global land cover map was produced in 2010 by the European Space Agency (ESA) and the Université catholique de Louvain based on data collected from January to December of 2009. The map is based on MERIS (Medium Resolution Imaging Spectrometer Instrument) fine resolution surface reflectance mosaics (Bontemps et al., 2011). MERIS is a wide field of view pushbroom imaging spectrometer on ENVISAT, an ESA environmental satellite. It measures 15 spectral bands of solar radiation that is reflected by the earth (Rast et al., 1999). The map is in geographic coordinates in a Plate-Carrée projection (WGS84 ellipsoid) (Bontemps et al., 2011).

The land cover map goes through a number of pre-processing (geometric corrections, cloud screening, atmospheric corrections, bidirectional reflectance distribution function correction and time compositing) and classification stages (spectro-temporal classification, labelling, validation). The final land cover map has a 300m resolution and counts 22 GlobCover categories within the 8 land cover classes defined with the United Nations LCCS, listed above (Bontemps et al., 2011).

GlobCover 2009 data was downloaded from the ESA GlobCover Portal (European Space Agency, 2010). The data was unzipped and added to the base map of Africa in ArcGIS version 10. The land cover categories for GlobCover2009 are presented in Figure 7.1. Those land cover categories that were assigned to the three categories used in this study: Agricultural, Forest and Other land cover are presented in Box 7.1-7.3. These categories are very similar to those used in site selection (Box 3.1 and 3.2) but are stricter on what constitutes biodiverse forest. The decision on what land cover categories were assigned to agricultural, forest and other land cover were based on the LCCS categories. The agricultural land cover is identical to the croplands used for site selection and it covers all the categories of the LCCS 'Cultivated Terrestrial Areas and Managed Lands'. The forest land cover category includes 6 out of the 8 categories of the LCCS 'Natural and Semi-natural Terrestrial Vegetation: Woody-Trees' and all three of the LCCS 'Natural and Semi-natural Aquatic Vegetation'. The other land cover category includes 2 categories from LCCS 'Natural and Semi-natural Terrestrial Vegetation: Woody-Trees' and all those in the Shrub and Herbaceous sub-section of LCCS 'Natural and Semi-natural Terrestrial Vegetation' as well as those in LCCS 'Artificial Surfaces', 'Bare Areas' and 'Inland Water bodies, snow and ice'.

Box 7.1. Land cover types classified as agricultural land cover

When assessing the land cover in the surrounding 5km of the DHS EA centre points in ArcGIS the following codes were used to represent 'agricultural' land cover:

- 1) Post-flooding or irrigated croplands (none present in Tanzania)
- 2) Rainfed croplands
- 3) Mosaic Cropland (50-70%) / Vegetation (grassland, shrubland, forest) (20-50%)
- 4) Mosaic Vegetation (grassland, shrubland, forest) (50-70%) / Cropland (20-50%)

Box 7.2. Land cover types classified as forest land cover

Whether biodiverse land falls within 5km of the DHS centrepoint is also of interest in this

chapter. The land cover classes used to represent the more biodiverse 'forest' areas are:

1) Closed to open (>15%) broadleaved evergreen and/or semi-deciduous forest (>5m)

2) Closed (>40%) broadleaved deciduous forest (>5m)

3)Open (15-40%) needleleaved deciduous or evergreen forest (>5m);

4) Mosaic Forest/Shrubland (50-70%) / Grassland (20-50%)

5) Mosaic Grassland (50-70%) / Forest/Shrubland (20-50%)

6) Closed (>40%) broadleaved forest regularly flooded - Fresh water

7) Closed (>40%) broadleaved forest or shrubland permanently flooded – Saline or brackish water (none present in Tanzania)

8) Closed to open (>15%) vegetation (grassland, shrubland, woody vegetation) on regularly flooded or waterlogged soil - Fresh, brackish or saline water.

Box 7.3. Land cover types classified as other land cover

The land cover categories that fall into the 'other' category, considered to be not biodiverse, include:

1) Open (15-40%) broadleaved deciduous forest (>5m);

2) Closed to open (>15%) mixed broadleaved and needleleaved forest (>5m);

3) Closed to open (>15%) shrubland (<5m)

4)Closed to open (>15%) grassland

- 5) Sparse (>15%) vegetation (woody vegetation shrubs, grassland)
- 6) Artificial surfaces and associated areas (urban areas >50%)
- 7) Bare areas
- 8) Water bodies



GlobCover2009 land cover categories







Figure 7.2. Globcover 2009 map of Tanzania with DHS EA centre points

7.3.2 Data analysis

7.3.2.1 Descriptive statistics

Figure 7.2 shows the land cover map with the location of the DHS EA used in the analysis. DHS EA are not distributed evenly throughout the country but are designed to collect representative data on the population at national and regional levels.

Basic descriptive statistics from the DHS data were calculated using Stata (version 12). Means (95% confidence intervals) and percentages (95% confidence Intervals) are presented for these descriptive results. BMI cut-offs of <20, 25-30 and >30 were used to determine whether the

respondent was categorised as underweight, overweight or obese. A cut-off of -2 was used to define whether a child was stunted (height for weight z-scores) or wasted (weight for height z-scores).

7.3.2.2 Linear regression analysis

Linear regression was used to investigate the relationships between demographic, social, agricultural and dietary factors and dietary diversity and nutritional status variables (chapter objective 7A). Factors potentially associated with nutritional status were identified through the literature and the results presented in chapters 4-6. The DHS is a rich dataset with many similar variables likely to be reflecting the same underlying construct. Multi-collinearity negatively effects the integrity of the regression model (UCLA: Statistical Consulting Group, 2014). In order to avoid this factors were grouped into categories (wealth; marital status; education; literacy; employment; urban/rural; sanitation; breastfeeding; complementary feeding; health; supplementation) and redundant variables within these categories were disregarded. This meant that there were only between 1 and 3 variables per category in the models (see Table 7.1). Univariate and multivariable regression with all relevant variables included in the model are presented.

nutritional status	
Area	Employment
Urban or rural	 Respondent's occupation
Distance to nearest health facility	 Husband's occupation
Demographics	Wealth
Current marital status	Wealthindex
Education	Drinking water/sanitation
 Education (years) 	 Source of drinking water
	 Something done to make water safe
 Husband's education (years) 	to drink
Literacy	Vaccinations and medication
Literacy	Received vitamin A in last 6 months

Table 7.1. The 17 covariates in 10 categories identified as being potential determinants of nutritional status

• Health

- Children < 5 slept under bednet last night
- Had diarrhoea recently

Breastfeeding and complementary feeding Given foods/liquids other than breastmilk in first 3 days

- Age when first fed with other food
- Currently being breastfed

As this chapter is specifically focused on the relationship between dietary diversity and nutritional status separate regression analyses were conducted to investigate these

relationships (chapter objective 7B). Variables that could potentially confound the relationship between dietary diversity and height for age and weight for height z-scores in the youngest child were identified using a DAG, as described in chapter 3 (page 73, figure 3.16). Then, for the multivariable regression, as in chapter 4-6, variables were added to the model one by one to determine whether they influenced the regression coefficients for dietary diversity. When the potential confounders modified the coefficient substantially they were included in the model. Variables included were: respondent's education in single years; frequency of watching television; currently breastfed; age when first fed with other food; wealth index; respondent employed all year or seasonally; respondent works for family, other, self.

Cluster analyses were carried out in Satscan version 9.1.1 in order to investigate whether dietary diversity and nutritional status vary spatially (chapter objective 7C). This program uses the GPS co-ordinates collected at DHS EA level to identify clusters of both high and low levels of the dietary diversity and nutritional status variables. Satscan clusters can be described as areas where the value of the characteristic of interest is unusual as compared to the area surrounding it. Clusters of the continuous variables dietary diversity, height for age and weight for age z-scores in children and dietary diversity and BMI in the respondents were analysed using a continuous normal model. This model, designed for continuous data, uses a likelihood function based on a normal distribution (Kulldorff, 2005). For the binary variables of whether the children were stunted or wasted a Bernoulli Model was used. The Bernoulli Model codes individuals as cases or controls, in this instance a case would be a child that was stunted and a control would be a child that was not stunted. For the analysis Satscan draws all possible windows, of variable sizes, centred on the DHS EA midpoints calculating a scan statistic for the normal models and prevalence for the Bernoulli models and these statistics are compared between within the window and outside the window. Those windows that have the scan statistic which is least likely due to chance is the most likely spatial cluster. P-values are assigned to the clusters (Kulldorff, 2005). The cluster output of the most likely clusters were then mapped by GPS co-ordinates using ArcGIS version 10 and the results which show where the clusters of high and low values are situated are presented as maps of Tanzania.

7.3.2.3 Land cover analyses

To investigate whether land cover is related to dietary diversity and nutritional status (chapter objective 7D) two analyses were carried out using data derived from the GlobCover 2009 land cover map using ArcGIS version 10. The dietary diversity and nutrition data averaged at DHS EA level was linked with this land cover data through the DHS EA GPS co-ordinates. These analyses

- 213 -

213

were carried out to determine whether surrounding land cover was associated with diet and nutrition outcomes in the DHS survey.

- 214 -

The first analysis aimed to present the percentage agricultural, forest and other land cover types surrounding the DHS EA in high and low Satscan clusters in order to determine whether land cover was different around areas with high dietary diversity or good nutritional status as compared to areas with poor dietary diversity and nutritional status. The proportion of the total area that was defined as agriculture, forest and other was calculated in buffer zones 5km from the DHS EA centre point. Figure 7.3. illustrates the buffer zones around the DHS centre points and shows the different areas each land cover type occupies. The light green area represents the 5km buffer around the DHS centre points (original categories are retained). ArcGIS was used to sum these areas by each of the 22 different land cover types. All the land cover types defined in this project as agricultural, forest were then added together in order to calculate the total proportion of the buffer they occupy. The same was done for all land cover types defined as other. For example, in the image on the right the majority of the land cover is "Rainfed croplands" (yellow) with some "closed to open (>15%) grassland" (orange) and "closed to open (>15%) shrubland (<5m)" (brown). A high proportion of the 5km buffer will be calculated to be rainfed croplands which is classed by this project as agricultural land cover.



Figure 7.3. Globcover2009 map with the 5km buffer zones around the DHS EA centre point marked in green; the image on the right is a close up look at the central cluster from the image on the left (different colours represent different land cover types)

This information was then linked to the Satscan clusters described above. The land cover surrounding the high and low clusters of dietary diversity and nutritional status variables were

summed and the average proportion of the surrounding land cover that fell into the agriculture, forest and other categories is presented. For example, the proportion of agricultural land cover in the surrounding 5km of the DHS centre points identified in the Satscan cluster of high dietary diversity and the proportion of agricultural land cover in the area surrounding those DHS in the low dietary diversity Satscan cluster were calculated and can be compared in the results table. This was repeated for the following variables with significantly high and low Satscan clusters; child dietary diversity, height for age z-scores, weight for height z-scores and mothers BMI.

In the second analysis, regression was used to determine whether land cover is significantly associated with average dietary diversity and nutritional status variables. The percentage of agricultural, forest and other land cover in the 5km surrounding the DHS centre points were used as the independent variables in the regression models. A verage dietary diversity and nutritional status variables. Both unadjusted models and models adjusted for whether the DHS EA was defined as urban or rural were run.

7.4 Results

7.4.1 Demographic characteristics of participants in the Tanzanian 2010 DHS

A total of 10139 female respondents took part in the DHS survey. Of the respondents that had children under the age of five, they had an average of 1.9 children each (Table 7.2). Over 60% of these respondents were married with approximately 30% having never married and about 10% being widowed, divorced or separated. The respondents and their husbands had, on average, 6 years of education. Approximately 70% of respondents fell in to the highest level of literacy; 'can read whole sentences' while almost 30% couldn't read at all. Fifty percent of respondents and 60% of their husbands worked in agriculture. Twenty-four percent of respondents and less than 1% of their husbands reported not working.

Total number of respondents	10139
Number of children under five per respondent (mean (95% CI))*	1.8 (1.8, 1.9)
Marital status (% (95% CI))	
Married/living together	62. (61.7, 63.5)
Nevermarried	26.7 (25.9, 27.5)
Widowed/divorced/separated	10.7 (10.1, 11.3)
Education (mean (95% CI))	
Average number of years of education	5.9 (5.8, 5.9)
Husband's average number of years of education	6.1 (6.0, 6.2)
Literacy (% (95% CI))	
Cannot read at all	27.4 (26.6, 28.3)
Able to read only parts of sentence	5.4 (5.0, 5.8)
Able to read whole sentence	67.1 (66.2, 68.0)
Respondent's employment (% (95% CI))	
Working in agriculture	48.8 (47.9 <i>,</i> 49.8)
Working in other area	26.8 (26.0, 27.7)
Notworking	24.3 (23.5, 25.1)
Husband's employment (% (95% CI))	
Working in agriculture	58.6 (57.5, 59.6)
Working in other area	41.1 (40.0, 42.2)
Notworking	0.4 (0.2, 0.5)

Table 7.2. Respondent's characteristics information in the Tanzanian 2010 DHS

*excluding those with no children under five

Table 7.3 shows the households characteristics from the DHS survey. Two thirds of households had a radio in the household and over half had mobile phones and a bicycle. Under 20% had electricity. Around a quarter of the households got their water from a public or neighbours tap/standpipe and another quarter from an open well. Twenty percent got their water from a borehole, river, dam, lake, pond, stream, canal or spring. Almost 15% used piped water. Over 60% of households took no action to make their drinking water safe. Almost 30% of households boiled their drinking water to make it safe. On average it took household members approximately 30 minutes to collect their drinking water. Over 50% of households used an open pit latrine toilet while 17% used a closed or improved pit latrine and another 17% had no facility available to them. 12% used a flush toilet.

Characteristics and ownership (% (95% CI))	
Radio in household	66.0 (65.0, 66.9)
Mobile phone in household	57.3 (56.4, 58.2)
Bicycle in household	51.1 (50.1, 52.1)
Household has electricity	19.1 (18.4, 19.9)
Television in household	17.1 (16.4, 17.9)
Refrigerator in household	9.5 (9.0, 10.0)
Motorcycle/scooter in household	4.9 (4.5, 5.3)
Car/truck in household	2.8 (2.5, 3.1)
Water source (% (95% CI))	
Public/neighbours tap/standpipe	27.7 (26.9, 28.6)
Open well	23.1 (22.2, 23.9)
Borehole/river/lake/spring etc	21.6 (20.8, 22.4)
Piped water	13.6 (12.9, 14.2)
Protected well	11.3 (10.7, 11.9)
Tank	1.7 (1.5, 2.0)
Rainwater	0.7 (0.6, 0.9)
Bottled	0.3 (0.2, 0.4)
Take action to make water safe (% (95% CI))*	
Nothing	61.8 (60.9, 62.7)
Boil	29.2 (28.4, 30.1)
Other e.g. Strain, let settle	9.0 (8.4, 9.5)
Time to get water minutes (mean (95% CI))	27.6 (26.9, 28.3)
Type of toilet (%)	
Open pit latrine	54.4 (53.4, 55.3)
Closed or improved pit latrine	16.5 (15.8, 17.2)
No facility	16.6 (15.9, 17.3)
Flush	12.4 (11.7, 13.0)
Other	0.2 (0.1, 0.3)

Table 7.3. Household characteristics in the 7	Fanzanian 2010 DHS
---	--------------------

Respondents had an average BMI of approximately 23kg/m²: 27% were underweight, 51% were normal weight, 16% were overweight and 6% were obese (Table 7.4). The height for age z-scores for the respondent's youngest child was -1.5 with 38% of these children being classed as stunted. The average weight for height z-scores was -0.1; 7% of the children were wasted.

Respondents(N)	10041		
BMI (kg/m², mean (95% CI))	22.7 (22.6, 22.8)		
Underweight (% (95% CI))	26.9 (26.0, 27.7)		
Normal weight (% (95% CI))	51.1 (50.2, 52.1)		
Overweight (% (95% CI))	15.7 (15.0, 16.3)		
Obese (% (95% CI))	6.3 (5.9, 6.8)		
Children under 5 (N)	4730		
Height for age z-scores (mean (95% CI))	-1.5 (-1.5, -1.4)		
Weight for height z-scores (mean (95% CI))	-0.1 (-0.1, -0.0)		
Stunted (% (95% CI))	36.4 (35.1 <i>,</i> 37.7)		
Wasted (% (95% CI))	7.4 (6.7, 8.1)		

- 218 -

Table 7.4. Respondent's and their youngest child's nutritional status

7.4.2 Demographic and social factors, dietary diversity and nutritional status.

The mean (95% CI) dietary diversity score for respondents and their youngest child was 3.03 (3.00, 3.07) and 2.56 (2.50, 2.61) out of 8 categories. Focusing on the adjusted regression model which was created including all variables in table 7.5 (respondent's education, husband's education, respondent's occupation, husband's occupation, current marital status, wealth index, literacy, urban/rural, time to get to water source, currently breastfed, age when first fed with other food, given foods/liquids other than breastmilk in first 3 days) the respondent's education remains significantly positively associated with both the child's and the respondent's dietary diversity score. For example, for each additional year the respondent was in education the child's dietary diversity scores increased by 0.06. The respondent's and their husband's occupation remains associated with the respondent's dietary diversity only. Diversity is highest for professional, technical, administrative and service occupations but interestingly, agricultural occupation is now positively associated with the respondent's dietary diversity diversity. A higher wealth index is associated with a lower dietary diversity in the respondents only.

On average (95% CI) the youngest child was breastfed until they were 19.3 (19.0, 19.5) months old. 29.8% of children were given food or drink other than breastmilk in their first three days of life. On average (95% CI) children were 3.8 (3.7, 3.8) months when they were first fed foods or drinks other than breastmilk. Children that are currently being breastfed have lower dietary diversity and children who were older when complementary feeding began have higher dietary diversity.

218

	Youngest child dietary diversity score		Respondent dieta	ary diversity score
	Unadjusted	Adjusted^	Unadjusted	Adjusted^
	Regression coefficient (95% CI)	Regression coefficient (95% CI)	Regression coefficient (95% CI)	Regression coefficient (95% CI)
Demographics				
Education (years)	0.08 (0.07, 0.10)	0.06 (0.03, 0.09)	0.08 (0.07, 0.09)	0.04 (0.01, 0.06)
Husband's education (years)	0.07 (0.05, 0.08)	0.01 (-0.01, 0.03)	0.07 (0.06, 0.08)	0.01 (-0.00, 0.02)
Respondent's occupation				
Not working	-0.05 (-0.20, 0.10)	0.28 (-0.23, 0.79)	-0.02 (-0.13, 0.09)	0.26 (-0.71, 0.19)
Professional, technical, managerial or clerical	1.14 (0.72, 1.57)	-	0.83 (0.53, 1.13)	-0.31 (-0.90, 0.28)
Agricultural	-0.31 (-0.43, -0.20)	0.23 (-0.28, 0.74)	-0.40 (-0.48, -0.32)	-0.06 (-0.51, 0.39)
Household or domestic	0.32 (-0.27, 0.91)	0.06 (-0.70, 0.82)	0.38 (-0.03, 0.80)	-
Services	0.44 (0.07, 0.82)	0.15 (-0.45, 0.75)	0.63 (0.37, 0.90)	-0.06 (-0.58, 0.46)
Manual	0.36 (0.20, 0.51)	0.36 (-0.15, 0.86)	0.48 (0.37, 0.59)	0.02 (-0.42, 0.48)
Husband's occupation				
Notworking	0.16 (-1.10, 1.41)	0.49 (-0.80, 1.77)	-0.03 (-0.98, 0.92)	0.35 (-0.68, 1.38)
Professional, technical, managerial or clerical	0.57 (0.31, 0.83)	0.38 (-0.34, 1.09)	0.68 (0.49, 0.86)	0.79 (0.24, 1.35)
Agricultural	-0.31 (-0.43, -0.19)	0.32 (-0.36, 1.00)	-0.42 (-0.50, -0.34)	0.56 (0.03, 1.09)
Household or domestic	0.44 (-0.34, 1.23)	-	-0.03 (-0.58, 0.52)	-
Services	0.72 (0.50, 0.95)	0.64 (-0.07, 1.34)	0.55 (0.39, 0.71)	0.77 (0.23, 1.32)
Manual	-0.02 (-0.16, 0.11)	0.14 (-0.55, 0.82)	0.16 (0.07, 0.26)	0.45 (-0.07, 0.98)

Table 7.5. Association between household and demographic variables and dietary diversity in the respondent and their youngest child

Table continued on page 211.

	Youngest child die	tary diversity score	Respondent dietary diversity score	
	Unadjusted	Adjusted^	Unadjusted	Adjusted^
	Regression coefficient (95% CI)	Regression coefficient (95% CI)	Regression coefficient (95% CI)	Regression coefficient (95% CI)
Current marital status				. ,
Never married	-0.02 (-0.27, 0.23)	-	0.15 (-0.02, 0.33)	-
Married/living together	-0.12 (-0.29, 0.04)	0.02 (-0.18, 0.21)	-0.04 (-0.16, 0.08)	-0.03 (-0.18, 0.12)
Widowed, divorced or separated	0.21 (0.00, 0.42)	-	-0.04 (-0.19, 0.11)	-
Wealth index (5 point scale)	0.23 (0.19, 0.27)	0.13 (0.08, 0.19)	0.27 (0.24, 0.30)	0.16 (0.12, 0.20)
Respondent's literacy				
Cannot read at all	-0.39 (-0.51, -0.27)	-1.21 (-3.89, 1.47)	-0.45 (-0.53, -0.37)	0.29 (-1.90, 2.48)
Able to read only parts of sentence	-0.08 (-0.32, 0.16)	-1.22 (-3.91, 1.48)	-0.17 (-0.34, 0.00)	0.29 (-1.91, 2.48)
Able to read whole sentence	0.39 (0.27, 0.50)	-1.30 (-3.99, 1.40)	0.47 (0.39 <i>,</i> 0.55)	0.33 (-1.87, 2.52)
Lived rurally (Y/N)	-0.52 (-0.66, -0.38)	-0.10 (-0.28, 0.07)	-0.70 (-0.80, -0.60)	-0.28 (-0.41, -0.15)
Time to get to water source (hours)	-0.08 (-0.16, 0.01)	-0.00 (-0.09, 0.08)	-0.18 (-0.24, -0.11)	-0.04 (-0.11, 0.02)
Breastfeeding and complementary feeding				
Currently breastfed (Y/N)	-1.00 (-1.14, -0.86)	-0.49 (-0.63, -0.36)		
Age when first fed with other food (months)	0.17 (0.14, 0.19)	0.16 (0.13, 0.19)		
Given foods/liquids other than breastmilk in first 3				
days (Y/N)	0.01 (-0.11, 0.14)	0.06 (-0.06, 0.18)		

- 220 -

^ Model adjusted for all other independent variables in the table. N is 2567 for the youngest child and 2563 for the respondent.

Variable with missing regression results in the adjusted model were excluded due to collinearity.

When the 17 covariates from the 10 categories identified as being important potential determinants of nutritional status (shown in Table 7.1) were added to the model simultaneously a number of significant associations were seen with height for age z-scores of the youngest child (Table 7.6). The wealthier the household the higher the child's height for age. For each additional point on the five point scale height for age improved by 0.14. Children in households where the respondent either did not work or worked in a manual job had higher height for age. There was no effect of her husband's occupation. The more literate the respondent was the higher the children's weight for height for height z-scores.

Children from households doing something to make water safe to drink had 0.16 higher height for age z-scores. Children from households using water that was piped in, was tank, rain or bottled water had higher height for age z-scores, as did those from households using well water. Children who had received vitamin A in the last 6 months had 0.31 lower height for age z-scores. Children who had diarrhoea recently had lower height for age by 0.14 and lower weight for height by 0.13 z-scores.

Those children currently being breastfed had higher height for age, by 0.52 z-scores, but lower weight for height z-scores, by 0.09 z-scores, compared to children not currently being breastfed. The older the child was when they were first fed foods or liquids other than breastmilk the lower their height for age z-scores were. Those first given food/liquids between 0 and 3 months of age had an mean height for age of -1.34, between 3 and 6 months; -1.57, between 6 and 12 months; -1.73 and more than 12 months; -1.93.

	Height for age z-scores		Weight for height z-scores	
	Unadjusted	Adjusted^	Unadjusted	Adjusted^
	Regression coefficient	Regression coefficient	Regression	Regression coefficient
	(95% CI)	(95% CI)	coefficient (95% CI)	(95% CI)
Demographics				
Education (years)	0.05 (0.04, 0.06)	0.01 (-0.01, 0.04)	0.00 (-0.01, 0.01)	-0.01 (-0.04, 0.01)
Husband's education (years)	0.04 (0.03, 0.05)	0.01 (-0.00, 0.02)	0.01 (-0.00, 0.02)	0.01 (-0.00, 0.03)
Respondent's literacy				
Cannot read at all	-0.17 (-0.26, -0.08)	2.06 (-0.72, 4.83)	-0.07 (-0.15, 0.01)	2.47 (-0.09, 5.02)
Able to read only parts of sentence	-0.09 (-0.27, 0.08)	1.90 (-0.88, 4.68)	0.08 (-0.08, 0.24)	2.63 (0.07, 5.19)
Able to read whole sentence	0.18 (0.09, 0.27)	1.96 (-0.82, 4.74)	0.05 (-0.03, 0.12)	2.60 (0.04, 5.16)
Respondent's occupation				
Not working	0.51 (0.40, 0.62)	0.53 (0.11, 0.96)	-0.32 (-0.42, -0.22)	-0.06 (-0.45, 0.34)
Professional, technical, managerial or clerical	0.39 (0.10, 0.68)	0.31 (-0.24 ,0.86)	0.11 (-0.15, 0.37)	0.26 (-0.25, 0.77)
Agricultural	-0.54 (-0.62, -0.45)	0.23 (-0.19, 0.65)	0.16 (0.8, 0.23)	0.22 (-0.17, 0.61)
Household or domestic	-0.10 (-0.46, 0.26)	-	-0.39 (-0.71, -0.07)	-
Services	0.38 (0.10, 0.65)	0.35 (-0.16, 0.85)	0.14 (-0.10, 0.39)	0.28 (-0.19, 0.74)
Manual	0.31 (0.20, 0.41)	0.44 (0.01, 0.86)	0.03 (-0.06, 0.13)	0.14 (-0.25 ,0.53)
Husband's occupation				
Not working	0.28 (-0.59, 1.15)	-	-0.43 (-1.20, 0.34)	-
Professional, technical, managerial or clerical	0.35 (0.16, 0.54)	-0.10 (-0.96, 0.76)	0.08 (-0.10, 0.25)	0.19 (-0.60, 0.98)
Agricultural	-0.47 (-0.56, -0.39)	-0.25 (-1.09, 0.59)	0.10 (0.02, 0.18)	0.30 (-0.47, 1.07)
Household or domestic	0.78 (0.22, 1.34)	0.41 (-0.61, 1.42)	-0.82 (-1.32, -0.33)	-0.52 (-1.46, 0.41)
Services	0.42 (0.25, 0.58)	0.01 (-0.84, 0.86)	-0.19 (-0.34, -0.04)	0.10 (-0.68, 0.89)
Manual	0.33 (0.23, 0.42)	-0.10 (-0.94, 0.75)	-0.05 (-0.14, 0.04)	0.23 (-0.55, 1.00)

Table 7.6. Association between household and demographic variables and nutritional status in the youngest child

Table continued on page 214.

	Height for age z-scores		Weight for h	eight z-scores
	Unadjusted	Adjusted^	Unadjusted	Adjusted^
	Regression coefficient	Regression coefficient	Regression	Regression coefficient
	(95% CI)	(95% CI)	coefficient (95% CI)	(95% CI)
Current marital status				
Never married	0.00 (-0.18, 0.19)	-	0.09 (-0.08, 0.25)	-
Married/living together	0.11 (-0.01, 0.23)	-0.02 (-0.17, 0.13)	-0.00 (-0.11, 0.10)	0.10 (-0.04, 0.24)
Widowed, divorced or separated	-0.16 (-0.30, -0.02)	-	-0.05 (-0.18, 0.08)	-
Wealth index (5 point scale)	0.18 (0.15, 0.21)	0.14 (0.10, 0.19)	0.01 (-0.02, 0.04)	0.02 (-0.02, 0.07)
Lived rurally (Y/N)	-0.53 (-0.63, -0.43)	-0.00 (-0.14, 0.14)	-0.02 (-0.11, 0.07)	-0.12 (-0.24 ,0.01)
Breastfeeding and complementary feeding				
Currently being breastfed (Y/N)	0.64 (0.56, 0.73)	0.52 (0.43, 0.61)	-0.04 (-0.11, 0.03)	-0.09 (-0.17, -0.01)
Age when first fed with other food (months)	-0.05 (-0.07, -0.03)	-0.03 (-0.05, -0.01)	0.02 (0.00, 0.04)	0.01 (-0.01, 0.03)
Given foods/liquids other than breastmilk in first 3 days(Y/N)	-0.07 (-0.17, 0.02)	-0.03 (-0.13, 0.07)	-0.03 (-0.11, 0.06)	0.02 (-0.07, 0.11)
Health				
Received vitamin A in last 6 months (Y/N)	-0.23 (-0.30, -0.16)	-0.31 (-0.41, -0.22)	-0.08 (-0.15, -0.02)	0.00 (-0.09 ,0.09)
Some/all children < 5 slept under bednet last night(Y/N)	-0.05 (-0.15, 0.04)	-0.04 (-0.14, 0.07)	0.10 (0.01, 0.19)	0.07 (-0.03, 0.17)
Distance to nearest health facility (km)	-0.02 (-0.03, -0.01)	0.00 (-0.01, 0.01)	0.01 (0.00, 0.02)	0.01 (-0.00, 0.02)
Something done to make water safe to drink (Y/N)	0.26 (0.17, 0.35)	0.16 (0.06, 0.25)	0.11 (0.03, 0.19)	0.08 (-0.01, 0.17)
Source of drinking water				
Borehole, river, lake, spring	-0.29 (-0.38, -0.19)	-	-0.04 (-0.13, 0.04)	-
Open or protected well	-0.12 (-0.20, -0.03)	0.17 (0.06, 0.29)	0.09 (0.01, 0.17)	0.04 (-0.06, 0.15)
Piped, tank, rainwater or bottled	0.34 (0.26, 0.43)	0.15 (0.02, 0.27)	-0.06 (-0.13, 0.02)	0.03 (-0.09, 0.14)
Had diarrhoea recently (Y/N)	-0.05 (-0.11, 0.01)	-0.14 (-0.26, -0.03)	-0.08 (-0.13, -0.03)	-0.13 (-0.24, -0.02)

^ Model adjusted for all other independent variables in the table . N is 3953.

Variable with missing regression results in the adjusted model were excluded due to collinearity.

7.4.3 Agricultural and dietary factors, dietary diversity and nutritional status.

In regression models adjusted for whether the DHS EA was rural or urban, the more meals per day, more days eating meat and days eating fish at the household level, the higher the respondent's and their youngest child's dietary diversity (Table 7.7). For each additional meal per day the child's and respondent's dietary diversity increased by 0.36. For each additional day in the last seven days eating meat and fish the child dietary diversity scores increased by between 0.16 and 0.03 respectively. Children and respondents in households reporting having problems meeting food needs often or always had lower dietary diversity than those from households who never, seldom or sometimes have problems meeting food needs. Children and respondents in households preparing *ugali* with maize flour or using oil for cooking in the last 7 days had higher dietary diversity.

	Dietary diversity score			
	Youngest child	Respondent		
	Regression coefficient	Regression		
	(95% CI)	coefficient		
	(55% Cl)	(95% CI)		
Own land usable for agriculture (y/n)	0.04 (-0.08, 0.17)	0.05 (-0.04, 0.13)		
Amount of agricultural land (Hectares)	0.01 (-0.01, 0.02)	0.01 (-0.00, 0.02)		
Meals per day (N)	0.36 (0.25, 0.46)	0.36 (0.28, 0.43)		
Days eating meat (N)*	0.16 (0.11, 0.21)	0.20 (0.17, 0.24)		
Days eating fish (N)*	0.03 (0.00, 0.05)	0.01 (-0.01, 0.02)		
Problem meeting food needs**	-0.35 (-0.49, -0.21)	-0.23 (-0.32, -0.13)		
Prepared <i>ugali</i> with maize flour (y/n)*	0.33 (0.19, 0.47)	0.45 (0.36 <i>,</i> 0.55)		
Used oil for cooking (y/n)*	0.55 (0.41, 0.68)	0.49 (0.40, 0.58)		

Table 7.7. A	Association between	agricultural and	dietary variables and	dietary diversity in t	۱e
responden	t and their youngest	child			

Models adjusted for living rurally. Nis 3334.

*in the past 7 days

**Problem meeting food needs: never, seldom or sometimes compared to often or always.

In models adjusted for whether the DHS EA was rural or urban, households owning land usable for agriculture had children with significantly lower height for age z-scores and significantly higher weight for height z-scores (Table 7.8). The more hectares of agricultural land a household had the higher the weight for height of the children.

In order to understand these associations additional analyses were conducted to compare wealth indices between those working in agriculture and those not working in agriculture. It

was found that a higher percentage of both respondents and their husbands who worked in agriculture fell into the poorest wealth category (24.9% and 26.0%) compared to those not working in agriculture (6.7% and 6.7%). The difference in wealth indices between agricultural and non-agricultural workers was significant for both the respondents and their husbands (chi squared tests, p<0.001).

Children in households having more meals per day had higher height for age z-scores. The more days in the last week the household ate meat the higher the child's height for age and weight for height. The more days the household ate fish in the last seven days the higher the child's height for weight but the lower their weight for height. Children from households who often or always have problems meeting food needs had lower weight for height z-scores than those from households who never, seldom or sometimes have problems meeting food needs. Children from households who prepared *ugali* with maize flour had higher weight for height and those from households using oil for cooking in the last seven days had higher height for age and lower weight for height.

	Height for age z-scores	Weight for height z- scores
	Regression coefficient (95% CI)	Regression coefficient (95% CI)
Own land usable for agriculture (y/n)	-0.18 (-0.28, -0.08)	0.21 (0.12, 0.30)
Amount of agricultural land (Hectares)	-0.00 (-0.01, 0.01)	0.02 (0.01, 0.03)
Distance to nearest market (km)	-0.00 (-0.00, -0.00)	0.00 (0.00, 0.00)
Meals per day (N)	0.14 (0.06, 0.22)	0.04 (-0.03, 0.12)
Days eating meat (N)*	0.04 (0.01, 0.08)	0.03 (0.00, 0.06)
Days eating fish (N)*	0.04 (0.02, 0.05)	-0.03 (-0.05, -0.01)
Problem meeting food needs**	-0.03 (-0.13, 0.07)	-0.10 (-0.19, -0.00)
Prepared <i>ugali</i> with maize flour(y/n)*	0.02 (-0.09, 0.12)	0.22 (0.12, 0.31)
Used oil for cooking(y/n)*	0.23 (0.14, 0.32)	-0.36 (-0.44, -0.28)

 Table 7.8. Association between agricultural and dietary variables and nutritional status

 variables in the youngest child

Models adjusted for living rurally. Nis 3999.

*in the past 7 days

**Problem meeting food needs: never, seldom or sometimes compared to often or always.

7.4.4 Dietary diversity and nutritional status

Dietary diversity was significantly negatively associated with both height for age and weight for height z-scores in the youngest children in the unadjusted models (Table 7.9). When models were adjusted for: respondent's education in single years; frequency of watching television; currently breastfed; age when first fed with other food; wealth index; respondent employed all year or seasonally; respondent works for family, other, self; the association between dietary diversity and height for age remained significantly negative (-0.11 (-0.15, -0.07). For each additional category of dietary diversity consumed the previous day children's height for age decreased by 0.11 z-scores.

When the individual components of the dietary diversity score (whether the child ate cereals, vitamin A rich fruit or vegetables, other fruit or vegetables, meat/fish, eggs, legumes, milk or oil/sweets the day before the interview) were added all together to the adjusted model instead of the composite score, whether the child ate legumes, meat, vitamin A rich fruit or vegetables or cereals were negatively associated with height for age. Oil and sweets were positively associated with height for age. Legumes were positively associated with weight for height.

	Height for age z-scores	Weight for height z-scores
	Regression coefficient (95% CI)	Regression coefficient (95% CI)
Dietary diversity scores		
Unadjusted [^]	-0.15 (-0.18, -0.11)	-0.04 (-0.07, -0.01)
Adjusted^^	-0.11 (-0.15, -0.07)	0.02 (-0.02, 0.06)
Dietary diversity score components	(adjusted only) ^^^	
cereals	-0.38 (-0.63, -0.14)	-0.02 (-0.27, 0.23)
vitamin A rich fruit or vegetables	-0.27 (-0.40, -0.15)	0.02 (-0.10, 0.15)
other fruit or vegetables	-0.05 (-0.21, 0.12)	-0.10 (-0.26, 0.07)
meat/fish	-0.19 (-0.32, -0.07)	0.07 (-0.06, 0.20)
eggs	0.11 (-0.14, 0.36)	-0.06 (-0.31, 0.19)
legumes	-0.22 (-0.35, -0.09)	0.14 (0.01, 0.27)
milk	0.10 (-0.04, 0.23)	0.06 (-0.07, 0.20)
oil/sweets	0.16 (0.02, 0.29)	-0.04 (-0.17, 0.10)

Fable 7.9. Association between dietary diversity scores and its components and nutritional
status variables in the youngest children

^Unadjusted model N is 3335.

^^Adjusted for respondent's education in single years, frequency of watching television, currently breastfed, age when first fed with other food, wealth index, respondent employed all year or seasonally, respondent works for family, other, self and whether they live rurally. Nis 2698.

^^Adjusted for all the other dietary diversity score components, respondent's education in single years, frequency of watching television, currently breastfed, age when first fed with other food, wealth index, respondent employed all year or seasonally, respondent works for family, other, self and whether they live rurally. Nis 2698.



Figure 7.4. Clusters of high and low dietary diversity in the youngest child and high dietary diversity in the mother

7.4.5 Spatial variation of dietary diversity and nutritional status

Figures 7.4 to 7.8 show the significant clusters of high and low dietary diversity, height for age, weight for height in the youngest child and dietary diversity and BMI in the mother (there was no significant cluster of low dietary diversity for the mother). There is a large cluster of high child dietary diversity in the North West half of the country. This cluster includes DHS EA near lake Victoria and other lakes in the North and stretches down to Singida region. This corresponds with a cluster of high dietary diversity in the respondent. Low dietary diversity is seen in the east of Tanzania including the capital, Dar es Salaam and Zanzibar.



Figure 7.5. Clusters of high and low height for age in the youngest child

The area of low dietary diversity (the east of Tanzania including the capital, Dar es Salaam and Zanzibar) is where a cluster of high height for age and low weight for age in the youngest children is seen as well as a cluster of high BMI in the respondents. Interestingly the cluster of low respondent BMI corresponds with the cluster of high dietary diversity in the respondents. These cluster analyses have shown Dar es Salaam, Zanzibar and Pemba to have low child dietary diversity, low weight for height and high height for age and high respondent BMI.



Figure 7.6. Clusters of high and low weight for height in the youngest child



Figure 7.7. Clusters of high and low BMI in the mother

7.4.6 Land cover, dietary diversity and nutritional status.

Agricultural, but not forest, land cover is associated with higher dietary diversity. The land cover surrounding the DHS EA of high child dietary diversity was approximately 50% agricultural while the land cover surrounding the cluster of low dietary diversity was only 30% agricultural (Table 7.10). The high dietary diversity clusters were surrounded with only 3% forest land cover compared to approximately 25% around the low dietary diversity clusters. The land cover surrounding the cluster of high respondent dietary diversity was very similar to the land cover seen around the high child dietary diversity cluster.
More agricultural and forest land cover is associated with lower height. The cluster of high child height for age was surrounded by approximately 20% agricultural land cover compared to 40% around the low height for age cluster. The high height for age cluster was surrounded by only 16% forest and 63% other land cover types. The low height for age cluster was surrounded by 32% forest and only 28% other land cover.

More agricultural land cover was associated with better weight for height. High children's weight for height clusters had approximately 50% agriculture, 18% forest and 30% other land cover in the 5km surrounding the DHS EA centre point and low weight for height clusters had 23% agriculture, 16% forest and 61% other land cover in their 5km buffer zone.

More agricultural land cover was associated with poorer respondent BMI. The majority of the land surrounding the high respondent BMI clusters was classed as other (70%), 21% was agricultural and only 9% was forest. Forty-nine percent of the land surrounding the low BMI cluster was classed as other, 48% as agricultural and 3% as forest land cover.

	Agriculture %^	Forest %^^	Other %^^^
	Agriculture //	10103070	
Children's dietary diversity			
High cluster (N=167)	51.2	3.0	45.8
Low cluster (N=224)	28.5	24.4	47.1
Children's height for age			
High cluster (N=115)	21.3	15.5	63.2
Low cluster (N=132)	39.9	31.7	28.4
Children's weight for height			
High cluster (N=80)	52.3	17.5	30.2
Low cluster (N=139)	23.3	16.2	60.5
Respondent's dietary diversity			
High cluster (N=187)	52.7	2.6	44.7
Respondent's BMI			
High cluster (N=62)	21.3	9.3	69.5
Low cluster (N=153)	48.4	2.5	49.0

 Table 7.10. Percentage of land cover falling into agriculture, forest and other land cover categories in the 5km surrounding the significantly high and low DHS EA centre points

Reported N is number of DHS EA in each Satscan cluster

^Rainfed croplands; Mosaic Cropland (50-70%) / Vegetation (grassland, shrubland, forest) (20-50%); Mosaic Vegetation (grassland, shrubland, forest) (50-70%) / Cropland (20-50%).

^^Closed to open (>15%) broadleaved evergreen and/or semi-deciduous forest (>5m); Closed (>40%) broadleaved deciduous forest (>5m); Open (15-40%) needleleaved deciduous or evergreen forest (>5m); Mosaic Forest/Shrubland (50-70%) / Grassland (20-50%); Mosaic Grassland (50-70%) / Forest/Shrubland (20-50%); Closed (>40%) broadleaved forest regularly flooded - Fresh water; Closed to open (>15%) vegetation (grassland, shrubland, woody vegetation) on regularly flooded or waterlogged soil - Fresh, brackish or saline water.

^^^ Open (15-40%) broadleaved deciduous forest (>5m); Closed to open (>15%) mixed broadleaved and needleleaved forest (>5m); Closed to open (>15%) shrubland (<5m); Closed to open (>15%) grassland; Sparse (>15%) vegetation (woody vegetation, shrubs, grassland); Artificial surfaces and associated areas (urban areas >50%); Bare areas; Water bodies.

The percentage of agricultural land cover in the 5 km surrounding the DHS centre point is positively associated with dietary diversity in the youngest child and the respondent; the more agricultural land cover the higher the dietary diversity (Table 7.11). There was no association between the amount of forest land cover in the surrounding area and dietary diversity. The higher the percentage of other land cover types, such as sparse vegetation, grassland and bare areas, the lower the dietary diversity in both the youngest child and the respondent.

The more agricultural land cover, the lower the youngest child's height for age and the higher the proportion of other land cover the higher the children's height. The higher the proportion of both agricultural and forest land cover (and the lower the other land cover) the higher the children's weight for height z-scores. In the respondent, having a higher proportion of agricultural land cover in the 5km surrounding the DHS EA centre points was associated with having a lower BMI and a higher proportion of forest land cover was associated with a higher BMI. The percentage of other land cover was positively associated with respondent BMI in the unadjusted model only, when the model was adjusted for whether the DHS EA was classed as rural or as urban, this significant association disappeared.

Table 7.11. Association between dieta	ry diversity and nutritional status and pe	ercentage land cover of agriculture, for	est or other land cover types in the 5km
surrounding the DHS centre poin	ts		

	Youngest child			Respondent			
	Dietary diversity	Height for age z- scores	Weight for height z- scores	Dietary diversity	BMI		
	Regression coefficient (95% CI)						
Percentage agricultural land cover (5km buffer)^							
Unadjusted	0.04 (0.02, 0.05)	-0.05 (-0.08, -0.03)	0.04 (0.01, 0.06)	0.04 (0.02, 0.06)	-0.20 (-0.28, -0.13)		
Adjusted*	0.03 (0.02, 0.05)	-0.04 (-0.06, -0.01)	0.04 (0.02, 0.06)	0.03 (0.02, 0.05)	-0.11 (-0.18, -0.05)		
Percentage forest land cover (5km buffer)^^							
Unadjusted	-0.01 (-0.03, 0.01)	-0.01 (-0.04, 0.02)	0.04 (0.02, 0.06)	-0.01 (-0.03 <i>,</i> 0.01)	0.08 (0.01, 0.15)		
Adjusted*	-0.01 (-0.03, 0.00)	0.00 (-0.02, 0.03)	0.04 (0.02, 0.06)	-0.01 (-0.03, 0.01)	0.12 (0.06, 0.18)		
Percentage other land cover (5km buffer)^^^							
Unadjusted	-0.02 (-0.04, -0.01)	0.07 (0.05, 0.09)	-0.07 (-0.08, -0.05)	-0.03 (-0.04, -0.01)	0.15 (0.09, 0.20)		
Adjusted*	-0.02 (-0.03, -0.01)	0.04 (0.02, 0.06)	-0.07 (-0.09, -0.05)	-0.02 (-0.03, -0.01)	0.02 (-0.02, 0.06)		

N for unadjusted and adjusted models are: 837 for agricultural land cover, 1084 for forest land cover and 1341 for other land cover.

*Adjusted for whether the cluster was urban or rural.

^Rainfed croplands; Mosaic Cropland (50-70%) / Vegetation (grassland, shrubland, forest) (20-50%); Mosaic Vegetation (grassland, shrubland, forest) (50-70%) / Cropland (20-50%).

^^Closed to open (>15%) broadleaved evergreen and/or semi-deciduous forest(>5m); Closed (>40%) broadleaved deciduous forest(>5m); Open (15-40%) needleleaved deciduous or evergreen forest (>5m); Mosaic Forest/Shrubland (50-70%) / Grassland (20-50%); Mosaic Grassland (50-70%) / Forest/Shrubland (20-50%); Closed (>40%) broadleaved forest regularly flooded - Fresh water; Closed to open (>15%) vegetation (grassland, shrubland, woody vegetation) on regularly flooded or waterlogged soil - Fresh, brackish or saline water.

^^^ Open (15-40%) broadleaved deciduous forest (>5m); Closed to open (>15%) mixed broadleaved and needleleaved forest (>5m); Closed to open (>15%) shrubland (<5m); Closed to open (>15%) grassland; Sparse (>15%) vegetation (woody vegetation, shrubs, grassland); Artificial surfaces and asso clated areas (urban areas >50%); Bare areas; Water bodies.

7.5 Discussion

7.5.1 Demographic and social factors, dietary diversity and nutritional status

Wealth, education and occupation all affected the diversity of the diets of mother and child as would be expected from the literature (Fouere et al., 2000, Torheim et al., 2004, Nyangweso et al., 2007). Children who were breastfed for longer, a WHO recommended practice (WHO, 2003), had higher dietary diversity. Children who were older when they were first given complementary foods or liquids had higher dietary diversity. As complementary foods were introduced at four months, on average instead of the recommended six months (WHO, 2003) these practices may reflect a better understanding of nutrition guidelines for both complementary feeding and improving nutrition through dietary diversity.

The respondent's and their husband's employment and the households wealth index were associated with children's nutritional status. A number of factors associated with infection were also significantly related with height for age and weight for height. Children who had access to water from a tank or tap as opposed to water from a borehole or river and children from households who treated their drinking water all had better markers of nutritional status. This shows that those children who had a decreased exposure to infection had better nutritional status. This finding is supported by the finding that children who had diarrhoea recently had poorer height for age and weight for height than those not experiencing diarrhoea recently. These findings are supported by other findings in the literature regarding the association between infection and nutritional status (Chen et al., 1980, Stephensen, 1999).

Children who were still being breastfed had better nutritional status than those who either had never been breastfed or had stopped being breastfed. Interestingly the older the child was when complementary foods were given the lower the height for age of the child. The majority of the children had complementary foods introduced early (mean (95% CI: 3.78 months (3.73, 3.84)). The highest height for age was seen in the group of infants that were first given complementary foods in the first three months of life, despite the WHO recommendations to exclusively breastfeed until six months of age (WHO, 2002). Whether the child was given foods or liquids other than breastmilk in the first three days of life was not associated with height for age or weight for height. The results presented in chapter 4 showed that the age liquids and solids were introduced as well as the types of foods children are weaned onto were associated with nutritional status. The data available in the DHS does not allow for these relationships to be interrogated. - 236 -

7.5.2 Agricultural and dietary factors, dietary diversity and nutritional status.

Having agricultural land was associated with poorer long term nutritional status and improved short term nutritional status in regression models adjusted for living rurally; Individuals in households with land that is deemed usable for agriculture have poorer height for age zscores. However, these children, and children in households with more agricultural land had higher weight for height. Lack of land ownership is used as a marker of poverty (Murthy et al., 2008) and land ownership is expected to be associated with better nutritional status (Swaminathan et al., 2012). This finding may be due to energy expended through farming and time away from childcare (Kumar and Hotchkiss, 1988) as discussed in chapter 5. It may also be linked to income as those working outside of agriculture had significantly higher wealth indices than those working in agriculture.

The more meals eaten per day and the more days eating meat and fish the higher the dietary diversity and the higher the children's height for age. Although this finding was not replicated in the primary data collection there is evidence of the association between animal product consumption and nutritional status in low income countries (Allen et al., 1992, Marquis et al., 1997, Krebs et al., 2011). The more days eating meat the higher the weight for height in the children while the more days the household ate fish the lower their weight for height. The consumption of small fish in small quantities (Thilsted et al., 1997) is common in Eastern Africa (Kawarazuka and Béné, 2011) and it may be that fish is not consumed in great enough quantities to impact on weight, which is more change able and strongly influenced by infection. The consumption of animal products in this population appears to be positively associated with nutritional status in the long term but has mixed association with nutritional status in the short term.

In households using oil for cooking, individuals had both higher dietary diversity and better nutritional status. This may be a reflection of additional calories from the oil, that cooking with oil allows the household to cook more nutritious foods such as green leafy vegetables or it could simply be a reflection of the socio-economic status of the household. A randomised controlled trial on stunted children (3-9 years old) in Gambia found no effect on height or weight of an oil supplement given in the form of a biscuit (1600KJ) for 12 months (Krähenbühl et al., 1998). Bajaj et al (2005) found that an oil supplement displaced breastmilk in the diet of 6-10 month old infants in India, decreasing overall energy intake. Although oil intake is often included in dietary diversity scores the effect of oil intake on nutritional status, outside of complementary feeding, has not been reported. Further research is needed to assess the impact of habitual oil intake on nutritional status in low income countries.

7.5.3 Dietary diversity and nutritional status

Dietary diversity was negatively associated with height for age, the opposite association than would be expected based on the literature (Arimond, 2004, Steyn et al., 2006) and shown in the project's conceptual framework (figure 3.2 a) and b)). This provide extra evidence that the association between dietary diversity and nutritional status is not as strong in Eastern Africa as in other low income countries. The specific foods that were negatively associated with height were cereals, vitamin A rich fruit or vegetables, legumes and meat, the latter three groups are considered to be essential additions to the staple diet in a low income country context. Although this regression model was adjusted for potential confounders captured by the DHS survey it is likely that there are a number of other factors influencing the relationship between dietary diversity and height in the survey that have not been captured, for example total energy intake, ethnicity (Habicht et al., 1974) and soil fertility (Byiringiro and Reardon, 1996). As was seen in the data collected in Minyenye and Mbwei, the results from the DHS data showed no significant relationship between the dietary diversity in the youngest children and their weight for height.

When the multiple determinants of nutritional status are acknowledged and other studies that have controlled for potential confounders are singled out, mixed results have been shown in the literature. Some have found no significant association between dietary diversity and weight for height z-scores in all populations studied (Hillbruner and Egan, 2008) or in some sub-groups of the studied population. Higher dietary diversity was associated with lower rates of stunting in urban but not rural populations (Hatloy et al., 2000), with height for age in boys but not girls (Eckhardt et al., 2005) and height for age in 11-23 month olds but there was no association seen for those 24-35 months or in any age group in weight for height (Sawadogo et al., 2006). However, Arimond (2004) found dietary diversity to be significantly positively associated with height for age when potential confounders were controlled for in seven out of the eleven countries studied through the DHS surveys (not including Tanzania). Positive associations between dietary diversity and height for age and weight for height was found in Indian infants aged 9-12 months (Garg and Chadha, 2009). Rates of stunting were associated with dietary diversity in Moroccan children (Aboussaleh and Ahami, 2009).

These mixed results in the literature as well as the results presented both in chapter 4 and the current chapter suggest that the relationship between dietary diversity and nutritional status in children in low income countries varies from context to context and should not be assumed. This is not to say that dietary diversity does not benefit health and is not important in and of itself but it does not appear appropriate for dietary diversity to be used as a marker of anthropometric nutritional status.

7.5.4 Spatial variation of dietary diversity and nutritional status

The Satscan, spatial analysis results did not find small pockets of dietary diversity or nutritional status variables which varied significantly from these variables in other DHS EA. The clusters were generally quite large and generally indicated that dietary diversity and nutritional status differs from West to East or from inland to coastal areas. There are a number of possible reasons for this variation. Dietary diversity was higher in the West or inland half of Tanzania. The clusters of low dietary diversity, high height for age and low weight for height in children were centred around Dar es Salaam and Zanzibar. Although it would be reasonable to expect the results to be influenced by the large population living in Dar es Salaam, this analysis did not show any significant effect of living in an urban environment. The results may also reflect different access to fish along the coastal areas or different cultural practices.

The inland cluster of high child dietary diversity and weight for height but low height for age compared to the coastal cluster of low dietary diversity and weight for height and high height for age generally corresponds to different agro-ecological zones (FAO, 2014b). There is limited evidence in the literature of the relationship between agro-ecological zones and diet and nutrition outcomes but it is well established that soil fertility varies between the different agro-ecological zones (Smaling and Braun, 1996) and that soil fertility is linked to food production (Arshad and Martin, 2002). This difference in soil fertility as well as the difference in other climatic factors between the agro-ecological zones may explain part of the difference seen in dietary diversity and nutritional status across Tanzania.

These analyses used all of the DHS data, both urban and rural, while the primary data collection focused on rural areas only. The inclusion of the urban data at a national level broadened the relevance of the work to cover Tanzania as a whole. Regression analyses were controlled for whether the area was rural or urban but if this analysis was focused on the rural areas only it is possible that different results would have been seen. For example, dietary diversity and nutritional status may have been positively associated as a more diverse diet in a rural setting is more likely to be beneficial while a more diverse diet in an urban setting may mean diets that are not necessarily associated with better health such as those that are processed or high in sugar (Popkin, 1998). Also, land cover may have shown different associations with dietary diversity and nutritional status as people relate to the land differently in rural compared to urban environments.

7.5.5 Land cover, dietary diversity and nutritional status

The results of these analyses show that agricultural land cover is positively associated with average dietary diversity. The more cropland land cover in the surrounding area, the higher the dietary diversity. When cropland is viewed as a proxy for agrobiodiversity this results provides support for the pathway from agrobiodiversity to dietary diversity illustrated in the conceptual framework (figure 3.2 a) and b)). These categories do not differentiate between subsistence farming which you would expect to be associated with greater agricultural diversity and large mono-crop farms. This result may reflect higher dietary diversity in subsistence farming areas or in areas with large cash crop farming. In either case, without being able to address the food security pillars of access or utilization, these results indicate a higher diversity in peoples diets with greater food availability.

This association did not, however, consistently carry through to the nutritional status variables. More agricultural land cover was associated with poorer average height for age in the child and lower respondent BMI. It was also associated with higher weight for height in the child. The fact that this agricultural land cover is positively associated with the short term measures of dietary diversity and weight for height at a group level but not with the long term nutrition measures of height for age is interesting. This mirrors the association that those households with agricultural land had better short term but poorer long term nutritional status. The reasons for this may be connected to the energy expenditure associated with farming impacting on the chronic measures of nutritional status as well as the time spent farming impacting on time for care of children, as discussed earlier (Kumar and Hotchkiss, 1988).

The results of these analyses show that more biodiverse land cover categorised as 'forest' land cover is not associated with average dietary diversity. However, more forest land cover was associated with a higher weight for height z-score in the youngest child and higher respondent BMI. In contrast to the dietary diversity results of this study, a similar study by Johnson et al (Johnson et al., 2013) found that communities experiencing loss in forest cover over the previous ten years had lower dietary diversity than children living in areas with no change in forest cover. Also, higher overall percentages of forest cover were associated with the consumption of more vitamin A rich foods and lower rates of diarrhoea. A study measuring tree cover and dietary diversity in the East-Usambara mountains in Tanzania found households with greater nearby tree cover were more likely to consume forest foods which in turn was linked to higher dietary diversity (Powell et al., 2011).

The current study found no association between forest cover and dietary diversity but did find a positive association between amount of forest cover and average weight for height in the children and BMI in respondents. This may reflect better access to nutritious foods available in

the forest (Johns and Maundu, 2006), as was shown in Johnson et al (2013) and Powell et al (2011). However, neither this study nor Johnson et al (2013) found positive associations between forest cover and height for age.

7.5.6 Limitations and strengths

The findings from this chapter should be interpreted in the context of the limitations of the analyses. All analyses are based on the demographic and health survey data which will be subject to all the biases associated with survey data. For example, questions on complementary feeding of children may be biased as they rely on memory of past behaviour and the honesty of the responses given may be effected by the presence of the research team. Additionally, it is a cross-sectional survey so the direction of causality between the variables found to be significantly associated is not know, it can only be said that the variables are associated. This is a large study which uses a number of different research assistants to collect data, inconsistencies between data collected in different areas in Tanzania are possible.

For the DHS data used in the spatial and land cover analyses the data was aggregated at the DHS EA level and attached to the DHS EA centre point. DHS EA vary in geographic size as they are based on population rather than distance. The 5km buffer zones around the DHS EA centre points used in the land cover analysis may be less appropriate for some of the geographically larger clusters than the smaller clusters. As the borders of the DHS EA areas were not available there was no way to get around this limitation. The land cover data itself also has its own limitations. The land cover map was validated using land cover experts from around the world. The validation report states the accuracy of the map to be 67.5% and states that globally there was 5% missing data of inland surfaces that were filled in using the 2005 map (Bontemps et al., 2011).

Individual data has been summarised at the DHS cluster level for comparison with land cover data. Interpretations about the implications of these associations at a DHS cluster level for individuals must be made with caution (Morgenstern, 1995). This ecological study uses groups, the DHS clusters, as the unit of analysis and it is the group exposure and how this relates to the groups average outcome that is examined. Individuals within DHS clusters with high levels of surrounding agricultural or forest land cover may not, themselves be exposed to factors associated with these land covers. Using these group associations to make conclusions about these associations at the individual level would be misleading, referred to as committing the ecological fallacy.

- 240 -

The DHS is a large nationally representative sample in which data collection has been developed and tested over many years to ensure the data collected is as accurate as possible. It collects very detailed data ideal for analyses investigating nutritional status. This data has been spatially linked to land cover data that has been pre-processed and classified by specialists in their fields and validated by local experts.

7.6 Conclusions

A number of factors such as wealth, education, occupation, breastfeeding and the timing of the introduction of complementary foods were associated with dietary diversity in this large, nationally representative survey. Wealth, occupation, currently being breastfed, timing of complementary foods, source of drinking water, taking action to make water safe to drink and having recently experienced diarrhoea were associated with height for age in children. Contrary to previous findings, dietary diversity itself was not significantly associated with nutritional status.

The results from this chapter indicate that a higher intake of animal products corresponds to a higher dietary diversity and a better nutritional status in children. Other factors such as having more meals and oil being used in cooking were also positively associated with nutritional status.

Households with agricultural land had better markers of short term nutritional status and worse markers of long term nutritional status. This raises some interesting questions about some of the potential consequences of farming such as increased energy expenditure and impact on caregivers time which warrant further investigation.

Both dietary diversity and nutritional status vary spatially and are associated with the surrounding land cover, especially the amount of agricultural land cover. This has implications for the distribution of intervention and support across the country. Perhaps communities in areas with low amounts of agricultural land in the vicinity would require more interventions focusing on dietary diversification while communities in areas with high amounts of agricultural land support to counteract their higher rates of stunting. Proximity to biodiverse forested areas was not associated with dietary diversity but was associated with acute measures of malnutrition. This provides some evidence in support of the benefits of access to forests in subsistence farming environments.

Chapter 8: Overall discussion

- 242 -

8.1 Chapter summary

Detailed results from the primary data collection were presented in chapters 4 to 6 and results using national data from Tanzania were presented in chapter 7. This chapter aims to pull together the findings from these two different approaches. The main lessons learnt will be synthesised and discussed to arrive at the final conclusions reached by the study, which will be presented in chapter 9. This chapter will also discuss the limitations and strengths of the two approaches, some methodological insights for future research, the contribution of this research and suggestions for future research. The combination of detailed data collection from data collection in the two villages and nationally representative data complements each other and provides interesting insights in to the determinants of nutritional status and the relationship between agrobiodiversity and dietary diversity and nutritional status.

8.2 Key messages

- This project is one of only two projects investigating the relationship between agrobiodiversity, dietary diversity and nutritional status, the last being conducted in Mexico and published in 1981 (Dewey).
- Dietary diversity was not associated with nutritional status at local village scales and was negatively associated with height for age in the national DHS analysis.
- Agrobiodiversity and land cover showed mixed associations with nutritional status.
- The results of the primary data collection generally show positive associations between eating and rearing animals and dietary diversity but negative or no associations with nutritional status.
- Selling crops was positively associated with dietary diversity but showed a mixed association with nutritional status at the local village scale.

- The results suggest that the effectiveness of agricultural interventions aimed at improving nutrition through improvements in agrobio diversity can only be evaluated in light of the multiple determinants of nutritional status.
- Standard definitions and methods for measuring agrobiodiversity would benefit future research in this area.

8.3 Dietary diversity and nutritional status

The results of the primary data collection points to the complexity of the determinants of nutritional status in sub-Saharan Africa. It has been well established in the literature that dietary diversity and nutritional status are related (Arimond, 2004, Marriott et al., 2012, Rah et al., 2010). It is biologically plausible that a diversity of different foods including foods high in energy, protein and different micronutrients would contribute to the health and growth of children (Daniels et al., 2007, Onyango, 2003). This combined with the wealth of literature in this area means that it was reasonable to hypothesise that dietary diversity and nutritional status would be related in this study.

Neither dietary diversity nor food variety were significantly associated with nutritional status in Minyenye or Mbwei. Additionally, dietary diversity and height for age was negatively associated in the national DHS analysis. It seems likely that this is due to the multitude of other factors that impact on nutritionals status. Education, occupation and wealth were associated with dietary diversity and nutritional status in the DHS analysis. As discussed, one of the main determinants of nutritional status is infection ((Chen et al., 1980, Stephensen, 1999). Although rates of infection were not measured in the primary data collection there were a number of risk factors for infection present in both villages. These include poor access to clean water, poor sanitation, early introduction of potentially unsafe complementary foods, unprocessed milk consumption, no refrigeration for meat, free roaming animals within the house and yard, and no facilities for hand washing after using the toilet. The DHS analysis showed a number of factors associated with nutritional status that may reflect infection, specifically access to tap water and treating drinking water. It found that children with recent diarrhoea had poorer nutritional status. Data at both the local scale and the national scale points to infection being a cause of malnutrition in Tanzania.

It is also reasonable to hypothesise that the age complementary foods are introduced to an infant's diet as well as the type of foods that are introduced have an effect on their growth. Firstly for the nutrients these foods provide to the infant and secondly for the potential exposure to infection. The results of the primary data collection support this theory. Children

- 243 -

who had already had liquids and/or solids introduced to their diets had poorer nutritional status in Minyenye and Mbwei. The type of foods first added to the diet also had an impact; children who were given multiple flour porridges with additions such as beans, oil or fish or millet juice had poorer nutritional status. These additions may act as vehicles for exposure to infection (Motarjemi et al., 1993). In the DHS results, children who were younger when complementary feeding began had higher height for age suggesting earlier introduction of complementary feeding is beneficial. At a national level the potential negative impact of introduced infection appears to be outweighed by the extra nutrients provided to the diet by these complementary foods.

There were other dietary factors identified by this study that may have impacted on the nutritional status of the children in the villages. There was quite a significant difference in the intake of vitamin A rich fruit and vegetables between the two villages. Ninety-five percent of participants consumed vitamin A rich fruit or vegetables in the day before the interview in Minyenye while only 53% of those did in Mbwei. In the majority of households in Minyenye this was a wild plant, *Mlenda*, which was readily available and also dried and used throughout the year. This may have had a positive effect on the micronutrient intake and potentially health of individuals in this village as compared to Mbwei where consumption of *Mlenda* was not common.

Mlenda highlights an important aspect of diet in Tanzania; the use of wild foods. These results showed that wild food contributed to dietary diversity in Minyenye year round, shifting the average dietary diversity score in the period of lowest diversity from approximately 1 to 1.5. The fact that wild foods consumed in this area tend to be green leafy vegetables high in micronutrients means that this could have a significant impact on the health of those in this area, filling an important gap at times of food shortage.

Another marked difference in consumption between the two villages was cassava. As discussed in chapter 4 cassava acts as an important food source in times of food scarcity. It is not as nutritious as other staples (Stephenson et al., 2010) but it is resilient, can grow in poor conditions and can be stored in the ground (El-Sharkawy, 2007). This means it can be used as an important tool against crop failure and food insecurity (Prudencio and Al-Hassan, 1994). It can be grown and left until times of food shortage when it can bridge the shortfall in energy intake.

Dietary diversity is likely to have many health benefits in low income countries, such as increased intake of vitamins. However, more diverse diets do not necessarily lead to better growth, the main outcome for this project. Diversifying diets in low income countries has the potential to improve markers of health such as risk of anaemia (Gibson et al., 2003).

Diversifying crops grown may have additional benefits to food security through resilience to drought and crop failure and improved distribution of food throughout the year. The results of this project combined with the findings of other projects (Ekesa et al., 2011, Hatloy et al., 2000, Eckhardt et al., 2005) illustrate that the relationship between dietary diversity and nutritional status vary depending on the area and characteristics of the population of interest. It cannot be assumed that increasing the diversity of the diet will improve growth or nutrition outcomes in all settings.

A high proportion of children in Minyenye and Mbwei (53%) as well as nation ally (36%) are stunted. There were a number of factors that are linked to nutritional status occurring in the two villages and nationally. Other, more significant determinants of nutritional status, such as infection, may have masked any relationship between dietary diversity and nutritional status that was present. Additionally, the variation in dietary diversity in these two villages was very low (mean (95%CI): 4.8 (4.6, 5.0)). Dietary diversity and nutritional status were measured at the same timepoint in both these studies. The combination of the low variety in dietary diversity and the cross sectional nature of the studies may have prevented any association between dietary diversity and nutritional status being seen in these studies in Tanzania.

8.4 Plant agrobiodiversity, dietary diversity and nutritional status

This study aimed to investigate the relationship between agrobiodiversity and dietary diversity and nutrition in these two villages in rural Tanzania and it addresses the main pathway from agrobiodiversity to dietary diversity to nutritional status in the conceptual framework (Figure 3.2a) and b)). The results suggest that there either is not an association between agrobiodiversity and dietary diversity or nutritional status or these associations are being masked by the other determinants of dietary diversity and nutritional status. The results of this study illuminated the complexity of the pathway from foods grown in the household farms though diet to the children's nutritional status.

As there were no significant associations between dietary diversity and nutritional status it is not that surprising that there were no significant associations between plant agrobiodiversity and nutritional status. However, it was surprising that there were significant negative associations between plant agrobiodiversity and nutritional status. Crop and vegetable diversity indices, which showed the diversity of crops planted over the previous year, were significantly positively associated with dietary diversity and food variety in both Minyenye and Mbwei. However, as with the plant agrobiodiversity indices these were negatively associated

with nutritional status in Minyenye. Additionally, the DHS and land cover analyses found that more agricultural land cover in the surrounding area was associated with higher dietary diversity and weight but poorer height in children. Higher proportions of high diversity forest land cover was not associated with dietary diversity but was associated with higher weight in the youngest child.

It is also possible that there are no associations between plant agrobiodiversity or high diversity land cover and dietary diversity or nutritional status due to their multiple determinants, that the variety of crops grown or the diversity of the surrounding land cover is too distant to the growth of children on a hypothetical causal pathway. There are many steps, decisions and biological processes between what is grown in a hou sehold farm and the growth of a child. Figure 8.1 takes the study's conceptual framework (figure 3.2a) and b)) and adds the decisions and agricultural, biological and socio-demographic factors that may confound the relationships between agrobiodiversity, dietary diversity and nutritional status. The figure firstly acknowledges the political, social, economic, care and health factors that are operating in this environment that are likely to be effecting agrobiodiversity, dietary diversity and nutritional status and potentially confounding the relationships of interest. Measuring the majority of these factors was beyond the scope of this project.

The figure illustrates factors that could potentially be interfering with the relationship between agrobiodiversity and dietary diversity. The cross sectional measurement of plant agrobiodiversity does not take into account seasonal variation in agrobiodiversity and crop failure, both of which would impact on the food available in the household farms. Once the crops leave the farm they could either go to the household for consumption by the family or they may be sold. The effect of selling crops was explored by the primary data collection but is not captured within the measure of plant agrobiodiversity. Finally, if the crops are consumed within the household, how these are distributed would also have an impact on the mother and young children's dietary diversity (Gittelsohn and Vastine, 2003). These factors may help account for the lack of association seen between plant agrobiodiversity indices and dietary diversity in the primary data collection.

The figure also illustrates factors that were not captured in this study that may have affected the relationship between dietary diversity and child nutritional status. The amount of food available from the household farms is an important factor that was not measured by this study. Annual variation in dietary diversity is another factor that needs to be considered and has been discussed throughout this thesis. Any anti-nutrients and illness that would reduce the absorption of micronutrients from the diet would limit the effectiveness of eating a diversified

diet (Molla et al., 1982, Stephensen, 1999, Gibson et al., 2006). All these factors have the potential to impair higher dietary diversity impacting positively on nutritional status.

Figure 8.1 also shows the potential pathway from agrobiodiversity to dietary diversity through household income, which is supported by the results of the primary data collection which showed that selling crops was positively associate with dietary diversity. Increased income from selling crops did not, however, translate into improved nutritional status in these households. Finally, this framework illustrates the specific socio-demographic factors measured in the primary data collection that were found to be significantly associated with child nutritional status. This framework illustrates the complexity of the pathway from foods grown by the household through foods eaten by the household to health outcomes in the household members and helps explain the lack of associations seen between plant agrobiodiversity, dietary diversity and nutritional status in the primary data collection.

As outlined in the literature review (Table 2.3 and 2.4) six other cross sectional studies investigated the relationships between agrobiodiversity and dietary diversity and two the relationships between agrobiodiversity and nutritional status. The two studies using a systematic measure of agrobiodiversity and reporting statistical significance of the associations in household farms or vegetable gardens found no significant association between agrobiodiversity and dietary diversity (Remans et al., 2011a, Akrofi et al., 2010). The one study using a crop diversity score as reported by a household member and reporting statistical significance found a positive association with dietary diversity (Bhagowalia et al., 2012). In the two studies investigating the relationship between agrobiodiversity or crop diversity scores and nutritional status, one found no association (Shack et al., 1990) and the other found a significant association in one of the villages but not the other (Dewey, 1981). As a whole, these results are very similar to those found in the current study; mixed associations with no strong message. It is possible that the study design of these previous studies, as well as the current study has limited the ability of the studies to detect these associations. In all the studies the exposure and outcomes are measured at the same time and agrobiodiversity is measured at only one time point, missing out on any seasonal effects. Future studies with stronger study designs such as longitudinal studies or randomised controlled trials may reveal a clearer picture of these associations.

- 247 -



- 248 -

Figure 8.1. Conceptual framework of factors impacting on the relationship between agrobiodiversity, dietary diversity and nutritional status

Many interventions attempting to increase agrobiodiversity in order to have an effect on dietary diversity focus on creating or improving vegetable gardens (Cabalda et al., 2011, Talukder et al., 2000, Talukder A, 2010). The results of the primary data collection provide some interesting insights into the relationship between vegetable gardens, dietary diversity and nutritional status. Households with vegetable gardens grew a higher number of crops throughout the year than those households without but there were no significant difference in any of the dietary diversity or nutritional status variables between these households. Agrobiodiversity indices measured in Mbwei's vegetables gardens were not associated with dietary diversitybut they were negatively associated with children's height; the higher the diversity, the lower the height. These results are not encouraging of vegetable garden based interventions and, as with many of the projects results, suggests simply encouraging the use of vegetable gardens in subsistence farmers may not lead to the improvements in diet and health that are expected.

Another interesting insight the primary data collection affords us is on the effect of land degradation on food production. Mbwei had poorer quality farms and were more likely to report not being able to eat the crops they had planted. These households had children with poorer nutritional status. This study adds evidence of the effect on nutritional status to the body of evidence on the association between soil quality and food production (Ezaza, 1988, Doran and Zeiss, 2000).

It is also important to consider the relationships between plant agrobiodiversity and food security. More diverse agricultural fields are more resilient against changes in the climate Frison et al (2011). Growing a variety of crops suited to different conditions is therefore less risky than growing one or two main crops. Due to this increased resilience, higher agrobiodiversity reduces the risk of food insecurity (Frison et al., 2011, Thrupp, 2000).

In areas where agricultural projects aiming to improve diet and nutritional outcomes are being set up, thorough investigations of social and dietary factors likely to impact on nutritional status, such as wealth, education, nutrition knowledge, sanitation and complementary feeding practices present in these areas should occur. Interventions focused on improving agrobiodiversity are likely to be more effective as part of a broad, long term, multi-component intervention. Increasing agrobiodiversity while simultaneously reducing crop failure, ensuring food produced by the household is available year round, ensuring enough of the produced food reaches the household and the vulnerable members within the household, ensuring food is prepared with minimal impact of anti-nutrients in the diet and given to individuals who have decreased risk of infection due to improved sanitation practices will smooth the pathway

between agrobio diversity and nutritional status, allowing improvements in agrobio diversity to improve health as intended.

8.5 Animal agrobiodiversity and animal product consumption

The results of this study generally show positive associations between eating and rearing animals and dietary diversity and food variety. But they show negative associations between animal diversity and animal consumption and nutritional status in Minyenye children with no associations seen in Mbwei. These similar results were seen in both the villages, which are geographically and socially quite different from each other.

In the DHS analysis it was found that the more days eating meat and fish the higher the dietary diversity and the higher the children's height for age. Although this finding was not replicated in the primary data collection there is evidence of the association between animal product consumption and nutritional status in low income countries (Allen et al., 1992, Marquis et al., 1997, Krebs et al., 2011). Comparing the results from the primary data collection, the DHS analysis and the literature, again, demonstrates the complexity between food produced by the household and the health of the household members.

Given these complex results it is important to consider what motivates people to raise animals. Many people keep animals so they can be sold at times of need, for example to cover the costs of school fees or to buy food in times of food shortage. Keeping certain animals provides an important source of fertiliser to be used on household farms. Households raising animals for these reasons would not benefit from the consumption of meat and animal products.

The difference between just supplying children with extra meat or dairy in order to improve nutritional status in children and encouraging households to produce more animals for meat and animal products needs to be fully explored. The available literature shows that although increased intake of animal products often improves health outcomes, (Gibson et al., 2003, Whaley et al., 2003, Murphy et al., 2003, Siekmann et al., 2003, Neumann et al., 2013), the evidence is less clear on the effect of increased animal production (Leroy and Frongillo, 2007). The further up the path from food production to nutritional status a project attempts to intervene the more barriers the intervention will face in attempting to meet its aims. These barriers need to be addressed and a good understanding of these complexities are needed or projects that could have positive effects on health are likely to fail.

Researchers need to consider the reasons farmers make the decisions they do, for example whether animals raised for food or to provide fertiliser (Powell and Williams, 1994). The primary data collection found that many households raised animals they did not own, the main

- 250 -

benefit for this was using the manure for fertilisers for their farms. Another example is whether animals are raised to feed their family or to be able to sell them in order to send one of their children to school (Mazzeo, 2011). Without an understanding of why individuals in sub-Saharan Africa follow certain agricultural practices researchers will be unable to provide the support needed to make improvements that can lead to improvements in diet and health in these in these environments.

8.6 Selling household produce

Selling crops was positively associated with dietary diversity but, showed mixed associations with nutritional status. A similar effect was seen with selling animals, meat, eggs or milk. It is expected that the generation of additional income through the sale of household produce would have a positive effect on diet and growth. As with the other findings in this study this points to the complexity of the determinants of nutritional status. It is possible that money generated from the sale of produce is not being used to buy additional food for the household or perhaps that food is being sold rather than being consumed in the household. Neither scenario would be beneficial for the nutritional status of young children.

8.7 Limitations

In the primary data collection, data was collected at just one time point, at a time of year that the villages were experiencing relatively high amounts of food availability. If data was collected at more time points throughout the year and at times of food shortages the results may have been different (Poskitt EME). Additionally, both the primary data collection and the DHS analysis were cross-sectional studies and dietary diversity and nutritional status were measured at the same time. Height for age reflects long term nutritional trends so would not be influenced by current diet. Weight for age and weight for height are influenced by shorter term factors (Rowland et al., 1988). The reason why dietary diversity and these measures of nutritional status might be expected to be associated when measured cross-sectionally is because diet is very stable in these communities. The assumption that dietary diversity is the same the year of data collection as it was the preceding five years is not unreasonable and has been shown to be the case in a number of other studies (Arimond, 2004, Garg and Chadha, 2009, Savy et al., 2005, Sawadogo et al., 2006, Onyango et al., 1998, Saibul et al., 2009, Hatloy et al., 2000).

Questionnaire data from the primary data collection was collected using a translator and will be subject to errors in translation between Swahili and English. The dynamics between the researcher who is white and comes from a high income country and the respondents may have impacted on the honesty of the participants responses. It has previously been documented (Twyman et al., 1999) that participants may alter their responses either to make their situation better or worse than it actually is; either to avoid judgement by the researcher or in order to be more eligible for aid. There were other family and community members present for many of the interviews which may have impacted on the honesty of the results given. So me of the data, for example the complementary feeding and food sources data, relied on the respondent's memory and would be subject to recall bias. The accuracy of anthropometric measurements may have been affected as there were no flat surfaces available in the villages to measure height and weight.

There are also a number of limitations associated with the collection of the agrobiodiversity data which are discussed in detail in chapter 5. Two different biodiversity experts collected the agrobiodiversity data in the two different villages. Data collection was planned to take place before harvest in each village. In many households in Mbwei, and some households in Minyenye, data collection occurred after their main harvest which decreases the agrobiodiversity indices for these households. The fact that agrobiodiversity was only measured at one timepoint was another limitation of this study. Ideally agrobiodiversity would be measured at a number of different times in the year to counteract any seasonal effect s. Farm size differed between the two villages necessitating more frequent data collection methods in Mbwei compared to Minyenye. As the farms were much further away in Mbwei it was not possible to collect data on every farm and those farms that were an hour or more away were not visited for practical reasons.

Based on the site selection process it was expected that Minyenye would have lower plant agrobiodiversity than Mbwei. Due to hotter and dryer weather in Mbwei than is typical for the surrounding area this was not the case. This should not impact on the accuracy of the results but gives less contrast between the two villages than expected.

As with all survey methods, the demographic and health survey data is likely to be affected by bias. Some of the data, for example on complementary feeding, relies on the memory of the respondent and the accuracy of the responses may have been effected by recall bias. As with the primary data collection, the honesty of the responses may have been effected by the presence of the research team. The DHS survey is cross sectional so, as in the primary data collection, no causation can be applied to the results of the analyses. The DHS data is summarised for the DHS EA centre point and linked to the land cover data surrounding the

- 252 -

centre point. As the DHS EA vary in geographical size this technique may introduced error in to the analyses conducted; It is possible that a surveyed household could be outside of the 5km buffer and their surrounding land cover would not be included in the analysis.

8.8 Strengths

For the primary data collection households were selected within the villages using randomised household lists so the results of the analyses are representative of the villages. The primary data collection was multi-disciplinary, employing mixed methods which were tailored to the study objectives. The data collection methodology was very thorough, collecting detailed data on the areas of interest. All measures; agrobiodiversity, dietary diversity and nutritional status were measured on the same households allowing the regression to be carried out at the household level. The interviewer spoke to the female head of household, through a translator, so issues could be explored and points could be clarified on data collection. In the same way all data collected was checked as the data collection took place ensuring little missing data.

The project employed a biodiversity expert in each village to ensure the accurate collection of species data on the household farms. Local contacts were hired for each sub-village to find the selected participants and introduce them to the research team and to help with data collection issues to ensure the smooth running of the study.

The national analysis was based on the DHS survey data which is a large, nationally representative data set with detailed information on demographic, social and dietary data. Nutritional status measures were taken following strict protocols. This data was collected in 2010 and temporally matches the land cover data which was collected in 2009 quite closely. This land cover data has been validated using land cover experts from around the world. The validation report states the accuracy of the map to be 67.5% and states that globally there was 5% missing data of inland surfaces that were filled in using the 2005 map (Bontemps et al., 2011).

8.9 Implications and contributions of this study

There are very few projects investigating the relationship between agrobiodiversity, dietary diversity and nutritional status. One other project looks at the relationship between all three factors (Dewey, 1981) and four papers investigate the relationship between agrobiodiversity or

the number of crops grown and dietary diversity (Walingo and Ekesa, 2013, Bhagowalia et al., 2012, Akrofi, 2010, Ekesa et al., 2008). One other published study reported the relationship between agrobiodiversity and nutritional status (Shack et al., 1990). This project looks directly at these relationships, at both a local level in two villages in rural Tanzania, and less directly at a national level, using land cover as a proxy for diversity.

Many discussion papers point to agrobiodiversity as a way to increase the diversity of peoples diets in the hope of improving health. The project provides evidence against the theory that increasing agrobiodiversity will increase dietary diversity and mixed evidence about the relationship between agrobiodiversity and nutritional status. It also shows that dietary diversity is not related to nutritional status in these local settings or at the national level. The main implication of these findings are that future nutrition interventions need to address the multiple determinants of nutritional status in order to test whether improvements in agrobiodiversity can lead to improvements in diet and nutritional status.

8.10 Future research and methodological and study design insights

Important questions around whether improvements in agrobiodiversity can lead to improvements in diet and health remain unanswered. In this study, results on the relationship between agrobiodiversity and nutritional status differ between the two different villages indicating the relationship is context specific within rural Tanzania. The DHS study also showed mixed results with agricultural but not forest land cover being positively associated with dietary diversity, both agricultural and forest land cover being positively associated with weight for height and agricultural land being negatively associated with height for age.

The conceptual framework proposed in chapter 3 is largely not supported by the results of either the primary data collection or the national analysis. More research is required to investigate the barriers between agrobiodiversity and dietary diversity as well as why, in some areas such as those studied in this project, dietary diversity is not related to nutritional status. More detailed investigation into how higher agrobiodiversity effects household income and how selling household produce can benefit child nutritional status is needed.

Collecting detailed biodiversity data in the household farms resulted in data that was not associated with dietary diversity and food variety while asking the respondents what they have grown in the past year resulted in a simple count of crops that was associated with food variety. This raises the question; is this simple count a more useful measure than collecting time consuming detailed data on all the different plants that are growing in the household

farms at the time of data collection? This measure would be quicker, easier and cheaper to collect and it would represent foods grown over the whole year which is potentially more useful that a measure taken at a limited number of cross-sectional time points. There would be disadvantages to this approach though, it would rely on the recall of the respondent, it requires more interview time and it is not as objective a measure as the systematic collection of agrobiodiversity data in the household farms.

For studies investigating the relationships between plant agrobiodiversity and dietary diversity and health it is important to measure what is grown in household farms and vegetable gardens and not to just rely on the recall of the household members as these two methods provide different information. To improve on the reliability of the plant agrobiodiversity data collected in this study, data should be collected at more than one timepoint throughout the year. These data collection times should match the growing seasons for the area to be confident all crops grown in the household farms and gardens are captured.

Capturing dietary diversity at these different times will increase the understandings of annual variation in dietary diversity and how this relates to the current and previous agrobiodiversity. Capturing both agrobiodiversity and dietary diversity longitudinally and linking these measures to either nutritional status at the end of the data collection or growth over this time period would allow for causal inferences to be made. If it was established that higher agrobiodiversity was associated with higher dietary diversity and nutritional status longitudinally then RCT intervention trials to improve agrobiodiversity in order to improve dietary diversity and nutritional status would be appropriate.

However, for monitoring and evaluating agricultural interventions in resource limited research environments, the simpler crop and vegetable diversity scores could be useful. If it is established that increasing the number of different types of crops grown throughout the year is a desired outcome then asking how many different crops types are grown annually, before and after any given intervention, would provide evidence towards determining whether an intervention had been effective in improving agrobi odiversity.

Due to the multiple determinants of nutritional status in rural sub-Saharan Africa single interventions focused on agrobiodiversity may miss important findings if the results are confounded by, for example, infection. An example of this occurred in an food based vitamin A intervention to improve dietary practices and serum retinol in Tanzania. The intervention area had higher incidence of *helminths* infection and the authors concluded that this was a reason for not seeing any improvements in serum retinol in the intervention group (Kidala et al., 2000). The most successful interventions have been broad, multi-component interventions (Berti, 2004, Leroy and Frongillo, 2007, Remans et al., 2011b). The Millennium villages project 255 found improvements in food security, dietary diversity and stunting with a long term, intersectorial project with agricultural, health, education and infrastructure components (Remans et al., 2011b). More research is needed to continue to unpick the associations seen between agrobiodiversity, dietary diversity and nutritional status in the varied contexts in rural sub-Saharan Africa. Longitudinal research is needed to better understand these associations and comprehensive multi-component interventions are needed in order to determine whether increasing agrobiodiversity improves dietary diversity and nutritional status.

Chapter 9:

Overall conclusion

9.1 How the study fulfils the aim and objectives

This project has fulfilled its aim to **investigate the relationship between agrobiodiversity**, **dietary diversity and nutritional status in Tanzania**. Before carrying out the primary data collection the methodology and data collection tools were designed and piloted in a village in the Usambara mountains in Tanzania (Objective 1). Methodology was tailored to assess the diversity and abundance of cultivated and wild plants growing on household farms in the two villages in Tanzania (Objective 2). On carrying out the primary data collection, dietary diversity and nutritional status were measured in household members living in these two villages (Objective 3). Other demographic, social and local environmental data were collected to explore the effect these factors had on both dietary diversity and nutritional status. These data were compiled and statistical analyses were used to investigate how these different factors were related and which were related to nutritional status (Objective 4).

In the second phase of this study data from the Demographic and Health Survey carried out in 2010 in Tanzania and land cover data from the Globcover 2009 map were analysed to investigate the relationships between land cover, social and demographic data, dietary diversity and nutritional status (Objective 5).

The results from the primary data collection and the DHS and land cover analyses are collated and discussed in chapter 8 in order to draw conclusions that can inform understanding of food security and the determinants of nutritional status in Tanzania (Objective 6).

9.2 Dietary diversity is not associated with nutritional status in crosssectional data

No association was found between dietary diversity and nutritional status in Minyenye or Mbwei. When these analyses were repeated on a national level using the Tanzanian DHS data it was found that children with higher dietary diversity actually had lower height for age. Despite the many papers published on data from low income countries showing positive

associations between dietary diversity and nutritional status this study found no or opposite associations. It is not appropriate to suggest that there is not an association between dietary diversity and nutritional status from the evidence presented in this thesis but it does illustrate that the association between dietary diversity and nutritional status is not universal and should not be assumed.

9.3 Agrobiodiversity is not associated with dietary diversity in crosssectional data

No significant associations were found between the agrobiodiversity found in household farms and dietary diversity or food variety in either village. There was however an association between crop and vegetable diversity scores and dietary diversity.

Looking at both agricultural land cover and forest land cover separately in the DHS analysis is a different approach to looking at agrobiodiversity of household farms. This approach, however, provides some interesting insights. Agricultural land cover is linked with higher dietary diversity. Interestingly the more diverse forest land cover is not associated with increased dietary diversity which supports the lack of association seen between the household diversity indices and dietary diversity in the primary data collection.

9.4 Agrobiodiversity has mixed associations with nutritional status

Different associations between agrobiodiversity and nutritional status were seen between the two villages. In Minyenye, agrobiodiversity was positively associated with height in children while in Mbwei it was negatively associated with BMI z-scores. Factors such as farm size, distance to the farms, soil fertility and crop failure, as well as the non-agricultural determinants of nutritional status vary between these two villages and are likely to be the reasons behind these differences. The results of land cover and nutritional status were mixed and revealed no clear picture overall.

9.5 Simply increasing agrobiodiversity will not necessarily improve dietary diversity and nutritional status

Both the DHS analysis and the primary data collection results suggest the often cited association between agrobiodiversity, dietary diversity and nutritional status should not be taken for granted. Areas with access to greater biodiversity, both in the surrounding area and on the households farms will not necessarily be better off nutritionally than those with less diversity. These results have shown that there are many different demographic and social factors that are associated with nutritional status and suggest that, in the context of high rates of poverty and infection, improvements in agrobiodiversity in isolation will not necessarily have the desired positive impact on diet and nutritional status in low income countries.

9.6 Streamline monitoring agrobiodiversity through diversity scores

Asking the female head of household about the variety of foods the household has eaten over the past year and whether these foods were produced within the household gave a simple measure of the crops grown and animals raised by the household. These measures were significantly associated with the dietary diversity of the respondent and her youngest child where measures of agrobiodiversity were not.

Measuring agrobiodiversity in the household farms was a time consuming process which required a research assistant with expertise in biodiversity and knowledge of local species. The data represented the diversity at the time point of data collection but did not give any indication of the other foods grown throughout the year. Although a simple count of food grown by a household annually does not represent agrobiodiversity it is an easier, less resource intensive method which gives data which is associated with dietary diversity. This suggests that this simple count may be more useful to evaluate projects which are, for example, aiming to increase the number of crops grown in order to increase dietary diversity.

9.7 Animal production and consumption will not necessarily improve dietary diversity and nutritional status

The results from the primary data collection found that eating meat and animal products were generally positively associated with dietary diversity but negatively associated with nutritional

status. Also, households raising animals for consumption of their meat or products had worse nutritional status in most cases than those not raising animals. Chapter 6 outlines a number of hypotheses of why this might be but the major conclusion from this work is that it cannot be assumed that encouraging subsistence farmers to raise animals will have a positive effect on the health of family members. Again, it must be acknowledged that the effects raising animals has and the uses animals are put to are more complex than may be expected. These results, along with the results from the plant agrobiodiversity work suggest local understanding and collaboration and detailed scoping and monitoring are essential in projects aiming to improve nutritional status through modifications in food production practices in subsistence far mers.

Looking at the national results from the DHS survey indicates that consuming meat and fish has a positive effect on children's height. Relying on data averaged at a national level means differences between regions and villages are lost. This again highlights the importance of local understanding. Although increased consumption of animal products has the potential to improve nutritional status these interventions need to be very carefully implemented and monitored.

9.8 Selling produce will not necessarily improve dietary diversity and nutritional status

Selling crops was positively associated with dietary diversity. This was not found in the majority of cases for selling animals or their products. Selling crops and products from animals raised in the household showed mixed associations with nutritional status. Again this provides caution to those development agencies and nutrition projects operating on the assumption that households raising additional income from selling food products that they produce will have a positive impact on the health of the family members. There are a number of reasons why households will chose to sell crops and animal products and the context these households are living in need to be well understood before any particular intervention is encouraged.

9.9 Agricultural interventions designed to improve nutritional status need to be piloted, context specific and regularly monitored

In order to improve the effectiveness of agricultural interventions aiming to impact on nutrition outcomes in low income countries it is important for their environmental, political,

social and cultural context is taken into consideration at all stages of the project. More people need to be fed nutritious diets with less land and resources than in the past and funding for nutrition research in low income countries is limited. Programs cannot be instituted without evidence of effectiveness. Factors determining the dietary intake of individuals are complex and context specific. Researchers cannot assume that interventions will have the effect they are intending or that effect alone.

For these reasons thorough background research, piloting of methodology, locally led research or at least community involvement is essential. Projects also need to be regularly and effectively monitored in order to determine the effect the project is having on the participants and their communities.

9.10 Interventions need to be broad to successfully address malnutrition

This project has provided much evidence and discussion on the multiple determinants of nutritional status. The fact that many of these determinants are still to be addressed in Minyenye and Mbwei could have limited the association seen between agrobiodiversity, dietary diversity and nutritional status. Following on from this hypothesis it is important for interventions designed to improve diet and nutritional status to address as many of these determinants as possible in order to truly test the effectiveness of the intervention of interest. It is likely that many interventions have failed to show an effect due to confounding from, for example, infection and consequently potentially effective interventions may have been disregarded. That agrobiodiversity, dietary diversity and nutritional status were not associated in this cross-sectional study provides evidence of the complexity of these relationships. Agricultural interventions designed to improve nutritional status need to be broad in scope, addressing food security, diet, risk of infection among other factors rele vant to the local context. They will likely need to be multi-component interventions investing much time and resources in order to have a positive effect on the health of the target group.

References

- ABOUD, F. E., SINGLA, D. R., NAHIL, M. I. & IVELINA, B. 2013. Effectiveness of a parenting program in Bangladesh to address early childhood health, growth and development. (Special Issue: Critical periods in early human development.). *Social Science & Medicine*, 97, 250-258.
- ABOUSSALEH, Y. & AHAMI, A. 2009. Dietary determinants of stunting and anaemia among preadolescents in Morocco. *African Journal of Food, Agriculture, Nutrition and Development*, 9, 728-747.

ABRAHAMS, Z., MCHIZA, Z. & STEYN, N. P. 2011. Diet and mortality rates in Sub-Saharan Africa: Stages in the nutrition transition. *BMC Public Health*, 11, 801.

- ABUYA, B. A., CIERA, J. & KIMANI-MURAGE, E. 2012. Effect of mother's education on child's nutritional status in the slums of Nairobi. *BMC Pediatrics*, 12, 80.
- ADAIR, L. S., FALL, C. H., OSMOND, C., STEIN, A. D., MARTORELL, R., RAMIREZ-ZEA, M., SACHDEV, H. S., DAHLY, D. L., BAS, I. & NORRIS, S. A. 2013. Associations of linear growth and relative weight gain during early life with adult health and human capital in countries of low and middle income: findings from five birth cohort studies. *The Lancet*.
- AFRICASCOPE. 2014. Africascope, Information for Africa homepage [Online]. Available: www.africascope.net [Accessed 29 April 2014.
- AKROFI, S., BROUWER, I. D., PRICE, L. L. & STRUIK, P. C. 2010. Home gardens contribute significantly to dietary diversity in HIV/AIDS afflicted households in rural Ghana. *Journal of Human Ecology*, 31, 125-134.
- AKROFI, S. B., ID. PRICE, LL. STRUIK, PC. 2010. Home gardens contribute significantly to dietary diversity in HIV/AIDS afflicted households in rural Ghana. . *Journal of Human Ecology*, 31, 125-134.
- ALDERMAN, H., HOOGEVEEN, H. & ROSSI, M. 2006. Reducing child malnutrition in Tanzania: Combined effects of income growth and program interventions. *Economics & amp; Human Biology, 4*, 1-23.
- ALLEN, L. H., BACKSTRAND, J. R., STANEK, E. J., III, PELTO, G. H., CHAVEZ, A., MOLINA, E., CASTILLO, J. B. & MATA, A. 1992. The interactive effects of dietary quality on the growth and attained size of young Mexican children. *American Journal of Clinical Nutrition*, 56, 353-364.
- ALTIERI, M. A. 1999. The ecological role of biodiversity in agroecosystems. *Agriculture, ecosystems & environment,* 74, 19-31.
- ALTIERI, M. A. 2009. Agroecology, small farms, and food sovereignty. *Monthly Review*, 61, 102-113.
- ALVARADO, B. E., ZUNZUNEGUI, M. V., DELISLE, H. & OSORNO, J. 2005. Growth trajectories are influenced by breast-feeding and infant health in an Afro-Colombian community. *Journal of Nutrition*, 135, 2171-2178.
- ANGULO, F. J., JOHNSON, K. R., TAUXE, R. V. & COHEN, M. L. 2000. Origins and consequences of antimicrobial-resistant nontyphoidal Salmonella: implications for the use of fluoroquinolones in food animals. *Microbial Drug Resistance*, 6, 77-83.

- ARIMOND, M. R., MT. 2004. Dietary diversity is associated with child nutritional status: evidence from 11 Demographic and Health Surveys. *Journal of Nutrition*, 134, 2579-2585.
- ARSHAD, M. A. & MARTIN, S. 2002. Identifying critical limits for soil quality indicators in agro-ecosystems. *Agriculture, ecosystems & environment,* 88, 153-160.
- AUSTIN, G. E., THOMAS, C. J., HOUSTON, D. C. & THOMPSON, D. B. 1996. Predicting the spatial distribution of buzzard Buteo buteo nesting areas using a Geographical Information System and remote sensing. *Journal of Applied Ecology*, 33, 1541-1550.
- AWUMBILA, M. 2003. Social dynamics and infant feeding practices in Northern Ghana. Institute of African Studies Research Review, 19, p. 85-98.
- BAJAJ, M., DUBEY, A., NAGPAL, J., SINGH, P. & SACHDEV, H. P. 2005. Short-term effect of oil supplementation of complementary food on total ad libitum consumption in 6-to 10-month-old breastfed Indian infants. *Journal of pediatric gastroenterology and nutrition*, 41, 61-65.
- BARRY, B., OLALEYE, A. & ADEOTI, A. 2006. Soil water conservation and rain water harvesting technologies in the semi-arid zone of sub-Saharan Africa. *GRID-Magazine of the IPTRID Network*.
- BEASLEY, N. M., HALL, A., TOMKINS, A. M., DONNELLY, C., NTIMBWA, P., KIVUGA, J., KIHAMIA, C. M., LORRI, W. & BUNDY, D. A. 2000. The health of enrolled and non enrolled children of school age in Tanga, Tanzania. Acta Tropica, 76, 223-9.
- BÉLANGER, J. & JOHNS, T. 2008. Biological diversity, dietary diversity, and eye health in developing country populations: Establishing the evidencebase. *EcoHealth*, 5, 244-256.
- BENEFICE, E., LOPEZ, R., MONROY, S. L. & RODRIGUEZ, S. 2007. Fatness and overweight in women and children from riverine Amerindian communities of the Beni River (Bolivian Amazon). *American Journal of Human Biology*, 19, 61-73.
- BERINYUY, J. & FONTEM, D. 2011. Evaluating post harvest opportunities and constraints to utilization and marketing of African leafy vegetables in Cameroon. *African Journal of Food, Agriculture, Nutrition and Development*, 11.
- BERTI, P. K., J. FITZGERALD, S. 2004. A review of the effectiveness of agriculture interventions in improving nutrition outcomes. *Public Health Nutrition*, 7, 599-609.
- BERTI, P. R., MILDON, A., SIEKMANS, K., MAIN, B. & MACDONALD, C. 2010. An adequacy evaluation of a 10-year, four-country nutrition and health programme. *International Journal of Epidemiology*, 39, 613-29.
- BHAGOWALIA, P., HEADEY, D. & KADIYALA, S. 2012. Agriculture, Income, and Nutrition Linkages in India: Insights from a Nationally Representative Survey.
- BHARUCHA, Z. & PRETTY, J. 2010. The roles and values of wild foods in agricultural systems. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 365, 2913-2926.
- BHASKARAM, P. 2002. Micronutrient malnutrition, infection, and immunity: an overview. *Nutrition Reviews*, 60, S40-S45.
- BHUTTA, Z. A., AHMED, T., BLACK, R. E., COUSENS, S., DEWEY, K., GIUGLIANI, E., HAIDER, B. A., KIRKWOOD, B., MORRIS, S. S. & SACHDEV, H. 2008. What works? Interventions for maternal and child undernutrition and survival. *The Lancet*, 371, 417-440.

- BHUTTA, Z. A., DAS, J. K., RIZVI, A., GAFFEY, M. F., WALKER, N., HORTON, S., WEBB, P., LARTEY, A. & BLACK, R. E. 2013. Evidence-based interventions for improvement of maternal and child nutrition: what can be done and at what cost? *The Lancet*, 382, 452-477.
- BLACK, R. E., ALLEN, L. H., BHUTTA, Z. A., CAULFIELD, L. E., DE ONIS, M., EZZATI, M., MATHERS, C. & RIVERA, J. 2008. Maternal and child undernutrition: global and regional exposures and health consequences. *The Lancet*, 371, 243-260.
- BONTEMPS, S., DEFOURNY, P., VAN BOGAERT, E., ARINO, O., KALOGIROU, V. & RAMOS PEREZ, J. 2011. GlobCover 2009: products description and validation report. URL: <u>http://ionia1</u>. esrin. esa. int/docs/GLOBCOVER2009_Validation_Report_2, 2.
- BORK, K., CAMES, C., BARIGOU, S., COURNIL, A. & DIALLO, A. 2012. A summary index of feeding practices is positively associated with height-for-age, but only marginally with linear growth, in rural Senegalese infants and toddlers. *Journal of Nutrition*, 142, 1116-1122.
- BOVET, P., ROSS, A. G., GERVASONI, J.-P., MKAMBA, M., MTASIWA, D. M., LENGELER, C., WHITING, D. & PACCAUD, F. 2002. Distribution of blood pressure, body mass index and smoking habits in the urban population of Dar es Salaam, Tanzania, and associations with socioeconomic status. *International Journal of Epidemiology*, 31, 240-247.
- BROOKFIELD, H. & STOCKING, M. 1999. Agrodiversity: definition, description and design. *Global Environmental Change*, 9, 77-80.
- BROWN, M. E. 2009. Biophysical remote sensing and climate data in famine early warning systems. *Geography Compass*, 3, 1381-1407.
- BRUN, T., REYNAUD, J. & CHEVASSUS-AGNES, S. 1989. Food and nutritional impact of one home garden project in Senegal. *Ecology of food and nutrition*, 23, 91-108.
- BUKUSUBA, J., KIKAFUNDA, J. K. & WHITEHEAD, R. G. 2007. Food security status in households of people living with HIV/AIDS (PLWHA) in a Ugandan urban setting. *British Journal of Nutrition*, 98, 211-217.
- BUKUSUBA, J., KIKAFUNDA, J. K. & WHITEHEAD, R. G. 2010. Nutritional knowledge, attitudes, and practices of women living with HIV in eastern Uganda. *Journal of health, population, and nutrition,* 28, 182.
- BURLINGAME, B. A. & DERNINI, S. 2012. Sustainable diets and biodiversity.
- BUTZLER, J. P. 2004. Campylobacter, from obscurity to celebrity. *Clinical microbiology and infection*, 10, 868-876.
- BWIBO, N. O. & NEUMANN, C. G. 2003. The need for animal source foods by Kenyan children. *Journal of Nutrition,* 133.
- BYIRINGIRO, F. & REARDON, T. 1996. Farm productivity in Rwanda: effects of farm size, erosion, and soil conservation investments. *Agricultural economics*, 15, 127-136.
- CABALDA, A. B., RAYCO-SOLON, P., SOLON, J. A. A. & SOLON, F. S. 2011. Home Gardening Is Associated with Filipino Preschool Children's Dietary Diversity. *Journal of the American Dietetic Association*, 111, 711-715.
- CAULFIELD, L. E., HUFFMAN, S. L. & PIWOZ, E. G. 1999. Interventions to improve intake of complementary foods by infants 6 to 12 months of age in developing countries: impact on growth and on the prevalence of malnutrition and potential contribution to child survival. *Food & Nutrition Bulletin*, 20, 183-200.

- CHEN, L. C., CHOWDHURY, A. & HUFFMAN, S. L. 1980. Anthropometric assessment of energy-protein malnutrition and subsequent risk of mortality among preschool aged children. *The American journal of clinical nutrition*, 33, 1836-1845.
- CHILTON, M., CHYATTE, M. & BREAUX, J. 2007. The negative effects of poverty & food insecurity on child development. *Indian Journal of Medical Research*, 126, 262.
- CHIVIAN, E. 2002. Biodiversity: its importance to human health. *Harvard Medical School, Boston.*
- COATES, J., FRONGILLO, E. A., ROGERS, B. L., WEBB, P., WILDE, P. E. & HOUSER, R. 2006. Commonalities in the experience of household food insecurity across cultures: what are measures missing? *The Journal of nutrition*, 136, 1438S-1448S.
- COLE, T. J., FLEGAL, K. M., NICHOLLS, D. & JACKSON, A. A. 2007. Body mass index cut offs to define thinness in children and adolescents: international survey. *BMJ*, 335, 194.
- CORBETT, E. L., BUTTERWORTH, A. E., FULFORD, A. J. C., OUMA, J. H. & STURROCK, R. F. 1992. Nutritional status of children with schistosomiasis mansoni in two different areas of Machakos District, Kenya. *Transactions of the Royal Society of Tropical Medicine and Hygiene*, 86, 266-273.
- CORDEIRO, L. S., WILDE, P. E., SEMU, H. & LEVINSON, F. J. 2012. Household food security is inversely associated with undernutrition among adolescents from Kilosa, Tanzania. *Journal of Nutrition*, 142, 1741-7.
- COULLOUDON, B., ESHELMAN, K., GIANOLA, J., HABICH, N., HUGHES, L., JOHNSON, C., PELLANT, M., PODBORNY, P., RASMUSSEN, A. & ROBLES, B. 1999. Sampling vegetation attributes. *BLM Technical Reference*, 1734-4.
- CROMWELL, E., COOPER, D. & MULVANY, P. 1999. Agriculture, Biodiversity and Livelihoods: issues and entry points for development agencies. Seeds & Crop Diversity Programme, Overseas Development Institute.
- DANCER, D., RAMMOHAN, A. & SMITH, M. D. 2008. Infant mortality and child nutrition in Bangladesh. *Health economics*, 17, 1015-1035.
- DANIELS, M., ADAIR, L., POPKIN, B. & TRUONG, Y. 2007. Dietary diversity scores can be improved through the use of portion requirements: an analysis in young Filipino children. *European Journal of Clinical Nutrition*, 63, 199-208.
- DE ONIS, M., BLÖSSNER, M. & BORGHI, E. 2012. Prevalence and trends of stunting among pre-school children, 1990–2020. *Public Health Nutrition*, 15, 142-148.
- DE ONIS, M., BLÖSSNER, M. & ORGANIZATION, W. H. 1997. WHO global database on child growth and malnutrition, World Health Organization Geneva.
- DECKELBAUM, R. J., PALM, C., MUTUO, P. & DECLERCK, F. 2006. Econutrition: implementation models from the Millennium Villages Project in Africa. *Food & Nutrition Bulletin*, 27, 335-342.
- DEWEY, K. G. 1981. Nutritional consequences of the transformation from subsistence to commercial agriculture in Tabasco, Mexico. *Human Ecology*, 9, 151-187.
- DHARANI, N. 2002. *Field guide to common trees & shrubs of east Africa*, Struik.

- DI GREGORIO, A. & JANSEN, L. J. 1998. Land Cover Classification System (LCCS): classification concepts and user manual. *FAO, Rome.*
- DISHA, A. D., RAWAT, R., SUBANDORO, A. & MENON, P. 2012. Infant and young child feeding (IYCF) practices in Ethiopia and Zambia and their association with child nutrition: analysis of Demographic and Health Survey data. *African Journal of Food, Agriculture, Nutrition and Development*, 12, 5895-5914.
- DORAN, J. W. & ZEISS, M. R. 2000. Soil health and sustainability: managing the biotic component of soil quality. *Applied Soil Ecology*, 15, 3-11.
- DOVIE, D. B. K., SHACKLETON, C. M. & WITKOWSKI, E. T. F. 2006. Valuation of communal area livestock benefits, rural livelihoods and related policy issues. *Land Use Policy*, 23, 260-271.
- DUE, J. M. & GLADWIN, C. H. 1991. Impacts of structural adjustment programs on African women farmers and female-headed households. *American Journal of Agricultural Economics*, 73, 1431-1439.
- EBOMOYI, E., ADETORO, O. O. & WICKREMASINGHE, A. R. 1991. Birthweight and sociobiological factors in llorin, Nigeria. *Journal of Biosocial Science*, 23, 417-23.
- ECKER, O., WEINBERGER, K. & QAIM, M. 2010. Patterns and determinants of dietary micronutrient deficiencies in rural areas of East Africa.
- ECKHARDT, C. L., SUCHINDRAN, C., GORDON-LARSEN, P. & ADAIR, L. S. 2005. The association between diet and height in the postinfancy period changes with age and socioeconomic status in Filipino youths. *The Journal of nutrition*, 135, 2192-2198.
- EKESA, B., WALINGO, M. & ABUKUTSA-ONYANGO, M. 2008. Influence of agricultural biodiversity on dietary diversity of preschool children in Matungu division, Western Kenya. *African Journal of Food, Agriculture, Nutrition and Development*, **8**, 390-404.
- EKESA, B. N., BLOMME, G. & GARMING, H. 2011. Dietary diversity and nutritional status of pre-school children from Musa-dependent households in Gitega (Burundi) and Butembo (Democratic Republic of Congo). *African Journal of Food, Agriculture, Nutrition and Development,* 11, 4896-4911.
- EL-GHANNAM, A. R. 2003. The global problems of child malnutrition and mortality in different world regions. *Journal of health & social policy*, 16, 1-26.
- EL-SHARKAWY, M. A. 2004. Cassava biology and physiology. *Plant molecular biology*, 56, 481-501.
- EL-SHARKAWY, M. A. 2007. Physiological characteristics of cassava tolerance to prolonged drought in the tropics: Implications for breeding cultivars adapted to seasonally dry and semiarid environments. *Brazilian Journal* of *Plant Physiology*, 19, 257-286.
- ENGLE, P. L. & NIEVES, I. 1993. Intra-household food distribution among Guatemalan families in a supplementary feeding program: behavior patterns. *Social Science & Medicine*, 36, 1605-1612.
- ENGLISH, R., BADCOCK, J., GIAY, T., NGU, T., WATERS, A. & BENNETT, S. 1997. Effect of nutrition improvement project on morbidity from infectious diseases in preschool children in Vietnam: comparison with control commune. *BMJ*, 315, 1122-1125.
- ESREY, S. A. 1996. Water, waste, and well-being: a multicountry study. *American journal of epidemiology*, 143, 608-623.
- EUROPEAN SPACE AGENCY. 2010. *ESA data user element* [Online]. Available: <u>http://due.esrin.esa.int/globcover/</u> [Accessed 25 April 2014.
- EZAZA, W. P. 1988. Geoecological Factors Influencing Over-Exploitation and Land Degradation in the Usambara Mountains of Northeastern Tanzania. *Mountain Research and Development*, 8, 157-163.
- FAFCHAMPS, M., UDRY, C. & CZUKAS, K. 1998. Drought and saving in West Africa: are livestock a buffer stock? *Journal of Development Economics*, 55, 273-305.
- FANTA 2004. Measuring household food insecurity: Workshop Report April 15– 16, 2004: Food and Nutrition Technical Assistance Project, Academy for Educational Development; 2005. .
- FAO. Rome Declaration on World Food Security, World Food Summit. World Food Summit, 1996. Rome.
- FAO. 2014a. *AGP Plant Production and Protection Home* [Online]. Available: <u>http://www.fao.org/agriculture/crops/en/</u> [Accessed 29 April 2014.
- FAO. 2014b. Agroecological zones of Africa. [Online]. Available: <u>http://www.fao.org/ag/againfo/programmes/en/lead/toolbox/Refer/africa.h</u> <u>tm</u> [Accessed 30 April 2014.
- FAO/WHO/IFPRI 2004. Workshop on dietary diversity and dietary quality. Rome, Italy.
- FAYER, R., MORGAN, U. & UPTON, S. J. 2000. Epidemiology of < i > Cryptosporidium </i>: transmission, detection and identification. *International journal for parasitology*, 30, 1305-1322.
- FERGUSSON, P. & TOMKINS, A. 2009. HIV prevalence and mortality among children undergoing treatment for severe acute malnutrition in sub-Saharan Africa: a systematic review and meta-analysis. *Transactions of the Royal Society of Tropical Medicine and Hygiene*, 103, 541-548.
- FERNANDEZ, I. D., HIMES, J. H. & ONIS, M. D. 2002. Prevalence of nutritional wasting in populations: building explanatory models using secondary data. *Bulletin of the World Health Organization*, 80, 282-291.
- FINK, G., GUNTHER, I. & HILL, K. 2011. The effect of water and sanitation on child health: evidence from the demographic and health surveys 1986-2007. *International Journal of Epidemiology*, 40, 1196-204.
- FORESIGHT 2011. The future of food and farming. Executive Summary.: The Government Office for Science, London.
- FOUERE, T., MAIRE, B., DELPEUCH, F., MARTIN-PREVEL, Y., TCHIBINDAT, F. & ADOUA-OYILA, G. 2000. Dietary changes in African urban households in response to currency devaluation: foreseeable risks for health and nutrition. *Public Health Nutrition*, 3, 293-301.
- FRISON, E., SMITH, I., CHERFAS, J., EYZAGUIRRE, P. & JOHNS, T. 2005. Using biodiversity for food, dietary diversity, better nutrition and health. *South African Journal of Clinical Nutrition,* 18, p. 112-114.
- FRISON, E. A., CHERFAS, J., EYZAGUIRRE, P. B. & JOHNS, T. Biodiversity, nutrition and health: making a difference to hunger and conservation in the developing world. Seventh Meeting of the Conference of the Parties to the Convention on Biological Diversity (COP7), International Plant Genetic Resources Institute, Rome, Italy, 2004.
- FRISON, E. A., CHERFAS, J. & HODGKIN, T. 2011. Agricultural biodiversity is essential for a sustainable improvement in food and nutrition security. *Sustainability*, 3, 238-253.
- FRISON, E. A., SMITH, I. F., JOHNS, T., CHERFAS, J. & EYZAGUIRRE, P. B. 2006. Agricultural biodiversity, nutrition, and health: making a difference

to hunger and nutrition in the developing world. *Food & Nutrition Bulletin,* 27, 167-179.

- GARG, A. & CHADHA, R. 2009. Index for measuring the quality of complementary feeding practices in rural India. *Journal of health, population, and nutrition,* 27, 763.
- GEGIOS, A., AMTHOR, R., MAZIYA-DIXON, B., EGESI, C., MALLOWA, S., NUNGO, R., GICHUKI, S., MBANASO, A. & MANARY, M. J. 2010. Children consuming cassava as a staple food are at risk for inadequate zinc, iron, and vitamin A intake. *Plant Foods for Human Nutrition*, 65, 64-70.
- GIBSON, R. S. & ANDERSON, V. P. 2009. A review of interventions based on dietary diversification or modification strategies with the potential to enhance intakes of total and absorbable zinc. *Food & Nutrition Bulletin*, 30, 108S-143S.
- GIBSON, R. S., PERLAS, L. & HOTZ, C. 2006. Improving the bioavailability of nutrients in plant foods at the household level. *Proceedings of the Nutrition Society*, 65, 160-168.
- GIBSON, R. S., YEUDALL, F., DROST, N., MTITIMUNI, B. M. & CULLINAN, T. R. 2003. Experiences of a community-based dietary intervention to enhance micronutrient adequacy of diets low in animal source foods and high in phytate: a case study in rural Malawian children. *The Journal of nutrition*, 133, 3992S-3999S.
- GITTELSOHN, J., THAPA, M. & LANDMAN, L. T. 1997. Cultural factors, caloric intake and micronutrient sufficiency in rural Nepali households. *Social Science & Medicine*, 44, 1739-1749.
- GITTELSOHN, J. & VASTINE, A. E. 2003. Sociocultural and household factors impacting on the selection, allocation and consumption of animal source foods: current knowledge and application. *the Journal of Nutrition*, 133, 4036S-4041S.
- GLYMOUR, M. M. 2006. Using causal diagrams to understand common problems in social epidemiology. *Methods in social epidemiology*, 393-428.
- GODFRAY, H. C. J., BEDDINGTON, J. R., CRUTE, I. R., HADDAD, L.,
 LAWRENCE, D., MUIR, J. F., PRETTY, J., ROBINSON, S., THOMAS, S.
 M. & TOULMIN, C. 2010. Food security: the challenge of feeding 9 billion people. *Science*, 327, 812-818.
- GOLLOGLY, L. 2009. World health statistics 2009, World Health Organization.
- GOTOR, E. I., C. 2010. The Impact of Bioversity International's African Leafy vegetables Programme in Kenya. *ImpactAssessment and Project Appraisal*, 28, 41-55.
- GREEN, R. E., CORNELL, S. J., SCHARLEMANN, J. P. & BALMFORD, A. 2005. Farming and the fate of wild nature. *Science*, 307, 550-555.
- GRILLENBERGER, M. 2006. Impact of animal source foods on growth, morbidity and iron bioavailability in Kenyan school children, Impact of animal source foods on growth, morbidity and iron bioavailability in Kenyan school children; 2006. :173 pp. many ref.
- GRILLENBERGER, M., NEUMANN, C. G., MURPHY, S. P., BWIBO, N. O., VEER, P. V. T., HAUTVAST, J. G. A. J. & WEST, C. E. 2003. Food supplements have a positive impact on weight gain and the addition of animal source foods increases lean body mass of Kenyan schoolchildren. *Journal of Nutrition*, 133.

- GROSS, R., GIAY, T., SASTROAMIDJOJO, S., SCHULTINK, W. & LANG, N. T. 2000. Premature complementary feeding is associated with poorer growth of Vietnamese children. *The Journal of nutrition*, 130, 2683-2690.
- GUÈYE, E. 2005. Gender aspects in family poultry management systems in developing countries. *World's poultry science journal*, 61, 39-46.
- HAAS, J. D., MURDOCH, S., RIVERA, J. & MARTORELL, R. 1996. Early nutrition and later physical work capacity. *Nutrition Reviews*, 54, S41-8.
- HABIB, N. A., LIE, R. T., ONEKO, O., SHAO, J., BERGSJO, P. & DALTVEIT, A. K. 2008. Sociodemographic characteristics and perinatal mortality among singletons in North East Tanzania: a registry-based study. *Journal of Epidemiology & Community Health*, 62, 960-5.
- HABICHT, J.-P., YARBROUGH, C., MARTORELL, R., MALINA, R. & KLEIN, R. 1974. Height and weight standards for preschool children: How relevant are ethnic differences in growth potential? *The Lancet*, 303, 611-615.
- HADGU, K. M., ROSSING, W. A., KOOISTRA, L. & VAN BRUGGEN, A. H. 2009. Spatial variation in biodiversity, soil degradation and productivity in agricultural landscapes in the highlands of Tigray, northern Ethiopia. *Food Security*, 1, 83-97.
- HADLEY, C., MULDER, M. B. & FITZHERBERT, E. 2007. Seasonal food insecurity and perceived social support in rural Tanzania. *Public Health Nutrition,* 10, 544-51.
- HARRIS, F. M. A. & MOHAMMED, S. 2003. Relying on nature: wild foods in northern Nigeria. *Ambio*, 32, 24-9.
- HATLOY, A., HALLUND, J., DIARRA, M. M. & OSHAUG, A. 2000. Food variety, socioeconomic status and nutritional status in urban and rural areas in Koutiala (Mali). *Public Health Nutrition*, 3, 57-65.
- HATLOY, A., TORHEIM, L. & OSHAUG, A. 1998. Food variety D a good indicator of nutritional adequacy of the diet? A case study from an urban area in Mali, West Africa. *European Journal of Clinical Nutrition*, 52, 891-8.
- HAWKES, C. & RUEL, M. 2006. The links between agriculture and health: an intersectoral opportunity to improve the health and livelihoods of the poor. *Bulletin of the World Health Organization*, 84, 984-990.
- HILLBRUNER, C. & EGAN, R. 2008. Seasonality, household food security, and nutritional status in Dinajpur, Bangladesh. *Food & Nutrition Bulletin*, 29, 221-231.
- HILLOCKS, R. J. 2011. Farming for balanced nutrition: an agricultural approach to addressing micronutrient deficiency among the vulnerable poor in Africa. *African Journal of Food, Agriculture, Nutrition and Development,* 11, 4688-4707.
- HODDINOTT, J., ALDERMAN, H., BEHRMAN, J. R., HADDAD, L. & HORTON, S. 2013. The economic rationale for investing in stunting reduction. *Maternal & child nutrition*, 9, 69-82.
- HOMER, C. G., EDWARDS JR, T. C., RAMSEY, R. D. & PRICE, K. P. 1993. Use of remote sensing methods in modelling sage grouse winter habitat. *The Journal of wildlife management*, 78-84.
- HYDER, A. A., MAMAN, S., NYONI, J. E., KHASIANI, S. A., TEOH, N., PREMJI, Z. & SOHANI, S. 2005. The pervasive triad of food security, gender inequity and women's health: exploratory research from sub-Saharan Africa. *African Health Sciences*, *5*, 328-34.

- JOHNS, T. 2003. Plant biodiversity and malnutrition: simple solutions to complex problems. *African Journal of Food, Agriculture, Nutrition and Development*, 3, 45-52.
- JOHNS, T. & EYZAGUIRRE, P. B. 2006. Linking biodiversity, diet and health in policy and practice. [Review] [42 refs]. *Proceedings of the Nutrition Society*, 65, 182-9.
- JOHNS, T. & EYZAGUIRRE, P. B. 2007. Biofortification, biodiversity and diet: A search for complementary applications against poverty and malnutrition. *Food Policy*, 32, 1-24.
- JOHNS, T. & MAUNDU, P. 2006. Forest biodiversity, nutrition and population health in market-oriented food systems. UNASYLVA-FAO-, 57, 34.
- JOHNS, T., MHORO, E. B. & SANAYA, P. 1996. Food Plants and Masticants of the Batemi of Ngorongoro District, Tanzania. *Economic Botany*, 50, 115-121.
- JOHNS, T. & STHAPIT, B. R. 2004. Biocultural diversity in the sustainability of developing-country food systems. *Food & Nutrition Bulletin*, 25, 143-155.
- JOHNSON, K. B., JACOB, A. & BROWN, M. E. 2013. Forest cover associated with improved child health and nutrition: evidence from the Malawi Demographic and Health Survey and satellite data. *Global Health: Science and Practice*, 1, 237-248.
- JOHNSON, R. K., DRISCOLL, P. & GORAN, M. I. 1996. Comparison of multiple-pass 24-hour recall estimates of energy intake with total energy expenditure determined by the doubly labeled water method in young children. *Journal of the American Dietetic Association*, 96, 1140-1144.
- JSTOR. 2014. *Global plants* [Online]. Available: <u>http://plants.jstor.org/</u> [Accessed 29 April 2014.
- KABANDA, T. & JURY, M. 1999. Inter-annual variability of short rains over northern Tanzania. *Climate Research*, 13, 231-241.
- KABUBO-MARIARA, J., NDENGE, G. K. & MWABU, D. K. 2009. Determinants of children's nutritional status in Kenya: Evidence from demographic and health surveys. *Journal of African Economies*, 18, 363-387.
- KARYADI, E., SCHULTINK, W., NELWAN, R. H., GROSS, R., AMIN, Z., DOLMANS, W. M., VAN DER MEER, J. W., HAUTVAST, J. G. J. & WEST, C. E. 2000. Poor micronutrient status of active pulmonary tuberculosis patients in Indonesia. *The Journal of nutrition*, 130, 2953-2958.
- KAWARAZUKA, N. & BÉNÉ, C. 2011. The potential role of small fish species in improving micronutrient deficiencies in developing countries: building evidence. *Public health nutrition*, 14, 1927-1938.
- KEDING, G. B., MSUYA, J. M., MAASS, B. L. & KRAWINKEL, M. B. 2011. Dietary patterns and nutritional health of women: the nutrition transition in rural Tanzania. *Food & Nutrition Bulletin*, 32, 218-226.
- KEENAN, D. P., OLSON, C., HERSEY, J. C. & PARMER, S. M. 2001. Measures of food insecurity/security. *Journal of Nutrition Education*, 33, S49-S58.
- KENNEDY, G. L., PEDRO, M. R., SEGHIERI, C., NANTEL, G. & BROUWER, I. 2007. Dietary diversity score is a useful indicator of micronutrient intake in non-breast-feeding Filipino children. *The Journal of nutrition*, 137, 472-477.
- KHATUN, M., STENLUND, H. & HÖRNELL, A. 2004. BRAC initiative towards promoting gender and social equity in health: a longitudinal study of child growth in Matlab, Bangladesh. *Public health nutrition*, 7, 1071-1079.

- KIDALA, D., GREINER, T. & GEBRE-MEDHIN, M. 2000. Five-year follow-up of a food-based vitamin A intervention in Tanzania. *Public Health Nutrition*, 3, 425-432.
- KIKAFUNDA, J. K., WALKER, A. F., COLLETT, D. & TUMWINE, J. K. 1998. Risk factors for early childhood malnutrition in Uganda. *Pediatrics*, 102, E45.
- KILARU, A., GRIFFITHS, P. L., GANAPATHY, S. & GHOSH, S. 2005. Community-based nutrition education for improving infant growth in rural Karnataka. *Indian Pediatrics*, 42, 425-432.
- KINABO, J., MNKENI, A., NYARUHUCHA, C., MSUYA, J., HAUG, A. & ISHENGOMA, J. 2006. Feeding frequency and nutrient content of foods commonly consumed in the Iringa and Morogoro regions in Tanzania. *International journal of food sciences and nutrition*, 57, 9-17.
- KNUEPPEL, D., DEMMENT, M. & KAISER, L. 2010. Validation of the Household Food Insecurity Access Scale in rural Tanzania. *Public Health Nutrition,* 13, 360-7.
- KRÄHENBÜHL, J.-D., SCHUTZ, Y. & JEQUIER, E. 1998. High fat versus high carbohydrate nutritional supplementation: a one year trial in stunted rural Gambian children. *European journal of clinical nutrition*, 52, 213-222.
- KREBS, N. F., MAZARIEGOS, M., CHOMBA, E., SAMI, N., PASHA, O., TSHEFU, A., CARLO, W. A., GOLDENBERG, R. L., BOSE, C. L., WRIGHT, L. L., KOSO-THOMAS, M., GOCO, N., KINDEM, M., MCCLURE, E. M., WESTCOTT, J., GARCES, A., LOKANGAKA, A., MANASYAN, A., IMENDA, E., HARTWELL, T. D. & HAMBIDGE, K. M. 2012. Randomized controlled trial of meat compared with multimicronutrient-fortified cereal in infants and toddlers with high stunting rates in diverse settings. *American Journal of Clinical Nutrition*, 96, 840-847.
- KREBS, N. F., MAZARIEGOS, M., TSHEFU, A., BOSE, C., SAMI, N., CHOMBA, E., CARLO, W., GOCO, N., KINDEM, M., WRIGHT, L. L. & HAMBIDGE, K. M. 2011. Meat consumption is associated with less stunting among toddlers in four diverse low-income settings. *Food and Nutrition Bulletin*, 32, 185-191.
- KULLDORFF, M. 2005. SaTScanTM User Guide.
- KUMAR, S. K. & HOTCHKISS, D. 1988. Consequences of deforestation for women's time allocation, agricultural production, and nutrition in hill areas of Nepal, Intl Food Policy Res Inst.
- LAMONTAGNE, J. F., ENGLE, P. L. & ZEITLIN, M. F. 1998. Maternal employment, child care, and nutritional status of 12-18-month-old children in Managua, Nicaragua. *Social Science & Medicine*, 46, 403-414.
- LARSON, B. A. & FRISVOLD, G. B. 1996. Fertilizers to support agricultural development in sub-Saharan Africa: What is needed and why. *Food Policy*, 21, 509-525.
- LEROY, J. L. & FRONGILLO, E. A. 2007. Can interventions to promote animal production ameliorate undernutrition? *The Journal of nutrition*, 137, 2311-2316.
- LEROY, J. L., RAZAK, A. A. & HABICHT, J.-P. 2008. Only children of the head of household benefit from increased household food diversity in northern Ghana. *The Journal of nutrition*, 138, 2258-2263.
- LI, Y., WEDICK, N. M., LAI, J., HE, Y., HU, X., LIU, A., DU, S., ZHANG, J., YANG, X., CHEN, C., HU, F. B. & MA, G. 2011. Lack of dietary diversity

and dyslipidaemia among stunted overweight children: the 2002 China National Nutrition and Health Survey. *Public Health Nutrition*, 14, 896-903.

LIN, C. A., BOSLAUGH, S., CILIBERTO, H. M., MALETA, K., ASHORN, P., BRIEND, A. & MANARY, M. J. 2007. A prospective assessment of food and nutrient intake in a population of Malawian children at risk for kwashiorkor. *Journal of pediatric gastroenterology and nutrition*, 44, 487-493.

LIPTON, M. & KADT, E. D. 1988. Agriculture-health linkages.

MA, J., ZHOU, L., HU, Y., LIU, J., LIU, S., ZHANG, J. & SHENG, X. 2012. A summary index of infant and child feeding practices is associated with child growth in urban Shanghai. *BMC Public Health*, 12.

MARQUIS, G. S., HABICHT, J. P., LANATA, C. F., BLACK, R. E. & RASMUSSEN, K. M. 1997. Breast milk or animal-product foods improve linear growth of Peruvian toddlers consuming marginal diets. *American Journal of Clinical Nutrition*, 66, 1102-1109.

MARRIOTT, B. P., WHITE, A., HADDEN, L., DAVIES, J. C. & WALLINGFORD, J. C. 2012. World Health Organization (WHO) infant and young child feeding indicators: associations with growth measures in 14 low-income countries. *Maternal and Child Nutrition*, 8, 354-370.

MARTEN, G. G. & ABDOELLAH, O. S. 1988. Crop diversity and nutrition in West Java. *Ecology of food and nutrition*, 21, 17-43.

MASSET, E., HADDAD, L., CORNELIUS, A. & ISAZA-CASTRO, J. 2012. Effectiveness of agricultural interventions that aim to improve nutritional status of children: systematic review. *BMJ: British Medical Journal*, 344.

MATHERS, C. D., SADANA, R., SALOMON, J. A., MURRAY, C. J. & LOPEZ, A. D. 2001. Healthy life expectancy in 191 countries, 1999. *The Lancet*, 357, 1685-1691.

MAZENGO, M. C., SIMELL, O., LUKMANJI, Z., SHIRIMA, R. & KARVETTI, R. L. 1997. Food consumption in rural and urban Tanzania. *Acta Tropica*, 68, 313-326.

MAZZEO, J. 2011. Cattle, Livelihoods, and Coping with Food Insecurity in the Context of Drought and HIV/AIDS in Rural Zimbabwe. *Human Organization* <u>http://sfaa</u>. metapress. com/openurl. asp? genre= article&issn= 0018-7259&volume= 70&issue= 4&spage= 405, 70, 405-415.

MBOYA, R., TONGOONA, P., DERERA, J., MUDHARA, M. & LANGYINTUO, A. 2011. The dietary importance of maize in Katumba ward, Rungwe district, Tanzania, and its contribution to household food security. *African Journal of Agricultural Research*, 6, 2617-2626.

MCMICHAEL, A. J. 2005. Integrating nutrition with ecology: balancing the health of humans and biosphere. *Public Health Nutrition*, 8, 706-715.

MEASURE DHS. 2014. DHS Program home page [Online]. Available: http://www.measuredhs.com/aboutdhs/ [Accessed 30 April 2014.

MILLER, K. V. & CONROY, M. J. 1990. SPOT satellite imagery for mapping Kirtland's warbler wintering habitat in the Bahamas. *Wildlife Society Bulletin*, 252-257.

MOLLA, A., MOLLA, A. M., RAHIM, A., SARKER, S. A., MOZAFFAR, Z. & RAHAMAN, M. 1982. Intake and absorption of nutrients in children with cholera and rotavirus infection during acute diarrhea and after recovery. *Nutrition Research*, 2, 233-242.

- MONTGOMERY, D. C., PECK, E. A. & VINING, G. G. 2012. Introduction to linear regression analysis, John Wiley & Sons.
- MORGENSTERN, H. 1995. Ecologic studies in epidemiology: concepts, principles, and methods. *Annual review of public health*, 16, 61-81.
- MOTARJEMI, Y., KÄFERSTEIN, F., MOY, G. & QUEVEDO, F. 1993. Contaminated weaning food: a major risk factor for diarrhoea and associated malnutrition. *Bulletin of the World Health Organization*, 71, 79.
- MOURSI, M. M., ARIMOND, M., DEWEY, K. G., TRÈCHE, S., RUEL, M. T. & DELPEUCH, F. 2008. Dietary diversity is a good predictor of the micronutrient density of the diet of 6-to 23-month-old children in Madagascar. *The Journal of nutrition,* 138, 2448-2453.
- MÜLLER, O., GARENNE, M., KOUYATÉ, B. & BECHER, H. 2003. The association between protein—energy malnutrition, malaria morbidity and all-cause mortality in West African children. *Tropical Medicine* & *International Health*, 8, 507-511.
- MURPHY, S. P., GEWA, C., LIANG, L. J., GRILLENBERGER, M., BWIBO, N. O. & NEUMANN, C. G. 2003. School snacks containing animal source foods improve dietary quality for children in rural Kenya. *Journal of Nutrition*, 133.
- MURTHY, R. K., SAGAYAM, J., RENGALAKSHMI & NAIR, S. 2008. Gender, efficiency, poverty reduction, and empowerment: reflections from an agriculture and credit programme in Tamil Nadu, India. *Gender & Development*, 16, 101-116.
- MYATT, M., KHARA, T. & COLLINS, S. 2006. A review of methods to detect cases of severely malnourished children in the community for their. *Putting the management of severe malnutrition back on the international health agenda,* 27, 7.
- NAKHAUKA, E. B. 2009. Agricultural biodiversity for food and nutrient security: The Kenyan perspective. *International Journal of Biodiversity and Conservation,* 1, 208-214.
- NATIONAL BUREAU OF STATISTICS 2011. Tanzania in figures 2010. The United Republic of Tanzania: Ministry of Finance.
- NEUMANN, C. G., JIANG, L. H., WEISS, R. E., GRILLENBERGER, M., GEWA, C. A., SIEKMANN, J. H., MURPHY, S. P. & BWIBO, N. O. 2013. Meat supplementation increases arm muscle area in Kenyan schoolchildren. *British Journal of Nutrition*, 109, 1230-1240.
- NIELSEN, H., ROOS, N. & THILSTED, S. H. 2003. The impact of semiscavenging poultry production on the consumption of animal source foods by women and girls in Bangladesh. *The Journal of nutrition*, 133, 4027S-4030S.
- NIRANJALA, A. M. S. & GUNAWARDENA, N. S. 2011. Nutritional status of adolescent females in estates in Haliela, Sri Lanka. *ICAN: Infant, Child & Adolescent Nutrition,* 3, 260-267.
- NJELEKELA, M., KUGA, S., NARA, Y., NTOGWISANGU, J., MASESA, Z., MASHALLA, Y., IKEDA, K., MTABAJI, J., YAMORI, Y. & TSUDA, K. 2002. Prevalence of obesity and dyslipidemia in middle-aged men and women in Tanzania, Africa: relationship with resting energy expenditure and dietary factors. *Journal of nutritional science and vitaminology*, 48, 352-358.
- NORDEIDE, M. B., HATLOY, A., FOLLING, M., LIED, E. & OSHAUG, A. 1996. Nutrient composition and nutritional importance of green leaves and wild

food resources in an agricultural district, Koutiala, in southern Mali. *International Journal of Food Sciences & Nutrition*, 47, 455-68.

- NOVOTNY, R. 1987. Preschool child feeding, health and nutritional status in Gualaceo, Ecuador. *Archivos Latinoamericanos de Nutricion*, 37, 417-443.
- NTI, C. A. & LARTEY, A. 2007. Effect of caregiver feeding behaviours on child nutritional status in rural Ghana. *International Journal of Consumer Studies*, 31, 303-309.
- NTI, C. A. & LARTEY, A. 2008. Influence of care practices on nutritional status of Ghanaian children. *Nutrition research and practice*, 2, 93-99.
- NUNGO, R. A., OKOTH, M. W. & MBUGUA, S. K. 2012. Nutrition status of children under-five years in cassava consuming communities in Nambale, Busia of western Kenya. *Food and Nutrition Sciences*, 3, 796-801.
- NYANGWESO, P., ODHIAMBO, M., ODUNGA, P., KORIR, M., KIPSAT, M. & SEREM, A. Household food security in Vihiga District, Kenya; Determinants of dietary diversity. African Crop Science conference proceedings, 2007. 1383-1389.
- OBATOLU, V. A. 2003. Growth pattern of infants fed with a mixture of extruded malted maize and cowpea. *Nutrition,* 19, 174-178.
- OGLE, B. M., HUNG, P. H. & TUYET, H. T. 2001. Significance of wild vegetables in micronutrient intakes of women in Vietnam: an analysis of food variety. *Asia Pacific Journal of Clinical Nutrition*, 10, 21-30.
- ONIS, M. D., FRONGILLO, E. A. & BLÖSSNER, M. 2000. Is malnutrition declining? An analysis of changes in levels of child malnutrition since 1980. *Bulletin of the World Health Organization*, 78, 1222-1233.
- ONYANGO, A., KOSKI, K. G. & TUCKER, K. L. 1998. Food diversity versus breastfeeding choice in determining anthropometric status in rural Kenyan toddlers. *International Journal of Epidemiology*, 27, 484-489.
- ONYANGO, A. W. 2003. Dietary diversity, child nutrition and health in contemporary African communities. *Comparative Biochemistry and Physiology Part A: Molecular & Integrative Physiology*, 136, 61-69.
- PADMAJA, G. & STEINKRAUS, K. H. 1995. Cyanide detoxification in cassava for food and feed uses. *Critical Reviews in Food Science & Nutrition*, 35, 299-339.
- PASRICHA, S.-R., CARUANA, S. R., PHUC, T. Q., CASEY, G. J., JOLLEY, D., KINGSLAND, S., TIEN, N. T., MACGREGOR, L., MONTRESOR, A. & BIGGS, B.-A. 2008. Anemia, iron deficiency, meat consumption, and hookworm infection in women of reproductive age in northwest Vietnam. *The American journal of tropical medicine and hygiene*, 78, 375-381.
- PAUDEL, R., PRADHAN, B., WAGLE, R. R., PAHARI, D. P. & ONTA, S. R. 2012. Risk factors for stunting among children: a community based case control study in Nepal. *Kathmandu University Medical Journal*, 10, 18-24.
- PENDAELI, E., CHALE, F., MWASHA, I. AND SHAO, S. 2010. Cooking with traditional leafy vegetables; Indigenous plants in Tanzania's kitchen.
- . In: SLOW FOOD FOUNDATION FOR BIODIVERSITY, D. E. S. R. (ed.).
- PETERS, C. R., O'BRIEN, E. M. & DRUMMOND, R. B. 1992. Edible wild plants of subsaharan Africa, Royal Botanic Gardens, Kew.
- PICKERING, A. J. & DAVIS, J. 2012. Freshwater availability and water fetching distance affect child health in sub-Saharan Africa. *Environmental Science & Technology*, 46, 2391-7.

- PLA, L. 2004. Bootstrap confidence intervals for the Shannon biodiversity index: a simulation study. *Journal of agricultural, biological, and environmental statistics,* 9, 42-56.
- PONCE, X., RAMIREZ, E. & DELISLE, H. 2006. A more diversified diet among Mexican men may also be more atherogenic. *The Journal of nutrition*, 136, 2921-2927.
- POPKIN, B. M. 1998. The nutrition transition and its health implications in lowerincome countries. *Public health nutrition*, 1, 5-21.
- POSKITT EME. Nutrition in the Rural African Woman: Studies from Keneba, The Gambia [Online]. Available: <u>http://www.ifuw.org/wp-</u> <u>content/uploads/2014/01/poskitt.pdf</u> [Accessed 3 May 2014.
- POUILLOT, R., GARIN, B., RAVAONINDRINA, N., DIOP, K., RATSITORAHINA, M., RAMANANTSOA, D. & ROCOURT, J. 2012. A risk assessment of campylobacteriosis and salmonellosis linked to chicken meals prepared in households in Dakar, Senegal. *Risk Analysis*, 32, 1798-1819.
- POWELL, B., HALL, J. & JOHNS, T. 2011. Forest cover, use and dietary intake in the East Usambara Mountains, Tanzania. (Special Issue: Forests, biodiversity and food security). *International Forestry Review*, 13, 305-317.
- POWELL, J. & WILLIAMS, T. 1994. An overview of mixed farming systems in sub-Saharan Africa. *Livestock and Sustainable Nutrient Recycling in Mixed Farming Systems of Sub-Saharan Africa*, 21-36.
- PRAZUCK, T., TALL, F., MACRO, B., ROCHEREAU, A., TRAORE, A., SANOU, T., MALKIN, J.-E., APAIRE-MARCHAIS, V., MASSON, D. & DUBLANCHET, A. 1993. HIV infection and severe malnutrition: a clinical and epidemiological study in Burkina Faso. *AIDS*, 7, 103-108.
- PROOS, L. 1993. Anthropometry in adolescence-secular trends, adoption, ethnic and environmental differences. *Hormone Research in Paediatrics*, 39, 18-24.
- PROSPERI, P., ALLEN, T., PADILLA, M., PERI, I. & COGILL, B. 2014. Sustainability and Food & Nutrition Security. *SAGE Open,* 4, 2158244014539169.
- PRUDENCIO, Y. C. & AL-HASSAN, R. 1994. The food security stabilization roles of cassava in Africa. *Food Policy*, 19, 57-64.
- RAH, J. H., AKHTER, N., SEMBA, R. D., DE PEE, S., BLOEM, M. W.,
 CAMPBELL, A. A., MOENCH-PFANNER, R., SUN, K., BADHAM, J. &
 KRAEMER, K. 2010. Low dietary diversity is a predictor of child stunting in rural Bangladesh. *European Journal of Clinical Nutrition*, 64, 1393-8.
- RAST, M., BEZY, J. & BRUZZI, S. 1999. The ESA Medium Resolution Imaging Spectrometer MERIS a review of the instrument and its mission. International Journal of Remote Sensing, 20, 1681-1702.
- REMANS, R., FLYNN, D. F., DECLERCK, F., DIRU, W., FANZO, J., GAYNOR, K., LAMBRECHT, I., MUDIOPE, J., MUTUO, P. K. & NKHOMA, P. 2011a. Assessing nutritional diversity of cropping systems in African villages. *PLoS ONE*, 6, e21235.
- REMANS, R., PRONYK, P. M., FANZO, J. C., CHEN, J., PALM, C. A., NEMSER, B., MUNIZ, M., RADUNSKY, A., ABAY, A. H. & COULIBALY, M. 2011b. Multisector intervention to accelerate reductions in child stunting: an observational study from 9 sub-Saharan African countries. *The American journal of clinical nutrition*, 94, 1632-1642.

- RIVERA, J. A., MARTORELL, R., RUEL, M. T., HABICHT, J.-P. & HAAS, J. D. 1995. Nutritional supplementation during the preschool years influences body size and composition of Guatemalan adolescents. *The Journal of nutrition*, 125, 1068S.
- ROMANOFF, S. & LYNAM, J. 1992. Commentary: Cassava and African food security: Some ethnographic examples.
- ROWLAND, M., ROWLAND, S. G. & COLE, T. J. 1988. Impact of infection on the growth of children from 0 to 2 years in an urban West African community. *The American journal of clinical nutrition*, 47, 134-138.
- RUBIN, D. 1987. Multiple imputation for nonresponse in surveys. *New York, USA: John Willey & Sons.*
- RUEL, M. T. 2003. Operationalizing dietary diversity: a review of measurement issues and research priorities. *The Journal of nutrition*, 133, 3911S-3926S.
- RUTSTEIN, S. O., JOHNSON, K. & MEASURE, O. M. 2004. *The DHS wealth index*, ORC Macro, MEASURE DHS.
- SAIBUL, N., SHARIFF, Z. M., LIN, K. G., KANDIAH, M., GHANI, N. A. & RAHMAN, H. A. 2009. Food variety score is associated with dual burden of malnutrition in Orang Asli (Malaysian indigenous peoples) households: implications for health promotion. *Asia Pacific Journal of Clinical Nutrition*, 18, 412-422.
- SAVY, M., MARTIN-PRÉVEL, Y., SAWADOGO, P., KAMELI, Y. & DELPEUCH, F. 2005. Use of variety/diversity scores for diet quality measurement: relation with nutritional status of women in a rural area in Burkina Faso. *European Journal of Clinical Nutrition*, 59, 703-716.
- SAVY, M., MARTIN-PRÉVEL, Y., TRAISSAC, P., EYMARD-DUVERNAY, S. & DELPEUCH, F. 2006. Dietary diversity scores and nutritional status of women change during the seasonal food shortage in rural Burkina Faso. *The Journal of nutrition*, 136, 2625-2632.
- SAWADOGO, P. S., MARTIN-PRÉVEL, Y., SAVY, M., KAMELI, Y., TRAISSAC, P., TRAORÉ, A. S. & DELPEUCH, F. 2006. An infant and child feeding index is associated with the nutritional status of 6-to 23-month-old children in rural Burkina Faso. *The Journal of nutrition*, 136, 656-663.
- SEN, A. 1981. Poverty and Famines: An Essay on Entitlement and Deprivation. Oxford: Clarendon Press.
- SHACK, K. W., GRIVETTI, L. E. & DEWEY, K. G. 1990. Cash cropping, subsistence agriculture, and nutritional status among mothers and children in lowland Papua New Guinea. Social Science & Medicine, 31, 61-68.
- SIEKMANN, J. H., ALLEN, L. H., BWIBO, N. O., DEMMENT, M. W., MURPHY, S. P. & NEUMANN, C. G. 2003. Kenyan school children have multiple micronutrient deficiencies, but increased plasma vitamin B-12 is the only detectable micronutrient response to meat or milk supplementation. *The Journal of nutrition*, 133, 3972S-3980S.
- SIMKO, M. D., COWELL, C. & GILBRIDE, J. A. 1995. *Nutrition assessment: A comprehensive guide for planning intervention*, Jones & Bartlett Learning.
- SKIDMORE, A., OINDO, B. & SAID, M. Biodiversity assessment by remote sensing. Proceedings of the 30th International symposium on remote sensing of the environment: information for risk management and sustainable development, 2003. 4.

- SMALING, E. M. & BRAUN, A. 1996. Soil fertility research in sub-Saharan Africa: New dimensions, new challenges. *Communications in Soil Science & Plant Analysis*, 27, 365-386.
- SMITH, L. C., RUEL, M. T. & NDIAYE, A. 2005. Why is child malnutrition lower in urban than in rural areas? Evidence from 36 developing countries. *World Development*, 33, 1285-1305.
- SOMMER, A., KATZ, J. & TARWOTJO, I. 1984. Increased risk of respiratory disease and diarrhea in children with preexisting mild vitamin A deficiency. *The American journal of clinical nutrition,* 40, 1090-1095.
- SRIVASTAVA, J., SMITH, N. J. & FORNO, D. 1996. *Biodiversity and agriculture: implications for conservation and development*, World Bank Publications.
- STAHL, M. 1993. Land degradation in East Africa. Ambio, 22, 505-8.
- STAT COMPILER. 2014. *The DHS Program STAT compiler.* [Online]. Available: <u>http://www.statcompiler.com/</u> [Accessed 30 April 2014.
- STEPHENSEN, C. B. 1999. Burden of infection on growth failure. *Journal of Nutrition*, 129, 534S-538S.
- STEPHENSON, K., AMTHOR, R., MALLOWA, S., NUNGO, R., MAZIYA-DIXON, B., GICHUKI, S., MBANASO, A. & MANARY, M. 2010. Consuming cassava as a staple food places children 2-5 years old at risk for inadequate protein intake, an observational study in Kenya and Nigeria. *Nutrition Journal*, 9, 9.
- STERNE, J. A., WHITE, I. R., CARLIN, J. B., SPRATT, M., ROYSTON, P., KENWARD, M. G., WOOD, A. M. & CARPENTER, J. R. 2009. Multiple imputation for missing data in epidemiological and clinical research: potential and pitfalls. *BMJ: British Medical Journal*, 338.
- STEYN, N., NEL, J., NANTEL, G., KENNEDY, G. & LABADARIOS, D. 2006. Food variety and dietary diversity scores in children: are they good indicators of dietary adequacy? *Public Health Nutrition*, 9, 644-650.
- SULLIVAN, J., NDEKHA, M., MAKER, D., HOTZ, C. & MANARY, M. J. 2006. The quality of the diet in Malawian children with kwashiorkor and marasmus. *Maternal and Child Nutrition*, 2, 114-122.
- SUPER, C. M., HERRERA, M. G. & MORA, J. O. 1990. Long-Term Effects of Food Supplementation and Psychosocial Intervention on the Physical Growth of Colombian Infants at Risk of Malnutrition. *Child development*, 61, 29-49.
- SWAMINATHAN, H., LAHOTI, R. & SUCHITRA, J. 2012. Women's Property, Mobility and Decision Making'. Discussion Paper 01188, Washington DC: International Food Policy Research Institute, available at: www. ifpri. org/sites/default/files/publications/ifpridp01188. pdf (accessed June 2013).
- SWINDALE, A. & BILINSKY, P. 2006. Development of a universally applicable household food insecurity measurement tool: process, current status, and outstanding issues. *The Journal of nutrition*, 136, 1449S-1452S.
- TALUKDER, A., KIESS, L., HUQ, N., DE PEE, S., DARNTON-HILL, I. & BLOEM, M. W. 2000. Increasing the production and consumption of vitamin Arich fruits and vegetables: Lessons learned in taking the Bangladesh homestead gardening programme to a national scale. *Food & Nutrition Bulletin*, 21, 165-172.
- TALUKDER A, N. J. H., A.K. OSEI, E. VILLATE, D. REARIO, H. KROEUN, L. SOKHOING, A. UDDIN, S. DHUNGE AND V. QUINN, 2010. Homestead food production model contributes to improved household food security

and nutrition status of young children and women in poor populations. *Field Actions Science Reports [Online],* Special Issue 1.

- TANZANIAN NBS & ICF MACRO 2011. Tanzania: Demographic and Health Survey, 2010- Final report.
- TANZANIAN NBS, MINISTRY OF AGRICULTURE AND FOOD SECURITY, MINISTRY OF WATER AND LIVESTOCK DEVELOPMENT, MINISTRY OF COOPERATIVES AND MARKETING, PRESIDENTS OFFICE, REGIONAL ADMINISTRATION AND LOCAL GOVERNMENT & MINISTRY OF FINANCE AND ECONOMIC AFFAIRS - ZANZIBAR 2006. United Republic of Tanzania. National Sample Census of Agriculture. 2002/2003. Small Holder Agriculture. . *Crop Sector - National Report.*
- TANZANIAN NBS & SINGIDA REGIONAL COMMISSIONER'S OFFICE 2005. Singida Region Socio-Economic Profile. *In:* 2ND (ed.). The United Republic of Tanzania.
- TANZANIAN NBS & TANGA REGIONAL COMMISSIONER'S OFFICE 2008. Tanga Regional Socio-Economic Profile. *In:* 2ND (ed.). The United Republic of Tanzania.
- TANZANIAN NBS, I. M. 2011. Tanzania Demographic and Health Survey 2010. Dar es Salaam, Tanzania: NBS and ICF Macro.
- TARWOTJO, I., KATZ, J., WEST, K. P., JR., TIELSCH, J. M. & SOMMER, A. 1992. Xerophthalmia and growth in preschool Indonesian children. *American Journal of Clinical Nutrition*, 55, 1142-1146.
- TESSEMA, M., BELACHEW, T. & ERSINO, G. 2013. Feeding patterns and stunting during early childhood in rural communities of Sidama, South Ethiopia. *The Pan African medical journal*, 14.
- THE DHS PROGRAM. 2014a. *Demographic and Health Surveys. Where we work.* [Online]. Available: http://www.measuredhs.com/countries/start.cfm [Accessed 30 April 2014.
- THE DHS PROGRAM. 2014b. *The DHS Program. Demographic and Health Surveys. Available datasets.* [Online]. Available: http://dhsprogram.com/data/available-datasets.cfm [Accessed 25 April 2014.
- THE DHS PROGRAM. 2014c. *Measure DHS. Demographic and Health Surveys. Download datasets.* [Online]. Available: http://legacy.measuredhs.com/myaccount/view_main.cfm [Accessed 30 April 2014.
- THE WORLD BANK 2008. World development report. Agriculture for development.
- THE WORLD BANK. 2014. Country classifications [Online]. Available: http://data.worldbank.org/about/country-classifications [Accessed 23 April 2014.
- THILSTED, S. H., ROOS, N. & HASSAN, N. 1997. The role of small indigenous fish species in food and nutrition security in Bangladesh. *Naga, The ICLARM Quarterly,* 20, 82-84.
- THORLAKSON, T., NEUFELDT, H. & DUTILLEUL, F. C. 2012. Reducing subsistence farmers' vulnerability to climate change: evaluating the potential contributions of agroforestry in western Kenya. *Agriculture & Food Security*, 1, 1-13.
- THORNE-LYMAN, A. L., VALPIANI, N., SUN, K., SEMBA, R. D., KLOTZ, C. L., KRAEMER, K., AKHTER, N., DE PEE, S., MOENCH-PFANNER, R. & SARI, M. 2010. Household dietary diversity and food expenditures are

closely linked in rural Bangladesh, increasing the risk of malnutrition due to the financial crisis. *The Journal of nutrition*, 140, 182S-188S.

- THRUPP, L. A. 2000. Linking agricultural biodiversity and food security: the valuable role of agrobiodiversity for sustainable agriculture. *International Affairs*, 76, 283-297.
- TILMAN, D., WEDIN, D. & KNOPS, J. 1996. Productivity and sustainability influenced by biodiversity in grassland ecosystems. *Nature*, 379, 718-720.
- TORHEIM, L., OUATTARA, F., DIARRA, M., THIAM, F., BARIKMO, I., HATLØY, A. & OSHAUG, A. 2004. Nutrient adequacy and dietary diversity in rural Mali: association and determinants. *European Journal of Clinical Nutrition*, 58, 594-604.
- TUCKER, C. & SELLERS, P. 1986. Satellite remote sensing of primary production. *International Journal of Remote Sensing*, 7, 1395-1416.
- TWYMAN, C., MORRISON, J. & SPORTON, D. 1999. The final fifth: Autobiography, reflexivity and interpretation in cross-cultural research. *Area*, 31, 313-325.
- U.S. DEPARTMENT OF AGRICULTURE, C. E. S., FOREST SERVICE. NATURAL RESOURCE CONSERVATION SERVICE, BUREAU OF LAND MANAGEMENT 1999. Sampling Vegetation Attributes. Interagency Technical Reference. Technical Reference. BLM/RS/ST-96/002+1730. .
- UCLA: STATISTICAL CONSULTING GROUP 2014. Chapter 2 Regression Diagnostics. . *Regression with Stata.*
- UNFAO. Expert Consultation on Nutrition Indicators for Biodiversity. . 2010a Rome.
- UNFAO 2010b. The State of Food Insecurity in the World. Addressing food insecurity in protracted crises.
- UNICEF. 1990. UNICEF Conceptual Framework [Online]. Available: http://www.unicef.org/nutrition/training/2.5/4.html [Accessed 25 April 14.
- UNICEF. 2006. Progress for children What is undernutrition? [Online]. Available: http://www.unicef.org/progressforchildren/2006p4/index_undernutrition

http://www.unicef.org/progressforchildren/2006n4/index_undernutrition.ht ml [Accessed 25 04 2014.

- UNICEF. 2014. *Definition of nutrition indicators* [Online]. Available: http://www.unicef.org/infobycountry/stats_popup2.html [Accessed 25 April 2014.
- UNITED NATIONS. 2014. *Millennium Development Goals* [Online]. Available: www.un.org/millenniumgoals/ [Accessed 25 April 2014.
- UPTON, M. 2000. The 'Livestock Revolution'–Implications for smallholder agriculture: A case study of milk and poultry production in Kenya. *Food and Agriculture Organization of the United Nations. Rome, Italy.*
- VAINIO-MATTILA, K. Wild vegetables used by the Sambaa in the Usambara Mountains, NE Tanzania. Annales Botanici Fennici, 2000. Helsinki: Societas Biologica Fennica Vanamo, 1964-, 57-68.
- VICTORA, C. G., ADAIR, L., FALL, C., HALLAL, P. C., MARTORELL, R., RICHTER, L. & SACHDEV, H. S. 2008. Maternal and child undernutrition: consequences for adult health and human capital. *Lancet*, 371, 340.
- VICTORA, C. G., BARROS, F. C. D., KIRKWOOD, B. R. & VAUGHAN, J. P. 1990. Pneumonia, diarrhea, and growth in the first 4 y of life: a

longitudinal study of 5914 urban Brazilian children. *The American journal of clinical nutrition*, 52, 391-396.

- VICTORA, C. G., DE ONIS, M., HALLAL, P. C., BLÖSSNER, M. & SHRIMPTON, R. 2010. Worldwide timing of growth faltering: revisiting implications for interventions. *Pediatrics*, peds. 2009-1519.
- WAHLQVIST, M. L. 2003. Regional food diversity and human health. Asia Pacific Journal of Clinical Nutrition, 12, 304-308.
- WAHLQVIST, M. L. & SPECHT, R. L. 1998. Food variety and biodiversity: Econutrition. *Asia Pacific Journal of Clinical Nutrition*, 7, 314-319.
- WALINGO, M. K. & EKESA, B. N. 2013. Nutrient intake, morbidity and nutritional status of preschool children are influenced by agricultural and dietary diversity in Western Kenya. *Pakistan Journal of Nutrition*, 12, 854-859.
- WALKER, R., STOMS, D., ESTES, J. & CAYOCCA, K. Relationships between biological diversity and multi-temporal vegetation index data in California. ACSM ASPRS ANNUAL CONVENTION, 1992. AMERICAN SOC PHOTOGRAMMETRY & REMOTE SENSING+ AMER CONG ON, 562-562.
- WEBB, P., COATES, J., FRONGILLO, E. A., ROGERS, B. L., SWINDALE, A. & BILINSKY, P. 2006. Measuring household food insecurity: why it's so important and yet so difficult to do. *The Journal of nutrition*, 136, 1404S-1408S.
- WELCH, R. M. & GRAHAM, R. D. 1999. A new paradigm for world agriculture: meeting human needs: productive, sustainable, nutritious. *Field Crops Research*, 60, 1-10.
- WHALEY, S. E., SIGMAN, M., NEUMANN, C., BWIBO, N., GUTHRIE, D., WEISS, R. E., ALBER, S. & MURPHY, S. P. 2003. The impact of dietary intervention on the cognitive development of Kenyan school children. *Journal of Nutrition*, 133.
- WHO 2002. Infant and young child nutrition. Global strategy on infant and young child feeding. Report by the Secretariat. *FIFTY-FIFTH WORLD HEALTH ASSEMBLY. A55/15.*
- WHO 2003. Global Strategy for Infant and Young Child Feeding. Geneva: World Health Organisation.
- WHO. 2004. World Health Report 2004: Changing history. Geneva: World Health Organization. [Online]. Available: http://www.who.int/whr/2004/en [Accessed 22 August 2014.
- WHO 2010a. WHO Anthro for personal computers, version 3.2.2, 2011: Software for assessing growth and development of the world's children.
- WHO 2010b. World Health Statistics.
- WORLD AGROFORESTRY CENTRE. 2014. World Agroforestry Centre [Online]. Available: http://www.worldagroforestry.org/ [Accessed 29 April 2014.
- WORLD VISION TANZANIA 2009. Evaluation Report (DRAFT), Mtinko area development programme. Tanzania: World Vision Tanzania.
- WORLD VISION TANZANIA 2011. Essential nutrition package Mtinko and Kinampanda. Annual report 2011. October 2010-September 2011. World Vision Tanzania.
- XU, J., SHARMA, R., FANG, J. & XU, Y. 2008. Critical linkages between landuse transition and human health in the Himalayan region. *Environment International*, 34, 239-247.

- YANG, W., LI, X., LI, Y., ZHANG, S., LIU, L., WANG, X. & LI, W. 2012. Anemia, malnutrition and their correlations with socio-demographic characteristics and feeding practices among infants aged 0-18 months in rural areas of Shaanxi province in northwestern China: a cross-sectional study. BMC Public Health, 12, 1127.
- ZACHARIAH, R., SPIELMANN, M., HARRIES, A. & SALANIPONI, F. 2002. Moderate to severe malnutrition in patients with tuberculosis is a risk factor associated with early death. *Transactions of the Royal Society of Tropical Medicine and Hygiene*, 96, 291-294.
- ZARIN, D. J., GUO, H. & ENU-KWESI, L. 1999. Methods for the assessment of plant species diversity in complex agricultural landscapes: Guidelines for data collection and analysis from the PLEC Biodiversity Advisory Group (PLEC-BAG). PLEC News and views, 13, 3-16.
- ZHANG, J., SHI, L., WANG, J. & WANG, Y. 2009. An infant and child feeding index is associated with child nutritional status in rural China. *Early Human Development*, 85, 247-252.
- ZONGRONE, A., WINSKELL, K. & MENON, P. 2012. Infant and young child feeding practices and child undernutrition in Bangladesh: insights from nationally representative data. *Public Health Nutrition*, 15, 1697-1704.

Appendix A: Ethical approval

Permissions obtained to carry out the primary data collection for Minyenye, Singida and

Mbwei, Lushoto

Level	Minyenye, Singida	Mbwei, Lushoto
National	COSTECH, Dar es Salaam	COSTECH, Dar es Salaam
Regional		Regional Administrative Secretary, Tanga
		Regional Medical Director, Tanga
District	District Executive Director, Singida	District Medical Director, Lushoto
		District Executive Director, Lushoto
		District Commissioner, Lushoto
		Agricultural Extension Officer, Lushoto
Ward	Ward Executive Officer, Mtinko	Ward Executive Officer, Malimbwe
Village	Village Executive Officer, Minyenye	Acting Village Executive Officer of Mbwei,
		Malimbwe

TAN	TANZANIA COMMISSION FOR SCIENCE AND TECHNOLOGY (COSTECH)								
Telepho Director Fax: (25 Email: 1	nes: (255 - 022) 2775155 - 6, 2700745/6 General: (255 - 022) 2700750&2775315 55 - 022) 2775313 relearance@costech.or.tz	Ali Hassan Mwinyi Road P.O. Box 4302 Dar es Salaam Tanzania							
	- RESEARCH PERMIT								
	No. 2012–216-NA-2012-93	08 th May 2012							
1.	Name : Christine L. Cleghorn								
2.	Nationality : New Zealander	ME (
3	Title : The Relationship betw Dietary Diversity and Nutri	veen Agrobiodiversity, ition in Rural Tanzania							
4	Research shall be confined to the following reg Singida and Tanga	gion(s): Dodoma, Pwani,							
5	Permit validity: 08 th May 2012 – 7 th May 2013								
6.	Local Contact/collaborator: Prof. Joyce Kinab Science and Technology, P.O. Box 3006, Agriculture, Morogoro	o, Department of Food Sokoine University of							
7.	Researcher is required to submit progress report on all Publications made after research.	quarterly basis and submit							
	M.Mushi								
	DIRECTOR GENERAL								

Appendix B: Information sheet and consent form

- 284 -

Information Sheet

Agrobiodiversity, dietary diversity and nutritional status in

Sub-Saharan Africa.

*Cristina Cleghorn is conducting this research project in order as part of her university degree. She is studying at and is being funded by the University of Leeds in the United Kingdom.

*You are being invited to take part in this research project. Before you decide it is important for you to understand why the research is being done and what it will involve.

*The researchers hope to provide information that will help people in rural Tanzania develop strategies to make sure they have enough food to eat.

*The purpose of this project is to look at how the food people grow effects what people eat in rural Tanzania.

*Participants have been randomly chosen from a list of people living in this village. We are hoping to interview about 60 people here in the next 2 weeks.

* To be a participant in this research means:

-To be interviewed by us for approximately 40 minutes on what you and you children eat, where this food comes from and some other basic information about your household.

-To have your height and weight measured.

-To have your children's height, weight and upper arms measured.

-To have 2 of our researchers walk around your household farm writing down the type and amount of plants growing there.

-To have the location of your house and farms measured.

*Whilst there are no immediate benefits for the people participating in the project it is hoped that this work will be useful to farmers in Tanzania in the future.

*Personal information that we collect will not be shared with anyone outside of the research team. Information will be summarised and anonymised for before being disseminated to Nutrition and Agriculture experts and/or used for publication. No personal information will be published.

*It is up to you to decide whether or not to take part in this research. If you decide to take part you can still withdraw at any time during the interview or measurements.

*Do you have any questions? Would you like more information on anything?

*Take the time you need to decide whether or not you wish to take part.

Consent Form

- 285 -

Agrobiodiversity, dietary diversity and nutritional status in Sub-Saharan Africa.

Selection of respondent within household:

Is there a female in this household who has children under the age of 5 living here and is responsible for the preparation of food in this household? Would it be possible to speak with her? Yes? No? If no, is there another female who lives in this household that has children under the age of 5 living here?

Would it be possible to speak with her? Yes? No?

If yes \rightarrow Go to Information Sheet

If no \rightarrow Go to next household

.....

Name:

ID Number:

Circle Participants response: Yes or No

Opportunity to ask questions?	Yes?	No?
Do you agree to take part in this research?	Yes?	No?
Can we record some of the interview using this tape recorder?	Yes?	No?
Can we measure your height and weight?	Yes?	No?
Can we measure the height, weight and mid upper arm circumference of your children?	Yes?	No?
Can we measure the location of your house and farm using this global positioning system?	Yes?	No?
Can we take pictures of your health records?	Yes?	No?

Appendix C: Data collection protocols

24 hour recall protocol

For the respondent and her oldest child under five:

- 1. This is designed to capture the diversity of what the individual is eating.
- 2. It will ask about all foods, drinks and snacks the respondent has eaten for the 24 hour period from when they got up yesterday to when they got up today.
- 3. We are not collecting the amount of food the respondent has had, just the different types.
- 4. We will be doing a 24 hour recall for the respondent and the respondent's oldest child under 5 years of age.
- 5. The 24 hour recall will be conducted in 3 passes:
 - a) A list of all foods, drinks and snacks consumed
 - b) Details of all the foods, drinks and snacks consumed
 - c) A review of all the foods, drinks and snacks consumed
- 6. Start by telling the participant that this recall covers all foods, drinks and snacks they had yesterday.
- 7. First pass:
 - a) List all the foods and drinks consumed between the time they got up yesterday morning and the time they got up this morning.
 - b) This should be a quick list with no details asked for. Any details they give however should be translated and I will write them down.
 - c) Don't ask about specific meals. Instead ask: "What was the next thing that you ate or drank yesterday?"
 - d) Don't show surprise or judgement in your responses.
- 8. Second pass:
 - a) Go through the quick list you have collected and probe for more detail about the foods and drinks they have consumed.
 - b) Ask "about what time did you eat/drink?"
 - c) Then "where did you eat/drink?"
 - d) The details of the actual food we want to collect are:
 - i. Cooking method.
 - ii. Additions to the food/drink before consumption. Prompt for vegetables eaten eg. Did you have any vegetables with this food?
 - iii. For recipes we want to know all the different foods that went into the dish.
 - iv. Brand names if packaged food/drinks were eaten.

- v. If they had salt on their food, ask if it was iodised.
- 9. Final pass:
 - a) This is a final check to see that you haven't missed any foods or drinks or any details about these foods and drinks. It is also to check that I have everything down correctly.
 - b) Read through all the information that I have written down and ask the respondent if it is correct.
 - c) Look for gaps in time and prompt the respondent eg. "did you eat anything between the porridge you had at 9am and the rice you had at 6pm?"

Anthropometry protocol

- 1. I need to measure the height and weight of the respondent, her husband/partner and the respondent's children. And the mid upper arm circumference of the respondent's children.
- 2. I will need you to help with these measures. The family may feel more comfortable with you taking the measures and this may be the easier option considering you can communicate with them.
- 3. There is no script for this section but you will have to facilitate the measurements and help do the actual measurements.
- 4. You will need to ask the child if it is ok to take their height and weight in a way that is appropriate to the child's age.
- 5. You will also have to ask if it is ok for us to look at the health records (after the anthropometry measures) We would have already asked consent to take photos of these.

6. Weight:

- a) Adults and children over 2 years: standing on the scales.
 - i. Place the scale on a flat surface in a well-lit area.
- b) Children under 2 years being held by their mother using the tear function on the scales.
 - i. The mother stands on the scale first, without the child. After the mother's weight appears on the display, tell her to remain standing on the scale.
 - ii. Write down the mothers weight.
 - iii. Push the tare button.
 - iv. Pass the child to be weighed to the adult on the scale. Record the child's weight indicated by the scale.

7. Height:

- a) Adults and children over 2 years: Stadiometer, standing up.
 - i. 2 people need to help take these measures.

- ii. Heels need to be against the back of the board. Heels must not be raised off the ground.
- iii. Chin needs to be level. The earholes should be in the same horizontal plane as the lower border of the eye sockets.
- iv. Participant needs to be standing comfortably but straight.
- v. Tell the participant to reach up as much as he/she can, take a deep breath and relax her/his shoulders.
- vi. Make sure the wooden slide that is brought down gently onto their head is flat against the back of the board.
- vii. Same person should take the measures to reduce measurement error.
- b) Children under 2 years: length measurement with the stadiometer laid down on the ground.
 - i. 2 people need to help take these measures. It would also be helpful if the mother could help calm the child and keep her/him still.
 - ii. Head should be against the fixed board.
 - iii. The ear/eye plane should be in line as for the standing height.
 - iv. The measuring tape should be closest to us for ease of reading.
 - v. The body should be as straight as possible. Ankles should be gently pulled to straighten child.
 - vi. Feet should be turned up vertically.
 - vii. Push the movable wooden slide into the heels to read the length measurement.

8. MUAC:

- a. MUAC measuring tape.
 - i. To be taken on the left arm of the child while the arm is hanging down the side of the body and relaxed.
 - ii. With the left arm bent, use a string to find the midpoint of the arm between the shoulder and the tip of the elbow. Mark the midpoint with a pen if this is acceptable to the child.
 - iii. Slide the tape around the midpoint of the arm and take the reading on the tape.
 - iv. Use enough tension to hold the tape against the skin but not so much that the skin is pinched.

Point intercept method:

- 1. Transect lines will be laid every 20 meters along the baseline.
- 2. The transect lines will run down the farm if the land is hilly.
- 3. Records will be made every 10 meters along the transect starting at a random starting point at either 1, 2, 3, 4 or 5 meters in from the baseline (to be randomly selected using tokens).
- 4. A metal pin 1meter in length and 2 mm in diameter will be inserted into the earth at the 10 meter points. All vegetation touching this metal pin will be recorded.

Information to be recorded:

- 1. Family, genus and species names will be recorded for each plant. This can be coded for ease of recording. If the pin hits only bare earth, this will be recorded as well.
- 2. Transect number, transect length and number of points along the transect need to be recorded in the margins of the data recording sheet. Use { to indicate what lines of information are for each transect.
- 3. The number of times the pin hits specific species and bare ground is recorded using a tally.
- 4. The photo numbers of the photos taken should be recorded in the notes section of the data collection form.
- 5. Take general photos of the farm from baseline down the farm. These photos should cover the whole farm.
- 6. Also take photos of specific plants that need to be identified later or specific points if necessary.
- 7. In addition to data collected on the data collection form some information needs to be collected in a notebook:
 - a. Make sure the household ID is clearly recorded on each page.
 - b. Record a description of the farm, what the landscape is like, how well it is kept, the main crops growing, whether crops have been recently harvested, any crop damage etc. Record anything that you find interesting.
 - c. Draw a sketch of the farm (A4 size) that shows the shape of the farm and where the transects are laid. GPS points along the edges need to be recorded on the sketch so the size of the farm can be calculated.
 - d. GPS co-ordinates at both ends of each transect line should also be recorded.
 - e. Record the latitude, longitude and elevation for each GPS point.
 - f. Add any trees to the sketch and write the family, genus and species of all the trees on the farm.
 - g. Mark where photos were taken.

Agrobiodiversity equipment:

- 1. Data entry sheets.
- 2. Notebook for the written description of the farm and the sketch with the GPS coordinates.
- 3. Transect string.
- 4. 1m x 2mm metal pin.
- 5. Camera (2 AA batteries plus 2 spares).
- 6. GPS (2 AA batteries plus 2 spares).
- 7. Randomisation tokens.
- 8. Stakes for either end of the transect.

Appendix D: Household Questionnaire

Date: Time: Interviewe Household ID: Respondent ID: Location:

Observations

House materials:	Interview:
Floor	
Roof	
Exterior walls	
Children:	Other of interest:

- 292 -

Household ID:

Respondent ID:

Demographic and livelihood information

1. How many people live in your household?

A. Family schedule

All family members				If >=5yrs If >=10yrs			¥						
Family member ID	Family member name	Present?	Relationship to respondent	Gender	Date of Birth		Educatio	n	Marital status		Emplo	oyment	
	 Please give me the names of your family members living in this household. 	2. Is this family member present in the village?	3. How is (NAME) related to you?	4. Is (NAME) male or female?	5. What is (NAME)'s Date (or year) of birth?	6. Has (NAME) ever attended school? If yes go to Q7 If no go to Q9	7. Is (NAME) still attending school?	8. What is the highest level of school (NAME) has attended?	9. What is (NAME)'s marital status?	10. Is (NAME) employe d? If yes go to Q11 If no go to Q12	11. Does (NAME) work througho ut the year, seasonall y or just once in a while?	12. Does (NAME) do any other income generatin g activities ? (prompt)	13. Does (NAME) have any other sources of income? (prompt)
(Respondent)			5 b) Respondents ethnicity:	MF		Y N	Y N						
Who looks after you	ur children when you	are working/do	ing these activities?	1	I	1	1	1	1	1	1		
(Husband/partner)			5 b) Husband/ partners ethnicity:	M F		Y N	Y N						
Others				M F		Y N	Y N						
				M F		Y N	Y N						
				MF		Y N	Y N						

- 293 -

Household ID:

Respondent ID:

Family member ID	Family member name	Present?	Relationship to respondent	Gender	Date of Birth	Education		Marital status	Employment		
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				
				M F		Y N	Y N				

Note: Others in household except nuclear family just enter name, relationship to respondent, gender and age

- 294 -

Household ID:

Respondent ID:

B. Household characteristics From now on I am going to be asking questions referring to you, your husband/partner and your children.

1. What is the main source of drinking water for members your family members? (don't read out options)

	piped water tube well or borehole dug well water from spring rain water		tanker truck cart with small tank river or lake bottled water other(specify)		
2.	Where is the water source located? (don't read out options)		in own dwelling in own yard/plot elsewhere	[Go to Q4 Go to Q4 Go to Q3
3.	How long does it take to get there, get water and come back?				minutes
4.	Do you do anything to the water to make it safer to drink?		yes Go to Q5	No Go to Q6	don't know Go to Q6
5 . ⊕	What do you usually do to make it safer to drink? (don't read ou	t options)		ſ	
	boil add bleach/chloride strain through a cloth use water filter		solar disinfection let it stand and settle don't know <u>other</u> (specify)		
6.	What kind of toilet facility do family members in your household	usually use? (don	't read out options)		
	flush or pour flush toilet pit latrine – open pit latrine - closed composting toilet		bucket toilet hanging toilet no toilet/bush/field <u>other</u> (specify)		
7.	Do you share this toilet facility with other households?		yes	no	don't know

- 295 -

Household ID:

Respondent ID:

	Does your household have:			
8.	Electricity?	yes	no	don't know
9.	A radio?	yes	no	don't know
10.	A television?	yes	no	don't know
11.	A mobile phone?	yes	no	don't know
12.	A non-mobile telephone?	yes	no	don't know
13.	A refrigerator?	yes	no	don't know

C. Livelihoods These questions refer to you, your husband/partner and your children.

1.	Do you, your husband/partner or your children own any agricultural land?	yes Go to Q2	NO Go to Q3	don't know Go to Q3
2.	How many hectares/acres of agricultural land do you own all together?			acres/hectares
3.	Do you, your husband/partner or your children grow food?	Yes Go to Q4	No Go to next sec.	don't know Go to next sec.
4.	Do you, your husband/partner and your children grow food mainly on your own land, on family land, on rented land or on someone else's land?	On your own land On family land On rented land On someone <u>elses</u> land		
÷ 5.	Where is this land?			

Where is this land?

Follow-up prompts: Do you just have x farms? Are these all the places you grow food? Do you have a vegetable garden?

- 296 -

Household ID:

Respondent ID:

Diet

A. 24 hour recall of foods eaten by t	he respondent I am going to ask you about food and drink you consu	imed yesterday, from when you woke up in the
morning to when you woke up this morning	g. What was the first thing you had to eat or drink when you woke up yeste	erday?

Time day	of Name of item or dish and ingredients	Cm	ooking tethod	Notes
Promp	t: Did you eat anything (meal or snack) outside of the home yesterday? Yes No IF YE	S: food included ab	ove?No?	ask details.
1.	Was yesterday a special or feast day where you ate unusual foods?	yes	no	don't know
2.	Were you feeling unwell yesterday?	yes	no	don't know
3.	Was yesterday a typical day for your food and drink intake?	yes	no	don't know

б

Household ID:

Respondent ID:

B. 24 hour recall of foods eaten by the respondents oldest child under 5 years (Household Member [D:......) Now I am going to ask you about food and drink consumed yesterday, from when he/she woke up in the morning to when he/she woke up this morning. What was the first thing they had to eat or drink when they woke up yesterday?

- 297 -

Tim day	ne of /	Name of item or dish and ingredients				Cooking method	Notes
Pron	npt: Did (l	NAME) eat anything (meal or snack) outside of the home yesterday?	Yes	No	IF YES: foo	d included ab	ove? No? ask details.
1.	Was y	esterday a special or feast day where (NAME) ate unusual foods?			yes	no	don't know
2. 3.	Was (I Was y	NAME) feeling unwell yesterday? esterday a typical day for (NAME'S) food and drink intake?			yes yes	no	don't know don't know

Househ	old ID: Re	spondent ID:		
C. Diet	questions (n the following question 'family' refers to you, your husband/partner and y	your children.		
1.	What do you think a healthy diet is?			
2.	Do you feel that your family has enough food to eat?	yes Go to Q3b	no Go to Q3a	don't know Go to Q3b
3.	a) What is stopping your family from having enough to eat? b) What factors influence what your family eats?			
4.	Do you feel that your family has enough different types of food to eat?	yes Go to Q5b	no Go to Q5a	don't know Go to Q5b
5.	a) What is stopping your family from having enough different types of food to e b) What factors influence what different types of food your family eats?	eat?		

- 298 -

D. Supplementation questions

Ia). Do you take any supplements? yes Go to Q1b no Go to Q2 don't know Go to Q2
 b). What supplements do you take?......

Househo	ld ID:		Respondent ID:		
2a).	Do your children take	any supplements?	yes Go to Q2b	no Go to sec. E	don't know Go to sec. E
b).	Which children and wh Childrens names	at supplements do they take? Supplements			

- 299 -

E. Breastfeeding questions These questions refer to your children under the age of 5.

+		Last birth: Name:	Second to last birth: Name:	Third to last birth: Name:
1.	Did you ever breastfeed (Name)?	Y N	Y N	Y N
2.	How old was (Name) when you stopped breastfeeding, or are you still breastfeeding (Name)?			
3a).	At what age did you introduce liquids (including water)?			
b) .	What liquids did you introduce?			
4a).	At what age did you introduce solids?			
b) .	What solids did you introduce?			
		9		

- 300 -

Household ID:

Respondent ID:

Food Sources

A. Food Sources table: Please think about the last year and tell me the foods that you, your husband/partner and children have eaten.

Food	Ea te n?	Food (if ot	eaten was? ther, give de	' (tick) tails)	GROWN			BOUGHT	WILD	Notes		
		Grown Go to column 6	Bought Go to column 11	Wild Go to column 13	How much of this food did you harvest in the last year? (in bags/kgg/plants/l ocal measures)	1 or 2 harves ts	What months did you have this crop available to eat?	Did you sell any of this grog?	If yes, how much? (in bags/kgs/plant s/local measures)	What months did you buy this food?	What months did you collect this food?	
Maize												
Mtama												
Ulezi												
Uwele												
Rice												
Sweet potatoes												
Yams												
Cassava root												
Cassava leaves												
Beans												

Respondent ID: Household ID: Bean leaves Spinach Avocado Bananas Groundn uts

Wild foods:

Jan Feb Mar Apr May June July Aug Sept Oct Nov Dec

11

- 301 -

- 302 -

Household ID:

Respondent ID:

Would you consider this normal compared to the last 5 years? No? Why?

Animals	Eater ? (tick	Foo) (if d	d eaten was other, give d	s? (tick) letails)			Notes			
		Raised	Bought	Wild	How many of these animals do you have in your household?	Do you own them?	How many have been killed for food in the last year?	Did you sell any of the meat/milk/egg s?	lf yes, how much?	
Chicken										
Eggs										
Cows										
Milk										
Sheep										
Goats										
Pigs										
Ducks										
Fish	_									
	_									
	_									

302
- 303 -

Respondent ID:

Household ID:

+

Did you r Goats Sheep Cows Chickens Would yo	raise ລູກູນູ; ອຸ່ມ Du consider this normal	Yes (add to table above) Yes (add to table above) Yes (add to table above) Yes (add to table above) I compared to the last 5 years?	No No No? Why?			
B. Food	Sources questio	ns				
1.	Did you grow any cro	ops to sell in the past year?		yes (add to ta	No ble)	don't know
2.	Is this new in the last	t 5 years?		yes	no	don't know
3.	From this information your family do for for	n it seems like you don't have r od during this time?	nuch food available to you ii	n		(months). What does
4 .	In the past year, did	any of the crops you grew go u	ineaten or unsold?	yes Go to Q6	No Go to next se	c. don't know Go to next sec.
5.	Why?					

13

- 304 -

Household ID:

Respondent ID:

Nutritional status of family members: Respondent, husband/partner and respondent's children

Name	Household member ID	Weight	Height	MUAC	Notes
	01 (Respondent)			XXX	
	02 (Respondent's husband/partner)			XXX	

Photographs taken of health records?

yes

no

What the photo is of:	Photo #	What the photo is of:
	What the photo is of:	What the photo is of: Photo #

14

- 305 -

Appendix E: Agrobiodiversity data collection sheet

Date: Time: Research Assistants: Household ID: Household member ID: Location: HH GOS co-ordinates:

Agrobiodiversity measures: "Agrobiodiversity, dietary diversity and Nutritional Status"

++						
	Transect number	Transect length	Photo number	Species	Number of plants	Notes
Γ						

Time:	
Research	Assistants:

Household ID: Household member ID Location: HH GOS co-ordinates:

1. How many hectares/acres is the farm?

acres/hectares

2.	Description of farm:
3.	Photo numbers of farm:

- 307 -

		h Assistants:	Location: HH GOS co-ordinates:		
+	4.	Sketch (including GPS co-ordinates at corners and key points)			
	5.	Key for sketch:			

307